Has opportunistic screening among young adults in England led to a reduction in *Chlamydia trachomatis* infection? Identifying and appraising outcome measures for the evaluation of chlamydia control programmes.

2015

Thesis submitted for the degree of Doctor of Philosophy

Sarah Charlotte Woodhall

UCL

#### **Statement of Authorship**

I, Sarah Woodhall, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

Date.
-------

The systematic review of chlamydia prevalence studies presented in Chapter 3 was carried out as part of a collaborative project on chlamydia control in Europe.<sup>2,3</sup> I contributed to the design and interpretation of the review and extracted data with two other reviewers. The other literature reviews presented in Chapter 3 were conducted independently. I planned, conducted, and interpreted all analyses in Chapters 4 to 8. I designed and implemented the pilot study in Chapter 4 including focus groups to develop the study materials, which were conducted with one other facilitator. Other investigators collected the data in three of the datasets used in this thesis. The National Surveys of Sexual Attitudes and Lifestyles (Natsal) used in Chapters 6 and 7 were collected by NatCen Social Research in collaboration with University College London, the London School of Hygiene and Tropical Medicine and the Health Protection Agency (now part of Public Health England).<sup>4,5</sup> Surveillance data used in Chapter 4 on tests carried out as part of the National Chlamydia Screening Programme (NCSP) (www.chlamydiascreening.nhs.uk) and in genitourinary medicine clinics, reported via the genitourinary medicine clinic activity dataset (GUMCAD)<sup>6</sup> were collected by the HIV & STI department of the Health

Protection Agency. Questionnaire data and serum samples in the Health Survey for England (HSE) used in Chapter 8 were collected by NatCen Social Research in collaboration with University College London<sup>7</sup>. Serum samples were tested for anti-chlamydia antibodies by Imperial College London. I carried out the data linkage between the questionnaire data and antibody test results.

#### Acknowledgements

Through the course of this PhD I have had the privilege of working with many individuals and teams within UCL, Public Health England and beyond. I take this opportunity to thank all of my co-investigators, collaborators and colleagues for their insights, contributions and encouragement over the years.

My especial and heartfelt thanks go to my supervisors. Thank you to Kate Soldan for invaluable professional and personal guidance and astute epidemiological insight, to Cath Mercer for endless methodological expertise and unswerving moral support and to Anne Johnson for always asking the right questions. It has been an honour and a pleasure to work with and learn from you all.

Among my other collaborators, my particular thanks go to: Soazig Clifton, Clare Tanton, Andrew Copas and Pam Sonnenberg for expert advice on using and interpreting Natsal data; Anthony Nardone, Gillian Wills, Myra McClure and Paddy Horner for making it possible to study Pgp3 seroprevalence and for making it happen; Alan Darbin for his contributions to the pilot survey of chlamydia prevalence and to Nicola Low and Sheila Redmond for inviting me to be part of the systematic review team.

To my colleagues in the HIV & STI Department, I extend thanks for the neverending task of compiling and improving national STI surveillance data and for advice on its uses and misuses, especially to Stephen Duffell, Ali Talebi and Bersabeh Sile. Thank you to all of the genitourinary medicine clinic and chlamydia screening staff and service users for contributing your time and information. I also thank the staff, teams and study participants responsible for

the Natsal and HSE surveys for creating such unique and invaluable resources, in particular Rachel Craig and Cathy Coshall from NatCen Social Research.

My thanks go to all members of the National Chlamydia Screening Programme team - past and present - for their continued dedication to the field and to their support of, interest in and contributions to my work. In particular, Paula Baraitser, Kevin Dunbar, John Saunders, Kate Folkard, Angie Bone, Mary McIntosh, Carla Naidoo, Nastassya Chandra, Paula Blomquist, Virginia Walker and all of the Sexual Health Facilitator team.

My thanks go to the Health Protection Agency and Public Health England for funding to undertake the PhD including the pilot survey of chlamydia prevalence and to Noel Gill for departmental support and for always reminding me of the need for action.

Thanks also to Lizzi Torrone, Gwenda Hughes, Kathy Lowndes, Ian Simms and Katy Turner for engaging and enlightening discussions about chlamydia, STI epidemiology and life. I thank Sandy Gale for always finding time. Thank you to the PWG in all its glory and to Ellie, Vicky, Jesse, Ibi, Sarika, Meaghan, Adamma and David for sharing the journey. To Fran and Andrew, thank you for setting me on this path. To my friends and the rest of my family, thanks for your patience; I'll be out to play again soon. And finally, to Paul for believing in me and for keeping me supplied with Kinder Hippos.

#### Abstract

Genital infection with *Chlamydia trachomatis* ('chlamydia') is the most commonly diagnosed sexually transmitted infection in England. Chlamydia is often asymptomatic and can lead to serious complications, especially in women. Chlamydia screening offers one approach to controlling chlamydia and its consequences. In England, chlamydia screening is offered opportunistically to sexually-active under-25 year-olds through the National Chlamydia Screening Programme, which was introduced in 2003 and nationally implemented by 2008.

Evaluating the real-world impact of chlamydia screening against its aims of interrupting transmission and reducing the prevalence of infection presents a considerable challenge, in part due to the absence of a robust outcome measure.

The research presented in this thesis sought to address this challenge. Four approaches to outcome measurement were investigated:

- Analysis of trends in percentage testing positive for chlamydia among 15-24 year-olds accessing chlamydia testing using surveillance data;
- Pilot of a postal survey of 17-18 year-old women to measure population prevalence;
- Analysis of chlamydia prevalence among 16-24 year-old participants in the second and third National Surveys of Sexual Attitudes and Lifestyles (Natsal-2: 1999-2000; Natsal-3: 2010-12);
- Application of a novel antibody assay to stored sera from 16-44 year-old participants in the Health Survey for England (HSE) between 1994 and 2012 to measure prevalence of antibodies in serum as a marker of previous *C. trachomatis* infection.

In summary, no definitive evidence was found in these or other published analyses to suggest that chlamydia screening, as delivered in practice, has led to a reduction in the incidence or prevalence of chlamydia infection among young adults in England up to 2012. Possible reasons for the absence of such evidence are discussed in light of findings presented in the thesis.

The strengths and limitations of these approaches to outcome measurement are discussed, and recommendations regarding the future evaluation and delivery of chlamydia control programmes are presented.

### **Table of Contents**

ABSTRACT					
GLOSSARY OF TERMS					
Т	TABLE OF ABBREVIATIONS				
1	INT	RODUCTION	. 16		
	1.1	Aims	. 17		
	1.2	Structure of the thesis	. 18		
2	BAC	CKGROUND	. 20		
	21	What is chlamydia?	20		
	2.2	What is chlamydia screening?	. 23		
	2.3	Questioning the evidence base for the effectiveness of chlamydia screening	. 30		
	2.4	How should chlamydia screening be evaluated?	. 34		
	2.5	How has chlamydia screening been implemented in practice in England?	. 40		
	2.6	Summary	. 49		
2	RE\		51		
5			. 51		
	3.1	Methods	. 51		
	3.2	Systematic review of chlamydia prevalence studies using population-based sampli	ng		
	metho	ds: Results	. 54		
	3.3	Repeated, cross sectional estimates of chlamydia prevalence in the general	~ ~		
	popula		. 68		
	3.4	I rends in percentage testing positive for chlamydia using clinical and routinely-	70		
	collect	ed data	. 70		
	3.5	I rends in <i>C. trachomatis</i> antibody seroprevalence as a marker of age-specific	77		
			. //		
	3.0	Summary	. 00		
4	TRE	INDS IN PERCENTAGE TESTING POSITIVE FOR CHLAMYDIA USING			
R	OUTIN	ELY-COLLECTED SURVEILLANCE DATA	. 82		
	4.1	Background	. 82		
	4.2	Aims & objectives	. 84		
	4.3	Methods	. 85		
	4.4	Results	. 92		
	4.5	Discussion	115		
5	PIL	OT OF A POSTAL SURVEY DESIGNED TO MEASURE THE POPULATION			
P	REVAL	ENCE OF CHLAMYDIA AMONG YOUNG WOMEN	121		
	E 1	Aime & chiestives	101		
	5.1 5.2	Aims & Objectives	121		
	5.Z	Populto	122		
	5.5 5.4		132		
	5.4		150		
6 ©'		AMYDIA PREVALENCE MEASURED IN THE SECOND AND THIRD NATIONAL	111		
31		S OF SERVER ATTIODES AND LIFEST TES (NATSAL-Z AND NATSAL-3)	141		
	6.1	Background	141		
	6.2	Aims & objectives	142		
	6.3	Methods	143		
	б.4 о.г	Kesuits	160		
	6.5	UISCUSSION	173		

7 CHLAMYDIA PREVALENCE, DIAGNOSIS AND TESTING AMONG 16 TO 24 YEAR- OLD PARTICIPANTS IN THE THIRD NATIONAL SURVEY OF SEXUAL ATTITUDES AND LIFESTYLES (NATSAL-3)					
7.1	Background	. 178			
7.2	Aims & objectives	. 179			
7.3	Methods	. 179			
7.4	Results	. 182			
7.5	Discussion	. 198			
8 T AGE-S PART	RENDS IN <i>C. TRACHOMATIS</i> ANTIBODY SEROPREVALENCE, AS A MARKER C SPECIFIC CUMULATIVE INCIDENCE, MEASURED USING STORED SERA FROM ICIPANTS IN THE HEALTH SURVEY FOR ENGLAND (HSE) (1994 TO 2012)	)F . 203			
01	Packground	202			
8.2	Aims & objectives	203			
8.3	Methods	205			
8.4	Results	. 213			
8.5	Discussion	. 223			
9 S	UMMARY AND DISCUSSION	. 230			
9.1	Critical appraisal of outcome measures for the evaluation of the impact of chlamy	dia			
scre	ening on incidence and prevalence of infection	. 231			
9.2	Summary of recommendations for future evaluation of the impact of chlamydia co	ntrol			
prog o 3	grammes in England	. 242			
chla	mydia infection?	243			
9.4	Implications for chlamydia control policies	254			
9.5	Future research	. 258			
9.6	Concluding statement	. 261			
10	REFERENCES	. 263			
APPEI POPU	NDIX 1: STUDY PAPERWORK USED IN PILOT POSTAL SURVEY TO MEASURE LATION PREVALENCE OF CHLAMYDIA AMONG YOUNG WOMEN	THE . 282			
APPEI MODE TESTI	NDIX 2: COMPARISON OF MAXIMAL AND PARSIMONIOUS MULTIVARIABLE ELS OF PREVALENT CHLAMYDIA INFECTION, RECENT DIAGNOSIS AND RECE NG	NT . 290			
APPENDIX 3: FACTORS ASSOCIATED WITH PREVALENT INFECTION, RECENT DIAGNOSIS OR RECENT TESTING AMONG 16 TO 24 YEAR-OLDS RESIDENT IN					
ENGL	AND (NATSAL-3)	. 298			
APPE	NDIX 4: THESIS OUTPUTS	. 303			

## List of figures

Figure 2-1: Reduced risk of pelvic inflammatory disease (PID) within one year associated with a single offer of a chlamydia screen among women: results of four randomised controlled trials 25
Figure 2-2: Age specific impact of three screening strategies in those aged under 25 on chlamydia prevalence in women using the base case parameter set
Figure 2-3: Timeline showing implementation of chlamydia screening 40
Figure 2-4: Schematic showing the chlamydia screening pathway according to National Chlamydia Screening Programme recommendations
Figure 2-5: Number of test reported among 15-24 year-olds in NCSP and other non-GUM settings, 2000 to 2012
Figure 2-6: Number of diagnoses reported among 15-24 year-olds in all settings (GUM clinics, NCSP and other) 2000 to 2012 <sup>a</sup>
Figure 2-7: Proportion of genitourinary medicine (GUM) clinic attendees seen within 48 hours 49
Figure 3-1a: Forest plot, estimates of chlamydia prevalence in women ≤ 26 years in EU/EEA and other high-income countries
Figure 4-1: Numbers of testing episodes and percentage testing positive by gender, NCSP dataset and GUMCAD
Figure 4-2: Adjusted* odds ratio (and 95% confidence intervals) of testing positive by venue of test and gender, compared to General Practice (NCSP testing episodes, 15-24 year-old women and men)
Figure 4-3: Trends in reported characteristics (NCSP dataset, 15-24 year-old women and men, 2008-2011)
Figure 4-4: Percentage of NCSP tests carried out in different testing venue types (15 to 24 year- old women and men, 2008 to 2011)
Figure 4-5: Trends in reported characteristics, GUMCAD (15 to 24 year-old women and men, 2009 to 2011)
Figure 4-6: Percentage testing positive by year and gender, stratified by age group (a), sexual risk group (b), and venue of test (c) (NCSP dataset, 15 to 24 yea-old women and men, 2008 to 2011)
Figure 4-7: Percentage testing positive by year and gender, stratified by age group. (GUMCAD, 15 to 24 year-olds)
Figure 4-8: Numbers of tests and percentage testing positive by calendar quarter, dataset and location of testing (15-24 year-old women and men)
Figure 4-9: Trends in reported characteristics by calendar quarter, NCSP dataset (15 to 24 year-old women and men, 2008 to 2011)
Figure 4-10: Unadjusted and adjusted odds ratios (and 95%CI) of testing positive per additional year, estimated from univariable, bivariable and multivariable models (NCSP tests, 15 to 24 year-olds 2008 to 2011)
Figure 4-11: Unadjusted and adjusted odds ratios (and 95%CI) of testing positive per additional year, estimated from univariable, bivariable and multivariable models (GUMCAD tests, 15 to 24 year-olds 2009 to 2011)
Figure 5-1: Reported sexual behaviours among pilot survey participants, same aged Natsal-2 participants, HSE 2010 participants and NCSP test carried out in participating PCTs in 2011 133
Figure 5-2: Participation rates by IMD quintile and randomisation group 135
Figure 5-3: Reported reasons for non-participation (n=48) 136
Figure 6-1: Flow chart showing participants included (16 to 44 year-old Natsal-2 & Natsal-3) 147
Figure 6-2: Impact of 'diagnosis weights' on estimated proportion diagnosed in the last year by age group (Natsal-2)

Figure 6-3: Difference between observed and true prevalence with different test characteristics
Figure 6-4: Flowchart showing samples included in analysis of non-vaccine HPV prevalence (18-24 year-old sexually-experienced women and men)
Figure 6-5: Odds ratio of chlamydia prevalence in Natsal-3, compared to unadjusted and 'detection strategy adjusted' prevalence in Natsal-2 (sexually-experienced 18-24 year-old women and men)
Figure 6-6: Hypothetical scenario, showing odds ratio of chlamydia prevalence plus 'screen- detected' diagnosis in Natsal-3, compared to unadjusted prevalence in Natsal-2 (sexually experienced 18-24 year-old women and men)
Figure 6-7 Hypothetical odds ratio of prevalent infection in Natsal-3 versus Natsal-2 under different assumptions of unmeasured participation bias (sexually-experienced 18-24 year-olds) 168
Figure 6-8: Prevalence of non-vaccine HPV and selected sexual behaviours among sexually experienced 18-24 year-olds in Natsal-2 and Natsal-3
Figure 6-9: Proportion reporting having had heterosexual sex before the age of 16 by year turned 16 (Natsal-3)
Figure 7-1: Flow chart showing participants included in analyses of self-reported chlamydia testing, self-reported diagnosis and prevalent infection detected in urine (16 to 24 year-old Natsal-3 participants)
Figure 7-2a: Bubble plot showing unadjusted odds ratios for prevalent chlamydia infection versus recent testing by sociodemographic and behavioural factors, and percentage of prevalent infections in each group (sexually-experienced 16-24 year-old women)
Figure 7-3: Percentage with a prevalent chlamydia infection, ever diagnosed with chlamydia, recent diagnosis of chlamydia and recently tested for chlamydia by country of residence (sexually-experienced 16-24 year-olds, Natsal-3)
Figure 8-1: Flow chart showing selection of stored sera from Health Survey for England (HSE) participants
Figure 8-2: Pgp3 seroprevalence by age group (sexually-experienced 16 to 44 year-olds, HSE2010 & HSE2012)
Figure 8-3: Pgp3 seroprevalence by years since first heterosexual sex, (sexually-experienced 16 to 44 year-olds, HSE2010 & HSE2012)
Figure 8-4: Pgp3 seroprevalence by reported numbers of lifetime sexual partners (16 to 44 year-olds, HSE2010 & HSE2012)
Figure 8-5: Pgp3 seroprevalence by year (16 to 24 year-old women, HSE 1994 to 2012) 221
Figure 8-6: Pgp3 seroprevalence by year and age group (16 to 24 year-old women, HSE 2008 to 2012)
Figure 8-7: Pgp3 seroprevalence by birth cohort and year of age (16 to 24 year-old women, HSE 1994 to 2012)

## List of tables

Table 3-1a: Summary of characteristics of studies included in systematic review of chlamydia    prevalence, studies with national coverage    55
Table 4-1: Comparison of data items included in the NCSP dataset and the genitourinary medicine clinic activity dataset (GUMCAD) used in the analysis
Table 4-2: Proportion of GUM attendances with a chlamydia testing episode recorded, by gender (15 to 24 year-old women and men, GUMCAD, 2009-2011)
Table 4-3: Percentage of tests with missing data on one or more variable (NCSP dataset, 15-24 year-old women and men, 2008-2011)
Table 4-4 (a): Number of tests, percentage testing positive, unadjusted and adjusted odds ratios    for testing positive by sociodemographic, sexual behavioural, and test characteristics (NCSP dataset, 15-24 year-old women, 2008-2011)
Table 4-5 (a): Number of tests, percentage testing positive, univariable and multivariable logistic regression results by reported characteristics (GUMCAD, women aged 15-24, 2008 to 2011)100
Table 5-1: Opportunistic screening uptake, diagnosis rates and PCT characteristics in theparticipating PCTs, 2011/12
Table 5-2: Invitations sent by randomisation group    125
Table 5-3: Sexual behaviour variables available in the pilot survey and comparator samples 129
Table 5-4: Unit costs of consumables
Table 5-5: Participation rates by randomisation group  132
Table 5-6: Costs per invitation sent and sample received by randomisation group
Table 6-1: Key features of the second and third National Surveys of Sexual Attitudes and    Lifestyles    144
Table 6-2: Diagnosis weights by month of interview
Table 6-3: Comparison of results from LCx and AC2 from 506 urine specimens
Table 6-4: Prevalence of chlamydia (detected in urine), and self-reported testing and diagnosis    by age group and sex (sexually-experienced 16-44 year-olds, Natsal-2 and Natsal-3)
Table 6-5: Percentage with a prevalent chlamydia infection, ever diagnosed with chlamydia andrecently diagnosed with chlamydia by survey and odds ratio (OR) for difference betweensurveys (sexually-experienced 16-24 year-olds)163
Table 6-6: Hypothetical effect of unmeasured participation biases on estimates of difference in prevalence in Natsal-2 and Natsal-3 (sexually-experienced 18-24 year-olds)
Table 7-1: Reason and location of most recent chlamydia test, among those tested forchlamydia in the last year, by whether diagnosed in last year (16-24 year-old sexually-experienced women and men)
Table 7-2a: Percentage, unadjusted and adjusted odds (OR/AOR) ratios for prevalentchlamydia infection, self-reported diagnosis in the last year and self-reported testing bysociodemographic and behavioural factors (16-24 year-old sexually-experienced women,Natsal-3)
Table 7-3a: Distribution of demographic and behavioural characteristics among a) individuals with a prevalent chlamydia infection, b) individuals with a recent chlamydia diagnosis and c) the sexually-experienced population (16-24 year-old sexually-experienced women, Natsal-3) 194
Table 8-1: Comparative sensitivities and specificities (and 95%CI) of the Indirect Pgp3 ELISA,    double-antigen sandwich ELISA and commercial ELISAs    209
Table 8-2: Comparison of indirect Pgp3 ELISA and double-antigen Pgp3 ELISA by absorbance    values on indirect Pgp3 ELISA.    210
Table 8-3: Numbers of women aged 16 to 24 contributing to seroprevalence estimates by birth cohort, by year of age

Table 8-4: Comparison of reported sociodemographic and behavioural variables between HSEparticipants and the study population by sex and age group (16-44 year-olds, HSE2010 &HSE2012)
Table 8-5a: Percentage Pgp3 seropositive by sociodemographic and sexual behavioural variables, and unadjusted and lifetime sexual partners-adjusted odds ratios for detection of Pgp3 antibody in serum (sexually-experienced 16 to 44 year-old women, HSE2010 & HSE2012)
Table 9-1: Hypothetical scenario showing point estimates, 95%CI and p values for prevalenceratio of Pgp3 seroprevalence achieved with different sample sizes (16 to 24 year-old women,HSE2008 to HSE2012)239
Table 9-2: Percentage with a) prevalent chlamydia infection, b) self-reported diagnosis in thelast year and c) self-reported testing and unadjusted and adjusted odds ratios for prevalentchlamydia infection by sociodemographic and behavioural variables (16-24 year-old sexually-experienced women)292
Table 9-3: Percentage with a) prevalent chlamydia infection, b) self-reported diagnosis in the last year and c) self-reported testing and unadjusted and adjusted odds ratios for prevalent chlamydia infection by sociodemographic and behavioural variables (16-24 year-old sexually-experienced men)
Table 9-4a: Percentage, unadjusted and adjusted odds (OR/AOR) ratios for prevalent    chlamydia infection, self-reported diagnosis in the last year and self-reported testing by    sociodemographic and behavioural factors (16-24 year-old sexually-experienced women    resident in England Natsal-3)

## Glossary of terms

<i>Chlamydia trachomatis; C. trachomatis;</i> chlamydia	This thesis follows the convention for microbial nomenclature set out by Low <i>et al.</i> <sup>8</sup> <i>Chlamydia trachomatis</i> (abbreviated to <i>C. trachomatis</i> ) is used to describe the organism. 'Chlamydia' is the name of the condition, and is used to describe genital infection with <i>C. trachomatis</i> .
Percentage testing positive	The percentage of tests for current infection with <i>C. trachomatis</i> that return a positive result. Percentage testing positive is distinct from chlamydia prevalence, as the tests are not necessarily drawn from a representative sample of the general population.
Prevalence	The percentage of a defined population who have chlamydia at a given point in time.
<i>C. trachomatis</i> antibody/ Pgp3 seropositive	The percentage of sera tested where <i>C. trachomatis</i> or Pgp3 antibodies are detected.
<i>C. trachomatis</i> antibody/ Pgp3 seroprevalence	The percentage of a defined population who test positive for <i>C. trachomatis</i> or Pgp3 antibodies in serum.
Incidence	The rate at which chlamydia infections occur in a defined population during a specified period.
Cumulative incidence	The probability of having been infected with chlamydia by a given point (i.e. by a given age or number of years after first sex).
Diagnosis rate	The number of diagnoses per 100,000 population (generally applied to 15 to 24 year-olds)
Coverage	The number of chlamydia tests divided by the population (expressed as a percentage of a given age group)
Opportunistic screening	Screening offered to people at the time of attending healthcare or other specified venues. Not register-based.
Register-based screening	Screening offered systematically via active invitation of the eligible population in a given age/demographic group.
Sexually-experienced	Reports at least one sexual partner by the time of measurement/interview.
Lifetime sexual partners	Number of sexual partners by the time of measurement/interview.

## Table of abbreviations

AC2	Aptima-Combo 2 assay (Hologic Gen-Probe)
AOR	Adjusted odds ratio
CDC	Centers for Disease Prevention and Control
CI	Confidence interval
ClaSS	Chlamydia Screening Studies
CMO	Chief Medical Officer
CSI	Chlamydia Screening Implementation project
CSO	Chlamydia screening office
DFA	Direct fluorescent antibody assay
ECDC	European Centre for Disease Prevention and Control
EIA	Enzyme immunoassay
ELISA	Enzyme-linked immunosorbent assay
EU/EEA	European Union/European Economic Area
GP	General practice
GUM	Genitourinary medicine
GUMAMM	Genitourinary medicine access monthly monitoring
GUMCAD	Genitourinary medicine clinic activity dataset
HIV	Human immunodeficiency virus
HSE	Health Survey for England
HPA	Health Protection Agency
HPV	Human papillomavirus
lgG	Immunoglubulin G
IMD	Index of Multiple Deprivation
IPP	Infertility Prevention Program
LCx	Ligase chain reaction assay (Abbott Diagnostics)
LGV	Lymphogranuloma venereum
LSOA	Lower super output area
MSM	Men who have sex with men
MSW	Men who have sex with women
NAAT	Nucleic acid amplification test
NAO	National Audit Office
Natsal-2	2nd National Survey of Sexual Attitudes and Lifestyles
Natsal-3	3rd National Survey of Sexual Attitudes and Lifestyles
NCSP	National Chlamydia Screening Programme
NHANES	National Health and Nutrition Examination Survey
OR	Odds ratio
PCT	Primary care trust
PHE	Public Health England
PID	Pelvic inflammatory disease
$R_0$	Basic reproductive number
RCT	Randomised controlled trials
SRH	Sexual and reproductive health
STI	Sexually transmitted infection
UK	United Kingdom
US	United States
USPSTF	United States Preventive Services Task Force
VVS	Vulvovaginal swab

#### 1 Introduction

Genital infection with *Chlamydia trachomatis* ('chlamydia') is the most commonly diagnosed bacterial sexually transmitted infection (STI) in England.<sup>9</sup> Most chlamydia infections are asymptomatic<sup>10,11</sup> and infection can lead to serious reproductive sequelae in women.<sup>12-15</sup> In England, the National Chlamydia Screening Programme (NCSP) recommends that sexually active under-25 year-olds are screened for chlamydia annually and on change of sexual partner with the aim of reducing transmission and preventing future complications. The programme was introduced in England in 2003 and nationally implemented by 2008.

As set out in the following chapter, chlamydia screening is expected to lead to a reduction in the incidence and prevalence of infection. However, the effectiveness of screening in practice has not yet been shown and evidence from previous research is inconclusive. When the NCSP was established no strategy was put in place to monitor the impact of screening on health-related outcomes. The impact of chlamydia screening to date on either the incidence or prevalence of chlamydia is unknown.

One of the major challenges in evaluating the impact of chlamydia screening is the availability of a robust and valid outcome measure. As the majority of infections with *C. trachomatis* are asymptomatic, true incidence of chlamydia (i.e. the rate at which chlamydia infections occur in a defined population during a specified period) cannot be directly measured without incredibly intensive studies of large samples with frequent and repeated chlamydia testing. Such an approach is not feasible for evaluating the impact of chlamydia screening at a

national level over a long time period. Instead, the research presented in this thesis focusses on three outcome measures: the prevalence of chlamydia among the general population (hereafter termed 'prevalence'); the percentage testing positive for chlamydia among people accessing testing; and the prevalence of *C. trachomatis* antibodies detected in serum (hereafter termed '*C. trachomatis* antibody seroprevalence') as a measure of the percentage of the population who have had at least one *C. trachomatis* infection by a given age (hereafter termed 'age-specific cumulative incidence').

#### 1.1 Aims

The aims of this PhD are:

1. to identify and appraise outcome measures, and methods of their measurement, for the purpose of evaluating the impact of opportunistic chlamydia screening on the incidence and prevalence of infection (or related measures); and

2. using the outcome measures and methods identified in (1), to examine whether widespread opportunistic chlamydia screening, as it has been delivered in practice, has led to a reduction in the incidence or prevalence of chlamydia among young adults in England up to 2012 that would otherwise have been seen in the absence of opportunistic chlamydia screening.

The findings from this research can contribute to public health policy by producing recommendations on methods to evaluate the impact of chlamydia screening and on the future development of chlamydia control policies in England.

#### **1.2** Structure of the thesis

This thesis is based on original analyses of nationally-collected surveillance data, a pilot of a chlamydia prevalence survey involving primary data collection, original analyses of data from the second and third National Surveys of Sexual Attitudes and Lifestyles (Natsal), and the application of a novel *C. trachomatis* antibody test to stored sera from the Health Survey for England (HSE).

Chapter 2 sets out the background and rationale for the research. It includes an overview of chlamydia and chlamydia screening, describes limitations of the evidence base for the effectiveness of chlamydia screening and sets out the challenges of evaluating the impact of chlamydia screening on the incidence and prevalence of infection. Key features of how chlamydia screening has been implemented in practice in England are presented.

Chapter 3 presents a review of the literature around methods of measuring a) the prevalence of chlamydia in the general population, b) changes over time in the percentage testing positive for chlamydia among people accessing testing and c) *C. trachomatis* antibody seroprevalence as a marker of age-specific cumulative incidence. The key methodological issues around selection bias and confounding are highlighted and discussed.

Chapter 4 uses nationally-collated surveillance data from women and men tested for chlamydia to explore the use of trends in percentage testing positive for chlamydia as a proxy for changes in chlamydia prevalence in the general population over time.

Chapter 5 describes the design and implementation of a pilot survey to measure prevalence of chlamydia among 17 to 18 year-old women.

Chapter 6 presents analyses of data from the second and third National Surveys of Sexual Attitudes and Lifestyles, conducted in 1999-2000 (Natsal-2) and 2010-12 (Natsal-3). Age-specific prevalence of *C. trachomatis* detected in urine is compared between Natsal-2 and Natsal-3. Differences between the design of the surveys and context in which they were carried out are examined to assess changes in chlamydia prevalence in the decade between the surveys.

Chapter 7 uses data from Natsal-3 and presents a detailed analysis of the epidemiology of prevalent *C. trachomatis* infection in relation to reported testing to explore the extent to which opportunistic chlamydia screening up to 2012 was reaching groups at risk of chlamydia.

Chapter 8 presents a *C. trachomatis* antibody seroprevalence survey using a novel assay to investigate change in age-specific cumulative incidence over time during a period of increasing chlamydia screening.

Chapter 9 brings together the findings from the previous chapters to summarise the strengths and weaknesses of each method and to examine whether there is any evidence for there having been a decrease in chlamydia prevalence or agespecific cumulative incidence since the implementation of the NCSP, up to 2012. This chapter summarises the contribution made by this research by discussing the implications for future monitoring and evaluation, for chlamydia control strategies and for future research.

#### 2 Background

In this chapter I present an overview of the epidemiology and clinical impacts of genital infection with C. trachomatis and the putative role of chlamydia screening in its control. I discuss the limitations in available evidence for the effectiveness of chlamydia screening with particular reference to its impact on the incidence and prevalence of infection, thereby setting out the rationale for the thesis. I also provide a summary of how chlamydia screening has been implemented in practice in England over the last decade.

#### 2.1 What is chlamydia?

*C. trachomatis* is a bacterium, belonging to the genus *Chlamydia*. Serotypes A-C cause ocular infections; L1, L2, L3 and L2b are responsible for lymphogranuloma venereum (LGV) and serotypes D to K cause urogenital tract infections,<sup>16</sup> which are the focus of this thesis. Genital infection with *C. trachomatis* (hereafter termed 'chlamydia') is the most commonly diagnosed STI in the UK and elsewhere.<sup>9,17</sup> Untreated chlamydia can persist for several months or years<sup>18</sup>, and can cause a range of complications (see section 2.1.1). The acute symptoms of chlamydia infection include pain and abnormal discharge,<sup>19</sup> but a large proportion of people who are infected with *C. trachomatis* remain asymptomatic.<sup>10,11,20</sup>

Highly sensitive and specific nucleic acid amplification tests (NAATs) that detect the presence of *C. trachomatis* are available in most diagnostic laboratories in England, and can be performed on non-invasive samples (urine in men, selftaken vulvovaginal swabs or urine for women).<sup>19</sup> Chlamydia testing can therefore be offered in a range of clinical and non-clinical settings. Once detected, chlamydia is easily treated with antibiotics.<sup>19</sup> Systemic and local antibodies to *C. trachomatis* can be detected in those with a current or previous chlamydia infection.<sup>21,22</sup>

The prevalence of chlamydia is highest among young adults. In the third National Survey of Sexual Attitudes and Lifestyles (Natsal-3, carried out in 2010-12), the prevalence of *C. trachomatis* (detected in urine) in the sexually-active adult British general population was 1.5% (95% confidence interval [CI] 1.1%-2.0%) among women and 1.1% (95%CI 0.7%-1.6%) among men aged 16 to 44 years old; prevalence among 16 to 24 year-olds was 3.1% (95%CI 2.2%-4.3%) in women and 2.3% (95%CI 1.5%-3.4%) in men.<sup>23</sup>

#### 2.1.1 Sequelae and natural history of infection

Although largely asymptomatic, chlamydia presents a serious public health problem, as genital infection with *C. trachomatis* can cause several severe complications which are associated with losses of quality of life<sup>24-26</sup> and incur substantial healthcare costs.<sup>27-30</sup> In women, infection with *C. trachomatis* can ascend the genital and reproductive tract and lead to pelvic inflammatory disease (PID), a spectrum of clinical disorders involving inflammation of the uterus, fallopian tubes, ovaries, or adjacent peritoneum. PID can resolve without any damage caused to the reproductive tract. However, PID can lead to scarring and fibrosis in the pelvic organs, which in turn can lead to serious long-term reproductive consequences including tubal factor infertility and ectopic pregnancy.<sup>11,15,31-34</sup> The scarring and fibrosis of pelvic organs occur as a result of the immunological processes involved in response to chlamydia infection, although the exact biological mechanism(s) by which genital chlamydia infection, cause tissue damage are not fully understood.<sup>35</sup> In men, chlamydia

can cause epididymitis (swelling of one of the tubes in the testicles).<sup>11</sup> Babies born to mothers with chlamydia infection are at risk of neonatal conjunctivitis and pneumonia.<sup>11,36,37</sup>

There are considerable uncertainties concerning the natural history of chlamydia.<sup>38</sup> However available data suggest that in the region of 10% to 15% of untreated genital infections with *C. trachomatis* result in diagnosed clinical PID;<sup>14,38,39</sup> 10% to 15% of these cases may then lead to tubal factor infertility.<sup>38</sup> Progression rates from chlamydia to other outcomes are less well understood.<sup>38</sup> In an economic evaluation of chlamydia screening in England, it was estimated that 7.6% of women with symptomatic PID would progress to ectopic pregnancy; 14.8% of babies born to mothers with chlamydia would develop neonatal conjunctivitis, 7% would develop neonatal pneumonia and 2% of men with asymptomatic chlamydia would develop epididymitis.<sup>28</sup>

More recently, chlamydia has been suggested as a possible cause of adverse birth outcomes including pre-eclampsia, spontaneous preterm birth or stillbirth,<sup>40-42</sup> although study findings vary and further work is needed to fully understand this relationship.<sup>43</sup> Chlamydia may also increase the risk of disease arising from other sexually transmitted pathogens, through facilitation of human immunodeficiency virus (HIV) transmission<sup>44</sup> and increasing the persistence of high-risk human papillomavirus (HPV).<sup>45,46</sup>

Genital infection with *C. trachomatis* confers, at best, only partial immunity to subsequent infection.<sup>47</sup> Therefore re-infections are possible either from untreated or new sexual partners. Re-infection with chlamydia is common and those who test positive for chlamydia are at greater risk of testing positive at

subsequent tests than those who test negative. Studies have found that around 10% to 15% of young adults diagnosed with chlamydia also test positive at their next test<sup>48-57</sup> and that the percentage testing positive at a repeat test is around two to three times higher in those with an initial positive than in those with an initial negative test.<sup>48-55</sup> Repeat diagnoses may be due to re-infection due to incomplete treatment of sexual partner(s), re-infection due to continuing risk behaviour (i.e. unprotected sex with new or existing partners) or detection of a persistent infection due to incomplete or ineffective treatment. The extent to which treatment affects the development of protective immunity is unclear.<sup>47,58,59</sup>

#### 2.2 What is chlamydia screening?

One way of trying to control chlamydia and reduce the adverse consequences associated with infection is to screen people for current chlamydia infection. Screening is a process of identifying apparently healthy people who may be at increased risk of a disease or condition. They can then be offered information, further tests and appropriate treatment to reduce their risk and/or any complications arising from the disease or condition.<sup>60</sup> In the case of chlamydia screening, people diagnosed with chlamydia following asymptomatic testing can be offered treatment and advised that their sexual partners should also be screened and treated (hereafter termed 'partner notification').

Criteria for determining whether a disease or condition is a suitable target for screening were originally set out by Wilson and Junger in 1968.<sup>61</sup> These criteria remain the basis of definitions of screening, although they have been adapted and developed to be more applicable to modern public health practice, emerging technologies and to emphasise the need for evidence of

effectiveness, evaluation and quality assurance in any screening programme.<sup>60,62-64</sup> In summary, in order for a disease or condition to be considered a suitable target for screening, it should present an important public health problem, have a recognisable latent or early symptomatic stage and its natural history including development from early/latent period to active disease should be understood. There should be a precise and acceptable test and an accepted and effective treatment. The cost of screening should be economically-balanced in relation to medical expenditure as a whole and the benefits of screening should outweigh the harms.<sup>60-64</sup>

Chlamydia presents a potential target for screening as a significant public health problem, especially among young people, as it is the most commonly diagnosed STI in the UK and as untreated chlamydia infections can have serious long term consequences. Accurate and acceptable<sup>65</sup> tests are available and safe and effective antibiotics to treat chlamydia infection are available and included in clinical guidelines.<sup>19</sup> As the majority of chlamydia infections are asymptomatic, chlamydia screening should result in the diagnosis and treatment of infections that would otherwise go undiagnosed, or be detected later in the course of infection, thereby reducing the average duration of infection.

By reducing the duration of infection, chlamydia screening is expected to reduce an individual's risk of developing complications such as PID, ectopic pregnancy or tubal factor infertility.<sup>39,66</sup> The potential for chlamydia screening to interrupt the development of tubal pathology has been shown in four randomised controlled trials (RCT) that have investigated the effectiveness of a single offer of a chlamydia screen on the risk of developing PID within one year (Figure 2-1). A recent meta-analysis of these studies reported the pooled risk ratio of

all-cause PID after one year of follow-up for women invited to have a chlamydia screen to be 0.64 (95%CI 0.45-0.90). Uptake of screening in the intervention arm varied between 29% and 100% and the reduction in the risk of PID was greater in studies with higher rates of uptake of chlamydia screening.<sup>2</sup>

Figure 2-1: Reduced risk of pelvic inflammatory disease (PID) within one year associated with a single offer of a chlamydia screen among women: results of four randomised controlled trials



Adapted from European Centre for Disease Control and Prevention Report<sup>2</sup>

In contrast to other screening programmes such as screening for breast cancer or for cervical cancer, screening for chlamydia, as an infectious disease, includes a strong element of infection control through treatment of the infected individual and their sexual partner(s). Chlamydia screening is therefore expected to confer benefits at a population level by interrupting transmission, thereby reducing the incidence of infection. The logical basis for this can be seen using the epidemiological concept of the 'basic reproductive number', denoted as  $R_0$ .  $R_0$  is defined as the average number of secondary infections caused by an infected person in a totally susceptible population. For STI,  $R_0$  is dependent on three parameters, such that  $R_0=\beta cD$ , where  $\beta$  denotes the average probability that an infected individual will infect a susceptible partner over the duration of their relationship; c denotes the average number of new partners acquired per unit of time; and D the average duration of infection.<sup>69</sup> When  $R_0$  is greater than one, infection will spread through a population and the larger the value of  $R_0$ , the more quickly the infection will spread. Chlamydia screening, which is expected to reduce the average duration of infection (D), should therefore reduce  $R_0$  and hence the incidence of infection.<sup>70</sup> Chlamydia screening is also expected to lead to a fall in the prevalence of chlamydia, given the relationship prevalence=incidence x duration.<sup>71</sup>

#### 2.2.1 The National Chlamydia Screening Programme (NCSP) in England

Although the validity of some of the earlier RCTs of chlamydia screening and PID was later questioned (see section 2.3), the landmark trial by Scholes *et al* in 1996 and subsequent trial by Ostergaard *et al* in 2000 provided strong evidence in the late 1990s and early 2000s in support of chlamydia screening, having reported a significant reduction of ~50% in PID among those invited to screen compared to the control arm.<sup>1,67</sup> Observational data from Sweden and the USA were also considered to support the argument for chlamydia screening.<sup>72</sup> Increases in testing in women in Sweden had been found to correlate with a fall in the number of diagnoses made<sup>73</sup> and in the US, a before and after study found the percentage testing positive among women attending family planning clinics in Wisconsin to be lower after the implementation of a selective screening policy.<sup>74</sup>

In 1998, the Chief Medical Officer (CMO) in England commissioned an Expert Advisory Group to review the case for chlamydia screening, the report of which concluded that:

"Action is required to reduce the prevalence and morbidity associated with chlamydial infection. The sequelae of chlamydial infection are

severe and can have lifelong implications. There is evidence that the effective management of chlamydial infection will result in considerable health benefit."<sup>75</sup>

And that,

*"The evidence supports opportunistic screening of sexually active women aged under 25."*<sup>75</sup>

Following this report and two pilots of chlamydia screening in 1999,<sup>75-77</sup> the Department of Health in England announced the planned roll-out of a national screening programme for chlamydia in targeted groups (women attending genitourinary medicine (GUM) clinics, seeking termination of pregnancy or having their first cervical smear) from 2002, with a broader national programme to follow.<sup>76</sup> The National Chlamydia Screening Programme (NCSP) was implemented on a phased roll-out basis in 2003, with national implementation by March 2008. The NCSP remains in place to this date and recommends that sexually active women and men aged under 25 are tested annually and on change of sexual partner, with the aim of reducing the incidence and prevalence of chlamydia and its consequences.<sup>78</sup> Although the original recommendation from the CMO's Expert Advisory Group referred to asymptomatic screening in women, the NCSP included recommendations for opportunistic screening in men from the start of the programme. Men were included to highlight the role of both sexes in controlling onward transmission and in preventing reproductive complications in women and to increase both sexes' ability to take responsibility for their sexual health.79

The delivery of chlamydia screening in England is described in detail in section 2.5. Briefly, chlamydia screening is offered to under-25 year-olds when they attend a range of clinical and non-clinical venues. This approach differs to

register-based screening programmes, where the eligible population are actively invited and recalled on a regular basis. The NCSP is therefore defined as an opportunistic screening programme and is referred to as such throughout this thesis to distinguish this approach from register-based screening.

The expected impact of opportunistic chlamydia screening on the prevalence of chlamydia was explored using a mathematical model developed by the then Health Protection Agency (HPA)<sup>i</sup> in 2006.<sup>80,81</sup> Turner *et al* modelled scenarios based on offering chlamydia screening to women only or women and men, attending general practice (GP) settings. The model explored the impact of different assumptions including: the age group screened; the proportion of people who accepted screening; different rates of partner notification; the proportion of the eligible population who attended GP settings and the interval between offers of screening. The results from three of the modelled screening strategies, using the base case assumptions of the model are shown in Figure 2-2. The model results suggested that opportunistic chlamydia screening of women or women and men aged <25 years could at least halve the prevalence of chlamydia within ten years, providing that the healthcare settings offered screening to the entire eligible population when they attend, that 50% of those offered screening accept the invitation and that 20% of partners were treated.<sup>81</sup> As seen in Figure 2-2, decreases in chlamydia prevalence in women aged 16 to 19 and 20 to 24 were predicted within the first few years of implementation of chlamydia screening, with more gradual declines seen in later years. These findings again supported the expectation that chlamydia screening could have a beneficial impact on population health.

<sup>&</sup>lt;sup>i</sup> The HPA was incorporated into Public Health England (PHE) in April 2013.

## Figure 2-2: Age specific impact of three screening strategies in those aged under 25 on chlamydia prevalence in women using the base case parameter set.

Reproduced with kind permission from Sexually Transmitted Infections from Turner et al 2006.<sup>81</sup> Age-specific impact of screening strategies in those aged less than 25 years on chlamydia prevalence in women using the base case parameter set. The different figures show different screening strategies: A (offer annual screen to women); B (offer annual screen to women and if changed their partner in the past 6 months) and C (offer annual screen to women and men). Time (years) shows years from implementation of hypothetical screening scenarios.



Several countries in Europe, North America and Australasia recommend asymptomatic testing for chlamydia among groups considered to have a higher risk of chlamydia. This includes young people (e.g. aged <25), men who have sex with men (MSM), pregnant women, and women undergoing abortion.<sup>82</sup> Scotland and Wales do not have an organised screening programme, but guidelines recommend opportunistic testing for chlamydia among young adults<sup>82</sup>, with a focus on those at high risk of infection, including individuals attending GUM clinics, those who have been previously diagnosed with chlamydia, or those who have a partner who has been diagnosed with chlamydia, and those with two or more sexual partners in the last year.<sup>83</sup> In the US, the United States Preventive Services Task Force (USPSTF) and the Centers for Disease Prevention and Control (CDC) recommend testing in asymptomatic young women and older women at increased risk.<sup>84</sup> However England is the only country with an organised – albeit opportunistic - chlamydia screening programme.<sup>82 ii</sup>

# 2.3 Questioning the evidence base for the effectiveness of chlamydia screening

As set out above, there is a strong logical basis for chlamydia screening and evidence from RCTs and mathematical modelling support the notion that chlamydia screening can benefit population health. However, the evidence base relating to the effectiveness of chlamydia screening is subject to some important limitations and has come under scrutiny in recent years.

While evidence from the available RCTs and observational data was used to support chlamydia screening in the early 2000s, from around 2006, the validity of evidence on the effectiveness of chlamydia screening began to be questioned in the public health literature.<sup>77,85,86</sup> It is now thought that HIV prevention messages in the late 1980s and early 1990s led to a reduction in sexual risk behaviour, which in turn reduced STI transmission during this period. It is therefore likely that the role of chlamydia screening in observational studies had been over-emphasised.<sup>86</sup> Re-appraisal of the early RCTs by Scholes *et al* and Ostergaard *et al* highlighted some important methodological issues.<sup>85</sup> In the Scholes study, more effort was made to invite women in the screening group to

<sup>&</sup>lt;sup>ii</sup> As defined by criteria set out by the European Centre for Disease Prevention and Control (ECDC), who define an organised screening programme as one *"that offers regular chlamydia screening to asymptomatic individuals in a well-defined target population. People found to be infected are managed according to guidelines for treatment and partner notification services. [The country also has] primary prevention activities<sup>82</sup>* 

take part, and they were followed up more rigorously than controls.<sup>1,85</sup> In the Ostergaard study, participants were randomised before they had consented to take part, almost half of the participants did not provide information at follow up, and assessment of whether someone had PID or not at follow up was not blinded.<sup>67,85</sup> Thus the effect of chlamydia screening may have been over- or underestimated in these studies.

Furthermore, more recent studies found smaller and non statistically-significant effect sizes. Oakeshott et al reported a 35% reduction in risk of PID at one year (risk ratio [RR] 0.65, 95%CI 0.34-1.24)<sup>39</sup> and Andersen et al reported an 11% reduction (RR 0.89, 95%CI 0.56-1.42). The meaning of these smaller effect sizes is subject to some uncertainty. In the well-conducted study by Oakeshott et al, around one fifth of women in both the intervention and control arms were tested outside of the study between the time of enrolment and follow up<sup>39</sup> and 9% of women in the Andersen et al study were tested in the first three months of the study.<sup>68</sup> This would have biased both studies towards a smaller effect size. Andersen et al used prescription information to measure cases of PID in community settings.<sup>68</sup> This means it is likely that a lot of cases of PID will have been missed,<sup>68,87</sup> which adds further uncertainty to the findings from this study. However, the magnitude of the effect seen in these two studies was substantially smaller than that reported in the two earlier RCTs by Scholes and Ostergaard that found ~50% reduction in risk of PID within one year of screening.

The effectiveness of chlamydia screening was further called into question with the publication of a cluster RCT of register-based chlamydia screening in the Netherlands in 2010. In the Chlamydia Screening Implementation (CSI)

project,<sup>88</sup> 16 to 29 year-old women and men living in three areas of the Netherlands were offered chlamydia screening annually for three years. Postal invitations were used to offer screening. Those who accepted used an internet site to request a home sampling kit, which was then posted to a laboratory for testing. The study was conducted with stepped-wedge implementation, with those in the first phase being sent three yearly invitations, and the final group participating only in the final round of screening. The study found no statistically significant reduction in the percentage testing positive for chlamydia among those tested or in estimated prevalence and no difference between areas that had participated in all three rounds of screening versus those in the comparison group who had been offered screening only once. This study had lower uptake than expected (16% in round one, decreasing to 10% in round three), which may have limited the impact of screening on transmission. However, the potential for chlamydia screening to reduce the incidence or prevalence of infection had yet to be demonstrated in practice.

The implications of the findings from these RCTs for the expected impact of chlamydia screening on population health in England are unclear. There are some important differences between the study interventions or populations which limit the generalisability of findings from the RCTs. None of the RCTs of chlamydia screening and PID investigated the impact of repeated offers of chlamydia screening and there are differences in the age groups, genders, and risk groups targeted. While the RCT in the Netherlands did investigate the impact of three rounds of annual offers of screening, the intervention was delivered using a registry to send invitations, which differs to the opportunistic approach in England and uptake of screening in the Netherlands was not comparable to that in England.<sup>88</sup>

At the outset of the NCSP, surveillance systems were established to collect data on numbers of tests and diagnoses among the target population tested through the NCSP (see Chapter 4). However, no system was established to monitor the health outcomes of the programme in terms of the incidence or prevalence of chlamydia or the incidence of chlamydia-related sequelae. No baseline survey of chlamydia prevalence was carried out at the start of the NCSP and there was no clear strategy as to how the programme would be shown to have delivered (or not) against its expected objectives.<sup>89</sup>

Thus ten years after widespread opportunistic chlamydia screening among young adults had been recommended in England,<sup>75</sup> there remained uncertainty about the expected or actual impact of the NCSP on population health. This uncertainty in the evidence base for chlamydia screening as it had been delivered in practice in England was epitomised in a National Audit Office (NAO) review of chlamydia screening among young people,<sup>89</sup> which was carried out in 2009, six years after the introduction of the NCSP. The NAO concluded that:

"A good understanding of two key aspects of chlamydia – the prevalence of the infection in the general population of young adults, and the probability of chlamydia leading to severe health complications – are crucial to any assessment of the Programme's impact and its cost-effectiveness. The scientific evidence in both these areas is limited and the interpretation of the existing data is subject to debate. Some of the studies which have been carried out since the Programme's launch have not strengthened the case for testing." <sup>89</sup>

The research presented in this thesis was undertaken in response to this evidence gap.

#### 2.4 How should chlamydia screening be evaluated?

There is a large body of literature on methods of evaluating public health and other policy interventions.<sup>90-97</sup> One of the key pieces of work in this area in the UK in recent years has been the publication and development of the Medical Research Council Framework on the Development and Evaluation of Complex Public Health Interventions.<sup>94</sup> The research presented in this thesis was not carried out within a specific theoretical evaluation framework. However, there are several key features of the literature on evaluation that are of particular relevance. Specifically: the choice of outcome measures, the role of non-experimental evaluation and the importance of process evaluation, each of which is dealt with in more detail below.

#### 2.4.1 Choice of outcome measure

Choosing a suitable outcome measure is crucial to impact evaluation of public health interventions. The choice of outcome measure should be guided by a theoretical understanding of the expected impact of an intervention. All public health programmes will (either implicitly or explicitly) be underpinned by a theory of change, which can direct the evaluator to the target outcomes of interest.<sup>90,98</sup>

As set out above, chlamydia screening is expected to have an impact on a number of health outcomes, specifically on the incidence and prevalence of chlamydia infection and the incidence of chlamydia-related complications such as PID and ectopic pregnancy. However achieving robust measures of these outcomes is a major challenge.

As chlamydia is largely asymptomatic, case reports per year do not equate to the annual incidence of infection. Nonetheless, measurement of the incidence of chlamydia infection has been attempted. For example, Lamontagne *et al* in their 2007 study of women attending GP, GUM and family planning clinics attempted to measure incidence of infection by inviting women to be tested for chlamydia and then re-tested either six (for those who originally tested negative) or three (for those who originally tested positive) months later. While the authors reported incidence rates per person year calculated on the basis of infections measured at follow-up, they could not take account of infections that had been acquired and cleared in between measurements.<sup>51</sup> Direct measurement of the true incidence of chlamydia is not possible without incredibly intensive studies of large samples with frequent and repeated chlamydia testing. Such an approach is not feasible for impact evaluation of chlamydia screening at a national level over a long time period, and is not, therefore, explored any further in this thesis.

The prevalence of infection presents another obvious target outcome measure. As described above, chlamydia screening is expected to reduce the prevalence of infection both by reducing the average duration of infection and by interrupting transmission. However, as described further in the next chapter, measures of chlamydia prevalence in the general population are hard to achieve. The percentage testing positive for chlamydia among people accessing testing or screening will not equate directly to population prevalence if the population accessing testing is not representative of the general population. Measures of population prevalence from surveys require robust sampling frames and ways of protecting against or dealing with non-response. This thesis explores the use of percentage testing positive (Chapter 4) and

population prevalence measured in cross-sectional surveys (Chapters 5 to 7) as outcome measures for the evaluation of screening.

Measuring *C. trachomatis* antibodies in serum has been proposed as an alternative approach for evaluating the impact of chlamydia screening.<sup>99</sup> The presence of antibodies to *C. trachomatis* in serum indicate that someone has been previously infected, even if they have been treated, or the infection has cleared on its own. The use of antibody seroprevalence as an outcome measure for evaluating chlamydia control programmes is promising for a number of reasons. Firstly, using stored or retrospectively collected blood samples that have been collected for a purpose other than chlamydia testing may avoid the bias associated with data from populations who are seeking or accepting chlamydia testing. Secondly, chlamydia antibodies persist and thus provide a longer-term marker of age-specific cumulative incidence.<sup>100</sup> Thirdly, as age-specific cumulative incidence will by definition be higher than the prevalence, smaller sample sizes should be needed to monitor change over time. The use of *C. trachomatis* antibody seroprevalence as a marker of age-specific cumulative incidence is the subject of Chapter 8.

Chlamydia-related outcomes such as rates of PID, ectopic pregnancy and tubal factor infertility are all also potential outcome measures for evaluation. As with infection-related measures, the use of these conditions as outcome measures is problematic as they are difficult to measure consistently across sites and over time,<sup>101,102</sup> they all have causes other than chlamydia<sup>38</sup> and infection doesn't necessarily lead result in these adverse outcomes. Ethical considerations of allowing a diagnosed infection to remain untreated also make it challenging to establish the incidence of complications following untreated compared to
treated infections.<sup>2</sup> Ectopic pregnancy and tubal factor infertility are rare outcomes that may be diagnosed a considerable time after infection, thus studies aimed at measuring these complications would require large sample sizes and long follow up times. Investigation of the impact of screening on sequelae, while warranted, is therefore beyond the remit of this thesis. Similarly, the thesis does not include an economic evaluation, since an understanding of the costs, benefits and harms of screening warrant detailed investigation in their own right and is beyond the scope of this thesis.

#### 2.4.2 The role of non-experimental approaches to evaluation

Another important consideration in evaluation is how to establish causal associations between any change in an outcome measure of interest and the intervention under evaluation. RCTs are considered the gold standard of evidence for investigating causal associations on specified outcomes among a defined population.<sup>103</sup> However the role of RCTs in the evaluation of public health interventions has received considerable attention in recent years,<sup>96,104,105</sup> and they are not feasible or desirable in all circumstances. As set out by Bonell and colleagues in their summary of a multi-disciplinary symposium held in 2006, interventions that have already been delivered across an entire area are not conducive to RCT methodology as it is not possible to establish a control group.<sup>103,106</sup> This is the case with chlamydia screening in England, as the NCSP has been operational across all areas of England since 2008. Thus estimating the effect of chlamydia screening in England via a RCT would not now be feasible without withdrawing existing service provision in order to establish a true control group.

Parts of the chlamydia screening pathway would be amenable to RCT methodology, for example comparing screening intervals, and the effectiveness of interventions to increase or target uptake in certain groups.<sup>107</sup> However the overall question of whether the NCSP has had a beneficial effect on population health, can no longer be answered through an RCT in England. Alternative, non-experimental approaches are therefore required. In this thesis I use several novel approaches to outcome measurement and identify the strengths and weaknesses of each. I then draw together the findings from these different approaches to evaluate what they, together, can tell us about whether chlamydia screening has reduced either the incidence or prevalence of chlamydia among young adults up to 2012.

There are other approaches to evaluating the impact of chlamydia screening on incidence and prevalence that could have been explored. Mathematical models of *C. trachomatis* transmission can help understand the potential impact of chlamydia screening.<sup>108</sup> The ability of such models to recreate actual transmission events over time is limited by the available data on chlamydia prevalence at baseline, tests, diagnoses, sexual behaviours and transmission dynamics over the last decade which are required for parameterisation and fitting, all of which are subject to considerable gaps<sup>109</sup> or uncertainty.<sup>108</sup> Mathematical modelling may be complementary to evaluation based on empirical outcome measurement by exploring the potential impact of screening frequency or target populations. However, this approach was beyond the remit of this thesis.

### 2.4.3 The importance of process evaluation

The importance of process evaluation has received more attention in the public health literature in recent years, in recognition of the multifactorial aetiology of many public health problems and of the multiple components of interventions.<sup>110</sup> Post-implementation process evaluation of an intervention includes a description of what was actually delivered, the interaction between the intervention and the people targeted and an appreciation of the context in which an intervention was implemented, in order to understand whether what was done would be expected to have had an impact given the theoretical understanding of mechanisms of action.<sup>90,111</sup>

The potential impact of chlamydia screening on incidence and prevalence will vary according to uptake of testing in different populations, the rates of treatment and testing and the index case and their sexual partner(s). In order to move beyond the generic question of 'does chlamydia screening work', we therefore need to understand what is meant by 'chlamydia screening' in the specific evaluation context. Section 2.5 below therefore describes how chlamydia screening has been delivered in practice in England via the NCSP since its implementation, thus providing the context in which the impact of chlamydia screening will be evaluated. Additionally, the analysis presented in Chapter 7 examines whether chlamydia testing and screening reported in 2010-12 was reaching 16 to 24 year-olds at risk of prevalent infection, thus exploring whether chlamydia screening has been implemented in such a way that we would expect there to have been an impact on transmission.

### 2.5 How has chlamydia screening been implemented in practice in England?

Figure 2-3 shows some of the key milestones and developments in the implementation of chlamydia screening in England. The way in which the NCSP has been implemented and delivered has changed substantially since the start of the programme in 2003. This variation has been driven by several factors, including local approaches to commissioning and service provision;<sup>112</sup> technological developments;<sup>113</sup> the development and constant review of national guidance on screening;<sup>114</sup> changes in the political and economic context in which screening is delivered and the re-organisation of health and public health services.<sup>115</sup> Some of the main features of chlamydia screening as delivered in practice, and changes in the implementation of chlamydia screening over time are described below.

Figure 2-3: Timeline showing implementation of chlamydia screening



\*Adapted from National Audit Office Report<sup>89</sup>

The NCSP recommends that sexually-active under 25 year-old women and men are tested for chlamydia annually and on change of sexual partner (Figure 2-4). Chlamydia screening in England is offered opportunistically to under-25 year-old women and men when they attend a range of settings. These include specialist sexual health services (GUM clinics, sexual and reproductive health services, and abortion services) where young adults may be attending for sexual health-related reasons. Screening is also offered in non-specialist, clinical settings such as GPs and pharmacies. The availability of testing in non-clinical settings varies by area, but includes testing offered in schools, colleges and universities, and in bars, pubs and clubs. In several areas it is also possible to order home sampling kits online, where samples are taken at home and then returned back to a laboratory for testing.<sup>113</sup> The relative contribution to tests and diagnoses made by each of these settings, and the percentage testing positive in each, varies. This is explored in more detail in Chapter 4.

Young adults who test positive for chlamydia should be provided with timely antibiotic treatment. Azithromycin (1g) and doxycycline (100mg twice a day for 7 days) are both recommended as first-line treatments.<sup>19</sup> In the first full year of the NCSP (2008-09), 88% of people who tested positive through the programme<sup>iii</sup> were recorded as having received treatment.<sup>89</sup> The proportion of cases that has been treated with azithromycin versus doxycycline or whether this has changed over time is unknown, although anecdotal evidence suggests 1g azithromycin is more widely used for asymptomatic genital infections with *C. trachomatis* as compliance with treatment is considered more likely (personal communication, Paula Baraitser).

<sup>&</sup>lt;sup>III</sup> Not including GUM clinics

### Figure 2-4: Schematic showing the chlamydia screening pathway according to National Chlamydia Screening Programme recommendations



Partner notification is the process of informing and treating sexual partners of individuals who have been diagnosed with STIs. Partner notification is an essential element of chlamydia control and confers benefit by reducing risk of re-infection and its adverse consequences in the original patient, preventing onward transmission of infection by infected sexual partner(s), and reducing risk of complications in infected sexual partner(s).<sup>116,117</sup> For asymptomatic chlamydia infection, an arbitrary period of six months (or to date of most recent sexual partner, whichever is longest) is used as a cut-off to determine which sexual partners should be followed up as contacts. Achieving high partner notification rates is notoriously challenging, and rates vary substantially by

clinic.<sup>118</sup> Up to 2012,<sup>iv</sup> the NCSP monitored partner notification performance against a standard of 0.4 partners treated per index case in a large city and 0.6 elsewhere<sup>89</sup>. In 2008-09 almost three-quarters of areas delivering chlamydia screening did not meet these standards.<sup>89</sup> Although rates of treatment and of partner notification are both subject to underreporting, the available data described above suggest that the screening pathway has not always been delivered in an optimal fashion, leaving infected individuals at risk of persistent or repeat infections.

The frequency of screening and timing of testing in relation to time since infection are also likely to affect the impact of chlamydia screening. The NCSP recommends that young adults are tested annually and on change of sexual partner, as young adults continue to be at risk of new and repeat infections after having been tested.<sup>51</sup> The proportion of young adults who get tested every year or upon change of sexual partner is unknown, as routinely collected data do not include a unique personal identifier, and to date this has not been investigated in national surveys. Given the frequency of chlamydia re-infections (section 2.1.1), following an evidence review and consultation with professionals and young people, the NCSP amended their case management guidelines in 2013 to recommend that those who test positive for chlamydia should be advised to re-test around 3 months after completing treatment.<sup>119</sup> An analysis of surveillance data from 2010 by Woodhall *et al* found the incidence rate of re-testing to be 18.4 and 26.1 per 100 person years in 15 to 24 year-olds testing

<sup>&</sup>lt;sup>iv</sup> National monitoring of treatment and partner notification for tests carried out as part of the NCSP was discontinued in 2012, as central data collection was deemed to place an unnecessary burden on local areas at a time when chlamydia screening offices were being integrated into sexual health services. Local areas are advised to monitor treatment and partner notification rates locally.

via the NCSP and in GUM clinics respectively.<sup>120</sup> Rates of re-testing before this period are not known.

## 2.5.1 Changes in performance management, public health indicators and standards

From the time when the NCSP had been nationally implemented in 2008, local areas were monitored against national targets for chlamydia screening coverage, defined as the proportion of the population tested each year. These 'vital signs' indicators were set by the Department of Health for 2008-09 to 2010-11 (Figure 2-3). The target for local areas increased from 17% of all 15 to 24 year-olds tested per year in 2008/09 to 35% in 2010/11 (tests performed in GUM clinics did not initially contribute towards this target, although were latterly included).<sup>89</sup> The introduction and implementation of these targets led to a step change in the number of tests and diagnoses reported in each year (Figure 2-5; Figure 2-6). In Natsal-3 (carried out in 2010-12) 57.1% of 16 to 24 year-old women and 37.3% of men living in England who reported at least one sexual partner over their lifetime reported having been tested for chlamydia in the last year.<sup>23</sup>

After 2011, following a change of government, the national target for screening coverage was removed. Chlamydia control remained a public health priority, and an indicator relating to chlamydia screening was included in the 2013 to 2016 Public Health Outcomes Framework for England.<sup>121</sup> Since 2013, the Department of Health has recommended that local areas aim to achieve a

diagnosis rate<sup>v</sup> (defined as the number of chlamydia diagnoses per head of the 15 to 24 year-old population) of 2,300 per 100,000 or higher. The change in focus from coverage to a diagnosis rate indicator for programme monitoring was partly because the coverage target had driven high volumes of testing in relatively low risk populations resulting. The diagnosis rate indicator was considered a more appropriate measure of programme performance as treating infections is thought to reduce subsequent ill health and as a higher diagnostic rate would be expected to lead to greater reductions in chlamydia prevalence.<sup>121</sup> As with the use of a coverage target there are some potential limitations of using a diagnosis rate to monitor programme performance. Diagnosis rate will depend on volumes of testing, who is being tested as well as the underlying prevalence in each area. It is therefore feasible that areas with high underlying prevalence in the population may be able to achieve the specified diagnosis rate of 2,300/100,000 more easily than those who have a low underlying prevalence of infection. In areas of particularly high or low prevalence this may create a perverse incentive where screening activity required to reach a specified diagnosis rate would be inverse to sexual health need. There is no evidence at present to support local geographical variation of population prevalence, so the extent to which diagnosis rate may operate as a perverse incentive is unclear. However local authorities are encouraged to consider local needs across sexual health when commissioning services to rather than consider this as a fixed target.<sup>121</sup>

<sup>&</sup>lt;sup>v</sup> The name of this indicator changed in 2014, and is now referred to as the 'detection rate'. For the purposes of consistency, 'diagnosis rate' is used throughout this thesis.

Since 2010, the NCSP has also encouraged the integration of chlamydia screening services into sexual health services for young adults.<sup>122</sup> In addition to continued provision in specialist GUM clinics, local areas are advised to focus on provision in general practice, community sexual and reproductive health services, pharmacies, termination of pregnancy clinics. Individuals tested in these services tend to have a moderate to high risk of chlamydia.<sup>20,123,124</sup> These settings present sustainable options for screening in services that can address other aspects of sexual health including contraception and other STI testing and interventions.<sup>124</sup>

## Figure 2-5: Number of test reported among 15-24 year-olds in NCSP and other non-GUM settings, 2000 to 2012



GUM: genitourinary medicine. NCSP: National Chlamydia Screening Programme. \*Tests reported from non-NCSP, non-GUM clinics were reported from April 2008 onwards only. Tests were not reported by age group in GUM clinics until 2009 so are not included. Tests reported of unknown age have been re-allocated according to year- and gender-specific distributions.

### Figure 2-6: Number of diagnoses reported among 15-24 year-olds in all settings (GUM clinics, NCSP and other) 2000 to 2012<sup>a</sup>



GUM: genitourinary medicine. NCSP: National Chlamydia Screening Programme. <sup>a</sup>Diagnoses reported from non-NCSP, non-GUM clinics were collected from 2008 onwards. Diagnoses reported of unknown age have been re-allocated according to year- and genderspecific distributions.

### 2.5.2 Changes in the context of chlamydia screening delivery

As shown in Figure 2-5 and Figure 2-6 chlamydia testing of young adults has increased substantially in England over the last decade. In evaluating the role of chlamydia screening on changes in prevalence over this period, the role of other interventions should also be considered. Along with increases driven by the national scale-up of the NCSP increases in testing occurred in GUM clinics as a result of improved access to sexual health services<sup>125-127</sup> and increased availability of diagnostic testing using non-invasive samples (urine or vulvovaginal swabs).<sup>128</sup>

Following the introduction of waiting time targets for patients attending a GUM clinic in 2004,<sup>76,129,130</sup> the proportion of GUM attenders who were seen within 48 hours increased from 38% in May 2004 to 87% in February 2009 (Figure 2-7). Reduced waiting times were achieved, to some extent, through increased capacity and the expansion of the role of primary care in sexual health service provision.<sup>130</sup> Numbers of chlamydia tests and diagnoses reported in GUM clinics increased from at least 2004 onwards (when data on tests in GUM clinics are first available), which likely indicates increased attendance as well as increased testing among attenders over this period. Findings from Natsal-2 and Natsal-3 demonstrate an increase in attendances at GUM clinics over this period: the percentage of 16 to 44 year-old women with at least one lifetime sexual partner who reported attendance at a GUM clinic in the past 5 years increased from 7% (95%CI 6%-8%) in Natsal-2 (carried out in 2000) to 21% in Natsal-3 (95%CI 20%-23%); and among men reported attendance increased from 8% (95%CI 7%-9%) to 20% (95%CI 18%-21%)<sup>23</sup>. This increase was seen across all age groups, but most notably among 16 to 24 year-olds.

### Figure 2-7: Proportion of genitourinary medicine (GUM) clinic attendees seen within 48 hours

Between May 2004 and August 2007, data were collected as part of the HPA GUM Waiting times audit, which was performed quarterly and based on questionnaire provided to attendees in a 1 week period<sup>125</sup>. From 2009 onwards, data are derived from the 48 Hour Genitourinary Medicine Access Monthly Monitoring (GUMAMM) dataset<sup>131</sup>. Data were reported on a monthly basis, but the corresponding months are presented here for comparison purposes. Data were not collected between November 2007 and December 2009. Data collection was discontinued after 2011.



#### 2.6 Summary

In summary, chlamydia presents an important public health problem in England. The implementation of the NCSP and increased access to sexual health services led to substantial increases in chlamydia testing in the decade up to 2012, with sustained rates of diagnoses among 15 to 24 year-olds since 2008. The way in which screening has been delivered has changed considerably during the course of the programe. Despite this moving target, there is a need to evaluate the impact of chlamydia screening - as it has been delivered in practice - against its objectives of reducing the incidence and prevalence of infection. However at the outset of the NCSP there was no system established to do this and obtaining robust measures of incidence, prevalence or related measures is challenging.

In the next chapter the challenge of outcome measurement is addressed in more detail through a review of the relevant literature.

### 3 Review of the literature

In the previous chapter I summarised the evidence and theoretical basis for a chlamydia screening programme. I set out the need and rationale for evaluating the impact of the NCSP on outcomes relating to the incidence and prevalence of infection. In this chapter I provide a summary of studies that have estimated chlamydia prevalence among general population samples or that have estimated changes over time in either chlamydia prevalence, percentage testing positive or antibody seroprevalence as a marker of age-specific cumulative incidence. Methodological considerations arising from these studies are discussed, which provide the basis by which the methods of outcome measurement presented in chapters 4 to 8 are appraised.

#### 3.1 Methods

## 3.1.1 Chlamydia prevalence studies using population-based sampling methods

Studies of chlamydia prevalence measured among general population samples were identified as part of a systematic review of chlamydia prevalence studies, which was carried out as part of a collaborative project on chlamydia control in Europe<sup>2</sup>, the full methodology and results of which have been reported (Appendix 4).<sup>3</sup>

Cross-sectional surveys that used population-based sampling and tested genital specimens from adult participants for *C. trachomatis* were eligible for inclusion. Population-based sampling was defined as studies that used a sampling frame and method that are potentially representative of the resident population of a

particular area. This includes surveys based on household sampling and/or population registers and GP registers (although the limitations of GP registers are discussed in section 3.2.2.2 below). Studies of populations attending clinical settings who were tested for chlamydia as part of routine clinical care were identified but not included.

The review focussed on studies of adults living in European Union (EU)/European Economic Area (EEA) Member States. Studies from other highincome countries (as defined by the Organisation for Economic Cooperation and Development) were also eligible for inclusion. Electronic databases (Ovid Medline, EMBASE, Popline and The Cochrane Library) were searched from January 1990 to 17 October 2011. Search strategies included terms for 'chlamydia infection' and 'prevalence' and names of the eligible countries. The titles and abstracts were screened by two reviewers (Redmond, Alexander-Kissling) and assessed against the predefined inclusion criteria. Two reviewers (Redmond and Woodhall or Redmond and Alexander-Kissling) extracted data independently in duplicate onto standardised data collection forms, compared the extracted data for each paper and resolved differences where necessary. Discrepancies were adjudicated by a third reviewer (Low). Study characteristics, numbers eligible, numbers tested and with C. trachomatis detected and estimated prevalence and 95% confidence intervals were extracted from each paper. Chlamydia prevalence estimates were compiled by sex, age group, country (EU/EEA or non EU-EEA), whether the study was conducted at a national or sub-national level and whether prevalence was estimated among the total population or limited to those who reported at least one sexual partner by the time of the survey (hereafter termed 'sexually-experienced'). Chlamydia prevalence and 95% confidence intervals were estimated using the number of

positive tests divided by the number of people tested unless stratified sampling methods were used in which case the estimates presented in the papers were used.<sup>3</sup>

The results of Natsal-3 were published after completion of the systematic review and therefore fell outside of the review's eligibility criteria. Results from Natsal-3 are included in the text and table below, but not in the meta-analyses. The systematic review also incorporated a meta-analysis to pool chlamydia prevalence estimates where appropriate and a meta-regression to examine the association between estimated chlamydia prevalence in  $\leq$ 25 year-old women and men and the calculated response rate.<sup>3</sup> These aspects of the review were carried out by other investigators, but results of these analyses are presented below for completeness.

### 3.1.2 Reviews of trends in outcome measures of interest

The reviews of literature presented in this chapter relating to: a) repeated, cross-sectional estimates of chlamydia prevalence using population-based sampling, b) trends in percentage testing positive using clinical/routinely-collected data and c) trends in *C. trachomatis* antibody seroprevalence, were not carried out as systematic reviews. For these reviews, electronic databases (including Ovid Medline, EMBASE) were searched using terms for chlamydia infection, prevalence, percentage testing positive and seroprevalence for papers up to November 2012. For the review of studies reporting trends in *C. trachomatis* antibody seroprevalence for papers up to November 2012. For the review of studies reporting trends in *C. trachomatis* antibody seroprevalence studies published up to the end of February 2014 were also included due to important contributions to the literature being published after the initial review had been carried out.<sup>132</sup> Studies of *C. trachomatis* antibody seroprevalence were included if they aimed to

investigate the burden of genital infection with *C. trachomatis*; studies aimed at monitoring *C. trachomatis* associated with ocular infections are not reported.

# 3.2 Systematic review of chlamydia prevalence studies using population-based sampling methods: Results

The search strategy identified a total of 1,003 reports. A total of 91 publications, describing 39 studies of chlamydia prevalence in the general population (25 EU/EEA, 14 non-EU/EEA) were included.<sup>3</sup> Table 3-1 presents a summary of included studies. Point estimates of chlamydia prevalence in these studies ranged from 0.2% to 8.0% in women and 0.4% to 6.9% in men. Five studies (not including Natsal-3) reported chlamydia prevalence in nationally-representative samples of sexually-experienced adults aged  $\leq$ 26 years (Figure 3-1). The pooled estimate of chlamydia prevalence in these studies was 4.3% (95%CI 3.6%-5.0%) among women and 3.6% (95%CI 2.8%-4.4%) in men. Point estimates of prevalence ranged from 3.0%<sup>133</sup> to 5.3%<sup>134</sup> in women and 2.4%<sup>135</sup> to 7.3%<sup>134</sup> in men. Further details of the studies carried out in the UK are provided below.

## Table 3-1a: Summary of characteristics of studies included in systematic review of chlamydia prevalence, studies with <u>national</u> coverage

Adapted from Redmond SM, Kissling KA, <u>Woodhall SC</u> et al.<sup>3</sup> The red shaded area indicates Natsal-3 survey, findings of which were not included in the original systematic review.

Country, year	Sex, age (years)	Prevalence estimated in whole study Sample, sexually experienced only or both	Number invited for testing (response rate overall, %)	Study name (acronym), if known; purpose of study, setting and sampling strategy
EU/EEA Co	untries			
Croatia 2011 <sup>134</sup>	W&M, 18–25	S-E	1005 participants. 861 sexually experienced. 280 provided urine sample (women 37.5%, men 27.9%)	Cross-sectional survey of sexual behaviour and STI prevalence. Nationally representative sample from all 21 counties in Croatia, with multi-stage probability sampling.
France 2010 <sup>135</sup>	W&M, 18–44	S-E	4957 eligible by age and sexual experience (women 54.4%, men 49.3%)	Sexual behaviour survey (subsample of Contexte de la Sexualité en France study, NatChla). Random subsample of sexually experienced people from a national population- based survey on sexual behaviour with two-phase stratified sampling. Urine testing kit only sent to women if no swab returned after 1 month.
Germany 2012 <sup>136</sup>	W&M, 12–17	Both	5755 in this age group (women and men 14– 17 years 63%)	General health survey (Kinder und Jugendgesundheitsstudie, KiGGS). Two-stage stratified cluster sampling, nationally representative sample of 0–17 year-olds. Only tested samples from participants in this age group.
Netherlands 2005 <sup>137</sup>	W&M, 15–29	Both	20791 (women 47.0%, men 33.0%)	Cross-sectional survey to estimate chlamydia prevalence and screening feasibility (CT PILOT). Stratified probability sample of randomly selected men and women in 4 regions of the Netherlands according to population density. Regions not sampled at random.
Slovenia 2004 <sup>138</sup>	W&M, 18–49	Both	2616 invited (women 60.0%, men 50.9%)	Sexual behaviour study. Stratified two stage probability sample of the general population of Slovenia in this age group. All participants invited to provide specimen for chlamydia testing.
Britain 2001 <sup>133</sup>	W&M, 18–44	S-E	5026 invited to give urine sample (women 71.1%, men 68.7%) <sup>a</sup>	Sexual behaviour study (National Survey of Sexual Attitudes and Lifestyles, Natsal-2), conducted 1999-2001. Random sample of sexually experienced people taking part in a stratified probability sample of people aged 16–44 years resident in Britain (total 11 161 interviewed).
Britain 2012	W&M, 16-44	S-E <sup>b</sup>	8047 eligible participants invited to provide a urine sample, 4828 (W 59.7%, M 60.4%) <sup>c</sup>	Sexual behaviour study (National Survey of Sexual Attitudes and Lifestyles, Natsal-3), conducted 2010-12. Random sample of sexually experienced people taking part in a stratified probability sample of people aged 16–74 years resident in Britain (total 15 162 interviewed).
Non-EU/EE	A counti	ies, including	g high-income OECD o	countries
USA 2002a <sup>139</sup>	м, 18–19, 22–26	WSS	from 470 aged 18–19, and 995 aged 22–26 who were aged 15– 19 in 1988 survey (insufficient data to calculate)	National Surveys of Adolescent Males (NSAM). Sexual health survey. Nationally representative sample of never- married, noninstitutionalised men aged 15–19 (1995 survey), and aged 22–26 (aged 15–19 in 1988 survey but re- interviewed in 1995). Oversampling of black and Hispanic youths.
USA 2004 <sup>140</sup>	W&M, 18–26	Both	Wave III: 14322 (women and men 84%)	Cohort study (US National Longitudinal Study of Adolescent Health, Add Health). Nationally representative sample of young people in the USA. Total in first survey, Wave I: 18924.
USA 2012 <sup>141</sup>	W&M, 14–39	WSS	20836 selected 17190 interviewed (women 80.4%, 2007–2008, men 74.5%, 2007– 2008) <sup>d</sup>	General health survey (US National Health and Nutrition Examination Surveys, NHANES). Stratified multistage probability cluster sampling. Data from five 2-year survey cycles.

## Table 3-1b: Summary of characteristics of studies included in systematic review of chlamydia prevalence, <u>sub-national</u> studies

Adapted from Redmond SM, Kissling KA, Woodhall SC et al.<sup>3</sup>

Country, year	Sex, age in years	Prevalence estimated in whole study Sample (WSS), sexually experienced only (S-E)	Number invited for testing (response rate overall, %)	Study name (acronym), if known; purpose of study, setting and sampling strategy
EU/EEA Co	untries	01117 (0 2)		
Denmark 1998 <sup>142</sup>	W&M, mean 18.0 women, 18.2 men	S-E	2603 women 928 eligible (women 33.3%) 1733 men 442 eligible (men 24.8%)	RCT of home sampling versus usual care. Random sample (half) of all high schools in Aarhus County. All students invited. Eligible if sexually experienced. (Only data from home sampling group included).
Denmark 1999 <sup>143</sup>	W, 20– 29	WSS	16345 eligible 11088 in cohort (women 67.8%)	Cohort study about risk factors for cervical cancer. Random sample of women born in Denmark, in catchment area of Righospitalet, Copenhagen taking part in a cohort study, who had cervical swab sample taken by gynaecologist.
Denmark 2001 <sup>144</sup>	M, 17– 32	Both	2500 (men 53.8%)	Cross-sectional survey to estimate chlamydia prevalence. All men in Northern Jutland, Aarhus or Copenhagen counties liable for military service and seen by a medical board.
Denmark 2002 <sup>145</sup>	W&M, 21–23	S-E	4000 women (women 32.5% group 1, Response rates from online results for 26.3% group 2) 5000 men (men 25.9% group 1, 15.4% group 2)	RCT on effectiveness of outreach screening strategies. Simple random sample from all residents of Aarhus County in this age group. Group 1 received sampling kit, group 2 had to request kit by post.
Estonia 2008 <sup>146</sup>	W&M, 18–35	WSS	1398 reachable (women 48%, men 32%)	Cross-sectional survey to estimate chlamydia prevalence. Stratified random sample of residents of Tartu county.
Netherlands 2000 <sup>147</sup>	W&M, 15–40	WSS	5714 women (women 50.8%) 5791 men (men 33.0%)	Cross-sectional survey to estimate chlamydia prevalence and screening feasibility. Simple random sample of patients on the lists of 16 general practices in Amsterdam.
Netherlands 2010 <sup>137</sup>	W&M, 16–29	S-E	140058 Amsterdam (women 22.4%, men 10.8%) 107806 Rotterdam (women 19.6%, men 10.5%)	Cluster controlled trial of chlamydia screening effectiveness (Chlamydia Screening Implementation, CSI). All 16–29 year- old residents of Amsterdam, Rotterdam, parts of South Limburg. Sexually active people invited to request test kit. South Limburg excluded because eligibility depended on response to questionnaire assessing risk of chlamydia.
Norway 2005 <sup>148</sup> Norway 2012 <sup>149</sup>	W&M, 18–29 W&M, 18–25	WSS S-E	646 reached (women 43.8%, men 25%) 10000 invited 1670 returned sample (women 18.9%, men 11.9%)	Cross-sectional survey to estimate chlamydia prevalence. All patients on the list of a group practice in Oslo. Cross-sectional survey to estimate chlamydia prevalence. Simple random sample of 10,000 people in this age group living in Rogaland county using unique personal identification number.
Spain 2007 <sup>150</sup>	W, 15– 44	S-E	1821 invited 916 reached or accepted (women 66.1%)	Cross-sectional multinational HPV prevalence survey. Random age stratified sample of the adult female general population from census list of 4 urban communities in metropolitan Barcelona.
Sweden 1992 <sup>151</sup>	W, 15– 34	S-E	543 reached and were sexually experienced women (68.9%)	Cross-sectional survey to estimate chlamydia prevalence. All women in this age group in a primary health care area in Nättraby invited, only sexually experienced screened.
Sweden 1995 <sup>152</sup>	W, 19, 21, 23, 25	WSS	816 reached 611 participated (68.3% women)	Cross-sectional survey to estimate chlamydia prevalence. All women of this age living in primary health care area of Ålidhem community centre in Umeå.
Sweden 2003 <sup>153</sup>	M, 22	WSS	1074 (men 35.6%)	Cross-sectional survey to investigate feasibility of chlamydia
Sweden 2004 <sup>154</sup>	W&M, 20–24	WSS	200 (women 65%, men 45%)	Cross-sectional survey to estimate chlamydia prevalence and cost-effectiveness of home sampling. Simple random sample of 100 men and 100 women in this age group living in Umeå.
Sweden 2007 <sup>155</sup>	M, 19–24	WSS	1936 reached (men 14.5%)	Cross-sectional survey to estimate chlamydia prevalence. Sampling method unclear, 1000 men living in Uppsala county (from population register), and 1000 Uppsala university students (from student register database).
United Kingdom 2000a <sup>156</sup>	M, 18–35	WSS	919 invited by post and reachable (men 45.3%)	Cross-sectional survey to estimate chlamydia prevalence and screening feasibility. Postal recruitment of all men aged 18–24 and a random sample of men aged 25–35 in 4 general practices in North West London.

United Kingdom 2000b <sup>157</sup>	W&M, 18–35	S-E	166 women reached (women 39%) 175 men reached (men 46%)	Pilot study of acceptability of home sampling. Simple random sample of patients on the lists of 3 general practices in North West London and Avon. Urine samples from random 50% of women, vulval swabs from other 50%.
United Kingdom 2007 <sup>10</sup>	W&M, 16–39	WSS	14382 reached (women 37.6%, men 27.9%)	Cross-sectional survey to estimate chlamydia prevalence and screening feasibility (Chlamydia Screening Studies project, ClaSS). Random sample of general population in Birmingham and Bristol areas, selected from 27 general practice lists.
United Kingdom 2012 <sup>158</sup>	W&M, 18–24	WSS	29917 invited (women 13.2%, men 9.8%)	Cross-sectional survey investigating feasibility of postal screening invitations. All people in this age group registered with any GP in North East Essex Primary Care Trust.
Non-EU/EE	A countries,	including	y high-income OECD o	countries
Australia 2003 <sup>159</sup>	W&M, 15–40+	WSS	6431 eligible 2862 participated (women and men 43.8%)	General health survey. All people living in 26 rural indigenous Australian and Torres Strait Islander communities in northern Queensland taking part in Well Person's Health Check.
Australia 2004 <sup>160</sup>	W&M, 15–35	WSS	2703 eligible listed 1219 screened (women 50.7%, men 39.3%)	Cross-sectional survey to estimate chlamydia and gonorrhoea prevalence. Indigenous Australian people aged 15–35 living in Alice Springs area
Australia 2006 <sup>161</sup>	W, 18–35	Both	1532 eligible households 979 women interviewed 657 gave urine sample (women 42.9%)	Cross-sectional survey to estimate chlamydia prevalence. Simple random sample from Melbourne residential telephone directory.
Australia 2008 <sup>162</sup>	W&M, 14–40	WSS	ca. 1300 in 1996 (insufficient data to calculate)	Cross-sectional survey in STI control programme screening for chlamydia, gonorrhoea and syphilis. All resident indigenous Australians living in the Anangu Pitjantjatjara Yankunvtiatiara Lands.
Canada 2002 <sup>163</sup>	W&M, 15–39	WSS	1075 women (women 29.3%) 1130 men (men 16.2%)	Chlamydia mass screening study. All adults from remote Inuit communities in Nunavik region. All sexually experienced or in this age group especially encouraged to take part.
Canada 2009 <sup>164</sup>	W&M, 15–65	WSS	224 estimated eligible (insufficient data to calculate) 181 screened (80.8% for women and men)	Chlamydia and gonorrhoea mass screening study. All men and women in this age group living in a rural Inuit community from Baffin Region, Nunavut.
New Zealand 2002 <sup>165</sup>	W&M, 16+	S-E	1582 invited 1136 consented 582 sexually active (insufficient data to calculate)	Cross-sectional survey to estimate chlamydia prevalence. Random sample of 50% of classes in all private and public high schools, Christchurch. Only sexually active had their samples tested.
Switzerland 2008 <sup>166</sup>	M, 18–26	Both	521 eligible and gave written consent (insufficient data to calculate)	Cross-sectional survey to estimate chlamydia prevalence. All young Swiss men attending obligatory medical board before army recruitment (French speaking region only).
USA 2001 <sup>167</sup>	W, 18–29	S-E	2148 eligible 1439 enrolled 1370 tested 1314 sexually active (women 61.2%)	Household survey of risk behaviour and chlamydia prevalence. All English- or Spanish speaking women in this age group in a random sample of low income housing blocks from the 1990 census (<10th percentile) in 3 counties in California.
USA 2002 <sup>c</sup>	W&M, 18–35	WSS	1224 adults aged 18– 45 reached 728 age- eligible for screening (women and men 79.5%)	Cross-sectional survey to estimate chlamydia and gonorrhoea prevalence. Stratified probability sampling of households in Baltimore; urine samples requested from those in study age group.
USA 2011 <sup>169</sup>	W&M, 15–35	Both	4998 eligible (women and men 42.7%)	Cross-sectional survey to estimate STI prevalence (Monitoring STI Survey Program). Probability sample of Baltimore residents

EU/EEA, European Union or European Economic Area Member States; M, men; OECD, Organization for Economic Cooperation and Development; S-E: Sexually-experienced; STI, sexually transmitted infections; W: women; WSS: Whole study sample. <sup>a</sup>Numbers from technical report Erens et al. 2001<sup>4</sup>. <sup>b</sup>Specimens were collected from all 16 to 17 year-olds regardless of reported sexual activity; prevalence of chlamydia infection was estimated among sexually-experienced only <sup>c</sup>Numbers from technical report Erens et al.<sup>5</sup> <sup>d</sup>Response rates from online results for 2007–2008 <u>www.cdc.gov/nchs/nhanes/response\_rates\_CPS.htm</u>

## Figure 3-1a: Forest plot, estimates of chlamydia prevalence in women ≤ 26 years in EU/EEA and other high-income countries

Reproduced from Redmond SM, Kissling KA, Woodhall SC et al.<sup>3</sup> For references, see Table 3-1 above.

National population, overall Germany 2012 Netherlands 2005 Netherlands 2005 Slovenia 2004 USA 2004 USA 2012 (1999-2000) USA 2012 (2001 2000)		(95% C1) 2.11 (1.36, 3.13) 2.60 (1.70, 3.40) 1.90 (1.20, 2.70) 4.10 (2.20, 7.40) 4.74 (3.93, 5.71) 4.10 (2.40, 6.80) 2.80 (1.80, 4.50)	15 15 20 18 18 18 14	17 19 24 24	1136 1657 1869
National population, overall Germany 2012 Netherlands 2005 Netherlands 2005 Slovenia 2004 USA 2004 USA 2012 (1999-2000) USA 2012 (2001 2002)		2.11 (1.36, 3.13) 2.60 (1.70, 3.40) 1.90 (1.20, 2.70) 4.10 (2.20, 7.40) 4.74 (3.93, 5.71) 4.10 (2.40, 6.80) 2.80 (1.80, 4.50)	15 15 20 18 18 14	17 19 24 24	1136 1657 1869
Germany 2012 Netherlands 2005 Netherlands 2005 Slovenia 2004 USA 2004 USA 2012 (1999-2000) USA 2012 (2001 2002)		2.11 (1.36, 3.13) 2.60 (1.70, 3.40) 1.90 (1.20, 2.70) 4.10 (2.20, 7.40) 4.74 (3.93, 5.71) 4.10 (2.40, 6.80) 2.80 (1.80, 4.50)	15 15 20 18 18 14	17 19 24 24	1136 1657 1869
Netherlands 2005         →           Netherlands 2005         →           Slovenia 2004         →           USA 2004         →           USA 2012         (1999-2000)           USA 2012         (2001-2000)		2.60 (1.70, 3.40) 1.90 (1.20, 2.70) 4.10 (2.20, 7.40) 4.74 (3.93, 5.71) 4.10 (2.40, 6.80) 2.80 (1.80, 4.50)	15 20 18 18 14	19 24 24	1657 1869
Netherlands 2005		1.90 (1.20, 2.70) 4.10 (2.20, 7.40) 4.74 (3.93, 5.71) 4.10 (2.40, 6.80) 2.80 (1.80, 4.50)	20 18 18 14	24 24	1869
Slovenia 2004 USA 2004 USA 2012 (1999-2000)		4.10 (2.20, 7.40) 4.74 (3.93, 5.71) 4.10 (2.40, 6.80) 2.80 (1.80, 4.50)	18 18 14	24	005
USA 2004		4.74 (3.93, 5.71) 4.10 (2.40, 6.80) 2.80 (1.80, 4.50)	18 14	~~	265
USA 2012 (1999-2000)		4.10 (2.40, 6.80) 2.80 (1.80, 4.50)	14	26	7555
		2.80 (1.80, 4.50)		25	NR
USA 2012 (2001-2002)		· · /	14	25	NR
USA 2012 (2003-2004)		4.30 (2.70, 6.70)	14	25	NR
USA 2012 (2005-2006)		1.80 (1.10, 2.90)	14	25	NR
USA 2012 (2007-2008)		3.80 (2.40, 6.00)	14	25	NR
Subtotal (I-squared = 75.4%, p = 0.000)					
National population, sexually experienced					
France 2010		3.60 (1.90, 6.80)	18	24	467
Slovenia 2004		4.70 (2.50, 8.50)	18	24	NR
United Kingdom 2001		3.00 (1.70, 5.00)	18	24	379
Croatia 2011		5.30 (2.30, 10.20)	18	25	151
USA 2004		4.70 (3.90, 5.70)	18	26	4874
Subtotal (I-squared = 0.0%, p = 0.438)		4.30 (3.59, 5.02)			
Sub-national population, overall					
Denmark 1999	$\longrightarrow$	10.70 (7.18, 15.20)	20	24	252
Netherlands 2000		3.82 (2.51, 5.54)	15	25	681
Sweden 1995		2.70 (1.50, 4.40)	19	25	557
United Kingdom 2007		6.20 (4.90, 7.80)	16	24	2132
United Kingdom 2012		4.40 (3.50, 5.40)	17	25	1951
Subtotal (I-squared = 81.1%, p = 0.000)					
Sub-national population, sexually experienced					
Denmark 1998		5.00 (3.61, 6.62)	16	19	867
Denmark 2002 /Group1		6.50 (4.70, 8.65)	21	23	649
Denmark 2002 /Group2		8.00 (5.82, 10.64)	21	23	526
Netherlands 2010		3.90 (2.75, 5.05)	16	19	3618
Netherlands 2010 -		3.95 (3.35, 4.54)	20	24	10783
Norway 2012		5.80 (4.48, 7.50)	18	25	930
Spain 2007		0.60 (0.00, 3.50)	15	24	157
United Kingdom 2000b [43]	$\longrightarrow$	8.00 (2.30, 20.00)	18	25	48
USA 2001		5.00 (2.80, 7.20)	18	21	424
USA 2001		2.30 (0.80, 3.70)	22	25	447
Australia 2006		3.70 (1.20, 8.40)	18	24	135
New Zealand 2002		2.30 (0.40, 4.20)	16	19	226
Subtotal (I-squared = 77.3%, p = 0.000)					
NOTE: Weights are from random effects analysis					
0 5 10	15				

Chlamydia prevalence, % (95% CI)

## Figure 3-1b: Forest plot, estimates of chlamydia prevalence in men $\leq$ 26 years in EU/EEA and other high-income countries.

Reproduced from Redmond SM, Kissling KA, Woodhall SC et al.<sup>3</sup> For references, see Table 3-1 above.

Country, year		Estimated CT prevalence in % (95% Cl)	Age N min max	lumber analyse x
National population, overall				
Germany 2012	F	0.38 (0.08, 1.11)	16 17	789
Netherlands 2005	◆	1.00 (0.40, 1.50)	15 19	916
Netherlands 2005	◆	1.30 (0.70, 1.90)	20 24	1023
Slovenia 2004	<b></b>	4.10 (2.20, 7.40)	18 24	252
USA 2004	<b>→</b> -	3.67 (2.93, 4.58)	18 26	6767
Subtotal (I-squared = 91.9%, p = 0.000)				
National population, sexually experienced				
France 2010	<b>_</b>	2.40 (1.00, 5.70)	18 24	322
Slovenia 2004	<b></b>	4.70 (2.50, 8.50)	18 24	NR
United Kingdom 2001	_ <b></b>	2.70 (1.20, 5.80)	18 24	301
Croatia 2011	<b>—</b>	7.30 (3.40, 13.40)	18 25	123
USA 2004	-	3.70 (3.00, 4.70)	18 26	4473
Subtotal (I-squared = 6.2%, p = 0.372)	$\diamond$	3.60 (2.77, 4.42)		
Sub-national population, overall				
Netherlands 2000	<b>→</b>	2.28 (1.05, 4.28)	15 25	395
Sweden 2003	♦—	1.10 (0.30, 2.80)	22 22	362
United Kingdom 2000a	•	1.50 (0.19, 5.56)	18 24	130
United Kingdom 2007	-	5.30 (4.40, 6.30)	16 24	1477
United Kingdom 2012	<b>_</b>	4.50 (3.50, 5.70)	17 25	1480
Subtotal (I-squared = 88.6%, p = 0.000)				
Sub-national population, sexually experienced				
Denmark 1998	<b>→</b>	2.60 (1.28, 4.53)	16 19	430
Denmark 2002 /Group1	<b>_</b>	5.90 (4.19, 7.97)	21 23	647
Denmark 2002 /Group2	<b></b>	5.70 (3.61, 8.50)	21 23	386
Netherlands 2010	<b>◆</b>	1.84 (1.17, 2.52)	16 19	1589
Netherlands 2010	<b>-</b>	3.84 (2.98, 4.70)	20 24	4500
Norway 2012	<b>_</b>	5.10 (3.80, 6.80)	18 25	605
New Zealand 2002	<b>→</b>	1.80 (0.20, 3.30)	16 19	240
Subtotal (I-squared = 84.5%, p = 0.000)				
OTE: Weights are from random effects analysis				
0	5 10	15		

Chlamydia prevalence, % (95% Cl)

### 3.2.1 UK-based studies of chlamydia prevalence

The National Surveys of Sexual Attitudes and Lifestyles (Natsal) are three, stratified probability sample surveys of the British general population (covering England, Scotland and Wales; Northern Ireland is not included). The first survey (Natsal-1) was conducted in 1990-1991<sup>170</sup>, Natsal-2 in 1999-2001<sup>171</sup> and Natsal-3 in 2010-12<sup>172</sup>. In both Natsal-2 and Natsal-3, chlamydia prevalence was estimated among a subset of participants<sup>23</sup>. Natsal-2 and Natsal-3 are the only studies to have estimated population prevalence at a national level within the UK<sup>t</sup>. In both surveys a sample of households was identified using a national postcode list, and interviewers visited the selected homes to identify eligible participants and carry out face to face interviews. The response rate to the overall survey was 65% in Natsal-2 and 58% in Natsal-3.9 Natsal-2 recruited 11,161 men and women aged 16 to 44; Natsal-3 recruited 15,162 16 to 74 yearolds. In Natsal-2, a sub-sample of sexually-experienced 18 to 44 year-olds in Natsal was also invited to provide a urine sample for chlamydia testing, 71% of whom (n=3,569) agreed. The prevalence of chlamydia among sexuallyexperienced 18 to 44 year-olds was estimated to be 2.2% (95%CI 1.5% to 3.2%) among men and 1.5% (95%CI 1.1% to 2.1%) among women. Among 18 to 24 year-olds, chlamydia prevalence was 3.0% (95%Cl 1.7%-5.0%) among women and 2.7% (95%CI 1.2%-5.8%) among men.<sup>133</sup> In Natsal-3, as well as all sexually-experienced 18 to 24 year-olds and a subset of 18 to 44 year-olds, all

<sup>&</sup>lt;sup>f</sup> The Natsal surveys were conducted in Britain (England, Scotland, Wales, not Northern Ireland). Studies in this review are labelled with the country of origin. UK is used to denote studies that were conducted in any of the countries of the UK.

<sup>&</sup>lt;sup>g</sup> As eligibility within households was not known for those households not contacted, these calculated response rates use an estimated denominator of eligibles using data from households that could be contacted.<sup>4,5</sup>

16 and 17 year-olds - regardless of reported sexual activity - were invited to provide a urine specimen for anonymous testing for *C. trachomatis* infection. The estimated chlamydia prevalence among sexually-experienced 16 to 24 year-olds was 3.2% (95%CI 2.2%-4.6%) in women and 2.6% (95%CI 1.7%-4.0%) in men.<sup>23</sup>

The other UK-based study to have estimated chlamydia prevalence using population-based methods was conducted on a sub-national basis. Chlamydia prevalence was estimated for two areas of England (West Midlands and Avon) as part of the Chlamydia Screening Studies (ClaSS) project in 2001/02. Lists of 16 to 39 year-old men and women registered with GPs were used to identify the eligible target population. Individuals were then randomly selected from these lists and sent a postal invitation to participate in chlamydia screening by providing a urine sample for chlamydia testing. A total of 35% of those who were successfully contacted agreed to a test. The estimated population prevalence was 3.6% in women (95%Cl 3.1%-4.9%) and 2.8% among men (95%Cl 2.2%-3.4%) aged 16 to 39. Prevalence was higher among 16 to 24 year-olds (women 6.2%; 95%Cl 5.2%-7.8%; men 5.1% 95%Cl 4.0%-6.3%).<sup>10</sup>

A small number of studies of population prevalence of chlamydia conducted at the national level have been conducted outside of the UK (Table 3-1; Figure 3-1). The most established of these is the National Health and Nutrition Examination survey (NHANES) in the US. NHANES is a series of crosssectional surveys using a stratified, multistage probability cluster design to select a representative sample of the US civilian, non-institutionalised population. Interviews are conducted in the household, and biological samples

are collected at mobile examination units. Recruitment is continuous, and data are presented in two-yearly cycles.<sup>141</sup>

### 3.2.2 Methodological issues in general population studies

Selection bias is error that arises from procedures used to select subjects or from factors that influence study participation<sup>173</sup>. Bias occurs when members of the study population differ in an important way from those who are not included in the study, with respect to the outcome or outcomes of interest. Some types of selection bias are particularly relevant for studies of chlamydia prevalence and are explored in more detail below.

#### 3.2.2.1 Non-response bias

In the absence of 100% response rates, non-response bias occurs when members of the study population differ in an important way from non-participants. In the case of chlamydia prevalence surveys, non-response bias occurs when the prevalence of chlamydia is different among participants and non-participants<sup>174,175</sup>.

The response rates in the prevalence surveys reported in section 3.2 ranged from 9.8%<sup>158</sup> to 84%<sup>140</sup>. The standard view in survey research has been that high response rates (ranging from more than 50% to more than 85% participation) are required in order to reduce the risk of non-response bias. However the relationship between response rate and non-response bias is not straightforward, and several studies have shown that non-response bias is not directly correlated with the survey response rate. Thus Groves argues that "there is no minimum response rate below which survey estimates are *necessarily* subject to bias"<sup>174</sup>. The extent to which an estimate among

participants will be a biased estimator of the target population depends on whether the factors affecting the likelihood that an individual will participate in a survey are related to the outcome of interest. If the factors relating to participation are entirely unrelated to the outcome, then estimates will be unbiased<sup>174</sup>. While this is theoretically possible, this is unlikely to happen in practice. An individual's propensity to participate is often related to their interest in the topic, or in this case risk of infection, and thus is likely to lead to a difference in the outcome measure between participants and nonparticipants<sup>176</sup>.

The studies among general population samples described in section 3.2 provide some important evidence to suggest that participation in surveys of chlamydia prevalence is indeed likely to depend on factors relating to risk of infection, including sociodemographic characteristics and sexual behaviours. In Natsal-2 and Natsal-3, chlamydia testing was performed among a subsample of respondents to the main survey. In both surveys the probability of providing a urine sample was associated with demographic factors including gender, ethnicity, and social class, as well as sexual risk factors, including history of condomless intercourse, having had a same-sex partner or reporting anal sex. Some of this differential participation would have led to participants being more at risk of having a prevalent infection than non-participants (e.g. those reporting more sexual partners without a condom and reporting a previous STI diagnosis were more likely to participate) whereas other factors may have tended towards a bias in the opposite direction (e.g. respondents of non-white ethnicity were more likely to participate in the urine study than those of other ethnicities).<sup>177,178</sup>

As part of the CSI project in the Netherlands, findings from a survey of nonrespondents versus respondents found some evidence that demographic factors associated with increased risk of infection (young age, ethnic minority groups, living in areas with high community risk), were associated with a decreased likelihood of participation. Conversely, individual sexual behavioural risk factors for chlamydia (casual or multiple sexual partners or a recent STI diagnosis), were associated with an *increased* likelihood of participation in screening, thus demonstrating the complexity of the factors affecting participation.<sup>179,180</sup> Uusküla et al used linked data from a healthcare utilisation database in Estonia to compare the characteristics of respondents and nonrespondents in their population-based survey of chlamydia prevalence. Individuals who had a STI or related diagnosis within the past 12 months were more likely to participate in the survey, suggesting those with an interest in the topic had a higher propensity to participate.<sup>181</sup> The higher rates of infections seen in populations attending clinical settings for testing are consistent with this evidence, suggesting that individuals who seek or accept testing are likely to be, on average, at higher risk of infection.<sup>123,182</sup> Further evidence of this is found in the meta-regression presented by Redmond et al in their systematic review of chlamydia prevalence studies, which found that surveys with lower response rates had higher estimates of chlamydia prevalence.<sup>3</sup> However, as the surveys were carried out in different countries and at different times, the extent to which this association is due solely to different participation rates is unclear.

Given the relationship between propensity to participate and risk of infection demonstrated above, the extent to which estimates are subject to non-response bias is likely to differ by study design. For example studies with a broad health focus (e.g. NHANEs), or those using stored samples collected for another

purpose, may be subject to different biases than those with a specific focus on sexual health (e.g. Natsal), as the decision to participate in the survey or provide a biological sample would be informed by considerations other than those that relate specifically to sexual risk behaviour. Similarly, those designed specifically as chlamydia screening interventions (e.g. CSI and ClaSS), are likely to be subject to selection biases, and may be more similar to study populations of individuals accepting testing in other clinical settings. Assessing the scale of bias in each case is, however, extremely difficult and the different methods (if any) used for assessing non-response bias make it infeasible to quantitatively compare bias between studies.

Where the variables that predict participation are perfectly measured in both the target population and the survey population, non-response bias can theoretically be eliminated.<sup>174,183</sup> However in practice, this is limited by the data available, and only minimal information is usually available from non-participants. Where the studies reported above have tried to address non-response, this has usually been limited to applying post-stratification weightings, to weight the sample population to a known distribution in the target population, for example age, gender and ethnicity. While this approach can go some way to correcting for non-response bias, this assumes that the prevalence of chlamydia among participants and nonparticipants is the same within specified demographic subgroups. Weights that only allow for basic demographic characteristics are likely to be inadequate, as the studies where non-response has been investigated show that these factors do not fully capture the risk difference between participants and non-participants. A few studies have applied more detailed weights. In Natsal-2 and Natsal-3, as well as weighting

for age, sex and geographical distribution of the target population, additional weights were applied to allow for the differential response patterns to the request to provide a sample for chlamydia testing.<sup>5,133,177</sup> However, even this approach is imperfect, as it does not allow for any self-selection in participating in the main Natsal-2 survey beyond age, sex and region.<sup>177</sup> In order to estimate prevalence, rather than percentage testing positive using data from participants in the CSI project in the Netherlands, weights were applied to match the age, gender, ethnicity and a community-level risk score of the population of the regions where the study took place. Further weights were applied to allow for an estimated proportion of the population who were sexually experienced, and to adjust for the use of a risk score to determine eligibility in one of the three participating regions (South Limburg, where population prevalence was thought to be lower than in the other two, largely urban regions). No weights were applied to account for any other sexual behaviour differences.<sup>88</sup>

#### 3.2.2.2 Coverage bias

Coverage bias is a form of selection bias that occurs when the sample from which a study population is selected is not representative of the target population, for which conclusions from the study are to be drawn.<sup>184</sup> The availability of adequate sampling frames varies by country, and identifying a representative population can be difficult and can have a large impact on the cost of the study.

The studies presented in section 3.2 above used a range of different sampling frames. Some studies were able to use a population registry, for example in Estonia, Uusküla *et al* selected a random sample of 18 to 35 year-olds living in Tartu from the national population registry in 2005/06. Forty percent of those

who were contacted participated, 86% of whom returned both a questionnaire and a sample for testing.<sup>146</sup> In the Netherlands, the CSI project<sup>137</sup> and the prevalence survey carried out before the implementation of the trial<sup>185</sup> used municipal population registers to access addresses of the eligible population.

However not all countries have reliable population registers available and alternative approaches have been used. For example, telephone directory listings have been used as a sampling frame in studies conducted in Australia<sup>161</sup> and France.<sup>141</sup> In Britain, in the absence of a readily-available population registry, Natsal-2 and Natsal-3 used the national Postcode Address File as the initial sampling frame, from which postcodes and subsequently addresses were selected. Interviewers at the household then assessed eligibility upon contact with the household.<sup>133</sup> Where the eligibility criteria are relatively broad (e.g. adults), this approach will potentially be more efficient than where more restrictive criteria are applied (e.g. 15 to 24 year-old women), as fewer households would have to be contacted in order to identify one eligible person.

One alternative sampling frame to population or household lists in England is to use GP registers, as used by the CLaSS survey.<sup>10</sup> Almost all residents of England are registered with a GP. In 2011, an estimated 98% of 15 to 24 year-old females were registered.<sup>h</sup> However GP lists have substantial limitations. The names and addresses held on GP registers may be inaccurate due to

<sup>&</sup>lt;sup>h</sup> Calculated using ONS mid-2011 population estimates, plus data from the NHS Information Centre Attribution Dataset of GP registered populations scaled to ONS population estimates, 2011, assuming that 1 registration = 1 person. http://www.ic.nhs.uk/statistics-and-datacollections/population-and-geography/gp-registered-populations/attribution-dataset-gpregistered-populations-scaled-to-ons-population-estimates-2011

registrations that relate to deceased patients, duplicate records or patients who have either moved out of the area or moved within the area but not updated their contact details (so-called 'ghost patients'). As part of the ClaSS study, intensive efforts were used to identify whether people actually received the postal invitation to screen. Individuals were classified as 'ghosts' if they were either confirmed as not resident at the address held by the practice or not contactable by any method. 26% of the invited 16 to 24 year-olds women were classed as 'ghosts'<sup>10</sup>. While there has been a move in recent years to improve the accuracy of these lists,<sup>186</sup> GP lists are still likely to be subject to inaccuracies.

In summary, few nationally-representative studies in EU/EEA or other high income countries have estimated chlamydia prevalence, although prevalence has been estimated in Natsal-2 and Natsal-3 in Britain. Surveys are subject to non-response bias and coverage bias. Although weighting has been used to correct for this in some surveys, this potential for bias remains a challenge for achieving robust estimates of prevalence in the general population. These biases are relevant for single estimates of chlamydia prevalence but are also important to consider when using repeated measures, which is the subject of the next section.

# 3.3 Repeated, cross sectional estimates of chlamydia prevalence in the general population

One option for monitoring changes in chlamydia prevalence would be to have regular, repeated measurements of a random sample of the population. Only two published studies using repeated sampling among a general population sample to measure chlamydia prevalence over time had been published at the time of carrying out this review.

In the US, NHANES has measured chlamydia prevalence since 1999. The estimated prevalence among 14 to 39 year-olds declined from 2.6% to 1.6% between 1999/2000 and 2007/2008. However there was no statistically significant fall measured among 14 to 25 year-old women, the population targeted for annual screening in the US.<sup>141</sup>

In the Netherlands, chlamydia prevalence was repeatedly estimated as part of the CSI project. As described in section 3.2.2.1, prevalence was derived from percentage testing positive using post-stratification weightings. No significant change was seen in either the overall percentage testing positive or estimated prevalence. A small decline was observed in one of the three regions (South Limburg), where the percentage testing positive declined from 5.1% to 3.1% and the estimated population prevalence declined from 3.2% to 1.8%.<sup>88</sup>

While these two studies used general population sampling frames, they have some limitations.

The sampling strategy of the NHANES survey means that different areas are sampled in each two year survey cycle. While this is based on probability sampling, and thus will be generally representative of the US population, prevalence estimates for diseases that are highly geographically clustered may be very sensitive to the chance selection of areas. For this reason, NHANES is no longer used for estimating the population prevalence of gonorrhoea, as selecting areas where a cluster of infections exists may lead to erroneous conclusions about the prevalence of infection in the general population.<sup>187</sup>

Chlamydia is more evenly distributed through the population,<sup>188</sup> but it is feasible that the clustering of infection within sexual networks may make such estimates more prone to measurement error.

In the CSI project in the Netherlands, participants had higher levels of sexual risk behaviour than the general population, and the proportion reporting sexual risk behaviours increased in each year (among those with available data on sexual behaviour).<sup>88</sup> Therefore not only are prevalence estimates within each year likely to be overestimated, but the change between years is subject to confounding by the change in the participating population in each year. Specifically, the change over time is likely to be underestimated, as the sample includes an increasingly higher-risk population over time.

The publication of findings from Natsal-3 in 2013, provided a third study which used repeated cross-sectional estimates of population prevalence. Comparisons between these two surveys are investigated in detail in Chapter 6.

### 3.4 Trends in percentage testing positive for chlamydia using clinical and routinely-collected data

Given that few studies have used repeated population-based measurement of prevalence, one potential alternative is to use data from populations accessing chlamydia testing to look at changes over time.

Two systematic reviews have demonstrated the existence of numerous studies of populations attending clinical settings who were tested for chlamydia as part of routine clinical care.<sup>2,123</sup> The percentage testing positive in these studies does not equate to the prevalence of chlamydia among the general population

as individuals seeking or accepting a chlamydia test will not be representative of the general population. This is clearly demonstrated in the systematic review by Adams et al, who found that estimates of percentage testing positive conducted in healthcare settings were higher than prevalence estimates conducted in population-based studies.<sup>123</sup> This distinction can also be seen in a comparison of NCSP data from 2008 and findings from Natsal-2 by Riha *et al*, which found that the percentage testing positive among women and men aged 18 to 24 years who were tested as part of the NCSP was higher than estimated population prevalence in Natsal-2 (8.5% versus 3.1% in women and 9.4% versus 2.9% in men).<sup>182</sup>

While studies using data from populations accessing testing cannot provide estimates of population prevalence of chlamydia, they may have some utility as outcome measures of changes in the burden of chlamydia infections over time, albeit in subgroups of the population.

Reports of trends in percentage testing positive for chlamydia using data from sexual health or family planning clinics have been published or included in routine surveillance reports for several countries, including the UK,<sup>189</sup> the US,<sup>190-192</sup> Australia,<sup>193-195</sup> New Zealand,<sup>196</sup> Finland,<sup>197</sup> the Netherlands,<sup>198</sup> Sweden<sup>73</sup> and Norway.<sup>199</sup> In the US, trends have also been reported among men and women tested through the National Job Training Program (NJTP), representing a high-risk subgroup of socioeconomically-disadvantaged 16 to 24 year-old men and women.<sup>200</sup> In contrast to trends seen in England (section 4.4.1), increasing trends in percentage testing positive for chlamydia have been seen since around 2000 in Denmark, Sweden and New Zealand.<sup>101</sup> In Australia, increases in percentage testing positive have been observed in several

populations,<sup>193</sup> although stable trends or decreases have been seen among men who have sex with men (MSM).<sup>194</sup> In the US, test positivity in family planning clinics and among young people entering the NJTP has been increasing in the last decade.<sup>200</sup> After adjusting for test technology and known changes in the test population however, either stable or declining trends were seen. Analyses at a sub-national level have shown increases in percentage testing positive during this period, even after adjustment.<sup>192</sup>

# 3.4.1 Methodological issues in studies of trends in chlamydia prevalence or percentage testing positive

### 3.4.1.1 Measured and unmeasured confounding

Confounding occurs when the estimated effect of an exposure of interest is distorted, as it is mistaken for the effect of another exposure (or exposures). In order for this to happen, the confounder must be correlated with both the outcome and the exposure of interest, and must not be part of the causal pathway from the exposure of interest to the outcome of interest.<sup>173</sup> Whether using repeated prevalence measures from population-based samples, or trends in percentage testing positive in populations accessing testing, estimates will be subject to confounding if there are any changes in the characteristics of the population tested or the way that infection status is measured that are also associated with risk of infection.<sup>201</sup> Where all confounders are perfectly measured, the main effect of time on chlamydia result can be measured by including confounders as covariates in multivariable models, or by carrying out stratified analyses. In practice, it is unlikely that all potential confounding on estimates of trends in infection over time.
### 3.4.2 Changes in demographics or sexual behaviour of those tested

The potential for confounding was seen in the CSI project in the Netherlands.<sup>88</sup> Among those who completed the questions on sexual behavioural at the time of screening, reported levels of sexual risk behaviours were higher than the general population and increased with each round of screening invitation, while the overall uptake of screening in the study decreased with each round from 16% in round one to 10% in round three.<sup>88</sup> Even though the project used population-based sampling (i.e. invited the total eligible population) it is reasonable to expect that the true prevalence of infections would be lower than that estimated in the CSI project's main findings and that the decline in prevalence would be steeper than was observed.<sup>88</sup>

Depending on the specific context, it is easy to see how trends in the percentage testing positive among tested populations may be even more subject to changes in who is tested than in population-based studies.<sup>201</sup> Changes in national or local screening policies or rates of partner notification may lead to different populations being tested. Media campaigns may increase awareness of chlamydia, and lead to people either requesting a test or accepting the offer of a test that may not have done before. While several reports have been published that examine the trends in percentage testing positive, relatively few have made any attempts to control for potential confounding introduced by the observational nature of the data.

In the US, Satterwhite *et al* examined trends in percentage testing positive in women tested in Infertility Prevention Program (IPP) family planning clinics<sup>191</sup> and antenatal settings.<sup>202</sup> In both settings, trends in percentage testing positive were adjusted for test technology, age, ethnicity and region. No sexual

73

behaviour data were available. The analysis was performed using the clinic as the unit of analysis. The unadjusted trend in percentage testing positive in the IPP clinics showed 2% annual increase in percentage testing positive (odds ratio (OR) 1.02, 95%CI: 1.01 to 1.03). After adjustment, the authors found that there was no change in percentage testing positive in women attending IPP FP clinics between 2004 and 2008 (adjusted odds ratio (AOR) 1.00, 95%CI: 0.99 to 1.00) and a decline in percentage testing positive among women attending IPP prenatal clinics between 2004 and 2009 (AOR 0.93, 95%CI 0.92-0.95). While this method was able to control for unmeasured confounding arising from a changing distribution of clinics, it assumes that the populations attending each clinic are largely stable over time, within the demographic stratum.

Fine *et al* explored trends in percentage testing positive among women attending IPP FP clinics in the Pacific Northwest region of the US between 1997 and 2004. Enhanced demographic and sexual behaviour data are collected at a regional level, and thus they were able to examine the trend in percentage testing positive adjusting for age, state, race/ethnicity, city size, test technology and several sexual behaviour variables. Adjusting for all these factors together had little effect on the measured trend in percentage testing positive, with both unadjusted and adjusted analyses showing a 5% increase per year in the odds of testing positive (AOR 1.05, 95%CI 1.03 to 1.05).<sup>192</sup>

Two papers have examined adjusted trends in percentage testing positive in Australian settings. O'Rourke *et al* reported trends in percentage testing positive among women attending the Melbourne Sexual Health Centre, and Vodstrcil *et al* reported trends among men from the same setting. Among women, the selfreported characteristics of attendees were found to be relatively stable over the analysis period, with only minor changes observed, such as an increase in the proportion reporting 2 or more sexual partners in the previous 3 months from 25% in 2003 to 30% in 2007.<sup>193</sup> More substantial changes were observed among men who have sex with women (MSW), among whom the proportion reporting 2 or more sexual partners increased from 42% in 2002 to 53% in 2009. Among men who have sex with men (MSM), the only notable change was the proportion reporting symptoms or contact with infection, which decreased from 39% in 2002 to 35% in 2009. Although these changes in the population tested were noted, adjusting for these characteristics made negligible difference to the measured trend in percentage testing positive, suggesting that the changes in the population were not sufficient to explain any of the trend in percentage testing positive over time. After adjusting for available risk factors, percentage testing positive was found to increase among women and MSW (women: AOR 1.12; 95%CI: 1.05-1.20; MSW: AOR 1.04, 95%CI: 1.01-1.07), but no change was found among MSM (AOR = 0.99, 95%CI: 0.93-1.06).<sup>194</sup>

### 3.4.3 Changes in test technologies

While relatively few studies have attempted to deal with potential changes in the demographic or behavioural characteristics of those tested, more have tried to address the impact of changes in test technologies. A change to a more sensitive test could result in an increase in percentage testing positive for chlamydia without a change in the prevalence of infection among the tested population, whereas a change to a more specific test could result in a decrease in the percentage testing positive, due to fewer false positive results. Changes in the test technology may also result in changes in the population tested. Less invasive sampling for both men and women may increase the likelihood of

someone either seeking a test, being offered a test, or accepting the offer of testing.

Three studies aimed specifically at determining the change in percentage testing positive that can be attributed to the change in test were identified in this literature review. Burckhardt *et al* used data from a GUM clinic in Scotland between 1992 and 2003, where the test used switched from culture to ligase chain reaction (LCR) in September 1998;<sup>203</sup> Dicker *et al* examined data from two regions of the Infertility Prevention Program (IPP), and compared percentage testing positive in the calendar quarters immediately before and after changes from a DNA probe to LCR in one region (1996-1997) and from direct fluorescent antibody (DFA) to enzyme immunoassay (EIA) with confirmatory DFA in another (1993-1994);<sup>204</sup> Gotz *et al* compared changes in percentage testing positive in laboratories that did and did not change their testing procedures in Sweden between 1996 and 1998, and 1998 and 1999.<sup>205</sup> In all cases, some, but not all of the increase in percentage testing positive was attributable to changes in the test used.

Consistent with the expected increase in percentage testing positive with increasingly sensitive test technology, Satterwhite *et al* found that adjusting for test technology had a substantial impact on the observed trend in percentage testing positive among men and women tested in the National Job Training Program (NJTP) in the US. Comparison of unadjusted data showed an increase in percentage testing positive between 2003 and 2007 (OR 1.11, 95%CI: 1.10 - 1.14), however after adjusting for test technology and other variables, the adjusted trend showed an average 5% decline in percentage testing positive per year (AOR 0.95, 95%CI: 0.93-0.97).<sup>200</sup>

76

In summary, in considering trends in prevalence or percentage testing positive for chlamydia, if the population tested or test used varies over time then changes in the measured outcome may be wrongly attributed to a change in the amount of infection in the population over time. This has been demonstrated in studies in the US and Australia, but the impact of differences in the population accessing testing has not been explored in England. This is the subject of Chapter 4.

# 3.5 Trends in *C. trachomatis* antibody seroprevalence as a marker of age-specific cumulative incidence

Seroepidemiology has been widely used for the monitoring of vaccinepreventable diseases where there is a clear correlation between previous vaccination and/or wild-type infection and detectable antibodies and to identify susceptible sections of the population.<sup>206,207</sup> As the presence of *C. trachomatis* antibodies in serum provides a marker of previous infection, serological analyses have been used to establish the association between previous chlamydia infection and reproductive complications including PID, ectopic pregnancies and tubal factor infertility.<sup>208</sup> However seroepidemiology has rarely been used to investigate trends in genital infection over time, or as a marker of control of genital chlamydia infections.<sup>99</sup>

This is, in part due to the lack of a suitable assay. Until recently, serological testing for *C. trachomatis*-specific antibodies was problematic as several assays were subject to cross-reactivity with other chlamydiae species<sup>21,208</sup> and sensitivity and specificity of commercially available assays was relatively poor.<sup>99</sup> In recent years, tests with higher specificity and which are not subject to cross-

reaction have become available<sup>209</sup> and present new opportunities for chlamydia seroepidemiology.

At the outset of this PhD, three studies had reported on trends in antibodies to chlamydia using samples collected from pregnant women. In Japan, the percentages testing positive for *C. trachomatis* antibodies (hereafter '*C. trachomatis* antibody seropositive') were compared between pregnant women in 1987, 1992 and 1996/97 (n=275, 297 and 9,652 respectively). Women in the later sample were routinely screened at the time, whereas the earlier results were obtained by testing stored serum samples. The percentage *C. trachomatis* antibody seropositive was found to decline<sup>i</sup> in all age groups, for example from 35% to 31% among women aged 20 to 24 years old.<sup>210</sup> Two studies in Finland have reported trends in *C. trachomatis* antibody seroprevalence among pregnant women up to the age of 28, using stored serum from the Finnish Maternity Cohort serum bank.<sup>211,212</sup> Both studies report a fall in antibody seroprevalence among 23 to 28 year-olds decreased from 22% (95%CI 19%-25%) in 1989-1991 to 12% (95%CI 10%-15%) in 1998 to 2000.<sup>212</sup>

More recently, in the Netherlands, van Aar *et al* have reported results from a study using stored sera from participants in two nationally-representative surveys of the general population in 1995-1996 and 2006-2007. Sera were tested using the Medac quantitative CT IgG ELISA. There was no statistically significant evidence of a difference in seroprevalence between birth cohorts.<sup>132</sup>

<sup>&</sup>lt;sup>i</sup> Statistical significance of the decline was not reported.

In England, two studies to date have investigated trends in percentage C. trachomatis antibody seropositive. Horner et al used residual sera submitted to laboratories in England for routine microbiological or biochemical investigations.<sup>213</sup> A second study by the same group used residual sera from individuals tested for syphilis at two GUM clinics in England.<sup>214</sup> In both studies, sera were tested using an "in-house" indirect immunoglobulin G (IgG) enzymelinked immunosorbent assay (ELISA) based on the Pgp3 antigen (hereafter 'indirect Pgp3 ELISA', see section 8.3.2.1 for more details), which have demonstrates superior sensitivity to commercially-available assays such as that used in the studies in Japan, Finland and the Netherlands above.<sup>209</sup> Reductions in percentage with Pgp3 antibodies detected in serum ('Pgp3 seropositive') among young women were observed in both studies, consistent with a decline in exposure to antibody inducing infection in the mid- to late-2000s, at a time when there was increasing uptake of chlamydia testing and treatment. In the first study using sera collected for routine testing in non-GUM settings, the agestandardised seroprevalence among 17 to 24 year-old women decreased from 20% (95% CI 17%–23%) to 15% (95%CI 12%–17%) in 2010 (p=0.0001).<sup>213</sup> In the second study that used sera from women attending GUM services, the percentage Pgp3 seropositive among women <20 years old decreased from 42% in 1998 to 30% in 2009.214

Antibody seroprevalence surveys are potentially less subject to selection bias as they can use stored or retrospectively collected blood samples that have been collected for a purpose other than chlamydia testing. However the studies above remain subject to bias and confounding. Pregnant women may have a different risk of infection to the general population and seroprevalence trends may be subject to confounding from changes in age-specific characteristics of

79

women who become pregnant. The studies from the UK were based on clinicattending populations rather than nationally-representative samples and in both studies it is also possible that the population who contributed serum samples changed over time. Sera in the study by Horner *et al* were excluded if they were listed as coming from GUM clinics. However it is feasible that the accuracy of coding improved over time meaning that the proportion of sera from high-risk populations may have changed (personal communication, Kate Soldan). Patterns in access to clinics may have affected the population being tested for syphilis testing in the UK study from GUM clinics.<sup>214</sup>

Along with general considerations of bias and confounding, seroprevalence studies are also subject to some specific limitations. The interpretation of agespecific trends in seroprevalence is complicated by the fact that serum antibody levels are likely to depend on several factors including the time since infection, the duration and seriousness of infection as well as number of infections. Thus trends in seroprevalence can only provide an indicator of trends in exposure to antibody-inducing infection, and all tests will underestimate the number of people who have ever been infected. These limitations are discussed further in Chapter 8).

### 3.6 Summary

Studies in Europe and the US that have used population-based methods to estimate chlamydia prevalence at a national level among sexually-experienced under 26 year-olds report prevalence estimates of 3.0% to 5.3% in women and 2.4% to 7.3% in men. Achieving robust measures of chlamydia prevalence is challenging due to the need for appropriate sampling frames with sufficient

80

information to adjust for non-response bias and changes in sampling or response over time.

The percentage testing positive among samples from clinical populations or those accessing testing is typically higher than prevalence estimated in population-based studies as those who access testing will be at higher risk than the general population.

In the absence of RCTs, interpreting trends over time in chlamydia-related outcome measures (chlamydia prevalence, the percentage testing positive, or antibody seroprevalence) is problematic due to changes in the population tested, changes in sexual behaviour and changes in tests. It is possible to control for confounding where all confounders are perfectly measured, but this is unlikely in practice. The confounding effect of changing test technology on trends in chlamydia infection has received some attention in the literature, but there have been relatively few studies that have attempted to control for potential confounding due to behavioural factors.

In the next chapter, I describe an analysis of nationally-collated surveillance data, which was carried out to address the issue of confounding and investigate the potential use of these data for monitoring changes in chlamydia infection over time. 4 Trends in percentage testing positive for chlamydia using routinely-collected surveillance data

As set out in the previous chapter, changes in the percentage testing positive for chlamydia among populations accessing testing may be due to changes in the population tested rather than a change in the underlying amount of infection in the population. Changes in percentages testing positive may also be due to changes in sexual behaviour rather than due to the effect of a chlamydia control intervention such as chlamydia screening. In this chapter, I investigate the use of surveillance data for the purposes of evaluating the impact of chlamydia screening. I explore the extent to which trends in percentage testing positive for chlamydia in England are affected by changes in the characteristics of the population tested and in turn whether these data provide any evidence of there having been a decline in chlamydia infections in the context of widespread chlamydia screening among under-25 year-olds.

### 4.1 Background

Public health surveillance involves the ongoing and systematic collection, analysis and interpretation of health-related data for the purposes of informing public health action, including planning and implementation of services or interventions and evaluation of public health practice.<sup>215,216</sup> Infectious disease surveillance encompasses a spectrum of activities from collation of case reports to active, population-based surveillance that aims to capture the total picture of infection or disease within a population.<sup>216</sup> These different approaches have different strengths and limitations. Systems based on case reports, even if augmented with additional demographic or clinical data, will often rely on routinely-collected information, the quality and completeness of which may vary depending on the setting and purpose for which it is collected. Data collected

82

from people who have accessed health services for testing or diagnosis of an infection will also be biased towards the health-seeking population. For asymptomatic infections such as chlamydia, this is particularly problematic. Systems based on more complete pictures of a population of interest that employ a sampling strategy to achieve a representative population of interest may avoid some of these problems but will often be much more resourceintensive and may still be subject to bias depending on who participates or from whom data are available.

During the course of the 2000s, a number of developments were made in STI surveillance in England. The KC60 data return, which reported aggregate data on STI diagnoses in GUM clinics with limited demographic data, was replaced in 2009 by the genitourinary medicine clinic activity dataset (GUMCAD). GUMCAD is a patient-level data return including all STI diagnoses made (including chlamydia) and services provided (including chlamydia tests) in STI clinics in England along with patient demographic information.<sup>6</sup>

Alongside the implementation of the NCSP from 2003 onwards, data reporting systems were set up to collect information on chlamydia tests carried out through the programme. From 2003 to 2011, all chlamydia tests performed outside of GUM clinics as part of the NCSP were reported by service providers to the Health Protection Agency (HPA)<sup>j</sup> (hereafter termed 'NCSP dataset').<sup>217</sup> The NCSP dataset includes tests performed in a range of settings including GP, community sexual and reproductive health services, community pharmacies,

<sup>&</sup>lt;sup>j</sup> The Health Protection Agency (HPA) was incorporated into Public Health England (PHE) from April 2013.

education and youth venues and tests accessed via the internet. From 2008 to 2011, the HPA also collected aggregate data on tests among 15 to 24 year-olds performed outside of GUM settings, but not reported to the NCSP, via a quarterly return from testing laboratories (the Non-NCSP, Non-GUM dataset, representing 14% to 17% of tests among 15 to 24 year-olds each year from 2008 to 2011).

Taken together, these datasets provide a complete picture of NHScommissioned chlamydia tests among 15 to 24 year-olds in England up to 2011. Since the start of the programme, these data have been used to monitor the performance of local areas and drive service improvement in terms of chlamydia testing coverage and, in more recent years, diagnosis rates.<sup>114,218</sup> While this is a key function of surveillance data, the extent to which this routinely-collected information from individuals accessing chlamydia testing can be used to monitor burden of infection within the population is unclear given both the bias associated with healthcare-attending populations and the considerable changes in the way that screening has been implemented in England over the past decade (section 2.5). As set out in the previous chapter it is feasible that trends in the number of case reports and percentage testing positive for chlamydia may be confounded by a change in the population tested, which in turn may limit the utility of routine data for monitoring the burden of infection.

### 4.2 Aims & objectives

The aim of this analysis was to determine whether trends in the percentage testing positive for chlamydia provide any evidence to suggest a decrease in

chlamydia prevalence since the national implementation of chlamydia screening in 2008. This aim was addressed through the following objectives:

- To describe trends in percentage testing positive for chlamydia among 15 to 24 year-olds tested for chlamydia as part of the National Chlamydia Screening Programme (between 2008 and 2011), or tested in GUM clinics (between 2009 and 2011).
- To investigate the extent to which time trends in percentage testing positive for chlamydia are subject to confounding due to changes over time in the characteristics of the population tested by adjusting trends for a range of sociodemographic and behavioural variables.

# 4.3 Methods

# 4.3.1 Data sources

Data on chlamydia tests among 15 to 24 year-old women and men in England reported through the NCSP dataset or GUMCAD were included<sup>k</sup>. The analysis period varied slightly between datasets: 2008 was included as the earliest year for analysis of the NCSP dataset as this was the first full year of the NCSP; the earliest year for analysis of GUMCAD was 2009 as this was the first year the system was nationally-implemented. From 2012 onwards, the NCSP dataset and the Non-NCSP Non-GUM dataset were replaced by a laboratory-based reporting system (the Chlamydia Testing Activity Dataset, CTAD). This dataset does not incorporate the same demographic and sexual behaviour variables as

<sup>&</sup>lt;sup>k</sup> Data are subject to minor revision due to retrospective submission of data, thus exact numbers may vary from published figures. These analyses are based on extracts dated February 2012 (NCSP dataset) and 17 August 2012 (GUMCAD).

the NCSP dataset, thus analyses were limited to tests performed up to the end of 2011.

Within the NCSP dataset, test records with a matching postcode of residence, date of birth and gender were considered to refer to the same person and were used for de-duplication.<sup>1</sup> In GUMCAD, individuals with more than one clinic visit were linked using their clinic number (this is unique within a clinic)<sup>m</sup>. All analyses were conducted on the basis of 'testing episodes' Test (and diagnosis) records within 6 weeks of a previous test were considered to be the same testing episode and were excluded. Individuals were allowed to contribute more than one testing episode during the analysis period. Analyses were carried out separately in the NCSP and GUMCAD datasets, as although individuals can attend more than one service they cannot be linked between the datasets.

Demographic, test and behavioural information are collected in the NCSP dataset and GUMCAD, although different items are collected in each (Table 4-1). An area-level indicator of socioeconomic deprivation (the Index of Multiple Deprivation, IMD<sup>220</sup>) was assigned to all test records by mapping postcodes of residence to lower super output areas (LSOA, an area of average population of 1,500 individuals<sup>221</sup>) of residence. Ranks of IMD scores were grouped into quintiles. Having a previous chlamydia diagnosis is a known risk factor for

<sup>&</sup>lt;sup>1</sup>Where a test record in the NCSP dataset had a postcode of residence equal to the postcode of testing venue, the postcode of residence was set to 'missing', as this was considered a likely data entry error.

<sup>&</sup>lt;sup>m</sup> Chlamydia tests and diagnoses are recorded separately in GUMCAD, and may be recorded on different dates. All recorded sexual health screens were considered to include a chlamydia test (as specified in coding guidelines<sup>219</sup>). A test was considered positive if there was a chlamydia diagnosis recorded on the same day or within the following 6 weeks. A chlamydia diagnosis reported without an accompanying sexual health screen was included as a positive test.

subsequent infection.<sup>48,51,56</sup> Testing episodes were therefore categorised as having a recorded chlamydia diagnosis in the past 12 months by linking tests records for individuals within each dataset. The sexual behaviour variables were combined into a single 'risk group' variable, in order to avoid collinearity in multivariable models (high risk: answered 'yes' to both sexual behaviour variables; medium risk: answered 'yes' to one sexual behaviour variable; low risk: answered 'no' to both sexual behaviour variables).

Table 4-1: Comparison of data items included in the NCSP dataset and the genitourinary medicine clinic activity dataset (GUMCAD) used in the analysis

	NCSP Dataset	GUMCAD
Date of test	1	1
Year of age	1	1
Ethnic group	1	1
Region (Strategic Health Authority)	1	1
LSOA of residence*	1	1
Testing venue type	1	Not applicable
Specimen type (urine/vulvovaginal swab/cervical swab)	1	×
Reports having a new sexual partner in the previous three months (yes/no)^	✓	×
Reports and more than one sexual partners in the previous 12 months (yes/no) <sup>^</sup>	1	×
Sexual orientation <sup>\$</sup>	×	1

\*LSOA: Lower super output area

<sup>^</sup>These sexual behaviour variables were included in the 'core dataset', which all local areas were required to submit as part of the NCSP dataset return. Other optional items were available as part of the NCSP dataset (including condom use), but were not included in this analysis as not all areas returned these items.

<sup>\$</sup>Sexual orientation is collected in GUMCAD at each clinic visit. Men who reported having sex with a man at any clinic visits were categorised as men who have sex with men (MSM).

# 4.3.2 Statistical analyses

Test, demographic and sexual behavioural characteristics were compiled for

NCSP and GUMCAD testing episodes. The total number of testing episodes,

percentage testing positive and reported test, demographic and behavioural characteristics, are presented by year, calendar quarter and dataset.

The relative annual change in the percentage testing positive was estimated using univariable logistic regression, with test result included as the outcome variable, and year as a continuous predictor variable. Odds ratios were interpreted as a reasonable proxy for relative risk, due to the generally low prevalence of chlamydia in these analyses.<sup>222</sup> Annual trends were calculated by gender and other factors including age, venue of test, sexual risk group (NCSP dataset) and sexual orientation (GUMCAD).

It is reasonable to assume that the likelihood of attending a GUM clinic with a symptomatic STI may be less subject to changes in sexual health policies and service provision than the likelihood of an asymptomatic attendance, meaning individuals attending with symptoms may provide a more stable population in which to monitor chlamydia infection. The proportion of attendances in 2009 to 2011 where a chlamydia testing episode was recorded was higher for those where a diagnosis of genital warts, candidosis, genital herpes or - for women - bacterial vaginosis was made (Table 4-2). Trends in percentage testing positive for chlamydia were therefore also explored in a subgroup analysis limited to GUM clinic attendances where a diagnosis of these selected symptomatic conditions was made.

	Wor	nen	М	en
	Ν	% tested	N	% tested
All GUM attendances	1,628,320	59%	996,862	63%
Attendances with diagnosis of selected symptomatic condition:				
Genital warts	68,148	83%	57,025	85%
Candidosis	121,518	88%	12,968	87%
Genital herpes	26,307	66%	10,782	85%
Bacterial vaginosis	137,552	92%		

# Table 4-2: Proportion of GUM attendances with a chlamydia testing episoderecorded, by gender (15 to 24 year-old women and men, GUMCAD, 2009-2011)

# 4.3.3 Factors associated with testing positive

The association between demographic and sexual behaviour variables and testing positive for chlamydia were explored using univariable and multivariable logistic regression. All variables significant in univariable analyses (p<0.05) were entered into a multivariable model.

# 4.3.4 Assessment of confounding of time trends in percentage testing positive introduced by changes in risk factors over time

In order to investigate potential confounding of trends in percentage testing positive for chlamydia arising from changes in the population tested, a series of 'bivariable' logistic regression models were constructed, which included test result as the outcome variable, and year (entered as a continuous variable) and one other variable as predictors (e.g. year and age, year and ethnicity, year and region). The estimated annual change in percentage testing positive was compared between the models when each variable was included or excluded to explore whether allowing for changes in each variable would change the estimated trend in percentage testing positive. Variables that were significantly associated (p<0.05) with testing positive in each of the bivariable models were included in a multivariable model, along with year.

NCSP testing episodes had considerable levels of item non-response for variables of interest (Table 4-3). The percentage of testing episodes with missing data increased between 2008 and 2011, from 42% to 58% among women and 58% to 69% among men, and varied by venue of testing service. The lowest levels of missing data were seen for testing episodes returned through remote testing (29% missing data among women, 32% among men). The percentage testing positive was higher among testing episodes with complete data compared to those with missing data for both women (6.9%) versus 6.0%) and men (6.4% versus 4.5%). As the purpose of the analysis was to compare the impact on the annual change in percentage testing positive when adjusting for different variables, this analysis was limited to testing episodes with complete data on all of the variables of interest. A sensitivity analysis was carried out to determine whether the same patterns were observed when using data that were complete on each individual item of interest. Sensitivity analyses were not carried out in the GUMCAD analysis as the percentage of GUMCAD tests with missing data was lower (13.9%).

All tabulations and models were constructed separately by gender and by dataset. All statistical analyses were performed using Stata version 12.0 (StataCorp, College Station, Texas, USA). Statistical significance is considered as p<0.05 for all analyses.

90

	Women	1	Men	
	Ν	%	Ν	%
	(Total=2,564,925)	missing	(Total=1,537,178)	missing
By variable of interest				
>1 sexual partner in previous year		43%		56%
At least 1 new sexual partner in previous 3 months		40%		53%
Ethnicity		25%		34%
IMD of LSOA of residence		10%		17%
Known chlamydia diagnosis in previous year		10%		17%
Specimen type		2%		n/a
One or more of the above		52%		66%
By venue type of testing				
Hospital	90,751	51%	15,281	54%
Outreach	305,175	74%	441,452	78%
Occupational Health	717	54%	1,298	73%
Pharmacy	71,300	41%	27,192	47%
Sexual and Reproductive	767 747	100/	207 617	E 40/
Conorol Proctico	101,141	40%	207,017	0470 400/
	475,269	47%	157,400	40%
	257,782	29%	151,556	32%
Military Prisons & Youth Offender	5,590	87%	55,242	93%
Institutions <sup>\$</sup>	4,758	89%	55,396	96%
Chlamydia Screening Offices	114,054	50%	87,836	60%
Abortion services	89,639	57%	884	56%
Education	297,569	69%	278,137	75%
Youth	84,574	53%	57,887	61%
By year of test				
2008	385,483	42%	181,539	58%
2009	645,276	48%	357,241	61%
2010	830,172	56%	569,272	68%
2011	703,994	58%	429,126	69%

# Table 4-3: Percentage of tests with missing data on one or more variable (NCSP dataset, 15-24 year-old women and men, 2008-2011)

*IMD:* Index of multiple deprivation, quintile based on lower super output area (LSOA) of residence; \*All tests had age, gender and venue of test; <sup>\$</sup>Missing data in these venues is largely accounted for by missing postcode of residence, as the validation algorithm for postcodes did not allow postcode of residence was not allowed to equal the postcode of testing venue.

### 4.4 Results

# 4.4.1 Number of tests and percentage testing positive in the NCSP dataset and GUMCAD

Between 2008 and 2011, 4,102,103 chlamydia testing episodes among 15 to 24 year-olds (62.5%, [2,564,925] in women) were reported in the NCSP dataset, of which 6.5% among women and 5.2% among men resulted in a positive diagnosis. Between 2009 and 2011 there were 1,585,395 testing episodes (60.5% [958,864] in women) reported in GUMCAD; 11.2% in women and 12.6% among men were positive.

Between 2008 and 2011, the number of NCSP chlamydia testing episodes almost doubled, from 567,022 to 1,133,120 (Figure 4-1a, Table 4-4). The percentage testing positive during this period decreased among both women (from 8.7% to 5.8%, OR per additional year 0.86, 95% confidence interval[CI] 0.86 to 0.87) and men (from 7.7% to 4.8%, OR 0.84, 95%CI 0.83-0.84).

The number of GUMCAD chlamydia testing episodes remained relatively stable during the analysis period (Figure 4-1b, Table 4-5). The percentage testing positive declined from 11.8% in 2009 to 10.8% 2011 among women (OR 0.95, 95%CI 0.94 to 0.95) and 12.9% to 12.4% among men (OR 0.98, 95%CI 0.97 to 0.99).





### 4.4.2 Factors associated with testing positive

All variables were statistically significantly associated with testing positive in univariable and multivariable analyses for NCSP (Table 4-4) and GUMCAD testing episodes (Table 4-5). Among NCSP testing episodes, there was substantial variation in the percentage testing positive by the type of testing venue, which remained after adjusting for the other variables in the multivariable analysis (Figure 4-2; Table 4-4). Testing episodes in educational venues and outreach settings were less likely to result in a positive diagnosis than those in general practice, whereas tests in sexual and reproductive health services, abortion services, prisons and hospital settings were more likely to be positive.

In NCSP testing episodes, reporting both a new sexual partner in the past three months and more than one sexual partner in the previous year was associated with at least a two-fold increase in the odds of testing positive compared with those reporting no more than one sexual partner in the last year and no new sexual partner in the last three months (women: AOR 2.24, 95%Cl 2.20-2.28; men: AOR 2.09, 95%Cl 2.03-2.16). In both the NCSP dataset and GUMCAD, the odds of testing positive were higher in those with a known chlamydia diagnosis than those without in both women (AOR NCSP: 1.13, 95%Cl 1.09-1.18; AOR GUMCAD: 1.04, 95%Cl 1.01-1.07) and men (NCSP AOR: 1.27, 1.17-1.37; GUMCAD AOR: 1.15, 1.10-1.20).

Being of black or mixed ethnicity and living in more deprived LSOA was significantly associated with having higher odds of testing positive after controlling for other variables, with the exception of in GUMCAD testing episodes among women, where those of black ethnicity had a lower odds pf testing positive than those of white ethnicity. MSM had a lower risk of testing positive compared to non-MSM in both univariable and multivariable analyses (AOR, 0.64 95%CI 0.62-0.67) of GUMCAD testing episodes.

Figure 4-2: Adjusted\* odds ratio (and 95% confidence intervals) of testing positive by venue of test and gender, compared to General Practice (NCSP testing episodes, 15-24 year-old women and men)



\*Adjusted for age, ethnicity, risk group, IMD quintile of LSOA of residence, known chlamydia diagnosis in the previous 12 months and year of test.

SRH: Sexual and reproductive health services; CSO: chlamydia screening office; YOI: Youth offending institution. Outreach: tests performed in non-clinical settings including entertainment and leisure venues; Remote testing: tests performed without face to face contact with a health professional at the time of test, includes tests performed through the internet.

Table 4-4 (a): Number of tests, percentage testing positive, unadjusted and adjusted odds ratios for testing positive by sociodemographic, sexual behavioural, and test characteristics (NCSP dataset, 15-24 year-old <u>women</u>, 2008-2011)

	Number of	%	Univariable analysis		Multivariable analysis			
	tests	positive	∍OR	95%CI	р	AOR	95%CI	p
Overall	2,564,925	6.5%						
Sociodemographic								
characteristics								
Age (years)								
15	146,897	5.2%	Ref		< 0.001	Ref		0.000
16	298,915	5.8%	1.11	1.08-1.14		1.22	1.17 -1.27	
17	343,314	6.8%	1.32	1.29-1.36		1.44	1.39 -1.50	
18	341,651	7.5%	1.47	1.43 -1.51		1.58	1.52 -1.64	
19	311,826	7.6%	1.50	1.46 1.54		1.60	1.54 -1.66	
20	278,653	7.2%	1.41	1.38-1.45		1.51	1.45 -1.57	
21	243,679	6.7%	1.30	1.26 - 1.34		1.38	1.33-1.44	
22	216,592	5.9%	1.13	1.10-1.16		1.21	1.16-1.26	
23	199,985	5.0%	0.96	0.93-0.99		1.03	0.98 -1.07	
24	183,413	4.6%	0.87	0.84 -0.89		0.95	0.91 -1.00	
Ethnic group								
White	1,638,982	6.8%	Ref		< 0.001	Ref		0.000
Black	107,978	8.1%	1.22	1.19-1.25		1.32	1.28 -1.36	
Asian	77,163	3.3%	0.47	0.45 -0.49		0.57	0.54 -0.60	
Chinese	9,401	4.7%	0.68	0.62-0.75		0.92	0.82 -1.03	
Other	12,486	5.5%	0.80	0.74 -0.86		0.95	0.86 -1.06	
Mixed	70,359	8.6%	1.29	1.26 - 1.33		1.28	1.24 -1.32	
Unknown	648,556	5.5%						
Region								
London	385,755	5.7%	Ref		0.000	Ref		0.000
North East	113,668	6.5%	1.15	1.12 - 1.18		1.47	1.41 -1.53	
North West	379,340	7.9%	1.40	1.38 - 1.43		1.35	1.32 - 1.39	
Yorkshire & Humber	232,092	7.3%	1.29	1.27 - 1.32		1.36	1.32 - 1.40	
East Midlands	203,013	6.2%	1.08	1.06 - 1.11		1.25	1.21 -1.29	
West Midlands	246,320	6.5%	1.14	1.12 - 1.17		1.28	1.24 -1.32	
East of England	229,985	5.6%	0.96	0.94 - 0.99	1	1.09	1.06 - 1.12	
South East Coast	152,567	5.6%	0.97	0.95 - 1.00	1	1.02	0.98-1.06	
South Central	169,175	5.7%	1.00	0.97 - 1.02		1.08	1.04 -1.12	
South West	202,890	6.9%	1.21	1.18 - 1.23		1.20	1.16-1.24	
Other / Unknown	250,120	6.2%						
IMD quintile of								
LSOA of residence <sup>y</sup>								
Least deprived	317,630	5.1%	Ref		<0.001	Ref		<0.001
2	356,775	5.7%	1.13	1.11 -1.15		1.09	1.06 - 1.12	
3	428,577	6.0%	1.19	1.17 -1.22		1.17	1.14 -1.20	
4	541,366	6.7%	1.33	1.30 - 1.35		1.28	1.25 - 1.31	
Most deprived	670,007	7.7%	1.54	1.51 -1.57		1.42	1.39-1.46	
Unknown	250,570	6.2%						

# Table 4-4(a) continued.

Text in Der of %     OR     PSK2Li     P     AOR 95%Cl     p       Sexual behaviour characteristics     >1 sexual partner in last year <td< th=""><th></th><th>Number of</th><th colspan="2">Univariable analysis</th><th colspan="3">Multivariable analysis</th></td<>		Number of	Univariable analysis		Multivariable analysis				
Sexual behaviour characteristics     Joint of 1     Original Prior 1     Internet of 1000 (1000)     P       >1 sexual partner in last year     >1.096.033     5.7%      <0.001         Yes     670.897     9.4%     1.97     1.95-2.00          Yes     670.897     9.4%     1.97     1.95-2.00 <t< th=""><th></th><th>tests</th><th>% nositive</th><th></th><th>able analys</th><th>n</th><th>AOR</th><th>95%CI</th><th>n</th></t<>		tests	% nositive		able analys	n	AOR	95%CI	n
Characteristics       Sexual partner in last year       No     670,897     9.4%     1.97     1.95-2.00       Yes     670,897     9.4%     1.97     1.95-2.00       Unknown     1,096,033     5.7%     Partners in last year     Sexual year     Sexual year     Sexual year       Partner in last 3 months     Sexual year     S.3%     Ref     < 0.001     Sexual year       Ves     721,027     8.9%     1.73     1.71-1.75     Sexual year       Unknown     1.05,701     5.5%     Ref     < 0.001     Ref     < 0.001       Medium     405,590     7.4%     1.66     1.63-1.69     1.65     1.62-1.68       Low     607,446     4.6%     Ref     < 0.001     Ref     < 0.001       Medium     405,590     7.4%     1.66     1.63-1.69     1.65     1.62-1.68       Low     475,269     6.1%     Ref     < 0.001     Ref     < 0.001       Mean     4075,262     6.4%     1.10     1.07-1.14     1.10	Sexual behaviour	10010	positive		,0,101	P			P
Secural partner in last year     No     797,995     5.0%     Ref     <0.001       No     670,897     9,4%     1.97     1.95-2.00	characteristics								
No     797,995     5.0%     Ref     <0.001       Yes     670,897     9.4%     1.97     1.95-2.00       Unknown     1,096,033     5.7%     1.95-2.00     1.95-2.00       ≥1 new sexual partner in last 3     months     -     -     -       No     828,197     5.3%     Ref     <0.001	>1 sexual partner in								
No   797,993   3.0.%   Ref   <0.001     Yes   670,897   9.4%   1.95   20.001     21 new sexual partner in last 3 months   5.3%   Ref   <0.001	No	707 005	E 00/	Dof		-0.001			
Test   070,097   34.46   1.97	NU	797,995	5.0% 0.4%	KU 1 07	1 05 2 00	<0.001			
Dirkholm     1,030,033     0.7 /b       2 <sup>+</sup> new sexual partner in last 3 months     -	Linknown	1 006 022	9.4 /0 5 70/	1.97	1.95-2.00				
Partner in last 3 months     Solution     Sexual risk group*     Co.001       Yes     721,027     8.9%     1.73     1.71 - 1.75       Unknown     1,015,701     5.6%     -		1,090,033	J.1 /0						
Participant in factor     Solution	<u>- nartner in last 3</u>								
No     828,197     5.3%     Ref     <0.001       Yes     721,027     8.9%     1.73     1.71-1.75       Unknown     1,015,701     5.6%     -     -     -     -     -     -     -     -     -     -     -     -     -     0.001     Ref     <	months								
Yes     721,027     8.9%     1.73     1.71 - 1.75       Unknown     1,015,701     5.6%     -	No	828 197	5.3%	Ref		<0 001			
Unknown   1,015,701   5.6% <td>Yes</td> <td>721 027</td> <td>8.9%</td> <td>1 73</td> <td>1 71 -1 75</td> <td><b>NO.001</b></td> <td></td> <td></td> <td></td>	Yes	721 027	8.9%	1 73	1 71 -1 75	<b>NO.001</b>			
Sexual risk group*     (101)     0.000       Low     607,446     4.6%     Ref     <0.001	Unknown	1 015 701	5.6%	1.70					
Low     607,446     4.6%     Ref     <0.001     Ref     <0.001       Medium     405,590     7.4%     1.66     1.63 + 1.69     1.65     1.62 + 1.68       High     493,167     9.8%     2.28     2.24 + 2.31     2.24     2.20 + 2.28       Unknown     1,058,722     5.6%     1.00     1.07 + 1.14     1.10     1.06 + 1.14       Venue of test <sup>6</sup> 6.0%     1.10     1.07 + 1.14     1.10     1.06 + 1.14       Outreach     305,175     4.2%     0.69     0.67 - 0.70     0.68     0.66 - 0.71       Occupational Health     717     7.8%     1.31     1.00 - 1.73     1.26     0.86 - 1.83       Pharmacy     71,300     7.0%     1.18     1.14 + 1.21     1.03     0.99 + 1.07       Sexual and     Reproductive Health     Services     767,747     8.4%     1.43     1.41 - 1.45     1.14     1.11 - 1.16       Services     767,747     8.4%     1.60     1.45 - 1.77     1.47     1.14 - 1.90       Offending Institutions 4,758     9.4%<	Sexual risk group <sup>x</sup>	1,010,101	0.070						
Medium     405,590     7.4%     1.66     1.63 + 1.69     1.65     1.62 + 1.68       High     493,167     9.8%     2.28     2.24 + 2.31     2.24     2.20 + 2.28       Unknown     1,058,722     5.6%     7.4%     1.66     1.61     1.65     1.62 + 1.68       Venue of test <sup>2</sup> 5.6%     7.8%     2.28     2.24 + 2.31     2.24     2.20 + 2.28       Venue of test <sup>2</sup> 6.6%     1.10     1.07 + 1.14     1.10     1.06 + 1.14       Occupational Health     717     7.8%     1.31     1.00 - 1.73     1.26     0.86 - 0.71       Occupational Health     717     7.8%     1.31     1.01 - 1.73     1.26     0.86 - 1.83       Pharmacy     71,300     7.0%     1.18     1.14 - 1.21     1.03     0.99 - 1.07       Sexual and     Reproductive Health     Services     767,747     8.4%     1.43     1.41 - 1.45     1.14     1.11 - 1.16       Offending Institutions     4.758     9.4%     1.60     1.45 - 1.77     1.47     1.14 - 1.90	Low	607,446	4.6%	Ref		< 0.001	Ref		<0.001
High   493,167   9.8%   2.28   2.24-2.31   2.24   2.20-2.28     Unknown   1.058,722   5.6%	Medium	405.590	7.4%	1.66	1.63-1.69		1.65	1.62 - 1.68	
Unknown     1,058,722     5.6%     International constraints       Characteristics     Venue of test <sup>2</sup> 6.1%     Ref     <0.001	High	493,167	9.8%	2.28	2.24 - 2.31		2.24	2.20-2.28	
Test and clinical characteristics	Unknown	1.058.722	5.6%	2.20	2.2 1 2.01			2.20 2.20	
characteristics       Venue of test <sup>2</sup> GP     475,269     6.1%     Ref     <0.001     Ref     <0.001       Hospital     90,751     6.6%     1.10     1.07 -1.14     1.10     1.06 -1.14       Outreach     305,175     4.2%     0.69     0.67 -0.70     0.68     0.66 -0.71       Occupational Health     717     7.8%     1.31     1.00 -1.73     1.26     0.86 -1.83       Pharmacy     71,300     7.0%     1.18     1.14 -1.21     1.03     0.99 -1.07       Sexual and     Reproductive Health     Services     767,747     8.4%     1.60     1.41 -1.45     1.14     1.11 -1.16       Remote testing     257,782     6.4%     1.06     1.04 -1.08     0.94     0.91 -0.96       Military     5,590     6.5%     1.08     0.97 -1.20     1.19     0.92 - 1.54       Prisons & Youth     0ffending Institutions     4,758     9.4%     1.60     1.45 -1.77     1.47     1.14 -1.90       Olfamydia Screening     114,054	Test and clinical	.,							
Venue of test <sup>2</sup> GP     475,269     6.1%     Ref     <0.001	characteristics								
GP   475,269   6.1%   Ref   <0.001   Ref   <0.001     Hospital   90,751   6.6%   1.10   1.07 - 1.14   1.10   1.06 - 1.14     Outreach   305,175   4.2%   0.69   0.67 - 0.70   0.68   0.66 - 0.71     Occupational Health   717   7.8%   1.31   1.00 - 1.73   1.26   0.86 - 1.83     Pharmacy   71,300   7.0%   1.18   1.14 - 1.21   1.03   0.99 - 1.07     Sexual and   Reproductive Health   1.18   1.41 - 1.45   1.14   1.11 - 1.16     Remote testing   257,782   6.4%   1.06   1.04 - 1.08   0.94   0.91 - 0.96     Military   5,590   6.5%   1.08   0.97 - 1.20   1.19   0.92 - 1.54     Prisons & Youth   4,758   9.4%   1.60   1.45 - 1.77   1.47   1.14 - 1.90     Offices   114,054   6.8%   1.13   1.10 - 1.16   1.01   0.98 - 1.05     Abortion services   89,639   7.1%   1.18   1.15 - 1.22   1.13   1.09 - 1.18     Education   207,569	Venue of test <sup>z</sup>								
Hospital   90,751   6.6%   1.10   1.07 - 1.14   1.10   1.06 - 1.14     Outreach   305,175   4.2%   0.69   0.67 - 0.70   0.68   0.66 - 0.71     Occupational Health   717   7.8%   1.31   1.00 - 1.73   1.26   0.86 - 1.83     Pharmacy   71,300   7.0%   1.18   1.14 - 1.21   1.03   0.99 - 1.07     Sexual and   Reproductive Health   Services   767,747   8.4%   1.43   1.41 - 1.45   1.14   1.11 - 1.16     Remote testing   257,782   6.4%   1.06   1.04 - 1.08   0.94   0.91 - 0.96     Military   5,590   6.5%   1.08   0.97 - 1.20   1.19   0.92 - 1.54     Prisons & Youth   Offending Institutions   4,758   9.4%   1.60   1.45 - 1.77   1.47   1.14 - 1.90     Chlamydia Screening   Offices   114,054   6.8%   1.13   1.10 - 1.16   1.01   0.98 - 1.05     Abortion services   89,639   7.1%   1.32   1.29 - 1.36   1.03   0.99 - 1.07     Specimen type   Urine   1,376,209<	GP	475,269	6.1%	Ref		<0.001	Ref		<0.001
Outreach Occupational Health     305,175     4.2%     0.69     0.67 - 0.70     0.68     0.66 - 0.71       Occupational Health     717     7.8%     1.31     1.00 - 1.73     1.26     0.86 - 1.83       Pharmacy     71,300     7.0%     1.18     1.14 - 1.21     1.03     0.99 - 1.07       Sexual and Reproductive Health     767,747     8.4%     1.43     1.41 - 1.45     1.14     1.11 - 1.16       Remote testing     257,782     6.4%     1.06     1.04 - 1.08     0.94     0.91 - 0.96       Military     5,590     6.5%     1.08     0.97 - 1.20     1.19     0.92 - 1.54       Prisons & Youth     916     1.45     1.47     1.47     1.14 - 1.90       Chlamydia Screening     4,758     9.4%     1.60     1.45 - 1.77     1.47     1.14 - 1.90       Chlamydia Screening     297,569     3.3%     0.52     0.55     0.51     0.49 - 0.53       Youth services     84,574     7.9%     1.32     1.29 - 1.36     1.03     0.99 - 1.07       Specimen type     Urine <td>Hospital</td> <td>90,751</td> <td>6.6%</td> <td>1.10</td> <td>1.07 - 1.14</td> <td></td> <td>1.10</td> <td>1.06 - 1.14</td> <td></td>	Hospital	90,751	6.6%	1.10	1.07 - 1.14		1.10	1.06 - 1.14	
Occupational Health     717     7.8%     1.31     1.00 - 1.73     1.26     0.86 - 1.83       Pharmacy     71,300     7.0%     1.18     1.14 - 1.21     1.03     0.99 - 1.07       Sexual and     Reproductive Health     Services     767,747     8.4%     1.43     1.41 - 1.45     1.14     1.11 - 1.16       Remote testing     257,782     6.4%     1.06     1.04 - 1.08     0.94     0.91 - 0.96       Military     5,590     6.5%     1.08     0.97 - 1.20     1.19     0.92 - 1.54       Prisons & Youth     Offending Institutions     4,758     9.4%     1.60     1.45 - 1.77     1.47     1.14 - 1.90       Chlamydia Screening     Offences     114,054     6.8%     1.13     1.10 - 1.16     1.01     0.98 - 1.05       Abortion services     89,639     7.1%     1.18     1.15 - 1.22     1.13     1.09 - 1.18       Education     297,569     3.3%     0.53     0.52 - 0.55     0.51     0.49 - 0.53       Youth services     84,574     7.9%     1.32     1.10 - 1.16	Outreach	305,175	4.2%	0.69	0.67 -0.70		0.68	0.66 -0.71	
Pharmacy Sexual and Reproductive Health   71,300   7.0%   1.18   1.14 - 1.21   1.03   0.99 - 1.07     Sexual and Reproductive Health   Services   767,747   8.4%   1.43   1.41 - 1.45   1.14   1.11 - 1.16     Services   767,747   8.4%   1.06   1.04 - 1.08   0.94   0.91 - 0.96     Military   5,590   6.5%   1.08   0.97 - 1.20   1.19   0.92 - 1.54     Prisons & Youth   Offending Institutions   4,758   9.4%   1.60   1.45 - 1.77   1.47   1.14 - 1.90     Chlamydia Screening   Offices   114,054   6.8%   1.13   1.10 - 1.16   1.01   0.98 - 1.05     Abortion services   89,639   7.1%   1.18   1.15 - 1.22   1.13   1.09 - 1.18     Education   297,569   3.3%   0.53   0.52 - 0.55   0.51   0.49 - 0.53     Youth services   84,574   7.9%   1.32   1.29 - 1.36   1.03   0.99 - 1.07     Specimen type   Urine   1,376,209   6.2%   Ref   <0.001	Occupational Health	717	7.8%	1.31	1.00-1.73		1.26	0.86 -1.83	
Sexual and   Reproductive Health     Services   767,747   8.4%   1.43   1.41 - 1.45   1.14   1.11 - 1.16     Remote testing   257,782   6.4%   1.06   1.04 - 1.08   0.94   0.91 - 0.96     Military   5,590   6.5%   1.08   0.97 - 1.20   1.19   0.92 - 1.54     Prisons & Youth   Offending Institutions   4,758   9.4%   1.60   1.45 - 1.77   1.47   1.14 - 1.90     Chlamydia Screening   Offices   114,054   6.8%   1.13   1.10 - 1.16   1.01   0.98 - 1.05     Abortion services   89,639   7.1%   1.18   1.15 - 1.22   1.13   1.09 - 1.18     Education   297,569   3.3%   0.53   0.52 - 0.55   0.51   0.49 - 0.53     Youth services   84,574   7.9%   1.32   1.29 - 1.36   1.03   0.99 - 1.07     Specimen type   Urine   1,376,209   6.2%   Ref   <0.001	Pharmacy	71,300	7.0%	1.18	1.14-1.21		1.03	0.99 -1.07	
Reproductive Health   Services   767,747   8.4%   1.43   1.41 - 1.45   1.14   1.11 - 1.16     Remote testing   257,782   6.4%   1.06   1.04 - 1.08   0.94   0.91 - 0.96     Military   5,590   6.5%   1.08   0.97 - 1.20   1.19   0.92 - 1.54     Prisons & Youth   Offencing Institutions   4,758   9.4%   1.60   1.45 - 1.77   1.47   1.14 - 1.90     Chlamydia Screening   Offices   114,054   6.8%   1.13   1.10 - 1.16   1.01   0.98 - 1.05     Abortion services   89,639   7.1%   1.18   1.15 - 1.22   1.13   1.09 - 1.18     Education   297,569   3.3%   0.53   0.52 - 0.55   0.51   0.49 - 0.53     Youth services   84,574   7.9%   1.32   1.29 - 1.36   1.03   0.99 - 1.07     Specimen type   Urine   1,376,209   6.2%   Ref   <0.001	Sexual and								
Services     767,747     8.4%     1.43     1.41 - 1.45     1.14     1.11 - 1.16       Remote testing     257,782     6.4%     1.06     1.04 - 1.08     0.94     0.91 - 0.96       Military     5,590     6.5%     1.08     0.97 - 1.20     1.19     0.92 - 1.54       Prisons & Youth     0     0     1.45 - 1.77     1.47     1.14 - 1.90       Offending Institutions     4,758     9.4%     1.60     1.45 - 1.77     1.47     1.14 - 1.90       Chlamydia Screening     0     0     1.18     1.10 - 1.16     1.01     0.98 - 1.05       Abortion services     89,639     7.1%     1.18     1.15 - 1.22     1.13     1.09 - 1.18       Education     297,569     3.3%     0.53     0.52 - 0.55     0.51     0.49 - 0.53       Youth services     84,574     7.9%     1.32     1.29 - 1.36     1.03     0.99 - 1.07       Specimen type     Urine     1,376,209     6.2%     Ref     <0.001	Reproductive Health								
Remote testing   257,782   6.4%   1.06   1.04 - 1.08   0.94   0.91 - 0.96     Military   5,590   6.5%   1.08   0.97 - 1.20   1.19   0.92 - 1.54     Prisons & Youth   0   1.45 - 1.77   1.47   1.14 - 1.90     Offending Institutions   4,758   9.4%   1.60   1.45 - 1.77   1.47   1.14 - 1.90     Chlamydia Screening   0   6.8%   1.13   1.10 - 1.16   1.01   0.98 - 1.05     Abortion services   89,639   7.1%   1.18   1.15 - 1.22   1.13   1.09 - 1.18     Education   297,569   3.3%   0.53   0.52 - 0.55   0.51   0.49 - 0.53     Youth services   84,574   7.9%   1.32   1.29 - 1.36   1.03   0.99 - 1.07     Specimen type   Urine   1,376,209   6.2%   Ref   <0.001	Services	767,747	8.4%	1.43	1.41 -1.45		1.14	1.11 -1.16	
Military   5,590   6.5%   1.08   0.97 - 1.20   1.19   0.92 - 1.54     Prisons & Youth   Offending Institutions   4,758   9.4%   1.60   1.45 - 1.77   1.47   1.14 - 1.90     Chlamydia Screening   Offices   114,054   6.8%   1.13   1.10 - 1.16   1.01   0.98 - 1.05     Abortion services   89,639   7.1%   1.18   1.15 - 1.22   1.13   1.09 - 1.18     Education   297,569   3.3%   0.53   0.52 - 0.55   0.51   0.49 - 0.53     Youth services   84,574   7.9%   1.32   1.29 - 1.36   1.03   0.99 - 1.07     Specimen type   Urine   1,376,209   6.2%   Ref   <0.001	Remote testing	257,782	6.4%	1.06	1.04 - 1.08		0.94	0.91 -0.96	
Prisons & Youth   Offending Institutions   4,758   9.4%   1.60   1.45 - 1.77   1.47   1.14 - 1.90     Chlamydia Screening   Offices   114,054   6.8%   1.13   1.10 - 1.16   1.01   0.98 - 1.05     Abortion services   89,639   7.1%   1.18   1.15 - 1.22   1.13   1.09 - 1.18     Education   297,569   3.3%   0.53   0.52 - 0.55   0.51   0.49 - 0.53     Youth services   84,574   7.9%   1.32   1.29 - 1.36   1.03   0.99 - 1.07     Specimen type   Urine   1,376,209   6.2%   Ref   <0.001	Military	5,590	6.5%	1.08	0.97 -1.20		1.19	0.92 -1.54	
Offending Institutions   4,758   9.4%   1.60   1.45 - 1.77   1.47   1.14 - 1.90     Chlamydia Screening   0ffices   114,054   6.8%   1.13   1.10 - 1.16   1.01   0.98 - 1.05     Abortion services   89,639   7.1%   1.18   1.15 - 1.22   1.13   1.09 - 1.18     Education   297,569   3.3%   0.53   0.52 - 0.55   0.51   0.49 - 0.53     Youth services   84,574   7.9%   1.32   1.29 - 1.36   1.03   0.99 - 1.07     Specimen type   Urine   1,376,209   6.2%   Ref   <0.001	Prisons & Youth								
Chlamydia Screening   Offices   114,054   6.8%   1.13   1.10 - 1.16   1.01   0.98 - 1.05     Abortion services   89,639   7.1%   1.18   1.15 - 1.22   1.13   1.09 - 1.18     Education   297,569   3.3%   0.53   0.52 - 0.55   0.51   0.49 - 0.53     Youth services   84,574   7.9%   1.32   1.29 - 1.36   1.03   0.99 - 1.07     Specimen type   Urine   1,376,209   6.2%   Ref   <0.001	Offending Institutions	4,758	9.4%	1.60	1.45 -1.77		1.47	1.14 -1.90	
Offices   114,054   6.8%   1.13   1.10 - 1.16   1.01   0.98 - 1.05     Abortion services   89,639   7.1%   1.18   1.15 - 1.22   1.13   1.09 - 1.18     Education   297,569   3.3%   0.53   0.52 - 0.55   0.51   0.49 - 0.53     Youth services   84,574   7.9%   1.32   1.29 - 1.36   1.03   0.99 - 1.07     Specimen type   Urine   1,376,209   6.2%   Ref   <0.001	Chlamydia Screening								
Abortion services   89,639   7.1%   1.18   1.15-1.22   1.13   1.09-1.18     Education   297,569   3.3%   0.53   0.52-0.55   0.51   0.49-0.53     Youth services   84,574   7.9%   1.32   1.29-1.36   1.03   0.99-1.07     Specimen type   Urine   1,376,209   6.2%   Ref   <0.001	Offices	114,054	6.8%	1.13	1.10-1.16		1.01	0.98 -1.05	
Education   297,569   3.3%   0.53   0.52 - 0.55   0.51   0.49 - 0.53     Youth services   84,574   7.9%   1.32   1.29 - 1.36   1.03   0.99 - 1.07     Specimen type   Urine   1,376,209   6.2%   Ref   <0.001	Abortion services	89,639	7.1%	1.18	1.15-1.22		1.13	1.09 -1.18	
Youth services   84,574   7.9%   1.32   1.29-1.36   1.03   0.99-1.07     Specimen type   Urine   1,376,209   6.2%   Ref   <0.001	Education	297,569	3.3%	0.53	0.52 -0.55		0.51	0.49 -0.53	
Specimen type     Urine     1,376,209     6.2%     Ref     <0.001     <0.001       Cervical swab     98,527     7.0%     1.13     1.10-1.16     1.04     1.00-1.08       Vulvovaginal swab     1,020,224     6.7%     1.08     1.07-1.10     1.07     1.05-1.09       Other     30,878     7.3%     1.19     1.14-1.24     1.27     1.19-1.35       Unknown     39,087     5.0%     Known chlamydia     4     4     4     4     4     4     4     4     5     4	Youth services	84,574	7.9%	1.32	1.29-1.36		1.03	0.99 -1.07	
Urine   1,376,209   6.2%   Ref   <0.001	Specimen type			<b>-</b> /					
Cervical swab   98,527   7.0%   1.13   1.10-1.16   1.04   1.00-1.08     Vulvovaginal swab   1,020,224   6.7%   1.08   1.07 - 1.10   1.07   1.05 - 1.09     Other   30,878   7.3%   1.19   1.14 - 1.24   1.27   1.19 - 1.35     Unknown   39,087   5.0%   5.0%   1.14 - 1.24   1.27   1.19 - 1.35     Wear   No   2,261,502   6.4%   Ref   <0.001	Urine	1,376,209	6.2%	Ref		<0.001			<0.001
Vulvovaginal swab   1,020,224   6.7%   1.08   1.07 - 1.10   1.07   1.05 - 1.09     Other   30,878   7.3%   1.19   1.14 - 1.24   1.27   1.19 - 1.35     Unknown   39,087   5.0%   5.0%   1.14 - 1.24   1.27   1.19 - 1.35     Known chlamydia   diagnosis in last   year   1.07   1.43 - 1.51   1.13   1.09 - 1.18     No   2,261,502   6.4%   Ref   <0.001	Cervical swab	98,527	7.0%	1.13	1.10-1.16		1.04	1.00 - 1.08	
Other   30,878   7.3%   1.19   1.14-1.24   1.27   1.19-1.35     Unknown   39,087   5.0%   5.0%   1.14-1.24   1.27   1.19-1.35     Known chlamydia   diagnosis in last   year   1.14-1.24   1.27   1.19-1.35     No   2,261,502   6.4%   Ref   <0.001	Vulvovaginal swab	1,020,224	6.7%	1.08	1.07 - 1.10		1.07	1.05 - 1.09	
Unknown   39,087   5.0%     Known chlamydia   diagnosis in last     year   No   2,261,502   6.4%   Ref   <0.001	Other	30,878	7.3%	1.19	1.14 - 1.24		1.27	1.19-1.35	
Known chiamydia     diagnosis in last     year     No   2,261,502   6.4%   Ref   <0.001	Unknown	39,087	5.0%						
diagnosis in last     year     No   2,261,502   6.4%   Ref   <0.001	Known chiamydia								
year   No   2,261,502   6.4%   Ref   <0.001   Ref   <0.001     Yes   53,921   9.1%   1.47   1.43 - 1.51   1.13   1.09 - 1.18     Unknown   249,502   6.2%   6.2%   6.2%   6.2%   6.2%     Year of test   2008   385,483   8.7%   Ref   <0.001   Ref   <0.001     2009   645,276   6.8%   0.77   0.76 - 0.78   0.81   0.80 - 0.83	diagnosis in last								
No   2,261,502   6.4%   Ref   <0.001	year	0.004.500	C 40/	Def		0.004	Def		0.004
Yes 53,921 9.1% 1.47 1.43-1.51 1.13 1.09-1.18   Unknown 249,502 6.2%   Year of test   2008 385,483 8.7% Ref <0.001	NO Xee	2,261,502	6.4%	Ret	4 4 9 4 5 4	<0.001	Rer	1 00 1 10	<0.001
Vear of test     249,502     6.2%       2008     385,483     8.7%     Ref     <0.001	Yes	53,921	9.1%	1.47	1.43 - 1.51		1.13	1.09-1.18	
2008   385,483   8.7%   Ref   <0.001   Ref   <0.001     2009   645,276   6.8%   0.77   0.76 - 0.78   0.81   0.80 - 0.83		249,502	0.2%						
2000     385,483     8.7%     Rei     <0.001     Ref     <0.001       2009     645,276     6.8%     0.77     0.76 - 0.78     0.81     0.80 - 0.83		205 402	0 70/	Def		.0.004	Def		-0.004
2009 045,270 0.8% 0.77 0.76 0.80 0.81 0.00 0.83	∠000 2000	300,403 645 276	0.1%		0.76 0.70	<0.001		0 00 0 00	<0.001
	2009	040,270 920 172	0.0% 5.7%	0.11	0.10-0.10		0.01	0.00-0.03	
2010 030,172 0.7% 0.04 0.03 0.72 0.71 0.74 2011 703.007 5.8% 0.65 0.67 0.66 0.72 0.71 0.74	2010	703 001	5.8%	0.04	0.03 - 0.05		0.72	0.71 - 0.74 0.71 - 0.74	

Table 4-4 (b): Number of chlamydia tests, percentage testing positive, unadjusted and adjusted odds ratios for testing positive by sociodemographic, sexual behavioural, and test characteristics (NCSP dataset, 15-24 year-old <u>men</u>, 2008-2011)

	Number		Univariable analysis			Multivariable analysis		
	of tests	% nositive	OR 9	able allary	515 n	OR 9	anable ana 95%Cl	n
Overall	1,537,178	5.2%			٣			Ρ
Sociodemographic								
characteristics								
Age (vears)								
15	64.416	1.6%	Ref		<0.001	Ref		<0.001
16	180.426	1.9%	1.23	1.14-1.32		1.36	1.22-1.52	
17	211.271	3.4%	2.19	2.05-2.34		2.30	2.08-2.55	
18	225,775	4.8%	3.18	2.98-3.40		3.34	3.02-3.70	
19	206,138	6.2%	4.10	3.84-4.37		4.19	3.78-4.63	
20	179,374	6.7%	4.52	4.24-4.82		4.53	4.09-5.01	
21	145,461	7.1%	4.77	4.47-5.09		4.62	4.17-5.11	
22	121,871	7.2%	4.85	4.54-5.18		4.62	4.17-5.11	
23	107,345	6.7%	4.45	4.16-4.76		4.16	3.75-4.61	
24	95,101	6.6%	4.40	4.12-4.71		4.13	3.72-4.58	
Ethnic group	,							
White	834,785	5.7%	Ref		<0.001	Ref		<0.001
Black	66,682	8.8%	1.58	1.54-1.63		1.69	1.61-1.76	
Asian	71,650	2.0%	0.33	0.31-0.35		0.46	0.42-0.49	
Chinese	3,882	3.0%	0.51	0.42-0.61		0.65	0.52-0.81	
Other	8,533	4.1%	0.70	0.63-0.78		0.74	0.62-0.88	
Mixed	34,200	7.6%	1.36	1.30-1.42		1.33	1.26-1.41	
Unknown	517,446	4.2%						
Region								
London	246,329	4.1%	Ref		< 0.001			<0.001
North East	97,200	4.7%	1.13	1.09-1.18		1.93	1.83-2.05	
North West	160,533	7.2%	1.81	1.76-1.86		1.68	1.61-1.75	
Yorkshire & Humber	131,424	6.8%	1.70	1.65-1.75		1.88	1.79-1.97	
East Midlands	125,824	4.5%	1.09	1.05-1.12		1.48	1.40-1.56	
West Midlands	161,981	4.5%	1.11	1.07-1.14		1.49	1.42-1.56	
East of England	120,757	4.6%	1.13	1.09-1.17		1.32	1.26-1.39	
South East Coast	63,808	4.7%	1.14	1.09-1.19		1.33	1.25-1.41	
South Central	77,906	4.9%	1.18	1.14-1.23		1.32	1.25-1.40	
South West	85,516	6.7%	1.66	1.60-1.72		1.58	1.51-1.66	
Other / Unknown	265,900	5.1%						
IMD quintile of								
LSOA of residence <sup>y</sup>								
Least deprived	182,535	4.1%	Ref		<0.001	Ref		<0.001
2	195,929	4.7%	1.13	1.10-1.17		1.10	1.05-1.14	
3	234,047	4.8%	1.17	1.14-1.21		1.14	1.09-1.18	
4	293,142	5.3%	1.30	1.27-1.34		1.21	1.16-1.26	
Most deprived	365,386	6.2%	1.52	1.48-1.56		1.31	1.26-1.36	
Unknown	266,139	5.1%						

#### Table 4-4(b) continued.

	Number of tests	% positive	Univa OR	ariable analy	/sis n	Multi OR	variable ana	lysis
Sexual behaviour	0110010	positive	UN	00/001	P		507001	٢
characteristics								
>1 sexual partner in								
last year								
No	283,042	3.9%	Ref		<0.001			
Yes	397,964	8.3%	2.24	2.19-2.29				
Unknown	856,172	4.2%						
≥1 new sexual								
partner in last 3								
months								
No	307,386	4.7%	Ref		<0.001			
Yes	411,533	7.7%	1.71	1.68-1.75				
Unknown	818,259	4.1%						
Sexual risk group"		0.00/	<b>D</b> (		0.004	<b>D</b> (		0.004
Low	205,729	3.6%	Ref	4 74 4 64	<0.001	Ref		<0.001
Medium	186,169	6.3%	1.79	1.74-1.84		1.71	1.65-1.77	
High	311,664	8.5%	2.48	2.41-2.54		2.09	2.03-2.16	
Unknown Teet and elinical	833,616	4.1%						
lest and clinical								
Vonue of test <sup>2</sup>								
	157 400	6 20/	Pof		-0.001	Pof		-0.001
GF	157,400	0.2%	1 5 1	1 42 1 60	<0.001	1 4 4	1 22 1 55	<0.001
Outroach	10,201	9.1/0 2 70/	0.42	0.42-1.00		0.46	0.44.0.49	
	1 208	2.1 /0 1 2%	0.42	0.41-0.43		0.40	0.44-0.40	
Pharmacy	27 192	4.270 8.1%	1 33	1 27-1 40		1 33	1 25-1 42	
Sexual and	27,102	0.170	1.00	1.27 1.40		1.00	1.20 1.42	
Reproductive Health								
Services	207 617	11 1%	1 88	1 83-1 92		1 57	1 52-1 63	
Remote testing	151 556	6.5%	1.05	1 02-1 08		0.86	0.83-0.89	
Military	55.242	5.3%	0.84	0.81-0.88		0.92	0.82-1.03	
Prisons & Youth	,							
Offending Institutions	55.396	7.8%	1.27	1.22-1.32		1.33	1.15-1.53	
Chlamydia Screening	,							
Offices	87,836	7.2%	1.17	1.13-1.21		1.16	1.11-1.22	
Abortion services	884	10.3%	1.73	1.39-2.15		2.01	1.48-2.71	
Education	278,137	1.7%	0.26	0.25-0.27		0.34	0.32-0.36	
Youth services	57,887	5.8%	0.92	0.89-0.96		1.04	0.98-1.11	
Known chlamydia								
diagnosis in last								
year								
No	1,259,011	5.2%	Ref		<0.001	Ref		<0.001
Yes	12,601	11.2%	2.33	2.202.46		1.27	1.17-1.37	
Unknown	265,566	5.1%						
Year of test								
2008	181,539	7.7%	Ref		<0.001	Ref		<0.001
2009	357,241	5.8%	0.73	0.71-0.75		0.76	0.74-0.79	
2010	569,272	4.3%	0.54	0.53-0.55		0.68	0.65-0.70	
2011	429,126	4.8%	0.60	0.59-0.62		0.72	0.69-0.74	

OR: Odds ratio; AOR: Adjusted odds ratio; <sup>×</sup>Sexual risk group: <u>High</u>: more than 1 sexual partner in the last year and at least one new sexual partner in the last 3 months; <u>Medium</u>: either more than 1 sexual partner in the last year or at least one new sexual partner in the last 3 months; <u>Low</u>: neither more than 1 sexual partner in the last year nor at least one new sexual partner in the last 3 months; <sup>v</sup> IMD: Index of multiple deprivation, quintile based on lower super output area (LSOA) of residence; Outreach: tests performed in non-clinical settings including entertainment and leisure venues. Remote testing: tests performed without face to face contact with a health professional at the time of test, includes tests performed through the internet. Table 4-5 (a): Number of tests, percentage testing positive, univariable and multivariable logistic regression results by reported characteristics (GUMCAD, <u>women</u> aged 15-24, 2008 to 2011)

		%	Univariable analysis			Multivariable analysis		
	Total (n)	positive	OR	95%CI	р	OR	95%CI	р
Overall	958,864	11.2%						
A								
Age (years)	47.000	40.404	Б (		0.004	- <i>(</i>		0.004
15	17,866	13.4%	Ref		<0.001	Ref		<0.001
16	40,984	15.2%	1.15	1.10-1.21		1.17	1.11-1.24	
17	70,783	14.6%	1.10	1.05-1.15		1.13	1.07-1.19	
18	100,665	14.5%	1.09	1.04-1.15		1.14	1.08-1.19	
19	124,957	13.2%	0.98	0.93-1.02		1.02	0.97 -1.07	
20	133,549	11.5%	0.84	0.80-0.88		0.87	0.83-0.91	
21	131,255	10.4%	0.75	0.72-0.79		0.78	0.74 -0.82	
22	122,945	9.2%	0.65	0.62-0.68		0.69	0.66 -0.73	
23	113,470	8.2%	0.58	0.55-0.61		0.62	0.58-0.65	
24	102,390	7.4%	0.52	0.49-0.54		0.55	0.52-0.58	
Ethnicity	- ,							
White	725 062	11.6%	Ref		<0.001	Ref		<0.001
Mixed	41 600	11.8%	1 02	0 99 - 1 06	\$0.001	1 03	1 00 -1 06	10.001
Asian	21 256	7.5%	0.62	0.50-0.66		0.70	0.66-0.73	
Rlock	21,250	10 10/	0.02	0.03-0.00		0.70	0.00-0.75	
Othor	16 626	0.10/	0.00	0.04-0.00		0.92	0.90-0.95	
	10,030	9.4%	0.00	0.70-0.04		0.91	0.00-0.90	
Unknown	68,360	9.8%			0.004			0.004
Region	047 700	0.00/	4 50		<0.001		4 40 4 50	<0.001
London	217,730	9.2%	1.58	1.54 - 1.63		1.47	1.42-1.52	
North East	51,094	13.7%	1.57	1.53-1.60		1.50	1.46-1.54	
North West	109,266	13.6%	1.53	1.49-1.57		1.43	1.39-1.47	
Yorkshire & Humber	85,812	13.3%	1.59	1.55-1.63		1.56	1.51 -1.61	
East Midlands	61,558	13.8%	1.34	1.31-1.37		1.31	1.28-1.35	
West Midlands	91,282	11.9%	1.20	1.17 -1.23		1.28	1.24 -1.32	
East of England	81,575	10.8%	1.05	1.02-1.08		1.10	1.07 -1.13	
South East Coast	74,954	9.6%	1.00	0.97-1.03		1.05	1.02-1.08	
South Central	79,494	9.1%	1.16	1.13-1.19		1.20	1.16-1.24	
South West	81,707	10.4%	1.58	1.54-1.63		1.47	1.42-1.52	
Unknown	24,392	11.7%						
IMD quintile of	2 1,002							
LSOA of residence <sup>x</sup>								
Least deprived	130 793	9.6%	R≏f		~0.001	Rof		~0.001
2	142 850	10.2%	1 07	1 04 -1 10	<b>NO.001</b>	1 05	1 03 -1 08	<b>\U.UU</b>
2	142,000	10.2 /0	1.07	1.04-1.10		1.00	1.03-1.00	
3	211 202	10.0 /0	1.12	1.09-1.15		1.10	1.12-1.10	
4 Moot deprived	211,293	11.470	1.22	1.19-1.20		1.20	1.23-1.30	
Most deprived	235,443	12.8%	1.39	1.36-1.42		1.42	1.38-1.45	
Unknown	65,570	11.6%						
Known chlamydia								
diagnosis in last								
year								
No	921,201	11.2%	Ref		<0.001	Ref		<0.001
Yes	37.663	12.2%	1.11	1.07-1.14		1.04	1.01 - 1.07	
Year		/0						
2009	317 081	11 8%	Ref		<0.001	Ref		<0.001
2010	309 295	11.0%	0.92	0.91-0.94	101001	0.93	0.91-0.95	10.001
2011	332.488	10.8%	0.90	0.88-0.91		0.91	0.89-0.92	

		%	Univariable analysis		Multivariable anal		lysis	
	Total (n)	positive	OR	95%CI	р	OR	95%CI	р
Overall	626,531	11.29	%					
Age (years)								
15	3,777	8.0%	Ref		< 0.001	Ref		<0.001
16	11.819	10.9%	1.40	1.22 - 1.59		1.48	1.28-1.70	
17	26,134	12.7%	1.67	1.47 - 1.89		1.73	1.52-1.98	
18	48,424	13.9%	1.85	1.64 - 2.09		1.91	1.68-2.18	
19	72,174	13.8%	1.83	1.63-2.07		1.93	1.69-2.20	
20	88,315	13.6%	1.80	1.60 - 2.03		1.89	1.66-2.15	
21	95,575	13.3%	1.75	1.56 - 1.98		1.83	1.61-2.08	
22	96,274	12.4%	1.63	1.45 - 1.84		1.74	1.53-1.98	
23	93,941	11.4%	1.48	1.31 -1.66		1.58	1.39-1.80	
24	90,098	10.7%	1.38	1.22-1.55		1.48	1.30-1.69	
Ethnicity								
White	466,201	12.4%	Ref		<0.001	Ref		<0.001
Mixed	24,701	14.5%	1.20	1.16-1.25		1.24	1.19-1.29	
Asian	20,587	6.9%	0.53	0.50-0.56		0.55	0.52-0.58	
Black	57.530	16.3%	1.38	1.34 - 1.41		1.46	1.42-1.50	
Other	10.080	10.2%	0.81	0.76-0.86		0.88	0.82-0.95	
Unknown	47,432	11.8%						
Region	,							
London	128.002	10.9%	Ref			Ref		<0.001
North East	35.713	15.1%	1.46	1.41-1.51		1.59	1.54-1.66	
North West	78.871	13.9%	1.33	1.29-1.36		1.45	1.41-1.50	
Yorkshire & Humber	60 094	14 0%	1.34	1 30 -1 38		1 47	1 42 -1 52	
East Midlands	43,295	14.5%	1.40	1.35-1.44		1.57	1.51-1.62	
West Midlands	59 142	13.2%	1 25	1 21 -1 28		1.38	1 33-1 43	
East of England	53.295	12.0%	1.12	1.09-1.16		1.33	1.28-1.38	
South Fast Coast	46 545	11.2%	1 03	1 00 -1 07		1 29	1 24 -1 34	
South Central	49.477	10.7%	0.99	0.95-1.02		1.17	1.13-1.21	
South West	51,526	12.2%	1.15	1.11 -1.18		1.33	1.28-1.38	
Unknown	20.571	12.9%					1120 1100	
IMD quintile of LSO	A							
of residence <sup>x</sup>	-							
5 (Least deprived)	87,870	10.6%	Ref		< 0.001	Ref		<0.001
4	94,222	11.6%	1.10	1.07 - 1.14		1.10	1.06-1.13	
3	110,858	12.1%	1.16	1.13-1.20		1.17	1.14-1.21	
2	131,768	13.1%	1.27	1.24 - 1.30		1.30	1.26-1.33	
1 (Most deprived)	154,169	14.2%	1.40	1.36-1.43		1.38	1.34-1.42	
Unknown	47.644	12.4%	-				-	
Ever MSM <sup>y</sup>	,							
No	577.650	12.9%	Ref		< 0.001	Ref		<0.001
Yes	48.881	8.4%	0.62	0.60-0.64		0.64	0.62-0.67	
Known chlamydia	-,							
diagnosis in last vea	ar							
No	606.099	12.5%	Ref		< 0.001	Ref		<0.001
Yes	20.432	15.1%	1.25	1.20-1.30		1.15	1.10-1.20	
Year	, . <b>.</b> _							
2009	213.012	12.9%	Ref		<0.001	Ref		<0.001
2010	203.076	12.3%	0.95	0.93-0.97		0.96	0.94-0.98	
2011	210 443	12.4%	0.96	0.94 -0.98		0.97	0.95-0.99	
<u>X</u>	2:0,440	12.770	0.00			0.01	0.00 0.00	

Table 4-5(b): Number of tests, percentage testing positive, univariable and multivariable logistic regression results by reported test characteristics (GUMCAD, <u>men</u> aged 15-24, 2008 to 2011)

<sup>x</sup>*IMD*: Index of multiple deprivation, quintile based on lower super output area (LSOA) of residence; <sup>y</sup>Ever MSM: Any record in GUMCAD where reports having sex with a man. OR: Odds ratio; AOR: Adjusted odds ratio.

### 4.4.3 Trends in test, demographic and sexual behaviour characteristics

Among NCSP testing episodes, demographic and sexual behaviour variables remained relatively stable during the analysis period, although between 2010 and 2011 there was a small increase seen in the proportion where the individual tested reported having at least one new sexual partner in the last year and in the proportion of tests among those of non-white ethnicity (Figure 4-3). In both women and men there was a small increase between 2010 and 2011 in the percentage with a known previous diagnosis in the last year, increasing from 2.4% to 2.7% in women and from 0.9% to 1.2% in men.

The distribution of testing venue types within the NCSP dataset changed considerably over the analysis period (Figure 4-4). Among women, the percentage of tests which were from sexual and reproductive health services declined from 41% in 2008 to 29% in 2011, while the percentage from GP settings increased from 16% in 2008 to 20% in 2011. Among men, the percentage of tests reported through outreach settings (including testing in non-clinical venues such as bars and pubs, music festivals and other one off events), increased substantially between 2008 and 2010, from 18% to 35%, and then fell to 29% in 2011. The percentage of tests that were from sexual and reproductive health services decreased from 20% in 2008 to 14% in 2011.



Figure 4-3: Trends in reported characteristics (NCSP dataset, 15-24 year-old women and men, 2008-2011)

\*Proportions reported among tests with complete data on each variable; High risk group: reported more than 1 sexual partner in the previous year and at least one new sexual partner in the previous 3 months; SP: sexual partner; Resident in most deprived area: resident in one of the 20% most deprived LSOAs in England, as defined by the Index of Multiple Deprivation (IMD); Ct: Chlamydia.





\*Other: Hospital, occupational health, pharmacy, military, prisons & youth offenders institutions and abortion services (abortion services included in 'other' category in men only); CSO: Chlamydia Screening Office; SRH: Sexual and Reproductive Health services There was no change during the analysis period in the proportion of GUMCAD testing episodes that were in people of non-white ethnicity, nor in that among those living in the most deprived areas (Figure 4-5). A slight decline in the percentage of testing episodes from 15 to 19 year-olds was observed among both women and men (39% to 36% among women; 27% to 25% among men). The percentage of testing episodes among men that were in MSM increased from 7% to 9%. In both women and men there was a small decrease between 2010 and 2011 in the percentage with a known previous diagnosis in the last year, decreasing from 4.8% to 4.6% in women and from 4.0% to 3.8% in men.

Figure 4-5: Trends in reported characteristics, GUMCAD (15 to 24 year-old women and men, 2009 to 2011)



\*Proportions reported among tests with complete data on each variable; Resident in most deprived area: resident in one of the 20% most deprived LSOAs in England, as defined by the Index of Multiple Deprivation (IMD); Ct: Chlamydia; MSM: Men who have sex with men.

### 4.4.4 Trends in percentage testing positive by subgroup

Among NCSP testing episodes, the percentage testing positive declined during the analysis period across all subgroups defined by age group, sexual risk group and venue of testing (Figure 4-6). In several subgroups among men, increases in the percentage testing positive were observed between 2010 and 2011.

Among GUMCAD testing episodes, between 2009 and 2011 the percentage testing positive declined among both 15 to 19 year-old and 20 to 24 year-old women (Figure 4-7). Among men, the percentage testing positive declined among 20 to 24 year-olds, but there was no change among 15 to 19 year-olds. In contrast to other subgroups, the percentage testing positive among MSM in GUMCAD increased from 7.9% in 2009 to 9.0% in 2011 (OR 1.08 95%CI 1.03 to 1.12). A declining trend in percentage testing positive was observed in all the symptomatic condition subgroups (i.e. in attendances where a diagnosis of genital warts, bacterial vaginosis, candidosis or genital herpes was made).



Figure 4-6: Percentage testing positive by year and gender, stratified by age group (a), sexual risk group (b), and venue of test (c) (NCSP dataset, 15 to 24 yea-old women and men, 2008 to 2011)

<sup>x</sup>Sexual risk group: <u>High</u>: more than 1 sexual partner in the last year and at least one new sexual partner in the last 3 months; <u>Medium</u>: either more than 1 sexual partner in the last year or at least one new sexual partner in the last 3 months; <u>Low</u>: neither more than 1 sexual partner in the last year nor at least one new sexual partner in the last 3 months; <u>SRH</u>: Sexual and reproductive health services; Outreach: tests performed in non-clinical settings including entertainment and leisure venues; Remote testing: tests performed without face to face contact with a health professional at the time of test, includes tests performed through the internet.



Figure 4-7: Percentage testing positive by year and gender, stratified by age group. (GUMCAD, 15 to 24 year-olds)

<sup>x</sup>Among attendances where a chlamydia test and one of the specified symptomatic conditions were recorded on the same date. For genital warts and genital herpes, data are presented for first episodes (as recurrences are possible).

# 4.4.5 Trends by calendar quarter

Quarterly trends in the percentage testing positive varied by dataset and venue type (Figure 4-8). In 2009 to 2011, there was a peak in numbers of testing episodes reported through the NCSP dataset between January and March of each year, which corresponded to a simultaneous drop in the percentage testing positive. This was also accompanied by a nadir in the proportion of tests coming from women in the 'high risk' sexual behaviour category (Figure 4-9). Among women, this seasonality in patterns of testing and of percentage testing positive was evident for tests in education settings, but was not seen in testing performed in sexual and reproductive health services or GP settings or in tests reported through GUMCAD.


## Figure 4-8: Numbers of tests and percentage testing positive by calendar quarter, dataset and location of testing (15-24 year-old women and men)

Figure 4-9: Trends in reported characteristics by calendar quarter, NCSP dataset (15 to 24 year-old women and men, 2008 to 2011)



# 4.4.6 Assessing the selection effects introduced by changes in risk factor variables over time

Figure 4-10 and Figure 4-11 show the estimated annual change in the percentage testing positive among NCSP and GUMCAD testing episodes before adjusting for other variables (unadjusted), after adjusting for one additional variable at a time (age, ethnicity, region, deprivation, sexual risk group, known chlamydia diagnosis in the last year, testing venue type and, for women, specimen type; the bivariable models) and after adjusting for all variables (multivariable analyses).

# Figure 4-10: Unadjusted and adjusted odds ratios (and 95%CI) of testing positive per additional year, estimated from univariable, bivariable and multivariable models (NCSP tests, 15 to 24 year-olds 2008 to 2011)\*

Open diamonds show the unadjusted odds ratio (OR) of testing positive per year, without adjusting for any other variables; closed diamonds show the adjusted OR of testing positive per year after including the stated risk factor variable of interest in the logistic regression model, either one additional variable at a time (bivariable analysis) or with all variables included (multivariable analysis). The dashed line shows the unadjusted OR.

#### Women



Greater <u>decrease</u> in odds of testing positive per year

#### Men Unadjusted Bivariable models, adjusted for: Age Ethnicity Region

IMD quintile of LSOA of residence Risk group

Known chlamydia diagnosis in last year Testing venue type

Multivariable model



← Greater <u>decrease</u> in odds of testing positive per year

Analyses are limited to tests with complete data on all variables included in the multivariable analysis. IMD: Index of Multiple Deprivation; Risk group: <u>High</u>: more than 1 sexual partner (SP) in the previous year and at least one new SP in the previous 3 months; <u>Medium</u>: either more than 1 SP in the previous year or at least one new SP in the previous 3 months; <u>Low</u>: neither more than 1 SP in the previous year nor at least one new SP in the previous 3 months.

Among NCSP testing episodes, unadjusted and adjusted ORs of testing positive were less than 1. Thus between 2008 and 2011, there was a reduction in the percentage testing positive, even after adjusting for all available demographic and behaviour variables (Figure 4-10).

In the bivariable models, adjusting for venue type had the largest and only notable effect on the estimated trend in percentage testing positive. Among women, a 14% annual decline was observed before adjustment (OR 0.86, 95%CI 0.86-0.87), compared to an 11% annual decline after adjusting for venue type of test (AOR 0.89, 95%CI 0.88-0.89). Among men, a 16% annual decline was observed in the unadjusted analysis, compared to a 12% decline after adjustment for venue testing type (unadjusted OR 0.84, 95%CI 0.83-0.84; AOR 0.88, 95%CI 0.83-0.84).

In the multivariable models, (adjusting for age, ethnicity, sexual risk group, IMD quintile of LSOA, having a known chlamydia diagnosis in the past year, testing venue type and specimen type [women only]), among women, there was a 12% annual decline before adjustment, and an 11% annual decline after adjustment (unadjusted OR 0.88, 95%CI 0.88-0.89; AOR 0.89, 95%CI 0.88-0.90). Among men there was a 12% annual decline before controlling for the other variables, and a 10% annual decline after adjustment (unadjusted OR 0.95%CI 0.89-0.91).

Sensitivity analyses using those NCSP testing episodes with complete data on each variable showed similar patterns, whereby adjusting for venue type of test was the most important variable, and changes in other variables led to little difference between unadjusted and adjusted ORs (data not shown).

Among tests reported to GUMCAD, all unadjusted and adjusted ORs were less than 1 among women and men, indicating a decline in the percentage testing positive during the analysis period (2009 to 2011). All ORs of testing positive among MSM were >1, indicating an increase in the percentage testing positive per year (Figure 4-11 (a-b)). Adjustment made a negligible difference to the measured trend in positivity among GUMCAD testing episodes in any of the bivariable or multivariable analyses (Figure 4-11 (c)).

# Figure 4-11: Unadjusted and adjusted odds ratios (and 95%CI) of testing positive per additional year, estimated from univariable, bivariable and multivariable models (GUMCAD tests, 15 to 24 year-olds 2009 to 2011)

Open diamonds show the unadjusted odds ratio (OR) of testing positive per year, without adjusting for any other variables; closed diamonds show the adjusted OR of testing positive per year after including the stated risk factor variable of interest in the logistic regression model, either one additional variable at a time (bivariable analysis) or with all variables included (multivariable analysis). The dashed line shows the unadjusted OR.

#### Women





← Greater <u>decrease</u> in odds of testing positive per year

#### 

Odds ratio • Greater <u>decrease</u> in odds of testing positive per year

#### Figure 4-11continued



Greater increase in odds of testing positive per year →

\*Analyses are limited to tests with complete data on all variables included in the multivariable analysis.

#### 4.5 Discussion

#### 4.5.1 Key findings

The number of 15 to 24 year-olds who tested for chlamydia increased between 2008 and 2011, while the percentage testing positive decreased. There was considerable variation over time in the relative contributions made by different testing venue types. Among those tested, people reporting more sexual partners and those reporting a previous chlamydia diagnosis were more likely to test positive, as were those living in more deprived areas.

Testing venue type was an important confounder in the relationship between test result and year among the tests reported to the NCSP. In men tested through the NCSP, there was evidence of confounding of the relationship between year and test result arising from differences in the distribution of age and ethnicity in each year. In women and in men tested in GUM clinics, there was no evidence of confounding arising from changes in other demographic or sexual behaviour variables for which data are available among those who tested as part of the NCSP or in GUM clinics, suggesting that these variables did not change sufficiently over time to introduce substantial error.

#### 4.5.2 Strengths and limitations

The main strength of this analysis is that it used two large datasets with national coverage and individual-level data that allowed investigation of the relationship between testing positive for chlamydia and trends in percentage testing positive and the characteristics of those tested. However it should be noted that the size of the datasets means that even small differences would likely be statistically significant, thus the size and direction of the associations should be considered when determining public health significance.

The analysis was limited by the completeness of data reporting. A large proportion of NCSP tests had missing data on one or more variable of interest and the primary analysis looking at the impact of changes in the population tested on trends in percentage testing positive had to be limited to tests with complete data. However, sensitivity analysis showed that the same patterns were seen among tests with complete data on each variable as when the limited dataset was used. The analysis was also limited as individuals could not be linked between datasets, meaning that the number with a previous known diagnosis is likely to be underestimated. It was interesting to note that the percentage of women and men with a known diagnosis in the last year increased in the NCSP dataset but decreased in GUMCAD between 2010 and 2011. While re-infections have a potentially important part to play in the

transmission, these divergent trends likely reflect changing patterns of service use with under 25 year-olds increasingly accessing testing outside of GUM services.

As described in Chapter 3, previous studies in the US, Scotland and the Netherlands have shown that trends in percentage testing positive for chlamydia among sentinel and clinic populations can be particularly affected by changes in test technology.<sup>203-205,223</sup> This is unlikely to have been influential in this analysis, as NAATs were universally in use during the analysis period<sup>128</sup>. It is feasible however that differences in the exact test used may have had a minor effect on the data which has not been accounted for in this analysis.

#### 4.5.3 Implications for evaluation of chlamydia control

Changes in the distribution of venue testing types can explain some, but not all, of the observed decline in percentage testing positive for chlamydia among 15 to 24 year-olds tested within the NCSP between 2008 and 2011. Thus this analysis of surveillance data up to 2011 provide some support for there having been a decrease in chlamydia infection over this period. However, this analysis also demonstrates that trends in the percentage testing positive in England are subject to measured and, potentially, unmeasured confounding. Specifically, venue type remained significantly associated with testing positive even after adjusting for all other available demographic and sexual behaviour variables. This demonstrates that the available data were not able to fully capture risk of testing positive for chlamydia. Thus, even after adjusting for sexual behaviour, trends in percentage testing positive will still be subject to unmeasured confounding and cannot be reliably attributed to there having been a change in the incidence or prevalence of infection. Stratifying analyses by venue of test

will reduce some of this potential error, but it is not possible with the available data to know whether this will fully eliminate the potential for confounding of trend estimates.

This raises the question as to what other data would be needed to improve the utility of surveillance data for monitoring the impact of chlamydia control programmes.

Firstly, more detailed sociodemographic and sexual behaviour variables may be useful. Only limited data were included in each dataset meaning it is possible that there were changes in the risk profile of those tested that were not captured. In the NCSP dataset, number of partners was recorded as a binary variable (<1 /  $\geq$ 2). Thus people with 2 partners in the last year could not be distinguished from those with many more sexual partners in the last year, even though their risk of infection would be very different.<sup>133</sup> A greater level of detail about sexual risk would be useful to allow adjustment for differences in characteristics of the population tested, which arise from changes in service use patterns or from underlying changes in sexual behaviour.

Secondly, data were not available on whether tests had been carried out as a result of partner notification. More recently-available data suggest that around 35% of people who attend GUM clinics as a result of partner notification (i.e. informed of having an infected partner) test positive for chlamydia.<sup>224</sup> Thus changes in partner notification rates over this period could also have had an effect on trends in percentage testing positive. Since 2012, GUMCAD has included a variable on whether tests were carried out as a result of partner notification. This variable should be incorporated into any future analyses of

trends in percentage testing positive using GUMCAD data. Data on numbers of partners tested were collected for tests performed through the NCSP up to 2012. However, these data were collected in a separate, aggregate dataset and as such could not be linked to data on tests and diagnoses. The data were also subject to considerable data quality issues and under-reporting (personal communication Alireza Talebi) and as such have not been reported in this chapter and were not incorporated into this analysis.

It was not possible to measure uptake of testing (i.e. number tested / number attending) among those attending all venues. This would have been possible with data from GUM clinics as a clinic-attending denominator is available. However as the offer and uptake of a chlamydia test is less likely to vary in GUM settings this was not investigated in this analysis. Developments in the GUMCADv2 surveillance system, whereby comparable information to that collected in GUM clinics is now being collected for other commissioned non-GUM sexual health services, mean that a measure of uptake of testing within sexual and reproductive health services and some GP clinics will be available in coming years. The extent to which percentage testing positive varies with test uptake within a setting would be useful to explore in future analyses.

#### 4.5.4 Summary

In summary, given the potential for residual unmeasured confounding, the observed decreases in percentage testing positive in the NCSP dataset or GUMCAD do not provide strong evidence of there having been a decrease in chlamydia incidence or prevalence between 2008 and 2011, in years with relatively high levels of screening among under 25 year-olds. While useful to identify volumes of testing and diagnoses and factors associated with being

infected at the time of testing, these surveillance data alone do not provide sufficiently reliable measures of chlamydia infection over time for the purposes of evaluating chlamydia screening. More robust outcome measures are needed to evaluate the impact of chlamydia screening. In the next three chapters, I turn to measures of chlamydia prevalence among general population samples.

### 5 Pilot of a postal survey designed to measure the population prevalence of chlamydia among young women

In the previous chapter I investigated the use of surveillance data for monitoring trends in percentage testing positive for chlamydia. I showed that such data alone are unsuitable for evaluating the impact of chlamydia screening, in part due to the potential for unmeasured confounding arising from changing patterns of service use over time that could not be adjusted for using the available data. One option to address this issue is to establish repeated cross-sectional surveys among young adults, as the population targeted by the NCSP. In this chapter I present a pilot study which I carried out to investigate the feasibility of a repeated postal survey of chlamydia prevalence with anonymous testing for chlamydia in England. The pilot was conducted between June and August 2011 in two primary care trusts (PCTs) in England and was carried out in conjunction with the Health Protection Agency (HPA)<sup>14</sup>.

#### 5.1 Aims & objectives

The aim of the study presented in this chapter was to determine whether repeat cross-sectional surveys using postal invitations and anonymous testing (i.e. without return of test result) could be a feasible method of population-based chlamydia prevalence monitoring in England. This aim was addressed through the following objectives:

<sup>&</sup>lt;sup>14</sup> Part of Public Health England since 2013.

- To design and pilot a cross-sectional survey of chlamydia prevalence using postal invitations and anonymous testing in two primary care trusts (PCTs) in England.
- To compare participation rates using different invitation approaches (+/provision of a test kit; +/- offer of small financial incentive).
- To investigate selection bias and costs associated with the different approaches.

#### 5.2 Methods

#### 5.2.1 Participants

Women aged 17 or 18 and resident in two PCTs (NHS Northamptonshire and NHS Sutton & Merton) were eligible for inclusion in the study.

This age group was selected for a number of reasons. Firstly, a high proportion of this age group would be expected to be sexually active<sup>133</sup> and thus be an appropriate target group for monitoring changes in prevalence. Secondly, the proportion of young adults living with their parents declines steeply with age over 16 years<sup>225</sup> meaning the reliability of address data reduces with age. Thirdly, chlamydia prevalence was expected to be relatively high among this age group,<sup>123</sup> thus providing a good opportunity to detect small changes in prevalence over time that might be expected from mathematical models<sup>81</sup>. However, only 18 year-old women were included in NHS Northamptonshire in order to comply with local guidance about research involving children (see section 5.2.4).

The sample was limited to women as levels of chlamydia screening are higher in women than men.<sup>226,227</sup> Any direct impact of screening on prevalence is therefore likely to be greater in women. Furthermore, the most serious complications of chlamydia such as PID, ectopic pregnancy and infertility, occur in women, making monitoring of prevalence in women arguable more important. Trends in chlamydia prevalence in women should be indicative of changes in prevalence among heterosexual men.

The two pilot sites were chosen for pragmatic purposes, to include one site in London and one outside of London and to include areas with different levels of screening activity. Table 5-1 shows the estimated proportion of 15-24 year-olds tested, the diagnosis rate (diagnoses per 100,000 population) and percentage testing positive in the two participating PCTs compared to England.

Table 5-1: Opportunistic screening uptake, diagnosis rates and PC
characteristics in the participating PCTs, 2011/12 <sup>228</sup>

	Northamptonshire	Sutton & Merton	England
Estimated coverage (number of tests per 100 population of 15-24 year-olds)	34%	24%	29%
Rank (1=PCT with highest coverage)	35/151	108/151	N/A
Diagnosis rate (diagnoses per 100,000 population of 15-24 year-olds)	2,457	2,008	2,090
Rank (1=PCT with highest diagnosis rate)	38/151	85/151	N/A
Percentage testing positive	7.2%	8.3%	7.3%
Classification	Rural	Urban	N/A

\*Urban: PCTs with 50 percent of their population in one of the 17 urban areas with a population of at least 250,000; \*Rural: PCTs with more than 26 percent of their population in rural settlements and larger market towns.

Eligible individuals were identified using lists of people registered with a GP in the two participating PCTs. A total of 3,857 women in NHS Sutton & Merton and

3,687 women in Northamptonshire were eligible.<sup>15</sup> The study aimed to establish a sampling frame that could be used to send postal invitation to a random sample representative of the general population of young women in England. The sampling frame therefore needed to have current address information for named individuals, of known age. As discussed in section 3.2.2.2, GP lists are known to be limited by potential inaccuracies. However GP lists were chosen as the most comprehensive source of this information in England that was available for use, and as sampling via household lists was considered too resource intensive.

#### 5.2.2 Study procedures

As described in the Chapter 3, response rate would likely be a key issue in determining feasibility and bias of any survey estimates. Participants in the pilot were therefore allocated into three different groups to allow comparison of response rates using different recruitment approaches (Table 5-2). All of the selected women were sent an invitation letter by post. Group A were sent a test kit with their invitation. The test kit included: an invitation letter; a cover letter from the participating PCT; a study information leaflet; a short questionnaire including questions on sexual behaviour; a vulvovaginal swab (VVS); an instruction leaflet on how to use the swab and return the sample and a pre-paid return envelope. Group B were sent a test kit and also offered a £5 voucher for high-street shops on return of sample. Group C were not sent a test kit but instead were invited to contact the study team by text message, email or return

<sup>&</sup>lt;sup>15</sup> In order to comply with regulatory approvals (section 5.2.4), invitations were sent out by staff at the participating PCTS; none of the study personnel at the HPA had access to name or address information of those invited.

of postcard to request a kit. A reminder letter was sent to non-responders three weeks after the initial invitation. Individuals who did not wish to participate in the survey were asked to complete and return a pre-paid postcard to the research team, indicating the reason they did not want to participate.

	Minimum number of invitations required*	Number of invitations sent per PCT	Number of invitations sent in total
Group A: Kit	968	500	1,000
Group B: Kit + voucher	484	250	500
Group C: No kit	484	250	500

Table 5-2: Invitations sent by randomisation	on group
--	----------

\*Sufficient to estimate the difference in response rate of 5 percentage points with 95% confidence. Randomised in a ratio of 1:2 (B:A and C:A).

The study questionnaire was designed to include questions that matched those in the Natsal-2 as closely as possible, to allow comparisons between the surveys (see Appendix 1). In order to maximise the acceptability of the study invitations, feedback was obtained on draft versions of the study paperwork (invitation letter and patient information leaflet) using two focus groups among a total of 17 women aged 16 to 18 years old. Focus group participants were recruited from a local higher education college and groups were facilitated by two female facilitators (Woodhall and Collander-Brown). Participants were provided with a £20 voucher as a token of appreciation. Feedback from the focus groups led to the information leaflet being redesigned to make it brighter, and more eye-catching, to reduce the amount of text in the leaflet, and to revise the invitation letter and information leaflet to emphasise that the invitation was being sent only on the basis of the age of the participant, and was not linked to any knowledge about the invitee's sexual activity. A total of 1,000 individuals from each PCT were selected at random for invitation, and were subsequently allocated to three study arms using simple randomisation.<sup>16</sup> The sample size was sufficient to identify a minimum 5% difference in response rate between three different types of postal invitation (Table 5-2). Group B was expected to have higher costs per invitation due to the offer of the £5 voucher. The main comparison was planned between group A and B and group A and C. The groups were therefore randomised in a ratio of 1:2 (B:A and C:A) for the purposes study costs and logistics.

#### 5.2.2.1 Return of biological sample for anonymous testing

Consenting participants returned the questionnaire and self-taken vulvovaginal swab to the HPA. The vulvovaginal swabs were sent to the Sexually Transmitted Bacteria Reference Unit (STBRU) for unlinked anonymous testing. Samples were tested using the APTIMA COMBO 2 assay for both chlamydia and gonorrhoea (Gen-probe Inc., San Diego, CA). Reactive samples were confirmed according to the manufacturers' instructions.

Test results could not be linked to any personal identifiable information, and participants were not sent any test results. Testing was carried out anonymously for a number of reasons. Firstly, as discussed in the Chapter 3, it is reasonable to expect that propensity to participate in a survey of chlamydia prevalence may be related to risk of having chlamydia. If test results had been

<sup>&</sup>lt;sup>16</sup> Random selection and allocation was performed by the local lead at the PCT using an Excel spreadsheet and written procedure to standardise the process across participating sites. Study sites entered the eligible participants into the spreadsheeet which assigned a random number to each eligible participant. Participants were then re-ordered by random number and the first 1,000 were selected. The first 500 were allocated to group A, the second 250 to group B and the third 250 to group C.

provided then it was expected that those who had taken a test recently or who did not consider themselves to be at risk from having chlamydia would be less likely to participate. There were also logistical considerations. In returning results to participants, it would have been necessary to adhere to governance standards for clinical testing, including timely return of results, and arranging partner notification and treatment.<sup>229</sup> It was not considered practical to try to achieve these standards within the resources of the pilot or future surveys. This aspect of the methodology formed an important part of the ethical review of the study (see section 5.2.4). The study was conducted in the context of widespread availability of chlamydia screening and testing. The study information leaflet provided detailed information on where a named test could be obtained locally. The Natsal-3 survey provided an important precedent for this decision, where anonymous testing for STIs without return of results was found to be an acceptable approach.<sup>230</sup>

#### 5.2.3 Analysis

Participation rates were defined as the number of samples returned with consent for testing divided by the number of invitations sent and were measured in each PCT and in each randomisation group. Reasons for non-response derived from the pre-paid postcard were compiled.

Reported sexual behaviours were compared between randomisation groups using a chi-squared test. In order to assess whether the recruited sample was comparable to the general population, I also compared reported sexual behaviours among study participants in each randomisation group to responses from 17 and 18 year-old female participants in Natsal-2 (n=359) and the Health Survey for England conducted in 2010 (HSE2010, n=108).<sup>17</sup> Sampling weights were applied according to the data analysis guides for each survey. The complex survey function in Stata was used to calculate standard errors, in order to allow for clustering and stratification.

In order to identify whether the population tested through this population-based postal survey with anonymous testing was different to the population being tested through the NCSP, the reported sexual behaviour in the pilot survey was compared to information from same-aged women (17 to 18 in Sutton & Merton, 18 in Northamptonshire), tested through the NCSP in the two participating PCTs in 2011 (n=3,485). Table 5-3 shows the variables that were available in each of the comparator datasets, and were used for analysis. All statistical analyses were performed using Stata version 12.0 (StataCorp, College Station, Texas, USA).

<sup>&</sup>lt;sup>17</sup> Although data from Natsal-3 would provide more timely results for comparison, the data were not available at the time of this analysis.

Table 5-3: Sexual behaviour	variables available in the pilot survey and
comparator samples	

Variable	Pilot	Natsal-2	HSE	NCSP
	survey		2010	2011
Ever had chlamydia test	V	x	~	×
Ever had sex	V	V	V	×
More than one sexual partner in the past year*	V	V	V	V
At least one new sexual partner in the past year*	V	V	×	×
More than one sexual partner in the past three months*	V	V	×	×
At least one new sexual partner in the past three months*	~	×	×	V
Condom used at last intercourse	V	~	×	×

\*Variables were collected as continuous variables in the pilot questionnaire (number of sexual partners in past year, number of new sexual partner per year etc). Due to the small numbers available, responses were categorised into binary variables for the purposes of comparison.

An area-level indicator of socioeconomic deprivation (IMD<sup>220</sup>) was assigned to all invited individuals by mapping postcodes of residence to LSOA of residence. Ranks of IMD scores were grouped into quintiles. The distribution of IMD quintiles was then compared between participants and the invited population.

The marginal costs per invitation and per sample received for each randomisation group were estimated (defined as the total of the unit costs of all consumables, postage and testing, divided by the number of invitations sent or the number of samples returned with consent for testing (Table 5-4). Staff and overhead costs were not included, as these were assumed to be equivalent for all recruitment methods.

#### Table 5-4: Unit costs of consumables

Item	Cost per unit
PRINTING: Letters & information sheets	
Printing of invitation letter	£0.05
Printing of PCO cover letter	£0.05
Printing of participant information	£0.10
Printing of participant instruction leaflet	£0.10
Printing of sample collection kit request card/voucher request card	£0.02
Printing of questionnaire	£0.11
Letter to accompany requested sample collection kits	£0.05
Reminder letters	£0.05
Letter to accompany music voucher	£0.05
ENVELOPES: Addressed envelopes for:	
Invitations (group C) (size: A5)	£0.01
Reminder letters (size: A5)	£0.01
Gift voucher/Kit request card (size: DL)	£0.01
Address label	£0.02
POSTAGE	
Invitation - without kit (2nd class letter)	£0.25
Invitation - with kit @ £0.44 (2nd class Large letter)	£0.44
Sample collection kits @ £0.44 (2nd class Large letter)	£0.44
Reminder letters (2nd class letter)	£0.28
Gift voucher (2nd class letter)	£0.28
Freepost for test request slip return (2nd class Freepost, standard response)	£0.32
Freepost for sample return (1st class Freepost, standard response)	£0.42
INCENTIVES	
Gift voucher (£5 each)	£5.00
SAMPLE COLLECTION KITS	
Rigid container+absorbent pad	£0.25
Outer envelope	£0.07
Return envelope	£0.15
Tube label	£0.06
Collection kits @ 1.39 plus VAT	£1.63
LABORATORY TESTING	
Testing (includes sample collection kit) @ £5 plus VAT	£5.88

#### 5.2.4 Regulatory approval

Participants indicated their consent for testing and storage of samples using a tick box on the questionnaire. Signed consent was not requested in order to maintain anonymity of participants.

The study was approved by North London Research Ethics Committee.<sup>18</sup> Research governance approval was obtained from both participating PCTs. Research governance approval was sought to invite both 17 and 18 year-olds. This age group was approved by the Research Ethics Committee. However research governance approval for NHS Northamptonshire was only granted for 18 year-olds due to local guidance on the use of personal information for under-18 year-olds.

National Information Governance Board (NIGB) approval was originally sought to obtain lists of names and addresses of eligible participants from the PCTs, to enable the study team to send invitations directly from the HPA. Approval was not granted, meaning that the initial invitation letter was sent by the PCT, and the HPA study team did not have access to any personal identifiable information from the women invited to the study.<sup>19</sup>

<sup>&</sup>lt;sup>18</sup> REC reference number: 10/H0717/57

<sup>19</sup> Except personal identifiable information provided by participants, in order to request kits or vouchers. This information was not linked to any test result or study information

#### 5.3 Results

#### 5.3.1 Participation rates

A total of 155/2,000 (7.8%) samples were returned with consent, 33 of which were returned after receipt of a reminder letter (Table 5-5). Consent was declined by a further 48 women, and 30 invitations were returned as undelivered. Participation rates were higher in Northamptonshire (90/1,000; 9.0%) than Sutton & Merton (65/1,000; 6.5%) (p=0.04). Participation rates varied by randomisation group; 78/1,000 (7.8%) of those in Group A, 72/500 (14.4%) in Group B and 5/500 (1%) in Group C returned a sample and provided consent (A v B: p<0.001). All received samples had sufficient material for testing; 3/155 (2%) tested positive for chlamydia (1 in Group A, 2 in Group B).

	Randomisation Group				Overa			
Participation rates	<b>A: K</b> i (n=1,00	i <b>t</b> )0)	<b>B: Kit +</b> (n=5	voucher	C: No (n=5	<b>o kit</b> 00)	(n=2,000	))
	n	%	n	%	n	%	n	%
NHS Northamptonshire	47/500	9.4%	41/250	16%	2/250	0.8%	90/1,000	9.0%
NHS Sutton & Merton	31/500	6.2%	31/250	12%	3/250	1.2%	65/1,000	6.5%
Overall	78/1,000	7.8%	72/500	14%	5/500	1.0%	155/2,000	7.8%

rabio o or randopation rateo by randonnoation group	Table 5-5: F	Participation	rates by	randomisation	group
---	--------------	---------------	----------	---------------	-------

#### 5.3.2 Assessment of selection bias

Figure 5-1 shows the reported characteristics among Groups A and B compared to 17 and 18 year-old participants in Natsal-2 and HSE2010, as well as data for 17 and 18 year-old females tested through the NCSP in the two participating PCTs in 2011.

Figure 5-1: Reported sexual behaviours among pilot survey participants, same aged Natsal-2 participants, HSE 2010 participants and NCSP test carried out in participating PCTs in 2011



\*Comparable questions are not available from all sources. Bars show proportions, with 95% confidence intervals shown as error bars. Everyone tested through the NCSP was assumed to be sexually experienced.

A total of 78% participants were sexually-experienced (reported at least one lifetime sexual partner, Figure 5-1). Due to the small sample size in Group C (n=5), no comparisons of reported behaviour or demographic characteristics are reported for this group. A higher proportion of participants in Group B than Group A were sexually-experienced (81% in Group B versus 75% in Group A, p=0.47), reported ever having had a chlamydia test (58% versus 54%, p=0.58) and reported more than one sexual partner in the past 12 months (45% versus 33% among sexually active participants, p=0.18). None of the observed differences were statistically significant (at the 0.05 level).

Responses from participants in the pilot survey were broadly similar to Natsal-2 responses, although there was some heterogeneity. A higher proportion of same-aged sexually experienced Natsal-2 participants reported condom use at last intercourse compared to pilot study participants (46% versus 31% in Group A, p=0.048; and versus 32% in Group B, p=0.045). The percentages of participants who reported ever having had a chlamydia test and who reported being sexually-experienced were higher than those reported among HSE2010 participants (Figure 5-1).

The percentage of participants who reported having had at least one new sexual partner in the last three months was lower than that reported among same-aged women tested through the NCSP in 2011 (40% in both Groups A and B versus 56% in NCSP tests. p=0.02).

The low response rates prevented a subgroup analysis of response rates by IMD quintile. However in both groups A and B, participation rates were higher among those living in less deprived areas (Figure 5-2). Participation rates varied less by IMD quintile in Group B.



Figure 5-2: Participation rates by IMD quintile and randomisation group

IMD: Index of multiple deprivation, quintile based on lower super output area of residence.

#### 5.3.3 Cost per invitations and cost per sample received

The marginal cost per initial invitation was £3.00 for group A, £3.10 for Group B and £0.50 for Group C. The marginal cost per sample received was £51 for Group A, £36 for Group B and £93 for Group C (Table 5-6).

	Group A (Kit)	Group B (Kit + voucher)	Group C (No kit)
Per invitation	£3.00	£3.10	£0.50
Per sample	£51	£36	£93

Table 5-6: Costs per invitation sent and sample received by randomisation group

#### 5.3.4 Reasons for non-participation

A total of 48 women declined consent by returning the pre-paid postcard (3% of all women who did not return a sample). The most commonly cited reasons for non-participation were that women did not want to use the swab (19/48, 40%), did not have the time (12/48, 25%), were not sexually active (9/48, 19%) or were not interested in chlamydia (7/48, 15%). Two respondents (4%) indicated that they wanted to receive their results (Figure 5-3).

#### Doesn't want to use the swab 40% Doesn't have the time 25% Not sexually active 19% Not interested in chlamydia 15% Took a test recently 6% Unwell or not able to take the test 6% Wants to receive results 4% Doesn't feel at risk of chlamydia 4% Recipient has moved 4% 5% 10% 15% 20% 25% 30% 35% 40% 45% 0%

#### Figure 5-3: Reported reasons for non-participation (n=48)

Respondents could choose more than one option. Responses marked \* were derived from freetext responses.

#### 5.4 Discussion

#### 5.4.1 Key findings

This pilot of a postal survey of young women with anonymous testing for chlamydia achieved a maximum response rate of 14%. Given this low response rate, the piloted methodology is not a feasible approach for obtaining regular measures of chlamydia prevalence among young women.

#### 5.4.2 Strengths and limitations

The main strength of this pilot study was that invitations were randomly allocated into different groups to allow investigation of the response rates that could be expected given different approaches. This showed that offering a small financial incentive slightly increased participation, and reduced the cost per sample received, although none of the piloted approaches achieved an acceptable response rate.

The study was subject to limitations. Firstly, it was not possible to determine whether participation rates would have been higher if participants had been offered their results. People who take part in named chlamydia testing are, on average, at higher risk of infection than the general population<sup>182</sup>. Therefore participants were not provided with their test results to reduce potential non-response bias (section 3.2.2.1). Among the 48 individuals who provided a reason for non-participation using the pre-paid postcard, only 2 respondents stated that they did not take part because they wanted to receive their test results. Although this is a small sample and only indicative of potential reasons for non-participation in the overall population, this suggests that providing test results would not have led to substantially higher response rates. This is consistent with the low participation rates in other recent postal chlamydia screening studies, where named testing was used<sup>88,158</sup>. However it is possible that offering test results may have led to a larger sample size.

A further limitation is that the age of those eligible and of participants was not known, meaning that not possible to determine whether the difference in participation rates between Northamptonshire and Sutton & Merton was associated with age or whether this was more likely to be due to local variation

in the willingness to participate or completeness of the GP registers. It was originally envisaged that year and month of birth would be provided from the participating PCTs, but this was unavailable. A lower response rate in the London PCT (Sutton & Merton) would be consistent with other studies that have shown lower participation rates within London compared to the rest of the country<sup>171</sup>.

The GP registers may not have been accurate in terms of registered patients' addresses. A total of 30/1,845 (1.6%) invitations among non-participants were returned to sender, indicating that they never reached the intended participant. This is unlikely to be the total number of undelivered invitations. These undelivered invitations were included in the denominator for participation rates. If the proportion of 'ghost' patients seen in the ClaSS study (26%, see section 3.2.2.2)<sup>10</sup> is applied to the results of this pilot, then the participation rate among those likely to have received the sample would be higher, but still low (11% in Group A, 19% in Group B, 1% in Group C).

The most commonly stated reason for non-participation was not wanting to use the self-taken VVS. While VVS have been found to be an acceptable method of sampling for diagnostic tests or chlamydia screening, as no clinical result was being returned, this may have reduced the willingness to use the swab. Urine samples are possible alternatives, but the sensitivity of NAAT tests among women is lower using urine compared to VVS samples<sup>231</sup>.

#### 5.4.3 Implications for evaluation of chlamydia control

Low response rates do not necessarily lead to biased estimates of outcome measures; bias will only occur if participants and non-participants differ in

respect to the outcome of interest. In order to gain some insight into the nonresponse bias in this survey, participation rates were compared by residence based deprivation, and responses from the pilot were compared to three other samples (Natsal-2, HSE2010 and women tested through the NCSP). Participation rates were higher among those in less deprived areas. As residence-based deprivation measures have been found to be a risk factor for chlamydia infection,<sup>133</sup> this suggests there may have been some important differences between participants and non-participants that may have introduced bias. Although limited by sample size, there was some evidence to suggest that these differences (and the potential resulting bias) were reduced with the offer of a voucher. There were minimal differences between our participants and the same aged Natsal-2 or HSE respondents. While this suggests that the findings from these three sources may be consistent, the utility of these comparisons is limited by the sample size in Natsal-2 and HSE for 17 and 18 year-old women and as Natsal-2 and HSE2010 provide national estimates that would mask local variation. Furthermore, Natsal-2 was carried out over 10 years ago, which may limit comparability of findings. Results from the more-recently conducted Natsal (Natsal-3, conducted 2010-12) were not available at the time of the pilot study. One alternative approach to explore the non-response bias would be to compare those who responded before or after a reminder was sent. If differences between groups are seen, then it is also reasonable to assume there to be important differences between participants and non-participants<sup>232</sup>. However numbers were too small for meaningful comparisons within this pilot survey.

Although it was not possible in this pilot to establish with certainty whether participants and non-participants had the same risk of having chlamydia, given

the low participation rates, a survey using the piloted methodology would be open to substantial, and potentially variable, selection bias. Even if the prevalence of infection among participants and non-participants did not differ, the costs of implementing repeated cross-sectional surveys would be prohibitively large, due to the number of invitations needed to achieve the sample size needed to measure relatively small changes in prevalence between surveys.

#### 5.4.4 Summary

In summary, this pilot study showed that repeated cross-sectional studies of chlamydia prevalence using postal invitations with anonymous testing for chlamydia is not a suitable method for measuring chlamydia prevalence in the general population. Other methods for measuring chlamydia prevalence are therefore required, and in the next chapter I explore the use of data from large nationally-representative sexual behaviour surveys as an alternative to chlamydia-specific surveys.

#### 6 Chlamydia prevalence measured in the second and third National Surveys of Sexual Attitudes and Lifestyles (Natsal-2 and Natsal-3)

The results of the previous chapter show that a postal survey of chlamydia prevalence is unfeasible, would incur unjustifiably high costs and would be open to considerable selection bias. Repeated cross-sectional postal surveys are not suitable for population-based monitoring of chlamydia prevalence among young women in England. Other methods for monitoring chlamydia prevalence over time are therefore required. In the following two chapters I investigate the use of the National Surveys of Sexual Attitudes and Lifestyles (Natsal) as a means of comparing agespecific chlamydia prevalence among a sample of the general population at different time points (presented in this chapter) and to explore how chlamydia screening had been delivered up to 2012, and what the epidemiology of infection in relation to testing can tell us about the actual or expected impact of chlamydia screening (Chapter 7).

#### 6.1 Background

The analyses in this and the subsequent chapter use data from the second and third National Surveys of Sexual Attitudes and Lifestyles (Natsal-2 and Natsal-3 respectively). Natsal are three, stratified probability sample surveys of the British general population. The first survey (Natsal-1) was conducted in 1990-1991,<sup>170</sup> Natsal-2 in 1999-2001<sup>171</sup> and Natsal-3 in 2010-12.<sup>172</sup> In both Natsal-2 and Natsal-3, chlamydia prevalence was estimated from urine samples among a subset of participants.<sup>23</sup> The surveys also provide a wealth of information on sexual behaviour, attitudes and lifestyles, experience of STI diagnoses (including chlamydia) and, in Natsal-3, chlamydia testing.

As both Natsal-2 and Natsal-3 measured the prevalence of *C. trachomatis* detected in urine (hereafter termed 'prevalent infection'), the surveys provide a unique opportunity to compare estimates of chlamydia prevalence among the general population of young adults before and after the widespread implementation of chlamydia screening. However, as described in detail below, there were differences in the detection strategy used in each survey, meaning that such comparisons should be made with caution. The analyses presented in this chapter were conducted to explore how this and other differences between the surveys might affect the conclusions that can be made about changes in chlamydia prevalence among young adults using a series of adjustments and counterfactual scenarios.

#### 6.2 Aims & objectives

To investigate whether the Natsal surveys can be used to compare chlamydia prevalence before and after the implementation of the NCSP and determine how differences between the surveys and underlying changes in sexual behaviour affect the conclusions that can be made about change in chlamydia prevalence between 1999-2001 and 2010-12. This aim was addressed through the following objectives:

- To compare chlamydia prevalence among 18 to 24 year-old participants in the Natsal-2 and Natsal-3 surveys.
- To adjust Natsal-2 and Natsal-3 prevalence estimates and comparisons for known and possible differences between the surveys.
- To describe differences in reported sexual behaviour and markers of sexual behaviour between the surveys and their potential impact on chlamydia prevalence.

#### 6.3 Methods

# 6.3.1 Summary of survey methodology and testing of biological specimens in Natsal-2 and Natsal-3

Full details of the survey methods and questionnaire have been described elsewhere.<sup>172</sup> In summary, both surveys used an independently designed, multistage, clustered, stratified probability sample, using the 'small-user' Postcode Address File, a list of all addresses (delivery points) in the Britain, as the sampling frame. Postcode sectors were selected as the primary sampling units, addresses within them were selected at the second stage, and finally one eligible adult per address was randomly selected at the final stage. In both surveys, participants were interviewed in their own homes using computerassisted face-to-face and computer-assisted self-interview for the most sensitive questions.<sup>4,5</sup> Most questions in Natsal-3 were identical in wording to those used in Natsal-2. However, there were some questions used in Natsal-3 that were not used in Natsal-2, including whether someone had been tested for chlamydia in the last year and reason for testing. Some questions had slightly altered in wording, the only relevant one for the analyses presented in this thesis being the wording for self-reported diagnosis in the last year (see Box 6-1). A summary of the key features and differences between Natsal-2 and Natsal-3 is provided in Table 6-1.

	Natsal-2	Natsal-3
Age range	16 to 44 years	16 to 74 years*
Number of participants	11,161 <sup>\$</sup>	15,162
Survey method	Computer-assisted personal ir computer-assisted se	nterviewing (CAPI) including elf-interview (CASI)
Response rate <sup><math>^</math></sup>	65%	58%
Self-reported chlamydia test in the last year collected?	No	Yes
Self-reported chlamydia diagnosis in the last year collected?	Yes – calendar year of last diagnosis.	Yes
Eligibility to provide a urine sample	A subset of sexually- experienced participants <sup>133</sup>	All 16 to 17 year-olds (regardless of reported sexual activity); all sexually-experienced 18 to 24 year-olds; a random subsample of 25 to 44 year-olds <sup>23</sup> .
Urine collection device	Urine cup	'FirstBurst' urine collection device
Diagnostic assay	Ligase Chain Reaction (LCx) (Abbott Diagnostics) <sup>133</sup>	Aptima Combo 2 (AC2) (Hologic Gen-Probe) <sup>23</sup>
Did participants receive their chlamydia test results?	Yes	No

## Table 6-1: Key features of the second and third National Surveys of Sexual Attitudes and Lifestyles<sup>5,172</sup>

\* Younger adults aged 16 to 34 years were oversampled to increase statistical analysis power for this group.

<sup>^</sup>As eligibility within households was not known for those households not contacted, these calculated response rates use an estimated denominator of eligible using data from households that could be contacted.<sup>4,5</sup>

<sup>\$</sup> Natsal-2 also included a boost sample of adults of Black and Asian ethnicity, but these data were not included in the analyses presented in this thesis.

In both surveys, a subset of participants was invited to provide a urine sample for testing for *C. trachomatis* testing. In Natsal-2, consenting participants were asked to collect a first-catch urine specimen in a plastic urine cup and transfer this to a 20ml universal container (containing 1 mg boric acid as a stabilising agent) for posting to the laboratory where they were tested for chlamydia using the ligase chain reaction test by Abbott Diagnostics (hereafter termed LCx)<sup>133</sup>.
Positive samples were confirmed through a repeat test using the LCx on the same sample (personal correspondence, C Carder. 2013). In Natsal-3, participants were asked to collect a urine specimen using a 'FirstBurst' urine collection device, which diverts the first 4-5mls of urine flow into specimen collection tube, without dilution from subsequent urine flow. This device increases the bacterial load in the specimen held in the specimen collection tube compared to a first catch specimen using a urine cup.<sup>233</sup> Urine samples were posted to Public Health England (PHE) where they were tested for *Chlamydia trachomatis* using the Aptima Combo 2 assay (Hologic Gen-Probe); all positive and equivocal results were confirmed with the Aptima chlamydia monospecific assay.<sup>23</sup>

In Natsal-2, participants received their chlamydia test results if they were positive whereas testing in Natsal-3 was carried out anonymously (i.e. without return of results) and participants did not receive their results for any of the other STI tests. The rationale for STI test results not being provided in Natsal-3 has been provided elsewhere.<sup>230</sup> Briefly, this was deemed appropriate and ethical for Natsal-3 given the availability of free STI testing and advice at the time of the survey. Furthermore, urine samples were tested for other STI (HPV, gonorrhoea, *Mycoplasma genitalium* and HIV), and for some of these (HPV and *Mycoplasma genitalium*) the result would be of uncertain clinical value. Timeliness of testing and in some cases test accuracy were also limited given the constraints of the survey conditions and specimen type meaning that STI testing would not be fully compliant with clinical standards. Participants were given information about where to obtain free diagnostic STI and HIV testing and sexual health advice.

In both surveys, urine specimens were also tested for HPV using an in-house Luminex-based genotyping assay for detection of HPV types 6, 11, 16, 18, 26, 31, 33, 35, 39, 45, 51, 52, 53, 56, 58, 59, 66, 68, 70, 73 and 82. In Natsal-2, this testing was carried out following storage at -80C as part of a separate, ethically-approved study<sup>234</sup>. In Natsal-3, participants consented to their specimen being anonymously tested for HPV, and testing was carried out shortly after collection.<sup>23,234</sup>

#### 6.3.2 Participants

All analyses in this chapter are based on data from sexually-experienced participants in Natsal-2 or Natsal-3. Apart from summary estimates of the key outcome measures (see below), which were produced for 16 to 44 year-olds, all analyses were conducted among women and men aged 18 to 24 year-old as the age group targeted for chlamydia screening by the NCSP and as urine samples were not collected from 16 to 17 year-old participants in Natsal-2.

The key outcomes of interest were prevalent infection, self-reported chlamydia diagnosis in the last year ('recent diagnosis'), self-reported chlamydia diagnosis at any time prior to the interview date ('ever diagnosed') and, for analyses among Natsal-3 participants, self-report of a chlamydia test in the last year ('recent testing').

Figure 6-1: Flow chart showing participants included (16 to 44 year-old Natsal-2 & Natsal-3)



#### 6.3.3 Statistical analysis

#### 6.3.3.1 Complex survey sampling and weighting

All analyses were carried out using Stata 12.1, accounting for the weighting, clustering and stratification of the Natsal data.<sup>23,172,235</sup> In both surveys, weights constructed and provided by the survey team were applied to adjust for unequal probability of selection and non-response to ensure the sample data were broadly representative of the British general population, according to the most recent census, in terms of gender, age group and Government Office

Region<sup>236,237</sup> and the differential provision of urine samples by demographic and selected sexual behaviours.<sup>5,172</sup>

Questions about self-reported diagnosis of chlamydia differed slightly between surveys. For estimates of diagnoses in the last year among Natsal-2 participants, I therefore constructed an additional weight to correct for the overestimate of recent diagnosis in the last year and to allow comparisons between the surveys (Box 6-1).

#### Box 6-1: Construction of additional weight for diagnosis in the last year

In Natsal-2, participants were asked the calendar year of their most recent chlamydia diagnosis, whereas in Natsal-3, participants were asked whether they had been diagnosed in the 12 months before the interview. A further weighting was therefore applied when estimating self-reported diagnosis in the last year in Natsal-2 to allow comparisons between the surveys. These additional 'diagnosis weights' were constructed as the number of days in the year specified that would have been in the 12 months before the interview. Weights were constructed for each month, assuming the date of interview was at the midpoint of the month (15th of the month, apart from February, where 14th was used).

For example, for a participant interviewed on the 15th of January 2000 who reported their last diagnosis was in 1999, 351 of the 365 days in 1999 would have been within the 12 months preceding the interview, thus the 'diagnosis weight' was 351/365=0.96. For a participant interviewed on the 15th of December 2000, only 16/365 days in 1999 would have been within the 12 months before the interview, resulting in a weight of 0.04. Table 6-2 below shows diagnosis weights calculated for each month. Diagnosis weights were then multiplied by the urine weights (which incorporate unequal probability of selection, nonresponse to main survey and urine sub-study), and used as final weights for estimate of diagnosis in the last year in Natsal-2. Figure 6-2 shows the difference in the estimated proportion with a chlamydia diagnosis in the last year by gender and age group.

Month of interview	January	February	March	April	Мау	June	July	August	September	October	November	December
Days within preceding year	351	321	291	260	230	199	169	138	107	77	46	16
'Diagnosis weight'*	0.96	0.88	0.8	0.71	0.63	0.55	0.46	0.38	0.29	0.21	0.13	0.04

#### Table 6-2: Diagnosis weights by month of interview

\*Number of days within preceding year / 365. Presented to 2 decimal places.





## 6.3.3.2 Comparison of chlamydia prevalence and diagnoses in Natsal-3 versus Natsal-2 among under-25 year-olds

The prevalence of infection and of self-reported diagnoses (ever or in the last year) in Natsal-3 was compared to Natsal-2 using univariable logistic regression with survey (i.e. Natsal-2 or Natsal-3) entered as the independent variable.

These comparisons were further explored by setting up a series of counterfactual scenarios to investigate the impact of differences between surveys on comparisons of chlamydia prevalence between them.

## 6.3.3.3 Correction for measurement error arising from imperfect diagnostic tests

As set out in section 6.3.1 above, different specimen collection devices and diagnostic tests were used in Natsal-2 and Natsal-3, reflecting changes in technologies between them. Differences in the detection strategy (i.e. the combination of tests, confirmation procedures and specimen collection devices) used in each survey may affect prevalence estimates and their comparability.<sup>140,238,239</sup> Imperfect tests lead to misclassification bias, whereby tests with <100% sensitivity result in individuals with an infection being incorrectly classified as negative ('false negatives') and tests with <100% specificity in individuals without an infection being categorised as having an infection ('false positives'). Thus imperfect tests can over or underestimate 'true' prevalence.<sup>240-243</sup> The extent and direction of this error will depend on both the sensitivity and specificity of the test used and the true prevalence of infection within the population (Figure 6-3).



Figure 6-3: Difference between observed and true prevalence with different test characteristics

Where the sensitivity and specificity of a diagnostic test is known, the 'true' prevalence of infection (i.e. that would have been measured, had a perfect test been used), can be calculated using the following formula set out by Rogan and Gladen:<sup>243</sup>

By applying this equation to observed prevalence estimates, the 'true' prevalence can be calculated. However this relies on knowing the true

sensitivity and specificity of the test, which in turn relies on having a perfect 'gold standard' test, i.e. a test (or combination of tests, also termed a 'reference standard') that is capable of classifying individuals into those who do and do not have an infection. Estimates of the sensitivity and specificity of chlamydia NAATs have been shown to vary according to the gold standard used.<sup>238,240,244</sup> Estimated performance characteristics of earlier NAATs (such as LCx) are particularly subject to error, where less sensitive assays (i.e. culture or enzyme immunoassays) have been used as the gold standard, or when discrepant analysis<sup>20</sup> has been applied.<sup>238,240,245-247</sup> Adjusting prevalence estimates to achieve a true prevalence is therefore problematic. To address this, the prevalence estimated in Natsal-2 was adjusted to estimate the prevalence of chlamydia that *would have been measured*, had the same detection strategy been used in Natsal-2 as was used in Natsal-3, thereby increasing the comparability of the prevalence estimates of true prevalence.

To do this, firstly the 'test-adjusted' prevalence was calculated by applying the Rogan & Gladen formula [1] to the prevalence estimates and 95% confidence intervals<sup>248</sup> in Natsal-2, using the sensitivity (91.1%) and specificity (99.1%) of the LCx compared to the AC2 NAAT from a head to head study of the two assays reported by Gaydos et al<sup>249</sup> (Box 6-2). A 'detection strategy-adjusted' prevalence was then calculated by applying this formula [1] to the 'test-adjusted' prevalence. The sensitivity (97.1%) and specificity (100%) estimates for the

<sup>&</sup>lt;sup>20</sup> A method introduced to overcome problems of the gold standard being less sensitive than the test under evaluation. Specimens that were positive for chlamydia on the assay under evaluation but negative by the gold standard (usually culture) were tested using an additional assay. This 'discrepant analysis' approach was subsequently discredited as it leads to systematic overestimation of the sensitivity and specificity of the test under evaluation.<sup>240</sup>

FirstBurst were taken from a study of urine specimens collected from 534 men using both a urine cup and the FirstBurst device and tested using the Amplicor CT/NG PCR assay (Roche Molecular Systems, Branchburg, NJ).<sup>233</sup>

As Natsal-2 and Natsal-3 are complex surveys, the standard errors are estimated using the complex survey function in Stata to account for the stratification and clustering of the surveys. However, as the 'detection strategyadjusted' estimates were calculated based on the previously estimated point estimates (and 95%Cls) of chlamydia prevalence, the complex survey analysis function in Stata could not be applied. An approximate method was therefore used to compare the adjusted Natsal-2 prevalence estimates to Natsal-3 prevalence, while also taking into account the original clustering of the sample. Firstly, the effective sample size (the sample size which equates to the given confidence intervals, if the survey had been conducted as a simple random sample), was derived using the adjusted prevalence estimates, and the design effects for each age group.<sup>250</sup> A simulated dataset was then created where the number of observations was equal to the effective sample size and the number with a positive chlamydia test result corresponded to the 'detection strategyadjusted' point estimates. The simulated dataset was then used to calculate the odds ratio for prevalence in Natsal-3 versus 'detection strategy adjusted' prevalence in Natsal-2 using logistic regression.

### Box 6-2: Literature review of sensitivity and specificity of ligase chain reaction (LCx) and Aptima Combo 2 (AC2) assays for *Chlamydia trachomatis* detection

In order to adjust the Natsal-2 prevalence to determine the prevalence that would have been observed had the same detection strategy been used as was used in Natsal-3, the relative sensitivity and specificity of the LCx compared to the AC2 was needed. A literature review was therefore carried out to identify i) systematic reviews of the sensitivity and specificity of the LCx and AC2 and ii) studies where the LCx had been compared directly to the AC2. Along with a search of an electronic database (MEDLINE), unpublished data were obtained via contact with experts in the field (personal communication, Prof Catherine Ison).

Two reviews were identified that provide summary estimates of the sensitivity and specificity of the LCx and AC2. In a systematic review by Nelson *et al*<sup>251</sup> estimated sensitivity of LCx among non-pregnant women ranged from 70% to 96% and specificity between 99% and 100%. In a review of non-invasive testing for chlamydia and gonorrhoea, Cook *et al*<sup>252</sup> included studies which compared urine to other sample types and reported pooled estimates for TMA (ACT or AC2) in urine of 92.5% (95%CI 88.0%-97.0%) sensitivity and 98.6% (95%CI 97.7%-99.6%) specificity in women and 87.7% (95%CI 80.1%-95.2%) sensitivity and 99.4% (95%CI 98.7% - 100%) specificity in men. However the gold standards used for the studies included in these reviews vary considerably, with earlier studies of the LCx often comparing to culture.

Five studies were identified which included a head to head comparison of LCx and AC2.<sup>249,253-256</sup> However only in the study by Gaydos et al, where 506 first catch urine samples were tested with both the LCx and AC2, can the performance characteristics of LCx in relation to AC2 (i.e. using AC2 as a gold standard) be calculated from the results reported.<sup>249</sup> In this study, 506 first catch urine samples from women and men aged 12 to 20 years old attending school-based clinics for routine screening were tested with both the LCx and AC2.<sup>249</sup> This resulted in a sensitivity and specificity of the LCx compared to the AC2 of 91.1% and 99.1% respectively.

(Based on Table 1, Gaydos et al 2004 )										
AC2 ('gold standard)										
	Positive Negative Total									
	Positive	72	4	76						
LCx	Negative	7	423	430						
	Total	79	427	506						

Table 6-3: Comparison of results from LCx and AC2 from 506 urine specimens(Based on Table 1, Gaydos et al  $2004^{249}$ )

Sensitivity of LCx relative to AC2: 91.1% Specificity of LCx relative to AC2: 99.1% Samples that were positive only on the AC2 were retested using the APTIMA chlamydia monospecific assay (ACT)<sup>249</sup>. Results shown for the AC2 are based on AC2 plus confirmation with ACT where both tests were available as this is closest to the detection strategy used in Natsal-3.<sup>23</sup>

AC2: Aptima Combo 2; LCx: Ligase chain reaction; ACT: Aptima chlamydia monospecific assay

#### 6.3.3.4 Combining urine infections with recent screen-detected diagnoses

The average duration of an untreated, asymptomatic chlamydia infection is estimated to be in the region of 16 months.<sup>18</sup> Therefore in both surveys, a proportion of the individuals who had been diagnosed with chlamydia in the last year might have had an infection at the time of the survey had they not been tested and appropriately treated (i.e., had their duration of infection not been shortened by treatment). A hypothetical scenario was constructed to estimate what the prevalence of infection would have been in Natsal-3, had there been no asymptomatic screening among young adults in the year before the survey.

In order to do this, diagnoses were categorised as being either 'screendetected' or not, based on the reason reported for most recent test. The Natsal-3 questionnaire did not ask where an individual was last diagnosed or why they had been tested. Reason of most recent test among those diagnosed in the last year was therefore assumed to refer to the test associated with the diagnosis. Reported tests in the last year were considered to be 'clinically-indicated' where the reason for last test was reported as either having had symptoms, having a partner diagnosed with chlamydia or who had symptoms, or a repeat test after a previous positive result. Individuals who reported having been diagnosed with chlamydia in the last year were considered to have a 'screen-detected' diagnosis if they reported a non-clinically indicated reason for their last test ("wanted a check-up"; "offered a test" or "worried about risk"). Reason for test was not collected in Natsal-2. For the purposes of this analysis, all reported diagnoses in Natsal-2 were considered to be a result of clinically-indicated testing, as the survey was carried out before the implementation of widespread chlamydia screening.

The proportion of Natsal-3 participants who had a 'screen-detected' diagnosis in the last year was estimated. As the simplest scenario, it is assumed that all of these 'screen-detected' infections would have persisted to the time of the Natsal-3 interview in the absence of screening. The estimated prevalence in Natsal-2 was then compared to the combined prevalence of infection and/or 'screen-detected' diagnoses in Natsal-3 using logistic regression.<sup>21</sup>

#### 6.3.3.5 Adjustment for hypothetical and unmeasured differences between participants and non-participants

Survey estimates may be biased where participants and non-participants differ with regard to the outcome of interest. As described in section 6.3.3, the weights applied to the Natsal-2 and Natsal-3 data were designed to minimise such participation bias arising firstly for non-response in the Natsal survey, and secondly for differential participation in the biological sampling part of the survey. In the weights to address the former, adjustments could be made to ensure the achieved sample was representative of the British population with regard to gender, age group and Government Office Region.<sup>236,237</sup> Weights could not adjust for differences in sexual behaviour, as no information was available from non-participants. As detailed demographic and behavioural data were available from survey participants who did and did not consent to provide a urine sample for chlamydia testing, survey weights could provide a more detailed adjustment for non-participation bias.<sup>5,172</sup> However it is feasible that participation bias remains in each survey due to unmeasured differences between participants and non-participants.

<sup>&</sup>lt;sup>21</sup> With additional weights applied for estimation of recent diagnoses in Natsal-2 as described in Box 6-1.

The potential impact of unmeasured/uncorrected-for participation bias was explored by calculating a range of 'participation-adjusted' prevalence estimates for Natsal-2 and Natsal-3, by varying the (hypothetical) participation bias in each survey from the estimated chlamydia prevalence being 50% lower to being 50% higher than among non-participants. The resulting odds ratios for each combination were then calculated and plotted against the participation bias in Natsal-2 and Natsal-3. In all scenarios response rates were assumed to be equivalent to those seen in each survey.<sup>22</sup> The resulting scenarios represent hypothetical differences between participants and non-participants which have not been captured by either the post-stratification weighting to account for non-participation in the main survey, or by the weights applied for differential provision of urine samples by demographic and behavioural characteristics.

#### 6.3.3.6 Indicators of change in sexual behavioural risk of transmission

Identification and treatment of chlamydia infections is only one of the factors that affect chlamydia transmission and prevalence. If sexual behaviours changed in the ten year period between the Natsal surveys, then in the absence of widespread screening, chlamydia prevalence may have either increased or decreased. In order to explore whether there is any evidence for a change in the behavioural risk of STI transmission, the estimated prevalence of reported sexual behaviours (condom use at last sex; numbers of new or condomless sexual partners in the last year; number of lifetime sexual partners) was compared between 18-24 year-old participants in Natsal-2 and Natsal-3. The proportion of Natsal-3 participants who reported having had their first

<sup>&</sup>lt;sup>22</sup> Overall participation rate in Natsal-2 (65.4%)<sup>171</sup>; 16-29 year-old boost sample in Natsal-3  $(67.3\%)^5$ .

heterosexual sexual intercourse before the age of 16 was estimated and plotted by year the participant turned 16.

The prevalence of non-vaccine-HPV types detected in urine was also compared between the surveys, as a biological marker of sexual behaviour change. Non-vaccine types comprised high-risk HPV types other than those included in the bivalent HPV vaccine (HPV 16, 18) and those with evidence of cross-protection (HPV 31, 33, 45).<sup>257</sup> HPV 6 and 11, which were also tested for, were considered non-vaccine types for the purposes of this analysis as although a quadrivalent vaccine is available that protects against HPV 6 and 11, the bivalent vaccine was used within the national vaccination campaign up to 2012.<sup>258</sup> Thus Natsal-3 participants would not have been eligible for quadrivalent vaccination. Participants who were HPV positive but where no type-specific data were available were excluded (Figure 6-4).

Figure 6-4: Flowchart showing samples included in analysis of non-vaccine HPV prevalence (18-24 year-old sexually-experienced women and men)

Natsal-2



#### 6.3.4 Regulatory approval

Natsal-2 obtained ethical approval from University College Hospital, North Thames Multicentre, and all local research ethics committees in Britain.<sup>133</sup> Natsal-3 was approved by Oxfordshire Research Ethics Committee A (reference 09/ H0604/27). Approval for use of these data for this analysis was granted by the Natsal-3 Project Management Team.

#### 6.4 Results

### 6.4.1 Summary of prevalence, diagnosis and testing by age group and gender (Natsal-3)

Table 6-4 shows the prevalence of chlamydia infection and of self-reported chlamydia testing and diagnosis in the last year among sexually-experienced 16 to 44 year-olds. Chlamydia prevalence and recent diagnoses peaked among 20 to 24 year-olds in men, and among 16 to 19 year-olds in women. No prevalent infections were detected among 16 to 18 year-old men.<sup>23</sup> Testing was highest in 16 to 19 year-olds among both men and women. The proportion who had ever been diagnosed with chlamydia peaked in an older age group than the peak for recent diagnoses (i.e in 25 to 29 year-olds in men and 20 to 24 year-olds in women). Reported diagnoses and testing were higher among women than men across all age groups.

Among 16 to 24 year-olds, 3.1% of women and 2.3% of men had a prevalent chlamydia infection, 62.5% of women and 43.2% of men had either been tested, or offered a chlamydia test in the last year and 12.3% of women and 5.3% men had ever been diagnosed with chlamydia.

 Table 6-4: Prevalence of chlamydia (detected in urine), and self-reported testing and diagnosis by age group and sex (sexually-experienced

 16-44 year-olds, Natsal-2 and Natsal-3)

Prevalent infection**		Ever diagnosed with chlamydia		Diagnosed with chlamydia in the		Tested in the last	Offered, not	Prevalent	Bases, Wt, U			<b>NWt</b> Diagnosed in	
-				· · · · <b>,</b> · · ·	last	year	year	tested in last year	infection	ever d	iagnosed	the la	ast year
Age group (years)	Natsal-2	Natsal-3	Natsal-2	Natsal-3	Natsal-2*	Natsal-3*	Natsal-3	Natsal-3	N2 N3	N2	N3	NZ	N3
Women													
16-19	3.9% (1.4-10.3)	3.8% (2.2-6.3)	1.3% (0.6-3.2)	7.4% (5.5-9.9)	1.1% (0.5-2.8)	3.4% (2.2-5.2)	56.6% (52.5-60.6)	10.4% (8.2-13.1)	86, 234, 73 343	451, 387	374, 582	451, 387	374, 582
20-24	2.8% (1.5-5.2)	2.7% (1.7-4.3)	5.3% (3.9-7.2)	14.9% (12.7-17.5)	1.7% (1.0-2.9)	2.7% (1.8-4.1)	52.8% (49.2-56.4)	7.1% (5.5-9.2)	271, 391, 209 497	773 <i>,</i> 598	629, 793	771, 598	629, 793
25-29	2.1% (1.1-4.0)	2.2% (1.2-4.0)	3.5% (2.6-4.8)	12.3% (10.5-14.3)	0.3% (0.1-0.6)	0.9% (0.5-1.6)	30.0% (27.2-32.9)	3.9% (2.9-5.3)	328, 422, 256 419	869, 777	651, 826	867, 777	651, 826
30-34	1.4% (0.6-3.0)	0.8% (0.4-1.9)	3.8% (2.8-5.0)	9.3% (7.6-11.3)	0.2% (0.1-0.5)	0.1% (0.0-0.5)	16.5% (14.2-19.2)	0.6% (0.3-1.3)	375, 385, 318 274	1008, 950	654 <i>,</i> 630	1008, 950	654 <i>,</i> 630
35-39	1.0% (0.4-2.5)	0.6% (0.1-2.6)	3.2% (2.4-4.3)	5.0% (3.5-7.1)	0.0% (0.0-0.1)	0.0% -	10.1% (7.8-12.9)	0.9% (0.4-2.2)	384, 393, 311 167	1059, 881	671, 382	1059, 881	671 <i>,</i> 382
40-44	0.2% (0.0-1.4)	0.0% -	1.5% (0.9-2.5)	4.6% (3.2-6.7)	0.1% (0.0-0.3)	0.0% -	8.3% (6.3-10.8)	0.2% (0.0-1.1)	305, 441, 269 185	946, 798	724 <i>,</i> 408	946, 798	724, 408
Men													
16-19	2.8% (0.7-11.1)	0.3% (0.1-1.4)	0.2% (0.0-1.4)	3.5% (2.2-5.7)	0.2% (0.0-1.4)	1.9% (1.0-3.7)	40.4% (35.9-45.1)	8.0% (5.9-10.8)	94, 214, 94 395	455, 410	344, 675	454, 410	344, 675
20-24	2.9% (1.1-7.4)	3.4% (2.2-5.2)	1.9% (0.9-3.8)	6.3% (4.8-8.4)	0.7% (0.2-2.0)	2.1% (1.2-3.5)	31.1% (27.8-34.7)	8.9% (6.8-11.4)	254, 383, 272 597	791, 778	624, 1065	785, 778	624, 1065
25-29	4.6% (2.3-9.0)	0.8% (0.3-1.9)	2.1% (1.2-3.4)	10.2% (8.2-12.7)	0.6% (0.2-1.5)	0.8% (0.3-2.0)	19.4% (16.6-22.5)	3.0% (1.9-4.6)	313, 412, 398 650	920, 1143	670, 1364	918, 1143	670, 1364
30-34	1.4% (0.5-3.5)	1.2% (0.5-3.0)	1.8% (1.1-2.9)	9.2% (7.0-12.1)	0.1% (0.0-0.4)	0.4% (0.1-1.4)	9.5% (7.2-12.3)	0.7% (0.3-1.7)	368, 397, 466 496	1065, 1325	653, 1033	1063, 1325	653, 1033
35-39	1.4% (0.5-4.0)	0.6% (0.1-4.4)	1.4% (0.8-2.4)	4.4% (2.7-7.1)	0.1% (0.0-0.4)	0.1% (0.0-0.9)	4.7% (3.1-7.2)	0.0% -	344, 421, 415 268	1085, 1252	673, 591	1084, 1252	673, 591
40-44	0.5% (0.1-3.3)	0.0% -	0.6% (0.2-1.6)	2.8% (1.5-5.1)	0.0% -	0.6% (0.2-1.6)	3.5% (2.1-5.9)	0.5% (0.1-3.2)	309, 457,	966,	751,	965,	751,

\*Estimates of proportion diagnosed in last year in Natsal-2 have been calculated using additional weights to adjust for over-estimation and to make comparable between surveys. See Box 6-1 for details. \*\*Natsal-2 prevalence estimates in the youngest age group are among 18-24 year-olds, as 16-17 year-olds were not tested for chlamydia.

## 6.4.2 Comparison of chlamydia prevalence and diagnoses in Natsal-3 versus Natsal-2 among under-25 year-olds

Table 6-5 compares the prevalence of self-reported chlamydia diagnoses (ever and in the last year) for 16 to 24 year-olds in Natsal-2 and Natsal-3 and the corresponding estimates of chlamydia prevalence for 18 to 24 year-olds. In univariable analyses, there was no significant difference between the prevalence of chlamydia among 18 to 24 year-olds in Natsal-3 compared to Natsal-2 (OR 1.04, 0.53-2.01 in women; OR 0.91, 0.36-2.27 in men). Among 16 to 24 year-olds, there was at least a three-fold increase in the odds of ever having been diagnosed with chlamydia (OR in women: 3.52, 2.47-5.01; OR in men: 4.40, 2.11-9.16), and at least a two-fold increase in the odds of having been recently diagnosed (OR in women 1.99, 1.12-3.54; OR in men 4.08, 1.40-11.8). There was no notable difference in these comparisons when limiting to participants resident in England (data not shown). Table 6-5: Percentage with a prevalent chlamydia infection, ever diagnosed with chlamydia and recently diagnosed with chlamydia by survey and odds ratio (OR) for difference between surveys (sexually-experienced 16-24 year-olds)

	Natsal-2		Natsal-3		Natsal-3 v Natsal-2			Bases (W,UW)		
	%	95%C.I	%	95%C.I	OR	95%CI	р	Natsal-2	Natsal-3	
Women										
Ever diagnosed with chlamydia	3.8%	(2.8-5.2)	12.3%	(10.6-14.1)	3.52	(2.47 - 5.01)	<0.001	1246, 1188	968, 1740	
Diagnosed with chlamydia in the last year	1.5%	(0.9-2.4)	3.0%	(2.2-4.0)	1.99	(1.12 - 3.54)	0.019	1239, 1188	968, 1740	
Prevalent chlamydia infection <sup>a</sup>	3.1%	(1.8-5.2)	3.2%	(2.2-4.6)	1.04	(0.53 - 2.01)	0.92	348, 366	513, 821	
Men										
Ever diagnosed with chlamydia	1.3%	(0.6-2.5)	5.3%	(4.1-6.7)	4.40	(2.11 - 9.16)	<0.001	1224, 985	1003, 1375	
Diagnosed with chlamydia in the last year	0.5%	(0.2-1.3)	2.0%	(1.3-3.0)	4.08	(1.40 - 11.84)	0.010	1222, 985	1003, 1375	
Prevalent chlamydia infection <sup>a</sup>	2.9%	(1.3-6.3)	2.6%	(1.7-4.0)	0.91	(0.36 - 2.27)	0.83	357, 282	533, 690	

<sup>a</sup>18 to 24 year-olds

## 6.4.2.1 Correction for measurement error arising from imperfect diagnostic tests

The difference between the 'test-adjusted' (i.e. adjusted for assay used) and the 'detection strategy-adjusted' (i.e. adjusted for assay and specimen collection device) prevalence estimates was negligible (<0.1% difference among men; 0.1% difference among women). This is to be expected, as nucleic acid amplification tests (NAATs) can detect chlamydia at very low organism loads,<sup>259,260</sup> thus the additional urine concentration provided by the FirstBurst device makes little difference to the results achieved using NAATs.<sup>233</sup> Given this negligible difference, the 'detection strategy-adjusted' estimates are presented for completeness.

The point estimate of the 'detection strategy-adjusted' prevalence was slightly lower than the unadjusted prevalence estimate in Natsal-2 among both women and men aged 18 to 24 (Figure 6-5). This might seem initially counterintuitive, as there is often a tendency to focus on imperfect sensitivity of tests and resultant false-negatives, thus creating an expectation that the lower sensitivity of LCx compared to AC2 would have underestimated chlamydia prevalence in Natsal-2. However, at such low levels of prevalence, small differences in the specificity dominate the direction of the measurement error.

After adjustment for differences in the detection strategy, the prevalence in Natsal-3 was non-significantly higher than the adjusted prevalence in Natsal-2 among both men (2.6% vs 2.2%, OR 1.12, 0.38-3.28) and women (3.2% vs 2.5%, OR 1.35, 0.69-2.68). Thus, all else being equal, if Natsal-2 tests had been carried out using the same detection strategy as in Natsal-3 the estimated prevalence in Natsal-3 would likely have been higher than that of Natsal-2, but this would not have amounted to a statistically significant difference.

# Figure 6-5: Odds ratio of chlamydia prevalence in Natsal-3, compared to unadjusted and 'detection strategy adjusted' prevalence in Natsal-2 (sexually-experienced 18-24 year-old women and men)

The solid markers and red dashed lines, show the unadjusted OR for prevalence in Natsal-3 compared to Natsal-2. Each open marker shows the OR of the unadjusted Natsal-3 prevalence compared to the 'detection strategy adjusted' prevalence in Natsal-2.



#### 6.4.2.2 Combining urine infections with recent screen-detected diagnoses

In Natsal-3, 0.9% of men and 1.6% of women aged 18 to 24 with a valid chlamydia test result had a 'screen-detected' diagnosis (i.e. reported a nonclinically indicated reason for their last test) in the last year. Thus in Natsal-3, 4.4% of women and 3.5% of men aged 18 to 24 years had a prevalent chlamydia infection and/or a screen-detected diagnosis in the last year. The proportion of Natsal-3 participants who had either a screen-detected diagnosis or a prevalent infection was non-statistically significantly higher than the proportion with a prevalent infection in Natsal-2 among both women (4.4% versus 3.1%, OR 1.45 95%CI 0.75-2.80) and men (3.5% versus 2.9%, OR 1.21 95%CI 0.49-2.97 ) (Figure 6-6). Thus, all else being equal, and assuming a) that in the absence of screening all these 'screen-detected' infections would have persisted to the point of the Natsal-3 interview and b) that there were no screendetected infections in Natsal-2, there would have been a greater increase in prevalence between surveys than was observed. This would not have been detectable as a significant difference.

# Figure 6-6: Hypothetical scenario, showing odds ratio of chlamydia prevalence plus 'screen-detected' diagnosis in Natsal-3, compared to unadjusted prevalence in Natsal-2 (sexually experienced 18-24 year-old women and men).

The solid markers and red dashed lines, show the unadjusted OR for prevalence in Natsal-3 compared to Natsal-2. Each open marker shows the OR of the 'screen-detected diagnoses'-adjusted prevalence in Natsal-3 compared to the unadjusted Natsal-2 prevalence. **Women aged 18-24** 





Prevalence lower in Natsal-3  $\leftarrow$   $\rightarrow$  Prevalence higher in Natsal-3

Prevalence lower in Natsal-3  $\leftarrow$   $\rightarrow$  Prevalence higher in Natsal-3

## 6.4.2.3 Adjustment for hypothetical unmeasured differences between participants and non-participants

Figure 6-7 shows the 'participation-adjusted' OR across the range of

hypothetical combinations of participation bias in Natsal-2 and Natsal-3

specified in section 6.3.3.5 for 18 to 24 year-old women and men. ORs

presented in red boxes are those where the comparison of 'participationadjusted' prevalence estimates (Table 6-6) resulted in an OR smaller than that using the unadjusted prevalence estimates; those in blue are where the participation-adjusted OR are greater than the unadjusted OR and those which are unshaded indicate where the participation-adjusted and unadjusted OR where equivalent (to 2 decimal places). Under the applied assumptions, the 'participation-adjusted' OR of prevalent infection in Natsal-3 compared to Natsal-2 ranged from 0.68 to 1.58 for women and 0.59 to 1.37 for men.

#### Figure 6-7 Hypothetical odds ratio of prevalent infection in Natsal-3 versus Natsal-2 under different assumptions of unmeasured participation bias (sexuallyexperienced 18-24 year-olds)

Figures show hypothetical, 'participation-adjusted' odds ratios of prevalent infection in Natsal-3 versus Natsal-2 using a weighted prevalence under different assumptions of the relative risk of having a prevalent infection in participants versus non-participants in each survey (see Table 6-6). The colours of the boxes show whether the participation-adjusted' OR is smaller (red boxes), larger (blue boxes) or the same (unshaded boxes) as the OR estimated using the unadjusted prevalence estimates.



non-participants Natsal-2

RR of prevalent infection, participants - v non-participants Natsal-2

#### Key:



OR of 'participation-adjusted prevalence is smaller than that using unadjusted prevalence (i.e. OR of difference between surveys would have been <u>under</u>estimated)



OR of 'participation-adjusted prevalence is equivalent to that using unadjusted prevalence (i.e. OR of difference between surveys would not have been affected.)

1.00

OR of 'participation-adjusted prevalence is larger than unadjusted prevalence (i.e. OR of difference between surveys would have been <u>over</u>estimated)

# Table 6-6: Hypothetical effect of unmeasured participation biases on estimates of difference in prevalence in Natsal-2 and Natsal-3 (sexually-experienced 18-24 year-olds)

	Natsal-2		Natsal-3					
(a) Relative risk of prevalent infection, participants v non- participants	(b) Hypothetical prevalence in non-participants	(c)Hypothetical 'participation- adjusted' prevalence	(d Relative risk of prevalent infection, participants v non- participants	(e) Hypothetical prevalence in non- participants	(f) Hypothetical 'participation- adjusted' prevalence			
0.5	6.2%	4.2%	0.5	6.4%	4.3%			
0.5	6.2%	4.2%	0.75	4.3%	3.6%			
0.5	6.2%	4.2%	1	3.2%	3.2%			
0.5	6.2%	4.2%	1.25	2.6%	3.0%			
0.5	6.2%	4.2%	1.5	2.1%	2.9%			
0.75	4.1%	3.5%	0.5	6.4%	4.3%			
0.75	4.1%	3.5%	0.75	4.3%	3.6%			
0.75	4.1%	3.5%	1	3.2%	3.2%			
0.75	4.1%	3.5%	1.25	2.6%	3.0%			
0.75	4.1%	3.5%	1.5	2.1%	2.9%			
1	3.1%	3.1%	0.5	6.4%	4.3%			
1	3.1%	3.1%	0.75	4.3%	3.6%			
1	3.1%	3.1%	1	3.2%	3.2%			
1	3.1%	3.1%	1.25	2.6%	3.0%			
1	3.1%	3.1%	1.5	2.1%	2.9%			
1.25	2.5%	2.9%	0.5	6.4%	4.3%			
1.25	2.5%	2.9%	0.75	4.3%	3.6%			
1.25	2.5%	2.9%	1	3.2%	3.2%			
1.25	2.5%	2.9%	1.25	2.6%	3.0%			
1.25	2.5%	2.9%	1.5	2.1%	2.9%			
1.5	2.1%	2.7%	0.5	6.4%	4.3%			
1.5	2.1%	2.7%	0.75	4.3%	3.6%			
1.5	2.1%	2.7%	1	3.2%	3.2%			
1.5	2.1%	2.7%	1.25	2.6%	3.0%			
1.5	2.1%	2.7%	1.5	2.1%	2.9%			

Notes on calculations

(b) Measured prevalence in participants (3.1%) x (a)

(c) Weighted prevalence of (b) and 3.1% using response rate (65.4%)<sup>171</sup>

(e) Measured prevalence in participants (3.2%) x (d)

(f) Weighted prevalence of (b) and 3.2% using response rate (67.3%)<sup>5</sup>

Figure 6-7 shows that where the extent and direction of the hypothetical participation bias was similar in each year, even though each estimate of prevalence may itself be biased, the impact on comparisons between the years was minimal. For example, if in both Natsal-2 and Natsal-3, the relative risk of

having a prevalent infection in participants v non-participants in 18 to 24 yearold women was 1.25 in each year, the 'participation-adjusted' prevalence would have been 2.9% in Natsal-2, compared to the 3.1% that was actually estimated in the survey; in Natsal-3, the 'participation-adjusted' prevalence would have been 3.0%, compared to the 3.2% actually measured. Although in this scenario the actual estimates in both Natsal-2 and Natsal-3 would have been biased towards a higher prevalence, the resulting ORs of testing positive in Natsal-3 versus Natsal-2 are equivalent (to 2 decimal places), being 1.04 for both the 'participation-adjusted' OR and the unadjusted OR.

Figure 6-7 also shows that the potential for comparisons between surveys to be affected was greater where the direction of the hypothetical bias was the same, but the extent was different. Again taking the example of 18 to 24 year-old women, if the relative risk of having a prevalent infection in participants v non-participants was 1.25 in Natsal-2 but 1.5 in Natsal-3, the 'participation-adjusted' OR would be 0.99, compared to the unadjusted OR of 1.04. The most extreme impact on comparisons between surveys arose from where the direction of bias was assumed to be different in each survey, and the difference between participants and non-participants in each year was large.

#### 6.4.2.4 Indicators of change in sexual behavioural risk of transmission

Reported numbers of sexual partners, non-condom use and prevalence of nonvaccine HPV types (see section 6.3.3.6), tended to be similar or slightly higher for women and lower for men in Natsal-3 compared to Natsal-2 (Figure 6-8), although none of the observed differences were statistically significant. In both women and men, the proportion reporting non-use of condoms at last sex was very similar between the surveys. However, after stratifying by partnership type, the percentage who reported not having used a condom at last sex increased between surveys from 43% to 56% in women and 34% to 38% in men whose last sex was with a non-steady partner. This was not a statistically significant difference.

There was no change in the median age at first intercourse among sexuallyexperienced 18 to 24 year-olds between Natsal-3 and Natsal-2 (16 years among women and men). However, among Natsal-3 participants, the proportion who reported having had sex by the age of 16 increased with decreasing age. The proportion of women who reported having sex by age 16 increased by an average of 3% per year (relative to that measured in the previous year) from 2000 to 2012 in both women (prevalence ratio 1.03, 1.02-1.05, p<0.001) and men (prevalence ratio 1.03, 1.01-1.06, p=0.003) (Figure 6-9).

### Figure 6-8: Prevalence of non-vaccine HPV and selected sexual behaviours among sexually experienced 18-24 year-olds in Natsal-2 and Natsal-3

Error bars show 95% confidence intervals on the prevalence of non-vaccine HPV or reported sexual behaviour. OR: Unadjusted odds ratio (95%CI shown in parentheses) for Natsal-3 versus Natsal-2, derived using logistic regression. Reference groups are those without the reported characteristic.

#### Women



### Figure 6-9: Proportion reporting having had heterosexual sex before the age of 16 by year turned 16 (Natsal-3)

The shaded area shows 95% confidence intervals. Participants who report having had sex before the age of 13 are excluded from the denominator.



#### 6.5 Discussion

#### 6.5.1 Key findings

There was no significant difference in chlamydia prevalence, as measured in the second and third Natsal surveys, in 2010-12 compared to that in 1999-2001. Adjusting for the different tests used in each survey did not alter the conclusions that can be made about the difference in chlamydia prevalence.

#### 6.5.2 Strengths and limitations

The main strength of this analyses was that it used data from two probability surveys with high response rates, with the two surveys having been carried out before (Natsal-2) and after (Natsal-3) the implementation of the NCSP. The sample size available was the main limitation for the comparisons of prevalence for each year. As a result, the Natsal surveys were consistent with a wide range of scenarios, from at least a halving of the odds of infection to more than doubling of the odds of infection between surveys (unadjusted 95%Cls ranged from at least 0.5 to 2 in both women and men). None of the adjusted estimates (i.e. adjusted for detection strategy, for screen-detected diagnoses or for hypothetical unmeasured participation bias) approached statistical significance, although the direction of the adjustments was informative. This analysis showed that in some scenarios, unmeasured participation bias may affect comparisons between surveys. However even after adjustment for a wide range of scenarios, the point estimates for prevalence differences lay within the 95% confidence intervals for odds ratios based on the unadjusted Natsal estimates, again emphasising the limitations arising from the available sample size. One alternative to investigating change in chlamydia prevalence relative to HPV prevalence and sexual behaviours would have been to include these variables as confounders in a regression model. The small absolute numbers of prevalent infections meant results from such an analysis would be subject to considerable error, so this was not pursued.

Adjustment for the differences in the detection strategy used in each survey was limited by the data available on the sensitivity and specificity of the LCx and the AC2 and the urine cup compared to the FirstBurst urine collection device. As the adjusted prevalence used the AC2 as the gold standard, this may be a biased estimator of the 'true' prevalence of infection in the population if the AC2 is not a perfect test. An alternative approach would have been to also adjust the Natsal-3 prevalence for AC2 performance characteristics. However this approach would introduce substantial uncertainty, as measuring the sensitivity

and specificity of diagnostic tests is highly dependent on the gold standard used, and there is no agreed reference standard for chlamydia NAATs.<sup>244</sup> The use of the point estimate of the sensitivity and specificity LCx versus AC2 is also an oversimplification, which did not incorporate the error within the sensitivity and specificity measurement. However as there was no significant difference when using these adjusted estimates, even wider confidence intervals would not alter the interpretation. There is only one published study which compares the FirstBurst to the urine cup. This was limited to samples from men and used a different NAAT (Amplicor CTY/NG PCR)<sup>233</sup>. However additional studies are unlikely to have affected these findings as low prevalences are more subject to error arising from imperfect specificity rather than sensitivity, and only marginal additional benefit would be expected from increased organism load when measuring chlamydia prevalence among the general population.

In comparing the prevalence plus 'screen-detected' diagnoses, it was assumed that all diagnoses in Natsal-2 had come about as a result of clinically-indicated testing. As no data were collected on reason for test, this is a simplification and it is feasible that a proportion of diagnoses may have come about as a result of asymptomatic screening activity.

The analysis did not account for all of the differences between the surveys. Degradation of specimens during transit (and storage, for HPV) may also have led to prevalence having been underestimated in Natsal-2 and Natsal-3. This was not investigated in detail, as quantitative estimates of the loss of precision were not available. This was acknowledged at the time of the Natsal-2 survey as a potential cause of underestimating prevalence.<sup>133,234</sup> Degradation is also

feasible in Natsal-3, although all specimens were transported in accordance with the manufacturer's instructions, and AC2 specimens are considered to be stable at room temperature for several months. Given the large confidence intervals on the unadjusted and adjusted estimates of prevalence difference, this potential difference is again unlikely to materially affect conclusions which can be made based on the Natsal surveys.

Prevalent chlamydia infection was measured in urine samples. This may have missed some infections in women among whom vulvovaginal swabs demonstrate marginally higher sensitivity.<sup>261</sup> Urine sampling will also have missed rectal infections, leading to underestimation of the total currently infected with chlamydia. However, the impact on our findings is likely minimal as men who have sex with men made up a small proportion of this sample of Natsal participants.<sup>235</sup>

#### 6.5.3 Implications for evaluation of chlamydia control

Although comparisons between the surveys had limited power due to the sample size available, there was no evidence to support there having been a decrease in prevalence between the surveys. However, there was some indication that risk behaviours relating to transmission risk may have increased among young women, but not in men. Additionally, the analysis that combined 'screen-detected' diagnoses with prevalent infections in Natsal-3 suggested that if all screen-detected infections in Natsal-3 had persisted to the time of the interview, then the prevalence in Natsal-3 would have been higher. Thus, in the absence of widespread chlamydia screening, chlamydia incidence and prevalence might have been expected to increase. However this remains an unproven hypothesis, as there is no evidence to determine how many, if any of

the 'screen-detected' infections would have persisted to the point of the interview and an increase in risk behaviours was not seen in all groups, or consistently across several behaviours, and there was no increase seen in nonvaccine high risk HPV types among women.

#### 6.5.4 Summary

In summary, there is no definitive evidence from the Natsal surveys to suggest that chlamydia prevalence has changed in either absolute terms or relative to indicators of sexual behavioural risk during a period of increased chlamydia testing and screening. Returning to the notion of process monitoring described in Chapter 2, it is important to understand how chlamydia screening has been delivered to consider whether we would have expected prevalence to have decreased over this period. In the next chapter, I use data from Natsal-3 to investigate the implementation of chlamydia screening in relation to risk of infection in more detail.

#### 7 Chlamydia prevalence, diagnosis and testing among 16 to 24 year-old participants in the third National Survey of Sexual Attitudes and Lifestyles (Natsal-3)

In the previous chapter I compared the prevalence of chlamydia estimated in the second and third Natsal surveys, which were conducted before and after the implementation of the NCSP, and found no definitive evidence to suggest that chlamydia prevalence had decreased in the ten years between the surveys. In this chapter I use Natsal-3 data to further our understanding of how chlamydia screening had been delivered in practice up to 2012.

#### 7.1 Background

As Natsal-3 measured prevalent infection and also collected data on selfreported chlamydia testing and diagnoses, the survey presents a unique opportunity to explore factors associated with testing and infection within the same population, three to five years after the NCSP had been nationallyimplemented. Sonnenberg *et al* have previously reported an overview of STI prevalence and service use using data from Natsal-3, including chlamydia prevalence and self-reported testing in the last year among 16 to 24 year-olds.<sup>23</sup> Although Sonnenberg *et al* reported prevalence by age group, factors associated with prevalent infection were assessed among all 16 to 44 year-olds and only a limited number of factors associated with chlamydia prevalence and testing were explored (age group, area-level deprivation, number of sexual partners in the last year, sexual partners in the last year without a condom [investigated for prevalence only], age at first sex and any same-sex experience). In this chapter, a detailed analysis of 16 to 24 year-olds in Britain (as the age group targeted by the NCSP) is presented. Factors associated with prevalent chlamydia infection, previous chlamydia diagnosis and chlamydia testing are described and compared to assess the extent to which, up to 2012, opportunistic chlamydia screening was reaching 16 to 24 year-olds at risk of chlamydia.

#### 7.2 Aims & objectives

To investigate whether chlamydia screening patterns indicate that chlamydia screening has been delivered in such a way that we would expect it to effect a reduction in the incidence or prevalence of infection. This aim was addressed through the following objectives:

- Using data from Natsal-3, to describe self-reported chlamydia testing, self-reported diagnosis and prevalent infection (detected in urine) among 16 to 24 year-olds living in Britain in 2010-12 by sociodemographic, clinical and behavioural variables.
- To compare factors associated with self-reported chlamydia testing, selfreported diagnosis and prevalent infection (detected in urine) among 16 to 24 year-olds in Natsal-3.

#### 7.3 Methods

#### 7.3.1 Participants

A flow chart of participants included in this analysis is presented in Figure 7-1. Analyses of recent testing (self-reported testing in the last year) and recent diagnosis (self-reported diagnosis in the last year) were based on sexuallyexperienced 16 to 24 year-olds (n=3,115). Analyses of prevalent infection (*C. trachomatis* detected in urine) were among those who provided a urine sample for STI testing and for whom a valid chlamydia test result is available (n=1,832,

62 of whom had a prevalent chlamydia infection detected in urine).

# Figure 7-1: Flow chart showing participants included in analyses of self-reported chlamydia testing, self-reported diagnosis and prevalent infection detected in urine (16 to 24 year-old Natsal-3 participants)



#### Key


#### 7.3.2 Statistical analysis

Analyses were carried out using Stata 12.1, accounting for the weighting, clustering and stratification of the Natsal data<sup>23,172,235</sup> as set out in the previous chapter (section 6.3.3).

Factors associated with prevalent infection, recent diagnosis and recent testing were investigated using univariable and multivariable logistic regression, for women and men separately. While the overall percentage of 16 to 24 year-olds who reported having been diagnosed with chlamydia (ever or in the last year) is reported among the sexually-experienced population, risk factors for recent diagnosis were investigated among those with a recent test so that results represent associations with infection at the time of test and not with testing per se. Sociodemographic and behavioural factors reported to be associated with STI risk were included as predictor variables.<sup>133,262,263</sup> Associations with deprivation were explored using both residence-based (quintile of IMD for the LSOA of residence) and individual-based (age left school) measures. Sexual behaviours investigated included numbers of sexual partners in the last year (total, new, without a condom), number of lifetime sexual partners and condom use at last sex. Frequency of binge drinking (>6/8 units on one occasion in women/men) was included as a proxy for sexual risk behaviour that may not be captured in reported numbers of sexual partners.

Two approaches to multivariable modelling were explored; maximal and parsimonious models were constructed for all outcomes of interest. To fit parsimonious models, variables found significant at p<0.1 in univariable analyses were entered into a multivariable model selection process based on backwards selection whereby variables were removed from the model if the

181

statistical significance of their association with the outcome had an associated p value of >0.05. Only those variables significantly associated with the outcome were retained in the final parsimonious models. For maximal models, all variables included in univariable models were included with two exceptions: number of sexual partners in the last year was not included due to collinearity with other sexual partnership variables; age left school was not included as data were unavailable for 16 year-olds. The results from the two models were largely comparable (see Appendix 2 for a detailed comparison). Results from the maximal models are therefore presented in the results section below to provide odds ratios (OR) adjusted for potential confounders.

In order to explore how chlamydia infections were distributed across different risk groups, the proportion reporting different demographic and behavioural factors was calculated among a) individuals with a prevalent infection, b) individuals with a recent diagnosis and c) the sexually-experienced population.

Sub-group analyses by country (England, Scotland and Wales) were performed when the difference in screening practice between countries was of potential relevance.

#### 7.4 Results

#### 7.4.1.1 Reason and location for last test (Natsal-3)

Among 16 to 24 year-olds who reported recently testing, fewer than 10% reported a clinical indication (symptoms; a partner with chlamydia/symptoms; check-up after a previous diagnosis) for their last test (Table 7-1). Around threequarters of women and half of men had last been tested in either a sexual health clinic, GP surgery, or a family planning clinic. When limiting to women and men recently diagnosed, 29.0% had last been tested due to having symptoms and 20.9% due to having a partner diagnosed with chlamydia or with symptoms (Table 7-1). Almost all of (95.4%) of those recently diagnosed had been most recently tested in a sexual health clinic, family planning clinic, or GP surgery. Whereas 17% of all recent tests had been carried out in educational settings, only 1.7% of those recently diagnosed reported their last test in this setting.

## Table 7-1: Reason and location of most recent chlamydia test, among those tested for chlamydia in the last year, by whether diagnosed in last year (16-24 year-old sexually-experienced women and men)

		Ву	sex		By whe	ether diagr yea	nosed in ar <sup>a</sup>	the last
	Wo	men	N	len	Diagnose y	d in the last ear	Not diag the la	jnosed in st year
	Percent	95%CI	Percent	95%CI	Percent	95%CI	Percent	95%CI
Denominator (W,UW)	523	, 943	347	, 475	48	, 81	816,	1330
Reason for most recent	test							
Had symptoms 4.2		(3.0-5.8)	4.2%	(2.7-6.5)	29.0% (	19.0-41.5)	2.7%	(1.9-3.8)
Partner diagnosed with chlamydia or had symptoms	2.8% (1.7-4.5)		3.8%	(2.4-6.1)	20.9% (	12.8-32.2)	2.2%	(1.4-3.4)
Check up after a previous positive	1.3%	(0.63-2.6)	0.95%	(0.33-2.7)	8.6% (	3.2-21.1)	0.7%	(0.4-1.4
Wanted a check-up / offered a test / worried about risk	84.9%	(82.1-87.4)	87.3%	(83.8-90.1)	37.2% (	26.2-49.86)	88.7%	(86.8-90.4)
Other	6.8%	(5.3-8.7)	3.7%	(2.3-6.0)	4.3% (	1.5-12.0)	5.7%	(4.5-7.1)
Location of most recent	chlamydi	a test						
Sexual health clinic	28.9%	(25.5-32.6)	30.5%	(25.9-35.5)	62.9% (	50.4-73.9)	27.6%	(25.0-30.4)
GP surgery	35.1%	(31.7-38.6)	17.0%	(13.6-20.9)	27.1% (	17.7-39.1)	28.0%	(25.3-30.8)
NHS FP clinic	9.2%	(7.4-11.4)	4.3%	(2.7-6.8)	5.4% (	1.6-16.3)	7.3%	(6.0-8.9)
School, college or university	11.6%	(9.4-14.2)	24.5%	(20.4-29.1)	1.7% (	0.4-7.2)	17.5%	(15.2-20.1)
Elsewhere	where 15.2% (12.9-17.8)			(19.3-28.9)	2.9% (	1.0-8.1)	19.6%	(17.2-22.2)

*GP:* General practice; *FP:* Family planning. <sup>a</sup>Women and men were combined due to small denominator for diagnosed in the last year.

#### 7.4.1.2 Factors associated with infection, diagnosis and testing

Table 7-2 explores associations between sociodemographic and behavioural variables and prevalent infection, recent testing and recent diagnosis. In univariable analyses, higher numbers of sexual partners (total/new/without a condom) in the last year were significantly (p<0.05) associated with prevalent infection among women and men. In women, area-level deprivation and frequency of binge drinking were also associated with prevalent infection. Among men, number of lifetime sexual partners, age group, age left school, age at first sex, and condom non-use at last sex were significantly associated with prevalent infection. Similar factors were associated with recent diagnosis among those tested. In multivariable analyses, living in more deprived areas and more frequent binge drinking remained significantly associated with having a prevalent infection in women (AOR 4.23, 95%CI 1.53-11.6, p=0.01; AOR 2.51, 95%CI 1.08-5.76, p=0.01 respectively). Being aged 20 to 24 years (AOR 7.54, 95%CI 1.37-41.3, p=0.02), living in more deprived areas (AOR 3.75, 95%CI 1.11-12.5, p=0.04) and higher numbers of lifetime sexual partners (>10 versus 1-4, AOR 8.69, 1.21-62.0, p=0.03) remained significantly associated with prevalent infection in men.

Table 7-2a: Percentage, unadjusted and adjusted odds (OR/AOR) ratios for prevalent chlamydia infection, self-reported diagnosis in the last year and self-reported testing by sociodemographic and behavioural factors (16-24 year-old sexually-experienced <u>women</u>, Natsal-3)

Prevalent chlamydia infection detected in urine (n=992)					Di (am	iagnosed nong tho	l with chla se tested i	nydia ir 1 the las	n the la st year)	st year ) (n=940)		Tes	sted for	chlamydia (n=1,736	in the 6)	last y	vear		Der (weighte	nominator d, unweighted	l) <sup>a</sup>				
	%	(95%CI)	OR	(95%CI)	р	$AOR^{b}$	(95%CI)	р	% (95%	6CI) O	R (95%C	I) p	AOR <sup>b</sup>	(95%CI)	р	% (95%Cl	) OR	(95%CI)	р	AOR	(95%CI)	р	Infection D	iagnosis Teste	ed
Age group 16-19 20-24	3.8% 2.7%	(2.2-6.3) (1.7-4.3)	1.00 0.71	- (0.35-1.46	0.36 ;)	1.00 0.71	- (0.27-1.87	0.50 )	6.0%(3.8-9 5.1%(3.4-7	9.2) 1 7.6) 0	00 - 86 (0.46-1.	0.62 60)	1.00 0.80	- (0.35-1.78)	0.58	56.6%(52.5-60.6 52.8%(49.2-56.4	) 1.00 ) 0.86	) 5 (0.69-1.06	0.16 )	1.00 0.82	- (0.62-1.07)	0.15	214, 395 383, 597	193, 375   343, 329, 565 623, 1	672 064
England Scotland Wales	2.9% 3.1% 5.3%	(2.0-4.3) (1.1-8.6) (1.9-13.8)	1.00 1.08 1.87	- (0.35-3.33 (0.63-5.60	0.53 6) 9)	1.00 1.34 1.88	- (0.43-4.14 (0.63-5.54	0.48 ) )								57.1%(54.1-60.1 32.4%(24.4-41.5 45.6%(36.2-55.4	) 1.00 ) 0.36 ) 0.63	) 5 (0.24-0.54 3 (0.42-0.94	<b>&lt;0.01</b> ) )	1.00 0.29 0.53	- (0.18-0.45) (0.32-0.85)	<0.01	504, 817 56, 103 37, 72	469, 832 823, 1 30, 58 91, 24, 50 52,	452 178 106
IMD quintile of L 2 least deprived Middle quintile 2 most deprived	SOA c 1.3% 1.8% 4.9%	f residend (0.5-3.4) (0.8-4.2) (3.3-7.3)	2 <b>e<sup>d</sup></b> 1.00 1.37 3.82	- (0.38-4.90 (1.35-10.8	<b>0.01</b>	1.00 1.40 4.23	- (0.39-4.98 (1.53-11.6	<b>0.01</b>	4.8%(2.8-8 3.5%(1.6-7 6.8%(4.6-1	8.1) 1 7.3) 0 0.0) 1	00 - 71 (0.27-1. 46 (0.73-2.	0.23 36) 93)	1.00 1.06 1.70	- (0.37-3.04) (0.73-3.91)	0.36	54.2%(49.5-58.8 54.4%(48.0-60.7 54.0%(49.8-58.2	) 1.00 ) 1.01 ) 0.99	) (0.74-1.38 ) (0.77-1.27	0.99 ) )	1.00 1.03 0.97	- (0.71-1.48) (0.73-1.29)	0.94	213, 355 111, 174 273, 463	183, 319   338, 102, 176   189, 236, 445   439,	595 324 817
Age left school <sup>e</sup> 17+ 16	3.2% 3.4%	(2.1-4.8) (1.9-6.0)	1.00 1.06	- (0.50-2.22	0.88 !)				5.2%(3.6-7 6.2%(3.7-1	7.5) 1 0.5) 1	00 - 21 (0.62-2.	0.58 37)				54.3%(51.0-57.6 55.5%(50.1-60.8	) 1.00 ) 1.05	) - 5 (0.82-1.35	0.70 )				445, 700 : 120, 229	387, 658 715, 1 109, 228 196,	217 405
Frequency of bir Never / <monthly Monthly &gt;weekly</monthly 	nge dr 2.5% 1.2% 7.9%	nking (1.5-4.1) (0.5-3.3) (4.7-13.1)	1.00 0.49 3.35	- (0.16-1.52 (1.55-7.25	<b>&lt;0.01</b> !) i)	1.00 0.46 2.51	- (0.13-1.51 (1.08-5.76	<b>0.01</b> )	3.7%(2.4-5 6.6%(3.6-1 9.4%(5.2-1	5.7) 1 1.8) 1 6.3) 2	00 - 85 (0.85-4. 69 (1.25-5.	<b>0.03</b> 01) 30)	1.00 1.91 2.06	- (0.74-4.89) (0.75-5.61)	0.28	52.1%(48.7-55.5 52.4%(46.6-58.2 64.4%(57.6-70.7	) 1.00 ) 1.01 ) 1.66	) (0.78-1.32 ) (1.22-2.27	<b>0.01</b> )	1.00 0.75 1.16	- (0.54-1.02) (0.81-1.64)	0.08	373, 598 130, 226 95, 168	312, 561 601, 1 112, 200 214, 97, 177 151,	085 375 274
Age at first heter 17+ 16 <16	osexu 1.6% 3.9% 4.0%	al sex (0.7-3.7) (2.1-6.9) (2.4-6.6)	1.00 2.52 2.65	- (0.86-7.36 (0.95-7.36	0.15 5) 5)	1.00 2.20 1.82	- (0.67-7.17 (0.60-5.42	0.42 )	4.1%(1.8-8 5.0%(2.8-8 6.9%(4.8-9	8.9) 1 8.9) 1 9.9) 1	00 - 24 (0.44-3. 76 (0.70-4.	0.37 55) 41)	1.00 0.80 0.89	- (0.26-2.42) (0.35-2.25)	0.93	43.9%(38.7-49.1 56.4%(51.6-61.1 63.9%(59.8-67.9	) 1.00 ) 1.66 ) 2.27	) 6 (1.25-2.20 7 (1.75-2.94	<b>&lt;0.01</b> ) )	1.00 1.39 1.44	- (0.99-1.92) (1.05-1.97)	0.05	188, 246 178, 304 213, 415	137, 215 313, 154, 272 273, 220, 429 344,	489 503 680
Condom used fo Yes No	or mos 2.8% 3.6%	t recent so (1.5-5.3) (2.4-5.4)	<b>ex with</b> 1.00 1.28	<b>most rec</b> - (0.59-2.80	ent par 0.53	t <b>ner</b> 1.00 1.59	- (0.67-3.74	0.29 •)	3.4%(1.6-7 6.3%(4.4-8	7.0) 1 8.9) 1	00 - 90 (0.81-4.	0.14 <b>19</b> )	1.00 1.88	- (0.81-4.32)	0.14	51.6%(46.9-56.3 58.0%(54.6-61.3	) 1.00 ) 1.30	) ) (1.03-1.63	<b>0.03</b>	1.00 1.01	- (0.74-1.35)	0.97	202, 314 352, 613	170, 303   330, 328, 592 567, 1	586 027
Number of sexua 0 or 1 2 3 to 4 5+	al part 2.5% 3.9% 1.9% 8.3%	ners in the (1.5-4.0) (1.8-8.5) (0.7-5.1) (3.9-16.8)	a last y 1.00 1.62 0.75 3.57	ear - (0.63-4.15 (0.24-2.33 (1.39-9.17	<b>0.03</b>				3.0%(1.7-5 6.6%(3.7-1 8.8%(5.2-1 11.6%(6.2-2	5.4) 1 1.6) 2 4.6) 3 20.8) 4	00 - 25 (0.98-5. 08 (1.34-7. 19 (1.69-10	<b>0.01</b> 18) 05) .4)				46.6%(43.3-50.0 65.2%(58.8-71.1 69.7%(62.0-76.4 74.8%(64.4-83.0	) 1.00 ) 2.15 ) 2.63 ) 3.40	) 5 (1.59-2.89 6 (1.82-3.80 9 (2.05-5.65	<0.01 ) )				387, 600 2 90, 161 63, 127 49, 93	291, 507 624, 1 93, 178 143, 76, 146 111, 58, 101 77,	096 275 210 135

#### Table 7-2a continued.

Number of new s	sexual partners in	n the la	ist year																		
0	2.2% (1.3-3.7)	1.00	-	0.03	1.00	-	0.70	2.8%(1.5-5.0)	1.00 -	<0.01	1.00-	0.11	45.6%(41.9-49.3)	1.00	-	<0.01	1.00 -	<0.01	313, 495 22	26, 397 4	495, 873
1	2.8% (1.2-6.3)	1.26	0.47-3.41	)	1.17 (0	0.38-3.52	)	4.8%(2.5-9.1)	1.76 (0.71-4	.38)	1.89(0.69-5.16)		59.2%(54.0-64.2)	1.73 (1	34-2.23)	)	1.69 (1.25-2.27)		160, 263 1	56, 287 2	264, 485
2+	5.9% (3.5-9.8)	2.73	(1.26-5.93	)	1.65 (0	0.50-5.39)	)	10.7%(7.3-15.6)	4.23 (2.04-8	.79)	3.09(1.07-8.86)		70.0%(63.8-75.6)	2.79 (2	.03-3.84)	)	1.46 (0.95-2.21)		118, 225 13	37, 249 1	197, 359
Number of sexua	al partners in the	last ye	ear withou	it a cond	lom																
0	2.9% (1.2-7.1)	1.00	-	0.03	1.00	-	0.18	4.2%(1.5-11.6)	1.00 -	<0.01	1.00-	0.84	36.4%(30.8-42.3)	1.00	-	<0.01		0.05	120, 173	76,130 2	210, 361
1	2.2% (1.4-3.6)	0.76	0.26-2.17	)	0.34 (0	0.10-1.10	)	3.8%(2.4-6.1)	0.90 (0.27-2	.94)	0.72(0.15-3.21)		54.7%(51.4-58.0)	2.11 (1	61-2.78)	)	1.52 (1.03-2.24)		368, 606 3 <sup>.</sup>	19, 567 58	35, 1049
2+	6.3% (3.5-11.2)	2.25	(0.73-6.93	)	0.49 (0	0.12-1.83)	)	10.3%(7.0-15.0)	2.60 (0.80-8	.46)	0.90(0.18-4.49)		74.2%(68.3-79.4)	5.04 (3	47-7.32)	)	1.86 (1.09-3.15)		108, 212 12	26, 241 1	169, 322
Concurrent partr	nerships in last y	earf																			
No	2.7% (1.8-4.2)	1.00	-	0.19	1.00	-	0.85	4.3%(2.9-6.4)	1.00 -	0.13	1.00-	0.46	51.0%(47.8-54.2)	1.00	-	<0.01	1.00 -	0.05	439, 706 36	61, 63971	10, 1256
Yes	6.3% (2.9-13.4)	2.40	0.94-6.15	)	1.34 (0	0.48-3.70	)	8.1%(4.1-15.3)	1.95 (0.84-4	.54)	0.75(0.27-2.00)		74.6%(67.1-80.8)	2.81 (1	93-4.09)	)	1.46 (0.93-2.28)		66, 134	78, 146 1	105, 196
Unknown	3.1% (1.3-7.2)	1.14	(0.43-3.04	)	1.13 (0	0.38-3.31)	)	7.8%(4.3-13.9)	1.88 (0.90-3	.94)	1.46(0.62-3.38)		64.6%(57.5-71.2)	1.76 (1	27-2.42)	)	1.53 (1.02-2.28)		73, 127	76, 144 1	17, 226
Number of lifetin	ne sexual partner	rs																			
1 to 4	2.4% (1.4-4.2)	1.00	-	0.14	1.00	-	0.67	2.1%(1.1-4.0)	1.00 -	<0.01	1.00-	0.08	43.5%(40.0-47.1)	1.00	-	<0.01	1.00 -	<0.01	321, 482 2 <sup>.</sup>	18, 391 5	503, 894
5 to 9	2.8% (1.4-5.2)	1.15	0.48-2.75	)	0.87 (0	0.32-2.32	)	6.2%(3.7-10.3)	3.07 (1.33-7	.07)	2.40(0.90-6.37)		63.8%(58.5-68.7)	2.29 (1	76-2.96)	)	1.96 (1.43-2.69)		150, 267 16	61,289 2	252, 453
10+	5.4% (2.9-9.7)	2.29	(0.97-5.39	)	1.39 (0	0.45-4.18	)	9.9%(6.5-14.7)	5.12 (2.31-1	1.3)	3.76(1.19-11.8)		69.6%(63.9-74.7)	2.97 (2	22-3.98)	)	2.11 (1.41-3.13)		121, 234 14	41,254 2	202, 373
Ever had any sar	me sex experienc	ce/cont	tact																		
No	3.1% (2.1-4.4)	1.00	-	0.92	1.00	-	0.49	5.3%(3.7-7.5)	1.00 -	0.72	1.00-	0.44	51.9%(48.8-55.0)	1.00	-	<0.01	1.00 -	0.24	473, 750 38	88, 70474	49, 1352
Yes	3.2% (1.4-7.1)	1.05	(0.42-2.63	)	0.74 (0	0.30-1.76	)	5.9%(3.5-10.0)	1.13 (0.57-2	.24)	0.71(0.30-1.68)		61.9%(56.0-67.5)	1.51 (1	15-1.98)	)	1.21 (0.88-1.65)		124, 242 13	34, 236 2	218, 384

Table 7-2b: Proportion, unadjusted and adjusted odds ratios for prevalent chlamydia infection, self-reported diagnosis in the last year and self-reported testing by sociodemographic and behavioural factors (16-24 year-old sexually-experienced <u>men</u>, Natsal-3)

	Prevalent chlamydia infection detected in urine (n=840)					Diagno (among	sed wit	h chlamyd ested in the	lia in 1 e last	the last year) (I	year 1=471)		Teste	d for c	hlamydia (n=1,37	in the 5)	last ye	ear		Der (weighte	nominate d, unwei	or ghted) <sup>a</sup>			
	% (95	%CI)	OR	(95%CI)	р	AOR⁵	(95%CI)	р	% (95%Cl)	OR	(95%CI)	р	AOR⁵	(95%CI)	р	% (95%CI)	OR	(95%CI)	р	AOR⁵	(95%CI)	р	Infection Di	iagnosis	Tested
Age group 16-19 20-24	0.3% (0.1- 3.4% (2.2-	-1.4) 5.2)	1.00 10.6 (	- [2.40-46.3]	<0.01	1.00 7.54	- (1.37-41.3	<b>0.02</b>	4.7% (2.4-9.0) 6.7% (3.9-11.1)	1.00 1.46 (	- (0.59-3.58)	0.41	1.00 0.76 (	- 0.26-2.15)	0.61	40.4%(35.9-45.1) 31.1%(27.8-34.7)	1.00 0.67	- (0.52-0.86	<b>&lt;0.01</b>	1.00 0.53	- (0.37-0.73	<b>&lt;0.01</b> )	234, 343 1 391, 497 1	51, 226 92, 245	374, 582 629, 793
<b>Country</b> <sup>c</sup> England Scotland Wales	1.9% (1.2- 5.7% (2.1- 1.7% (0.2-	·3.0) ·14.3) ·12.1)	1.00 3.13 ( 0.88 (	- (1.04-9.45) (0.11-7.02)	0.12	1.00 3.16 1.20	- (0.78-12.8 (0.18-7.63	0.27 ) )								37.3% (34.3-40.3) 22.2% (14.0-33.5) 12.8% (6.9-22.3)	1.00 0.48 0.25	- (0.27-0.85 (0.13-0.48	<b>&lt;0.01</b> ) )	1.00 0.33 0.19	- (0.16-0.64 (0.08-0.40	<b>&lt;0.01</b> ) )	532, 719 3 60, 72 33, 49	16, 440 20, 22 7, 9	859, 1181 89, 111 55, 83
IMD quintile of L 2 least deprived Middle quintile 2 most deprived	SOA of res 1.3% (0.4- 1.6% (0.6- 3.4% (2.1-	<b>idence</b> ·3.6) ·4.4) ·5.6)	e <sup>d</sup> 1.00 1.24 ( 2.71 (	- (0.26-5.88) (0.83-8.82)	0.14	1.00 1.01 3.75	- (0.15-6.68 (1.11-12.5	<b>0.04</b> ) )	5.2% (2.5-10.5) 5.0% (1.7-13.8) 6.5% (3.6-11.4)	1.00 0.96 ( 1.26 (	- (0.26-3.58) (0.48-3.33)	0.85	1.00 0.68 (i 1.06 (i	- 0.15-2.97) 0.42-2.64)	0.85	34.5% (30.0-39.2) 33.3% (27.4-39.9) 35.2% (31.1-39.5)	1.00 0.95 1.03	- 0.67-1.35 0.78-1.36	0.89 ) )	1.00 1.04 1.13	- (0.70-1.52 (0.82-1.53	0.75 ) )	241, 315 1 114, 164 269, 361 1	27, 180 60, 86 55, 205	369, 509 183, 263 450, 603
<b>Age left school</b> <sup>e</sup> 17+ 16	1.6% (0.9- 5.0% (2.7-	-2.7) -9.2)	1.00 3.28 (	- (1.38-7.82)	0.01				5.3% (3.1-9.0) 7.2% (3.6-13.8)	1.00 1.38 (	- (0.55-3.45)	0.49				33.6% (30.4-37.1) 37.8% (32.3-43.5)	1.00 1.20	- (0.91-1.58	0.21 )				439, 568 2 143, 206 {	33, 304 87, 134	703, 927 230, 334
Frequency of bin Never / <monthly Monthly &gt;weekly</monthly 	nge drinkin / 1.1% (0.4- 3.0% (1.4- 3.8% (2.0-	<b>g</b> ·2.7) ·6.2) ·7.0)	1.00 2.79 ( 3.58 (	- (0.85-9.18) (1.17-11.0)	0.08	1.00 1.47 2.03	- (0.33-6.48 (0.49-8.38	0.62 ) )	3.5% (1.6-7.7) 4.0% (1.5-10.3) 9.8% (5.6-16.6)	1.00 1.15 ( 2.98 (	- (0.31-4.18) (1.07-8.34)	0.07	1.00 0.60 ( 1.23 (	- 0.13-2.67) 0.48-3.14)	0.46	28.1% (24.6-32.0) 39.9% (33.4-46.7) 43.2% (37.5-49.1)	1.00 1.69 1.94	- [1.21-2.37 [1.44-2.62]	<b>&lt;0.01</b> ) )	1.00 1.48 1.50	- (1.01-2.16 (1.04-2.15	<b>0.04</b> ) )	322, 421 1 125, 177   { 177, 241 1	46, 204 80, 113 17, 154	527, 715 202, 289 273, 370
Age at first heter 17+ 16 <16	rosexual se 1.0% (0.3- 1.5% (0.5- 4.0% (2.4-	ex -2.8) -4.6) -6.5)	1.00 1.49 ( 4.18 (	- [0.31-7.15] [1.31-13.3]	0.03	1.00 1.14 1.65	- (0.27-4.74 (0.55-4.90	0.67 ) )	2.8% (0.7-9.9) 4.7% (1.8-11.8) 7.8% (4.7-12.6)	1.00 1.75 ( 2.99 (	- (0.33-9.27) (0.70-12.7)	0.26	1.00 1.14 (i 1.58 (i	- 0.24-5.30) 0.37-6.62)	0.74	25.6% (21.4-30.3) 33.4% (27.9-39.4) 45.3% (40.7-49.9)	1.00 1.46 2.40	- (1.03-2.06) (1.78-3.23)	<b>&lt;0.01</b> ) )	1.00 1.13 1.53	- (0.75-1.67 (1.07-2.19	0.05 ) )	210, 245    { 148, 205   { 238, 352  1	87, 112 84, 108 67, 243	340, 431 253, 351 376, 539
<b>Condom used fo</b> Yes No	r most rec 0.7% (0.2- 4.1% (2.6-	<b>ent se</b> x -2.0) -6.4)	<b>with</b> 1.00 6.03 (	most rece - (1.87-19.4)	ent par <0.01	tner 1.00 3.59	- (0.77-16.6	0.10 )	4.3% (2.0-8.9) 7.4% (4.5-12.0)	1.00 1.79 (	- (0.70-4.59)	0.22	1.00 1.06 (	- 0.35-3.21)	0.91	33.5% (29.7-37.6) 38.0% (33.9-42.4)	1.00 1.22	- (0.95-1.56	0.13 )	1.00 0.97	- (0.70-1.34	0.87 )	301, 391 1 283, 398 1	63, 221 72, 237	491, 671 458, 621
Number of sexua 0 or 1 2 3 to 4 5+	al partners 1.5% (0.7- 1.3% (0.4- 3.1% (1.1- 7.5% (3.7-	in the -3.0) -4.2) -8.5) -14.6)	last ye 1.00 0.86 ( 2.16 ( 5.47 (	ear [0.21-3.56] [0.60-7.78] [1.89-15.9]	0.01				3.4% (1.5-7.8) 0.9% (0.2-3.9) 1.5% (0.4-6.2) 21.2% (12.9-32.7)	1.00 0.27 ( 0.44 ( 7.54 (	- (0.05-1.41) (0.08-2.30) (2.66-21.3)	<0.01				26.0% (22.6-29.7) 40.3% (33.2-47.7) 43.0% (35.5-50.9) 60.9% (51.8-69.3)	1.00 1.92 2.15 4.42	- 1.34-2.75 1.49-3.10 2.93-6.67	<b>&lt;0.01</b> ) )				359, 466 1 123, 159 70, 110 67, 100	45, 196 74, 99 57, 83 63, 89	568, 768 185, 251 134, 194 103, 146

#### Table 7-2b continued.

Number of new s	exual partners in	n the last year														ĺ	
0	1.8% (0.9-3.8)	1.00 -	0.01	1.00 -	0.49	5.5% (2.3-12.6)	1.00 -	0.37	1.00 -	0.31	26.0%(22.0-30.5)	1.00 -	<0.01	1.00 -	0.37	263, 335-108, 136	6 416, 540
1	0.8% (0.2-2.5)	0.42 (0.10-1.7	2)	0.33 (0.05-2.06)	)	4.0% (1.7-9.0)	0.71 (0.21-2.47)		1.13 (0.09-13.9)		36.7% (31.8-41.8)	1.64 (1.21-2.24)		1.28 (0.88-1.85)		203, 270 115, 161	323, 452
2+	5.1% (2.9-8.8)	2.87 (1.08-7.6	3)	0.47 (0.09-2.45)	)	8.0% (4.5-13.7)	1.48 (0.50-4.35)		2.87 (0.26-30.7)		46.3% (40.7-52.0)	2.45 (1.78-3.37)		1.06 (0.67-1.68)		152, 229 115, 170	251, 366
Number of sexua	al partners in the	last year witho	ut a con	dom												l	
0	0.3% (0.1-1.3)	1.00 -	<0.01	1.00 -	0.09	1.8% (0.4-8.4)	1.00 -	0.04	1.00 -	0.45	27.0%(22.4-32.1)	1.00 -	<0.01	1.00 -	0.49	205, 248 88, 115	331, 450
1	1.7% (0.8-3.6)	5.26 (1.05-26.	2)	1.23 (0.09-15.2)	)	4.9% (2.6-8.9)	2.75 (0.50-15.0)		0.78 (0.19-3.15)		34.3%(30.2-38.7)	1.41 (1.02-1.96)		1.12 (0.72-1.71)		287, 396 160, 222	475, 640
2+	6.5% (3.9-10.9)	21.3 (4.67-97.	2)	4.95 (0.42-57.9)	)	10.9%(6.0-19.0)	6.51 (1.19-35.6)		0.46 (0.11-1.83)		48.8%(42.3-55.4)	2.59 (1.81-3.70)		1.37 (0.80-2.34)		130, 194 95, 134	194, 281
Concurrent partr	nerships in last y	vear <sup>f</sup>														l	
No	2.6% (1.6-4.2)	1.00 -	0.74	1.00 -	0.05	6.7% (4.1-10.8)	1.00 -	0.10	1.00 -	0.07	32.9%(29.6-36.3)	1.00 -	<0.01	1.00 -	0.24	418, 557 219, 298	676, 916
Yes	2.0% (0.7-5.4)	0.75 (0.24-2.3	7)	0.18 (0.04-0.71)	)	6.5% (2.7-14.8)	0.96 (0.34-2.72)		0.60 (0.19-1.79)		49.7%(41.0-58.4)	2.02 (1.37-2.96)		1.52 (0.92-2.50)		82, 121 67, 92	136, 188
Unknown	1.7% (0.5-5.6)	0.64 (0.17-2.3	7)	0.60 (0.11-3.00)	)	1.3% (0.3-5.5)	0.19 (0.04-0.88)		0.06 (0.00-0.71)		38.5%(30.9-46.7)	1.28 (0.88-1.86)		1.18 (0.77-1.80)		92, 122 52, 75	137, 193
Number of lifetin	ne sexual partne	rs														l	
1 to 4	0.4% (0.1-1.5)	1.00 -	<0.01	1.00 -	0.03	1.0% (0.3-3.1)	1.00 -	<0.01	1.00 -	<0.01	25.3%(21.8-29.2)	1.00 -	<0.01	1.00 -	<0.01	332, 412 133, 176	524, 706
5 to 9	1.2% (0.3-4.0)	3.21 (0.47-21.	9)	1.78 (0.20-15.5)	)	3.8% (1.3-10.8)	4.15 (0.82-21.0)		4.87 (0.58-40.2)		39.6% (33.6-45.9)	1.93 (1.39-2.69)		1.50 (1.01-2.21)		141, 200 84, 123	222, 314
10+	7.6% (4.8-11.7)	22.6 (4.92-104	ł)	8.69 (1.21-62.0)	)	12.3%(7.6-19.2)	14.6 (3.89-54.5)		19.80 (3.03-129.)		49.2%(43.2-55.2)	2.86 (2.09-3.92)		2.23 (1.45-3.42)		148, 224 121, 167	247, 342
Ever had any sar	ne sex experien	ce/contact														1	
No	2.3% (1.5-3.5)	1.00 -	0.89	1.00 -	0.37	5.8% (3.8-8.9)	1.00 -	0.87	1.00 -	0.82	33.9%(31.1-36.8)	1.00 -	0.11	1.00 -	0.02	577, 762 311, 427	922, 126
Yes	2.0% (0.4-9.0)	0.90 (0.18-4.4	3)	0.31 (0.02-4.16)	)	5.1% (1.0-22.0)	0.87 (0.17-4.54)		0.79 (0.10-6.09)		42.4%(32.2-53.4)	1.44 (0.92-2.26)		2.08 (1.12-3.84)		47, 78 31, 44	80, 115

<sup>a</sup>N in column headings shows unweighted denominators. Total denominators by characteristic and in multivariable models vary due to item-missingness. <sup>b</sup>AOR: Adjusted odds ratios, adjusted for all variables shown. <sup>c</sup>Results for recent diagnosis are not reported due to small sample size in Scotland and Wales when limited to those tested. <sup>d</sup>IMD: Index of multiple deprivation of LSOA (lower super output area) of residence. IMD scores for England, Scotland and Wales were adjusted before being combined and assigned to quintiles, using the method described by Payne and Abel<sup>264</sup>. <sup>e</sup>Excludes 16 year-olds. <sup>f</sup>Among those with <u>></u>1 sexual partner in last year. Figure 7-2 shows the unadjusted OR for prevalent infection and for recent testing by sociodemographic and behavioural factors. Groups in the upper right hand guadrant are those where both the odds of prevalent infection and of testing were higher than the reference group. Groups in the upper left hand guadrant show those where the odds of prevalent infection were higher, but odds of testing were lower than the reference group. The demographic and behavioural factors associated with recent testing were broadly similar to those associated with prevalent infection and with recent diagnosis, with some exceptions. Notably, whereas women living in one of the two most deprived IMD quintiles had almost a four-fold increase in the odds of prevalent infection compared to those living in less deprived areas (OR 3.82, 95%CI 1.35-10.79), the odds of recent testing did not differ by deprivation (OR 0.99, 95%CI 0.77-1.27). Among men, the odds of prevalent infection were higher among 20 to 24 versus 16 to 19 year-olds (OR 10.6, 95%CI 2.40-46.3), but odds of recent testing were lower in the older age group (OR 0.67, 0.44-0.84). In men, not having used a condom at last sex was associated with a six-fold increase in the odds of prevalent infection (OR 6.03, 95%CI 1.87-19.42), but was not associated with recent testing (OR 1.22, 95%CI 0.95-1.56). Similar patterns were seen when comparing adjusted ORs from multivariable models (Table 3).

Although the proportion recently tested was generally higher in those reporting risk factors for chlamydia, recent testing remained well below 100% in all sociodemographic and behavioural subgroups. For example, 25% (95%CI 20%-31%) of women and 47% (95%CI 41%-53%) of men with two or more new sexual partners in the last year, and 20% (95%CI 16%-27%) and 45% (95%CI 38%-51%) with two or more sexual partners without a condom in the last year had not been either recently tested or offered a test. There was evidence of

both repeat (or persistent) and incident infections. Among individuals with a prevalent chlamydia infection, 14% (95%CI 7%-14%) had ever been diagnosed with chlamydia and 5% (95%CI 2%-17%) reported a diagnosis in the last year (indicating either repeat or persistent infections). Fifty percent (95%CI 35%-64%) of those with a prevalent infection reported a recent chlamydia test (89% of whom did not report a recent diagnosis, thus indicating incident rather than repeat/persistent infections within the last year).





The area of the bubble represents the percentage of those with a prevalent infection in each group (percentage shown in parentheses). Factors in the upper right hand quadrant are those where both the odds of prevalent infection and of testing were higher than the reference group. Factors in the upper left hand quadrant show those where the odds of prevalent infection were higher, but odds of testing were lower than the reference group.

Letters indicate reference groups: [a] 16-19 years old; [b] Resident in lower super output area in the 2 least deprived quintiles, as measured by Index of Multiple Deprivation (IMD); [c] Left school at 17+ (among those aged  $\geq$ 16); [d] 17+ years at first heterosexual sex; [e] 0 or 1 sexual partners in the last year; [f] 0 new sexual partners in the last year; [g] 0 sexual partners in the last year without a condom; [h] 1 to 9 lifetime sexual partners; [i] Condom used at last sex; [j] No concurrent partnership in last year (among those with 1+ more sexual partners in last year); [k] Reports binge drinking never or less than monthly; [l] Never had same sex contact/experience.



Figure 7-2b: Bubble plot showing unadjusted odds ratios for prevalent chlamydia infection versus recent testing by sociodemographic and behavioural factors, and percentage of prevalent infections in each group (sexually-experienced 16-24 year-old <u>men</u>)

The area of the bubble represents the percentage of those with a prevalent infection in each group (percentage shown in parentheses). Factors in the upper right hand quadrant are those where both the odds of prevalent infection and of testing were higher than the reference group. Factors in the upper left hand quadrant show those where the odds of prevalent infection were higher, but odds of testing were lower than the reference group.

Letters indicate reference groups: [a] 16-19 years old; [b] Resident in lower super output area in the 2 least deprived quintiles, as measured by Index of Multiple Deprivation (IMD); [c] Left school at 17+ (among those aged  $\geq$ 16); [d] 17+ years at first heterosexual sex; [e] 0 or 1 sexual partners in the last year; [f] 0 new sexual partners in the last year; [g] 0 sexual partners in the last year without a condom; [h] 1 to 9 lifetime sexual partners; [i] Condom used at last sex; [j] No concurrent partnership in last year (among those with 1+ more sexual partners in last year); [k] Reports binge drinking never or less than monthly; [l] Never had same sex contact/experience.

## 7.4.1.3 Distribution of infections by demographic and behavioural characteristics

Table 7-3 shows the distribution of demographic and behavioural characteristics among individuals with a prevalent infection, those with a recent diagnosis and the sexually-experienced population. Among both women and men, prevalent infections were detected in those reporting no new sexual partners in the last year, and those reporting no condomless sexual partners in the last year. Infections in women tended to be more evenly distributed according to numbers of sexual partners and other risk factors than in men. For example, among the 62 individuals (37 women, 25 men) with a prevalent infection 38% of women and 55% of men reported two or more new sexual partners in the last year (versus 21% and 25% of the sexually-experienced population); 37% of women and 61% of men reported two or more condomless sexual partners in the last year (versus 18% and 19% of the population). Living in one of the 40% most deprived LSOA accounted for 73% and 65% of prevalent infections in women and men respectively. This difference in distribution of infections by sex is also illustrated in Figure 7-2, where the size of the markers indicates the proportion of prevalent infections found in that risk group. Reporting that no condom had been used at last sex was common among this age group (63% of sexuallyexperienced women and 48% of men), and among both women and men with a prevalent infection (69% in women and 85% in men).

Table 7-3a: Distribution of demographic and behavioural characteristics among a) individuals with a prevalent chlamydia infection, b) individuals with a recent chlamydia diagnosis and c) the sexually-experienced population (16-24 year-old sexually-experienced <u>women</u>, Natsal-3).

	(a) Prevale	nt infection	(b) Diagnosed wi	th chlamvdia in	(c) Sexually-	experienced
	detected in	urine (n=37)	the last ye	ar (n=54)	population	(n=1,740)
	Percent	95%CI	Percent	95%CI	Percent	95%CI
Age group						
16-19	43.6%	(26.7-62.1)	40.5%	(26.6-56.2)	35.6%	(33.2-38.0)
20-24	56.4%	(37.9-73.3)	59.5%	(43.8-73.4)	64.4%	(62.0-66.8)
Country		· · · · ·		· · · ·		,
England	79.8%	(62.5-90.3)	92.1%	(80.8-97.0)	85.1%	(83.1-86.9)
Scotland	9.5%	(3.3-24.5)	3.4%	(0.7-14.3)	9.5%	(8.0-11.2)
Wales	10.8%	(3.8-27.0)	4.5%	(1.3-14.3)	5.4%	(4.4-6.7)
IMD quintile of LSO	A of residence	a				( - )
2 least deprived						
guintiles	15.6%	(5.9-35.2)	30.9%	(18.4-46.9)	35.1%	(32.3-38.0)
middle quintile	11.0%	(4.5-24.7)	12.5%	(5.6-25.4)	19.5%	(17.3-21.9)
2 most deprived		( )		· · · ·		,
quintiles	73.4%	(54.9-86.3)	56.7%	(40.8-71.3)	45.4%	(42.5-48.4)
Age left school <sup>b</sup>		(/		( /		( )
17+	77.9%	(62.0-88.3)	74.9%	(60.2-85.4)	78.5%	(76.3-80.6)
16	22.1%	(11.7-38.0)	25.1%	(14.6-39.8)	21.5%	(19.4-23.7)
Age at first heteros	exual sex	(1.11 0010)	2011/0	(1.10.0010)	211070	()
17+	16.0%	(6.5-34.3)	19.5%	(8.9-37.7)	33.6%	(31.0-36.3)
16	37.3%	(21.0-57.1)	27.0%	(15 1-43 3)	29.5%	(27.0-32.0)
<16	46.8%	(28 8-65 6)	53.5%	(38.0-68.4)	36.9%	(34 3-39 6)
Number of sexual r	artners in the l	ast vear	00.070	(00.0 00.1)	00.070	(01.0 00.0)
0 or 1	52 1%	(33 7-69 9)	31 10%	(18 2-47 9)	65.3%	(62 7-67 9)
2	19.4%	(8 6-38 0)	21 70%	(12.3-35.5)	15.0%	(13.3-17.0)
2 3 to 4	6.4%	(2.1-17.5)	23.60%	(12.0 00.0)	11.6%	(10.0 17.0)
5+	22.2%	(9.8-42.7)	23.60%	(12,2-40,6)	8.1%	(10.1 10.2)
Number of new sex	ual nartners in	the last year	20.0070	(12.2 40.0)	0.170	(0.0 0.0)
0	38.0%	(22 4-56 6)	21 9%	(12 0-36 7)	51.8%	(49 0-54 5)
1	24.4%	(10.9-46.0)	26.1%	(12.0 00.7)	27.6%	(25 2-30 1)
1 2±	27.7/0	(10.3-40.0)	51.0%	(13.3-43.7)	20.6%	(20.2-30.1) (18.4-23.1)
Number of sexual r	artners in the l	ast year without :	a condom	(30.4-07.1)	20.070	(10.4-20.1)
	18.0%	(7 5-40 1)	11 /%	(3 9-28 6)	21.8%	(10 7-24 1)
1	10.978	(7.5-40.1)	11.470	(28 6-58 6)	21.0%	(19.7-24.1)
1 2±	36.9%	(20.6-57.0)	45.0%	(20.0-50.0)	17.5%	(15 6-19 6)
∠⊤ Number of sevual r	artners over th	(20.0-07.0)	+0.770	(01.0-00.0)	17.570	(10.0-10.0)
1 to A	12 20/	(25.1-61.4)	16 1%	(8 3-28 8)	52.6%	(10.0-55.3)
5 to 9	22.6%	(11 3-40 1)	35.0%	(21.6-51.2)	26.3%	(24 0-28 8)
10±	22.070	(11.3-40.1) (10.2-55.3)	48.0%	(21.0-51.2)	20.376	(24.0-20.0) (10.0-23.4)
Condom used for n	00.270	(19.2-00.0)	40.370	(33.3-04.3)	21.170	(19.0-23.4)
	31 10/	(16 6-50 5)	21 0%	(10.7-30.7)	36.8%	(34 2-30 6)
No	69.0%	(10.0-30.3)	79 10/	(10.7 - 39.7)	62 20/	(60 4 65 9)
Concurrent partner	oo.970 Sching in last vo	(49.5-65.4)	70.170	(00.3-09.3)	03.276	(00.4-05.8)
No		(15 1 90 6)	56.0%	(405704)	76 20/	(72 0 70 2)
NU	00.0%	(45.4-60.6)	00.0%	(40.5-70.4)	10.2%	(73.9-70.3)
i es Unknown	22.0%	(10.0-43.5)	22.0%	(11.0-39.3)	11.370	(9.7-13.1)
Erequency of bings	12.470	(5.0-27.6)	21.2%	(11.7-35.4)	12.0%	(10.9-14.4)
Frequency of binge	annking					
then monthly	E0 E9/	(22.4.69.6)	41 00/	(27.0.57.0)	60.00/	(50 7 64 0)
man monuny	0.3%	(32.4-00.0)	41.2%	(27.0-57.0)	02.3%	(59.7-64.9)
monuniy	ŏ.1%	(3.0-22.9)	20.4%	(14.7-43.0)	22.1%	(19.9-24.5)
weekly or more	40 70/		22 40/	(10 0 40 0)	45 00/	(10 7 47 0)
	40.7%	(24.2-59.6)	32.4%	(18.8-49.8)	15.6%	(13.7-17.6)
Ever nad any same	sex experience		00.40/	(40 5 40 0)	00 50/	
res	21.6%	(9.4-42.2)	28.1%	(16.5-43.6)	22.5%	(20.3-24.8)
NO	78.4%	(57.8-90.6)	71.9%	(56.4-83.5)	77.5%	(75.2-79.7)
Bases (Wt,				- 4	. · ·	0 4740
unwt)	18, 3	37	28,	54	96	8, 1740

Table 7-3b: Distribution of demographic and behavioural characteristics among a) individuals with a prevalent chlamydia infection, b) individuals with a recent chlamydia diagnosis and c) the sexually-experienced population (16-24 year-old sexually-experienced <u>men</u>, Natsal-3).

	(a) Prevale detected in	ent infection urine (n=25)	(b) Diag chlamydia i	nosed with n the last year	(c) Sexually populatio	/-experienced on (n=1,375)
	%	95%CI	(n %	95%CI	%	95%CI
Age group						
16-19	5.5%	(1.2-22.5)	35.6%	(17.9-58.3)	37.3%	(34.4-40.2)
20-24	94.5%	(77.5-98.8)	64.4%	(41.7-82.1)	62.7%	(59.8-65.6)
Country		, ,		, ,		,
England	71.8%	(47.3-87.9)	78.8%	(55.5-91.7)	85.7%	(82.7-88.2)
Scotland	24.2%	(9.5-49.3)	19.2%	(7.0-43.1)	8.8%	(6.8-11.4)
Wales	3.9%	(0.5-26.9)	2.0%	(0.2-14.5)	5.5%	(4.0-7.5)
IMD quintile of LSOA of						
residence <sup>a</sup>						
2 least deprived quintiles	22.0%	(7.3-50.1)	33.7%	(16.4-56.7)	36.9%	(33.6-40.2)
middle quintile	12.9%	(4.1-34.1)	15.3%	(5.1-38.0)	18.2%	(15.9-20.7)
2 most deprived quintiles	65.1%	(41.5-83.1)	51.0%	(29.9-71.8)	44.9%	(41.4-48.5)
Age left school <sup>b</sup>						
17+	49.3%	(27.7-71.1)	66.5%	(43.5-83.6)	75.4%	(72.6-77.9)
16	50.7%	(28.9-72.3)	33.5%	(16.4-56.5)	24.6%	(22.1-27.4)
Age at first heterosexual sex						
17+	15.1%	(4.9-37.9)	12.3%	(3.1-38.4)	35.1%	(32.1-38.2)
16	15.7%	(4.5-42.3)	20.6%	(7.7-44.7)	26.1%	(23.5-28.8)
<16	69.2%	(44.9-86.1)	67.1%	(43.1-84.6)	38.8%	(35.8-41.8)
Number of sexual partners in						
the last year				··· ·>		(
0 or 1	37.4%	(18.2-61.6)	25.2%	(10.7-48.7)	57.30%	(54.3-60.3)
2	11.0%	(3.0-33.3)	3.5%	(0.7-15.0)	18.70%	(16.5-21.2)
3 to 4	15.6%	(5.1-38.5)	4.4%	(1.0-18.0)	13.60%	(11.7-15.7)
5+	36.1%	(17.6-59.8)	66.9%	(44.4-83.6)	10.40%	(8.8-12.2)
Number of new sexual partners						
in the last year	04.40/		00.00/		40 40/	(00.0.45.0)
0	34.1%	(16.0-58.5)	30.3%	(13.1-55.4)	42.1%	(39.0-45.2)
1	11.1%	(3.0-33.2)	23.3%	(9.8-45.9)	32.6%	(29.7-35.6)
2+ Number of covuel pertners in	54.8%	(32.0-75.7)	46.5%	(26.3-67.9)	25.3%	(22.8-28.0)
the last year without a condom						
0	4.8%	(1.0-19.9)	8.2%	(1.6-33.0)	33.1%	(30.2-36.1)
1	34.7%	(16.1-59.5)	39.7%	(21.1-61.8)	47.5%	(44.4-50.5)
2+	60.5%	(36.7-80.2)	52.1%	(30.7-72.8)	19.4%	(17.2-21.8)
Number of sexual partners over the lifetime						
1 to 4	8.50%	(1.8-32.5)	6.5%	(1.8-21.3)	52.70%	(49.8-55.7)
5 to 9	11.60%	(3.0-35.8)	16.7%	(5.5-41.1)	22.40%	(20.0-25.0)
10+	79.90%	(55.8-92.6)	76.7%	(53.8-90.3)	24.90%	(22.5-27.4)
Condom used for most recent sex with most recent partner						
Yes	15.5%	(4.9-39.3)	35.3%	(17.2-58.9)	51.7%	(48.6-54.8)
No	84.5%	(60.7-95.1)	64.7%	(41.1-82.8)	48.3%	(45.2-51.4)
Concurrent partnerships in last		( )		( /		( )
year <sup>c</sup>						
No	77.6%	(55.8-90.5)	74.5%	(52.3-88.7)	71.2%	(68.4-74.0)
Yes	11.4%	(3.7-30.3)	22.0%	(9.0-44.6)	14.3%	(12.3-16.6)
Unknown	11.0%	(3.0-33.3)	3.5%	(0.7-15.0)	14.5%	(12.5-16.7)
Frequency of binge drinking						
never / less than monthly	25.1%	(9.9-50.6)	25.9%	(11.2-49.1)	52.6%	(49.6-55.5)
monthly	26.7%	(12.0-49.5)	16.2%	(5.8-37.7)	20.2%	(17.9-22.6)
weekly or more often	48.2%	(26.6-70.4)	57.9%	(35.8-77.3)	27.3%	(24.7-30.0)
Ever had any same sex						
experience/contact						
Yes	6.8%	(1.3-29.3)	8.2%	(1.6-33.0)	8.0%	(6.5-9.8)
No	93.2%	(70.7-98.7)	91.8%	(67.0-98.4)	92.0%	(90.2-93.5)
Bases (wt, unwt)		14, 25		20, 27		1003, 1375

<sup>a</sup>IMD: Index of multiple deprivation of LSOA (lower super output area) of residence. IMD scores for England, Scotland and Wales were adjusted before being combined and assigned to quintiles, using the method described by Payne and Abel<sup>9</sup>, <sup>b</sup>Excludes 16 year-olds,<sup>c</sup>Among those with 1+ more sexual partners in last year

#### 7.4.1.4 England versus Scotland/Wales

Women and men living in England were significantly more likely to have been recently tested than those living in Scotland or Wales (OR: 2.25, 95%CI 1.66-3.05 in women; 2.60, 1.64-4.12 in men, Figure 7-3). There was no significant difference in the prevalence of infection or proportion recently diagnosed by country of residence. When limiting to participants resident in England, there were no notable differences in either the factors associated with prevalent infection, recent diagnosis or recent testing, or in the relationships between these factors (Appendix 3).

Figure 7-3: Percentage with a prevalent chlamydia infection, ever diagnosed with chlamydia, recent diagnosis of chlamydia and recently tested for chlamydia by country of residence (sexually-experienced 16-24 year-olds, Natsal-3)



#### 7.5 Discussion

#### 7.5.1 Key findings

In 2010-12 chlamydia was a common, and commonly diagnosed, infection among young adults in Britain. Prevalent infections in women were more evenly distributed according to numbers of sexual partners than in men. Diagnoses had arisen following both opportunistic screening and clinically-indicated testing in roughly equal numbers. Living in more deprived areas was significantly associated with prevalent infection after adjusting for sociodemographic and behavioural factors. The proportion reporting chlamydia testing was generally greater among those reporting factors associated with chlamydia. However, substantial proportions of young adults reporting risk factors for chlamydia had not been recently tested and incident infections in the last year were evident.

#### 7.5.2 Strengths and limitations

The major strength of this analysis is that it used individual-level data from a nationally-representative sample. Behavioural and biological data were linked to examine a range of risk factors for different outcomes within the same survey.

Investigations using the Natsal-3 data were reliant on the questions asked in the survey. For example the survey asked about the location of the most recent test, but not of the most recent diagnosis. The finding that the vast majority of those recently diagnosed had last been tested in clinical settings may therefore reflect re-testing patterns with follow up tests done in sexual health settings, rather than low diagnosis rates among non-clinical settings. However this is likely to have minimal impact on these findings as routine re-testing before one year after a positive test was not recommended in England until 2013.<sup>119</sup>

Despite the size of Natsal-3 and oversampling of younger people,<sup>172</sup> the number of participants aged under 25 limited statistical power to explore all associations of potential interest including local area and ethnicity. Given the relatively small absolute numbers of prevalent infections (n=62), the proportions in specific subgroups should be interpreted with caution. These findings may be affected by who agreed to take part in the survey or provide a urine sample. Survey weights were used to minimise bias but unmeasured bias remains feasible.

Comparisons between risk factors for prevalent infection and recent testing may have been affected by the estimation of outcomes among different denominators. This was explored further in a sensitivity analysis, which showed no notable difference between ORs for testing when estimated in sexuallyexperienced participants versus urine study participants (data not shown). A further limitation is the accuracy of self-reporting. Detailed questions were answered via self-completion, which are expected to have minimised social desirability bias.

#### 7.5.3 Comparison to other studies/data

Recent testing was not associated with area-level deprivation in this analysis. This is contrary to an analysis of data from the South East of England, which found higher rates of chlamydia screening in more deprived areas in 2008.<sup>265</sup> This difference in findings may reflect the different study period, when screening coverage was lower, or regional variation in screening patterns.

National surveillance data on chlamydia tests and diagnoses among 15 to 24 year-olds are available for England for the period covered by Natsal-3. The average coverage of chlamydia testing in England in 2010-12 among 15 to 24 year-olds was 40% in women and 20% in men.<sup>189,266</sup> This is lower than the 57%

of women and 37% of men resident in England estimated to have been test in the last year in Natsal-3. Differences between denominators (all versus sexually-experienced only) and age ranges (surveillance data for this period use partly aggregated data and are not available for 16 to 24 year-olds) may partly explain these differences. Applying the proportion of 16 to 24 year-olds with one or more sexual partner estimated in Natsal-3 (80%<sup>235</sup>) to surveillance data results in an estimated coverage per year of 51% and 25% among sexuallyexperienced women and men respectively. This is more comparable, but still somewhat lower than the estimates presented in this chapter, especially since surveillance data may include repeat tests among from the same individual within one year.<sup>120</sup> This may indicate some residual bias arising from who took part in Natsal-3. The findings on location of last test among those recently diagnosed are consistent with 2011 surveillance data, where 42% of diagnoses among 15 to 24 year-olds were reported from GUM clinics, 15% from family planning services, 7% GPs, 2% from education and 33% from other/unknown settings.<sup>189</sup> The proportion of diagnoses from GPs was higher in Natsal-3 (27%) than in surveillance data. This may reflect the partially- aggregate nature of surveillance data as a large proportion of diagnoses made in other/unknown settings are likely to be from GPs.

#### 7.5.4 Implications for evaluation of chlamydia control

Those reporting risk factors for chlamydia were generally more likely to report having been recently tested. This is contrary to uptake patterns often seen in public health interventions, where those in most need are often least likely to access care.<sup>267</sup> However, at least one quarter of women and around half of men reporting a risk factor associated with prevalent infection had not been recently tested. This presents a clear potential for ongoing transmission of chlamydia from high risk but untested individuals. Almost all prevalent infections in men were among 20 to 24 year-olds, less than a third of whom reported recent testing. As young women tend to have slightly older male partners,<sup>268</sup> sexual mixing patterns by age may play a key role in transmission.

These findings also suggest that the likelihood of having an infection diagnosed and treated varies by deprivation, as although screening coverage was uniform by area-level deprivation, chlamydia prevalence was higher in those living in more deprived areas. This raises the question as to whether efforts to expand or intensify chlamydia screening should prioritise those living in more deprived areas to address this potential inequality. A high proportion of infections were found in those who had not used a condom at last sex, and around one fifth of recent diagnoses were made following a test prompted by a partner having chlamydia, which emphasises the importance of condom use and partner notification in chlamydia prevention and control.

#### 7.5.5 Summary

Comparison of factors associated with prevalent infection and with diagnosis and testing showed that those with risk factors for chlamydia were more likely to have accessed a chlamydia test in the last year. While this alignment between need and uptake of the intervention suggests the NCSP may be reaching those at risk of infection, this analysis demonstrated substantial opportunities for ongoing transmission of infection, and incident infections were evident. Given that comparisons between chlamydia prevalence in Natsal-2 and Natsal-3 did not provide any strong evidence for chlamydia prevalence to have decreased in

201

the decade up to 2010-12, these opportunities for transmission present a possible explanation for an absence of a decrease in prevalence.

The findings presented up to this point in the thesis still leave us in a position of considerable uncertainty as to whether there has or has not been any change in the incidence or prevalence of chlamydia during a time of widespread screening among under 25-year-olds. In the next chapter I therefore move to investigation of a third outcome measure, that of antibody seroprevalence as a marker of age-specific cumulative incidence.

#### 8 Trends in *C. trachomatis* antibody seroprevalence, as a marker of age-specific cumulative incidence, measured using stored sera from participants in the Health Survey for England (HSE) (1994 to 2012)

In previous chapters, my analysis of surveillance data showed that there was a decrease in percentage testing positive for chlamydia between 2008 and 2011 that remained after adjusting for known confounders, but that unmeasured confounding could not be ruled out. My comparison of data from the second and third Natsal surveys did not provide evidence to support there having been a decrease in chlamydia prevalence in the decade between the two surveys, which were conducted before and after the implementation of the NCSP. Evidence from Natsal-3 highlighted important opportunities for transmission of infection in 2010-12. In this chapter I present a study that applied a novel C. trachomatis antibody assay to stored sera from participants in a series of nationally-representative health surveys to investigate whether these data support there having been a change in age-specific cumulative incidence following the implementation of widespread screening.

#### 8.1 Background

As well as being resource-intensive and hard to achieve, population-based estimates of chlamydia prevalence using NAATs measure only *current* infections. Such studies are therefore unable to estimate the proportion of the population who have ever been infected and who are therefore at risk of chlamydia-related sequelae. Serology has therefore been proposed as an alternative approach for evaluating the impact of chlamydia control programmes.<sup>99</sup> Chlamydia antibodies persist and thus provide a longer-term marker of past infection.<sup>100,269</sup> Using blood samples collected for a purpose

203

other than chlamydia testing can also avoid the bias associated with data from populations accessing chlamydia testing.

As set out in Chapter 3, chlamydia seroepidemiology has been previously hampered by the lack of a suitable assay.<sup>99</sup> Commercially available assays to detect *C. trachomatis* antibodies suffered from poor sensitivity and specificity with several subject to cross-reactive with other species of *Chlamydia*. Assays also lacked robust validation against panels of sera from individuals with known previous infection and negative controls. In recent years tests with better specificity - not so affected by inter-species cross-reaction - have become available.<sup>209,269</sup>

The analysis presented in this chapter used data and stored sera from a nationally-representative household survey, the Health Survey for England (HSE), to explore sociodemographic and behavioural factors associated with serological evidence of a previous infection and to evaluate the impact of widespread opportunistic chlamydia screening on age-specific cumulative incidence of chlamydia in England up to 2012.

#### 8.2 Aims & objectives

The aims of this analysis were firstly, to explore the utility of chlamydia seroprevalence as an epidemiological tool for the evaluation of chlamydia control and secondly, to investigate trends in age-specific chlamydia incidence among 16 to 24 year-old women in England between before and after the implementation of the National Chlamydia Screening Programme. This aim was addressed through the following objectives:

- To describe the association between sociodemographic variables and selfreported health and sexual behaviours and the presence of chlamydia and antibodies in serum among HSE2010 and HSE2012 participants.
- To compare age-specific Pgp3 seroprevalence (defined as the prevalence of Pgp3 antibodies detected in serum) at selected years (1994 to 2012) during a period of increased chlamydia screening.
- To compare age-specific Pgp3 seroprevalence between birth cohorts exposed to differing levels of chlamydia screening.

#### 8.3 Methods

#### 8.3.1 Participants

HSE is a series of nationally-representative surveys carried out annually since 1991. Participants are invited to provide a blood sample for laboratory analyses and storage for future research. Details of HSE methodology are reported elsewhere.<sup>7,270</sup> In summary, each annual survey used a multi-stage stratified probability sampling design. In each year a random sample of postcode sectors was selected from the Postcode Address File, from which a random sample of postal addresses was drawn. All adults aged 16 years and over at each selected household were selected for interview. Information on participants' health and wellbeing, along with detailed sociodemographic information, was collected using a combination of face to face interviews and self-completed questionnaire booklets and a nurse visit.<sup>7</sup> HSE2010 and HSE2012 also included questions on sexual behaviours and chlamydia diagnosis history, which were collected using the self-completed booklet. Other household members could be

present during interviews and questionnaire completion but did not see others' booklets.<sup>7</sup>

Sera were obtained from HSE participants who provided consent for their blood sample to be stored and used in future anonymous analyses (Figure 8-1). Sera were obtained from a) 16 to 44 year-old HSE2010/12 participants to explore factors associated with testing positive for anti-Pgp3 antibodies in serum (hereafter 'Pgp3 seropositive') and b) female participants aged 16 to 24 who took part in HSE years when stored sera were available (1994-1996, 2001-02 and each year from 2008-2012), to examine trends in the prevalence of anti-Pgp3 antibodies in serum (hereafter 'Pgp3 seroprevalence') in the age group targeted by the NCSP. Trends over time among men were not investigated as lower assay sensitivity<sup>209,269</sup> complicates interpretation of trends and as monitoring chlamydia infections in women is of greater public health value given that most chlamydia-related complications are among women.

### Figure 8-1: Flow chart showing selection of stored sera from Health Survey for England (HSE) participants



Note: analysis groups are not mutually exclusive, therefore sum of number in each group is greater than total number tested (n=5,618).

#### 8.3.2 Laboratory testing

After collection, samples had been posted to the Royal Victoria Infirmary, Newcastle and stored in the HSE serum bank at -40°C. Specimens were sent to PHE Colindale where they were aliquoted, relabelled with the study ID and stored at -20°C. Specimens were transported frozen to Imperial College London and stored at -20°C until testing.

#### 8.3.2.1 Pgp3 assay

Pgp3 is a *C. trachomatis*-specific protein. Pgp3 is considered a useful immunogen for serological tests as it is highly conserved between strains and is rarely found in *C.pneumoniae* isolates.<sup>209</sup> In 2009, Wills *et al* developed an "inhouse" indirect immunoglobulin G (IgG) enzyme-linked immunosorbent assay (ELISA) based on the Pgp3 antigen (hereafter 'indirect Pgp3 ELISA'). Sensitivity was measured among adult patients attending GUM clinics with a known, previously diagnosed chlamydia infection and was estimated at 73.8% (95%CI 66.5%-79.9%) in women and 44.2% (95%CI 37.3%-51.3%) in men. Specificity was estimated using microimmunfluourescence (MIF)-negative paediatric sera to reduce likelihood of a previous sexually-acquired *C. trachomatis* infection, and was found to be 97.6% (95%CI 96.2%-98.6%).<sup>209</sup>

The sensitivity and specificity of the indirect Pgp3 ELISA has been compared to other commercially available assays, and found to be at least 14% more sensitive in women than any of the comparator tests (Table 8-1). Sensitivity of the indirect Pgp3 ELISA is equivalent to the comparator commercial tests in men. No evidence was found of cross-reactivity with *C. pneumoniae*.

### Table 8-1: Comparative sensitivities and specificities (and 95%CI) of the Indirect Pgp3 ELISA, double-antigen sandwich ELISA and commercial ELISAs

Assay	Sens	sitivity	Specificity (women & men combined)
	Women	Men	
Double-antigen Pgp3 ELISA	82.9% (76.3-88.0)	54.4% (47.1-61.5)	97.8% (96.1-98.7)
Indirect Pgp3 ELISA	73.8% (66.5-79.9)	44.2% (37.3-51.3)	97.6% (96.2-98.6)
Ani Labsystems	59.8% (52.1-67.0)	40.5% (33.8-47.6)	99.0% (97.7-99.6)
SeroCT	55.5% (47.8-62.9)	40.0% (33.3-47.1)	97.2% (95.7-98.2)
Medac	45.7% (38.3-53.4)	43.7% (36.8–50.8)	96.0% (94.3-97.2)

Adapted from Wills et al<sup>209</sup> and Horner et al<sup>269</sup>

Since the publication of the original performance of the indirect Pgp3 ELISA, investigators at Imperial College London and University of Bristol have developed a double-antigen sandwich ELISA for the detection of anti-Pgp3 antibody (hereafter 'double-antigen Pgp3 ELISA'. This assay has demonstrated equivalent specificity (97.8%, 95%CI 96.1%–98.7%) and higher sensitivity (82.9%, 95%CI 76.3%-88.0% in women; 54.4%, 95%CI 47.1%-61.5% in men)<sup>269</sup>.

Although the double-antigen Pgp3 ELISA has demonstrated higher sensitivity than the indirect Pgp3 ELISA to detect a previous known infection in both women and men, the double-antigen Pgp3 ELISA requires around a 25-fold higher volume of sera. A study of female participants in a cohort study in New Zealand tested sera using both the indirect and double-antigen Pgp3 ELISA.<sup>269</sup> Comparison of results from the two assays showed that the indirect Pgp3 ELISA has good agreement with the double-antigen Pgp3 ELISA at low (<0.1) and high (>1.0) absorbance values (Table 8-2). Imperial College London therefore recommend the use of the indirect Pgp3 ELISA for initial screening, with subsequent testing of sera with absorbance values between 0.1 and 1.0

using the double-antigen Pgp3 ELISA to resolve 'equivocal' specimens (personal communication, Myra McClure, Gillian Wills, Imperial College London).

In the study presented in this chapter, all specimens were tested for *C*. *trachomatis* anti-Pgp3 antibody using the indirect Pgp3 ELISA.<sup>209</sup> Specimens with an optical density between 0.1 and 1.0 on the indirect Pgp3 ELISA were retested using the double-antigen Pgp3 ELISA. Specimens which were positive on both assays (cut-off values were determined by the testing laboratory), positive on the double-antigen Pgp3 ELISA only, or which were above the cut-off value for re-testing (i.e. absorbance value over 1.0) were considered Pgp3 seropositive. Those which were negative on both assays, were positive on the lndirect Pgp3 ELISA but negative on the double-antigen Pgp3 ELISA, or were below the cut-off value for re-testing (i.e. absorbance value less than 0.1) were considered Pgp3 seronegative. This testing strategy represents a 98.0% sensitivity, and 99.8% specificity compared to the results that would have been achieved had all samples been tested on the double-antigen Pgp3 ELISA (Table 8-2).

### Table 8-2: Comparison of indirect Pgp3 ELISA and double-antigen Pgp3 ELISAby absorbance values on indirect Pgp3 ELISA.

Data are from a study of female participants in a cohort study in New Zealand where all sera
were tested with both the indirect Pgp3 ELISA and the double-antigen Pgp3 ELISA (n=2,641
sera). <sup>269</sup> (Personal communication, M McClure & G Wills, Imperial College London)

		Double-antige	en Pgp3 ELISA
		Negative	Positive
Indirect	Negative (<0.1)	1817	10
Pgp3 ELISA	Negative (0.1-0.4731)*	302	233
(Absorbance	Positive (0.4731-1.0)	15	96
range)	Positive (>1.0)	4	164

\*An absorbance (450-620nm) value of 0.473 is the cut-off for the indirect assay.

#### 8.3.3 Statistical analyses

Data were analysed in Stata 12.1 accounting for weighting, clustering and stratification. Weights were applied in line with HSE analysis guidelines and included weights to correct for uneven probability of selection and - from HSE2003 onwards - non-response weights to ensure the sample is representative with regard to age, sex and region and to adjust for differential participation in blood specimen collection by sociodemographic and general health variables.<sup>271</sup>

Pgp3 seroprevalence among HSE2010/12 participants was estimated by sex, number of lifetime sexual partners and reporting of a previous chlamydia diagnosis, regardless of reported sexual activity. Associations between being Pgp3 seropositive and sociodemographic and behavioural factors among sexually-experienced HSE2010/12 participants were investigated using univariable and multivariable logistic regression. Associations with deprivation were explored using a residence-based measure, the IMD quintile of the LSOA of residence. Sexual behaviour variables reflecting exposure over the lifetime (years since first heterosexual sex, number of lifetime sexual partners) and more recent exposures (number of sexual partners in the last year, condom used at last sex) were included. Odds ratios (ORs) adjusted for number of lifetime sexual partners were calculated to reduce confounding of the association between predictors and being Pgp3 seropositive. Pgp3 seroprevalence among individuals ever diagnosed with chlamydia was estimated to explore sensitivity of serological testing to detect a previous known infection in this population. Pgp3 seroprevalence in those reporting no lifetime sexual partners was also estimated to explore test specificity and validity of

211

reporting of sexual activity. Analyses were conducted on those with non-missing data on the variable(s) of interest.

Pgp3 seroprevalence was estimated among 16 to 24 year-old women for each year with available sera from 1994 to 2012. The trend in Pgp3 seroprevalence was estimated from 2008 (the first year when the NCSP was nationallyavailable) to 2012 using a generalised linear model with year entered as a continuous variable.

Pgp3 seroprevalence by birth cohort was explored among women aged 16 to 24, with birth cohorts grouped to reflect their relative exposure to widespread chlamydia screening. The median age at first sex reported among 16 to 24 year-olds in Natsal-3 (16 years)<sup>235</sup> was used as a proxy age of sexual debut for the purposes of categorisation by birth cohort. Women who were ≤16 years in 2008 (the first year of national implementation of the NCSP) were defined as having high exposure. Women aged 17 to 24 in 2008 were defined as having partial exposure as they would have had some of their years post sexual debut before the NCSP was nationally-implemented but would have still been within the target age group when national implementation occurred. Women aged over 24 years in 2008 were defined as having 'limited' exposure as they would have been outside the NCSP target age group when the programme was nationally implemented. The numbers in each group by year of age are set out in Table 8-3 below.

212

Year Year Number of women by year of age												
Exposure to widespread screening	born	turned 16	16	17	18	19	20	21	22	23	24	16-24
Limited (born 1966-1975)	1970	1986									66	66
25 or older in 2008	1971	1987								61	81	142
	1972	1988							63	76	87	226
	1973	1989						47	66	70		183
	1974	1990					52	50	69			171
	1975	1991				40	46	55				141
Total			0	0	0	40	98	152	198	207	234	929
Limited (born 1976-1983)	1976	1992				52	56					108
25 or older in 2008	1977	1993		51	41	44					45	181
	1978	1994	40	54	55					46	76	271
	1979	1995	68	61					40	87		256
	1980	1996	65					31	87			183
	1981	1997					48	82				130
	1982	1998				39	75					114
	1983	1999			37	69						106
Total			173	166	133	204	179	113	127	133	121	1349
Partial (born 1984-1991)	1984	2000		29	74						36	139
17 to 24 in 2008	1985	2001	37	107						27	9	180
	1986	2002	88						23	11	21	143
	1987	2003						21	9	17	19	66
	1988	2004					26	7	15	24	21	93
	1989	2005				16	8	15	11	20		70
	1990	2006			22	6	10	20	20			78
	1991	2007		28	14	13	12	17				84
Total			125	164	110	35	56	80	78	99	106	853
High (born 1992-1996)	1992	2008	27	5	14	15	6					67
16 or younger in 2008	1993	2009	8	17	13	18						56
	1994	2010	9	12	10							31
	1995	2011	16	11								27
	1996	2012	10									10
Total			70	45	37	33	6*	0	0	0	0	185

### Table 8-3: Numbers of women aged 16 to 24 contributing to seroprevalenceestimates by birth cohort, by year of age

\*Not included in Figure 8-7 due to small group size.

#### 8.3.4 Regulatory approval

HSE is approved by an NHS research ethics committee each year. The use of stored sera from HSE participants for this study was approved by the Yorkshire and the Humber–South Yorkshire research ethics committee (ref: 13/YH/0304). All analyses were carried out anonymously and Pgp3 results were not linked to any identifiable information.

#### 8.4 Results

A flow chart of included participants is shown in Figure 8-1. Of 14,762 eligible participants 6,882 (46.6%) provided a sample with consent for future testing. Of

these, samples for 1,264 were unavailable due to a missing sample or

insufficient residual volume. Overall, 1,111/5,618 eligible participants with a

valid chlamydia antibody test result were Pgp3 seropositive. Among

HSE2010/12 participants, 546/2,521 were seropositive. Among 16 to 24 year-

old women in selected HSE years from 1994 to 2012, 565/3,361 were

seropositive. HSE2010/12 participants included in the analysis were

comparable to the overall HSE population on a range of sociodemographic and

behavioural variables (Table 8-4).

# Table 8-4: Comparison of reported sociodemographic and behavioural variablesbetween HSE participants and the study population by sex and age group (16-44year-olds, HSE2010 & HSE2012)

			Wo	men					M	en		
	16	-24 yea	rs	25	5-44 yea	rs	16	6-24 yea	rs	25	5-44 yea	rs
	HSE participants	Study population	Difference									
IMD quintile of LSOA												
area of residence												
Least deprived	18.2%	16.6%	-1.6%	18.5%	17.8%	-0.7%	18.7%	17.8%	-0.9%	17.2%	16.1%	-1.0%
2	18.9%	19.6%	0.8%	18.9%	19.4%	0.5%	18.1%	19.2%	1.1%	19.7%	19.5%	-0.1%
3	21.0%	21.4%	0.3%	21.6%	21.4%	-0.2%	21.0%	19.9%	-1.1%	21.0%	21.8%	0.8%
4	21.2%	21.4%	0.2%	19.5%	19.3%	-0.1%	18.1%	17.6%	-0.5%	21.1%	21.5%	0.4%
Most deprived	20.6%	21.0%	0.4%	21.6%	22.1%	0.5%	24.2%	25.6%	1.5%	21.1%	21.0%	-0.1%
Ethnicity												
White	84.9%	85.5%	0.6%	84.5%	86.0%	1.5%	81.2%	82.2%	1.0%	86.2%	86.6%	0.4%
Black or Black British	4.0%	3.0%	-1.1%	3.6%	3.1%	-0.5%	2.3%	2.5%	0.2%	2.8%	2.8%	0.0%
Asian or Asian British	5.7%	5.3%	-0.4%	8.5%	7.8%	-0.7%	10.4%	9.7%	-0.6%	7.7%	7.6%	-0.1%
Mixed	1.7%	1.9%	0.2%	1.3%	1.2%	-0.1%	4.8%	4.0%	-0.8%	1.5%	1.4%	0.0%
Other ethnic groups	3.8%	4.4%	0.5%	2.0%	1.9%	-0.1%	1.3%	1.5%	0.2%	1.8%	1.6%	-0.2%
Marital status												
Single	78.5%	77.9%	-0.7%	18.5%	18.5%	-0.1%	86.7%	86.6%	-0.1%	22.7%	22.1%	-0.6%
Married	2.8%	2.5%	-0.2%	52.7%	52.0%	-0.7%	1.5%	1.7%	0.3%	48.6%	50.0%	1.4%
Separated	0.2%	0.3%	0.0%	8.2%	8.6%	0.4%	0.0%	0.0%	0.0%	5.2%	5.1%	-0.1%
Cohabiting	18.5%	19.3%	0.8%	20.6%	20.9%	0.3%	11.8%	11.7%	-0.2%	23.5%	22.8%	-0.7%
Ever had sex	80.4%	80.2%	-0.2%	98.4%	98.5%	0.0%	73.7%	76.5%	2.8%	97.3%	97.2%	-0.1%
Ever diagnosed with chlamydia	5.8%	4.2%	-1.5%	6.5%	6.7%	0.2%	2.1%	2.4%	0.4%	5.1%	5.5%	0.4%
Ever been tested for chlamydia	47.6%	47.7%	0.1%	33.5%	33.2%	-0.4%	39.1%	39.9%	0.9%	18.0%	17.7%	-0.4%

\*IMD: index of multiple deprivation; LSOA: Lower super output area

#### 8.4.1 Pgp3 seroprevalence in HSE2010 & HSE2012

The overall Pgp3 seroprevalence among 16 to 44 year-olds in HSE2010/12 was

24.4% (95%CI 22.0%-27.1%) in women and 13.9% (95%CI 11.8%-16.2%) in

men. A further 2.1% of women and 2.6% of men aged 16 to 44 reported having been diagnosed with chlamydia but were Pgp3 seronegative. Among individuals who reported a previous chlamydia diagnosis, 64.7% (95%CI 51.9%-75.6%) of women and 43.9% (95%CI 26.5%-63.0%) of men were Pgp3 seropositive.

Among sexually-experienced participants, Pgp3 seroprevalence increased with age (Figure 8-2), years since first sex (Figure 8-3) and with increasing numbers of lifetime sexual partners (Figure 8-4). Peak Pgp3 seroprevalence was seen in women aged 30 to 34 (33.5%) and in men aged 35 to 39 (18.7%).

### Figure 8-2: Pgp3 seroprevalence by age group (sexually-experienced 16 to 44 year-olds, HSE2010 & HSE2012)

Solid lines show point estimates; dashed lines show 95% confidence intervals. N shows unweighted denominator.



### Figure 8-3: Pgp3 seroprevalence by years since first heterosexual sex, (sexually-experienced 16 to 44 year-olds, HSE2010 & HSE2012)

Solid lines show point estimates; dashed lines show 95% confidence intervals.


Figure 8-4: Pgp3 seroprevalence by reported numbers of lifetime sexual partners (16 to 44 year-olds, HSE2010 & HSE2012)



Table 8-5 presents Pgp3 seroprevalence by selected sociodemographic and behavioural factors among sexually-experienced women and men. In women, those reporting  $\geq$ 10 lifetime sexual partners had almost four-fold higher odds of being Pgp3 seropositive than those with 1-4 partners (OR 3.84, 95%CI 2.68-5.51). Being Pgp3 seropositive was also significantly associated with living in more deprived areas, younger age at first sex, non-condom use in the last year and reporting a previous chlamydia diagnosis. All variables associated with being Pgp3 seropositive in univariable analyses remained statistically significant after adjusting for number of lifetime sexual partners. The lifetime sexual partner-adjusted odds ratio (AOR) of being seropositive was 1.68 (95%CI 1.18-2.39) in women aged less than 16 at first heterosexual sex versus those aged 16 or over. Women who reported non-condom use in the last 4 weeks had two-fold higher odds of being seropositive (AOR 2.17, 95%CI 1.41-3.35).

In sexually-experienced men, similar factors were associated with being seropositive as seen in women in univariable analyses although age group and

deprivation of residence were not statistically significant. After adjusting for number of lifetime sexual partners, only reporting a previous diagnosis of chlamydia remained a significant predictor of being seropositive in men (AOR 3.55, 95%CI 1.53-8.25).

Among Pgp3 seropositive individuals, 84.7% (80.3%-88.3%) of 16 to 44 yearolds did not report a previous chlamydia diagnosis. This proportion varied by age group, whereby 76.7% of 16 to 24 year-olds, 79.0% of 25 to 34 year-olds and 92.9% of 35-44 year-olds did not report a previous diagnosis. Furthermore, among 16 to 24 year-olds 36.3% of seropositive individuals reported having been tested for chlamydia, but had never been diagnosed. Among those aged 16 to 24 years who had any evidence of a previous *C. trachomatis* infection (i.e. those who were Pgp3 seropositive or who reported a previous chlamydia diagnosis), 75.5% (95%CI 70.8%-79.7%) did not report a previous diagnosis. Table 8-5a: Percentage Pgp3 seropositive by sociodemographic and sexual behavioural variables, and unadjusted and lifetime sexual partners-adjusted odds ratios for detection of Pgp3 antibody in serum (sexually-experienced 16 to 44 year-old <u>women</u>, HSE2010 & HSE2012)

	Percent Pgp3		Unadjusted OR		Lifetime S		ifetime SP-	р	Bases
	posi	tive (95%CI)		(95%CI)	p(uOR)	a	(95%CI)	(aOR)	(W,UW) <sup>a</sup>
Overall	25.8%	(23.1-28.6)					(00,00)		1423, 1194
Age group									
16-19	6.3%	(2.3-16.4)	1.00	-	0.002	1.00	-	0.009	114, 55
20-24	21.4%	(15.3-29.2)	4.05	(1.29-12.78)		3.70	(1.17-11.71)		255, 139
25-29	23.3%	(17.1-30.8)	4.50	(1.44-14.07)		3.92	(1.26-12.16)		229, 166
30-34	33.5%	(27.5-40.2)	7.50	(2.50-22.54)		6.34	(2.11-19.04)		284, 224
35-39	29.5%	(24.2-35.6)	6.23	(2.08-18.65)		5.07	(1.71-15.06)		240, 278
40-44	28.5%	(23.2-34.3)	5.91	(1.99-17.60)		5.47	(1.84-16.26)		302, 332
IMD quintile of LSOA of residence <sup>b</sup>		· · · ·		· · · ·			· · · ·		
Least deprived	22.7%	(17.1-29.5)	1.00	Ref	0.033	1.00	Ref	0.037	256, 238
2	25.8%	(20.3-32.2)	1.18	(0.74-1.89)		1.08	(0.67-1.73)		292, 248
3	19.6%	(15.1-25.2)	0.83	(0.51-1.35)		0.83	(0.50-1.38)		309, 256
4	32.1%	(26.1-38.8)	1.61	(1.02-2.53)		1.61	(1.00-2.61)		290, 230
Most deprived	28.9%	(22.7-36.0)	1.38	(0.86-2.24)		1.46	(0.87-2.43)		276, 222
Age at first heterosexual sex		, ,		. ,			,		
16+ at first intercourse	22.5%	(19.6-25.7)	1.00	Ref	0.000	1.00	Ref	0.004	1080, 911
<16 at first intercourse	39.6%	(33.0-46.6)	2.26	(1.61-3.16)		1.68	(1.18-2.39)		304, 246
Years since first heterosexual									
sex									
0 to 4	8.1%	(3.8-16.7)	1.00	-	<0.001	1.00	Ref	<0.001	184, 90
5 to 9	18.5%	(13.1-25.4)	2.56	(1.01-6.49)		2.05	(0.80-5.20)		289, 174
10 to 14	25.9%	(20.1-32.8)	3.95	(1.61-9.69)		2.79	(1.14-6.79)		256, 199
15 to 19	38.2%	(32.0-44.7)	6.96	(2.98-16.26)		5.40	(2.31-12.62)		271, 264
20+	31.1%	(26.5-36.0)	5.08	(2.19-11.77)		3.61	(1.55-8.37)		415, 461
Number of partners of the									
opposite sex in last year									
0	19.2%	(11.7-30.0)	1.00	Ref	0.370	1.00	Ref	0.028	91, 81
1	26.8%	(23.8-30.0)	1.53	(0.84-2.79)		1.58	(0.80-3.12)		1159, 993
2+	25.7%	(17.9-35.4)	1.45	(0.69-3.06)		0.89	(0.41-1.94)		149, 103
Number of lifetime sexual									
partners	40 50/	(40.0.00.0)	4		0.004				
	10.5%	(13.2-20.3)	1		<0.001				
5 to 9	30.0%	(25.0-35.6)	2.18	(1.51-3.14)					
Condom use in last 4 weeks	43.1%	(30.0-49.0)	3.04	(2.00-5.51)					
Used on every occasion	15 8%	(11 /-21 6)	1 00	Rof	0.001	1 00	Pof	0.001	262 207
Used on some occasions	10.0%	(11.4-21.0) (12.7-28.5)	1.00	(0.68-2.41)	0.001	1.00	(0.57-2.00)	0.001	133 95
Not used in last 4 weeks	30.1%	(12.7 - 20.3) (26.4 - 34.1)	2 30	(0.00-2.41) (1.49-3.53)		2 17	(0.37 - 2.03) (1.41 - 3.35)		735 630
Not had vaginal or anal sex in	50.170	(20.4 04.1)	2.00	(1.45 5.55)		2.17	(1.41-0.00)		700,000
last 4 weeks	26.8%	(20 5-34 3)	1 95	(1 15-3 31)		1 70	(1 03-3 11)		239 209
Ever diagnosed with	20.070	(20.0-04.0)	1.30	(1.10-0.01)		1.19	(1.00-0.11)		200, 200
chlamvdia?									
No	23.6%	(20.9-26.7)	1.00	-	< 0.001	1.00	Ref	< 0.001	1145, 960
Yes	65.5%	(52.7-76.3)	6.12	(3.57-10.51)		5.08	(2.79-9.23)		76, 68

 
 Table 8-5b: Percentage Pgp3 seropositive by sociodemographic and sexual
 behavioural variables, and unadjusted and lifetime sexual partners-adjusted odds ratios for detection of Pgp3 antibody in serum (sexually-experienced 16 to 44 year-old men, HSE2010 & HSE2012)

	Percent Pgp3 positive (95%CI)		Unadjusted OR (95%Cl)		p(uOR)	Li ac	Lifetime SP- adjusted OR (95%CI)		Bases (W,UW) <sup>a</sup>
Overall	14.6%	(12.3-17.1)					(		1424, 948
Age group									
16-19	4.1%	(1.0-15.0)	1.00	-	0.113	1.00	-	0.385	142, 66
20-24	10.1%	(5.8-17.0)	2.66	(0.57-12.40)		1.74	(0.35-8.61)		220, 114
25-29	14.3%	(9.2-21.5)	3.94	(0.88-17.68)		2.32	(0.48-11.20)		255, 139
30-34	18.1%	(12.6-25.3)	5.22	(1.19-23.00)		3.21	(0.70-14.76)		251, 165
35-39	18.7%	(13.4-25.6)	5.45	(1.24-23.96)		3.22	(0.70-14.76)		242, 214
40-44	16.7%	(12.4-22.1)	4.74	(1.09-20.65)		2.74	(0.60-12.62)		314, 250
IMD quintile of LSOA of residence <sup>b</sup>		· · · ·		<b>、</b>			· · ·		
Least deprived	13.7%	(8.9-20.5)	1.00	Ref	0.680	1.00	Ref	0.795	241, 184
2	12.4%	(8.5-17.8)	0.90	(0.47-1.72)		0.98	(0.50-1.92)		283, 195
3	14.8%	(10.0-21.3)	1.10	(0.56-2.14)		1.31	(0.65-2.61)		304, 199
4	13.8%	(9.5-19.6)	1.01	(0.54-1.88)		0.90	(0.44-1.82)		285, 179
Most deprived	17.7%	(12.7-24.0)	1.36	(0.73-2.52)		1.22	(0.61-2.45)		310, 191
Age at first heterosexual sex									
16+ at first intercourse	12.0%	(9.6-15.0)	1.00	Ref	0.002	1.00	Ref	0.233	980, 658
<16 at first intercourse	22.6%	(17.5-28.8)	2.14	(1.41-3.24)		1.50	(0.94-02.38)		357, 231
Years since first heterosexual									
sex									
0 to 4	3.5%	(1.1-10.5)	1.00	Ref	0.003	1.00	-	0.077	230, 109
5 to 9	10.1%	(6.2-16.2)	3.11	(0.86-11.18)		1.78	(0.47-6.82)		283, 154
10 to 14	16.3%	(10.7-24.0)	5.36	(1.51-19.03)		3.12	(0.85-11.42)		236, 148
15 to 19	22.3%	(16.2-29.8)	7.91	(2.31-27.02)		4.10	(1.15-14.66)		249, 186
20+	18.0%	(14.1-22.6)	6.03	(1.84-19.76)		3.04	(0.90-10.33)		409, 339
Number of partners of the									
	10 /0/	(2 0 25 5)	1 00	Pof	0 7 1 1	1 00		0 6 9 7	100 59
1	10.4%	(3.0-23.3)	1.00		0.741	1.00	-	0.007	1050 7/5
1	14.5%	(12.1-17.4)	1.40	(0.49 - 4.33)		1.02	(0.43 - 5.36)		215 115
Total number of lifetime sexual partners <sup>c</sup>	13.0%	(10.2-23.1)	1.59	(0.49-5.16)		1.20	(0.32-5.06)		210, 110
1 to 4	5.9%	(3.7-9.2)	1.00	Ref	<0.001				575, 361
5 to 9	9.9%	(6.6-14.4)	1.74	(0.90-3.33)					322, 223
10+	27.2%	(22.1-33.1)	5.95	(3.41-10.35)					446, 306
Condom use in last 4 weeks		(		( ,					
Used on every occasion	10.8%	(7.0-16.3)	1.00	Ref	0.046	1.00	-	0.272	285, 181
Used on some occasions	13.0%	(7.6-21.2)	1.23	(0.58-2.61)		1.24	(0.54-2.82)		173, 98
Not used in last 4 weeks	17.6%	(14.3-21.4)	1.76	(1.04-3.00)		1.51	(0.83-2.72)		657, 479
Not had vaginal or anal sex in		· · ·		,			· · ·		
last 4 weeks	9.7%	(5.7-15.8)	0.88	(0.43-1.81)		0.86	(0.39-1.90)		246, 151
Ever diagnosed with chlamydia?									
No	12.8%	(10.6-15.4)	1.00	Ref	<0.001	1.00	Ref	0.003	1145, 776
Yes	44.6%	(26.9-63.8)	5.47	(2.47-12.15)		3.55	(1.53-8.25)		62, 35

<sup>a</sup>Denominator totals vary due to item-missingness & include individuals where the specified characteristic is unknown <sup>b</sup>IMD: index of multiple deprivation; LSOA: Lower super output area

<sup>c</sup>Includes partners of both the opposite and of the same gender.

### 8.4.2 Pgp3 seroprevalence over time among 16 to 24 year-old women

Figure 8-5 shows Pgp3 seroprevalence among 16 to 24 year-old women in each year sampled. There was no significant difference in Pgp3 seroprevalence between the first (1994 to 1996) and second (2001 to 2002) time-periods sampled. Between 2008 and 2012, there was a non-significant decline in Pgp3 seroprevalence among 16 to 24 year-old women (prevalence ratio per year: 0.94, 95%CI 0.84-1.05; p=0.26). After stratifying by age group, there was no notable trend among 16 to 19 year-olds (prevalence ratio: 0.96, 95%CI 0.74-1.24; p=0.76), and a non-significant decline in Pgp3 seroprevalence among 20 to 24 year-olds (prevalence ratio: 0.92, 95%CI: 0.83-1.04; p=0.18, (Figure 8-6).

## Figure 8-5: Pgp3 seroprevalence by year (16 to 24 year-old women, HSE 1994 to 2012)





## Figure 8-6: Pgp3 seroprevalence by year and age group (16 to 24 year-old women, HSE 2008 to 2012)

Solid lines show point estimates; dashed lines show 95% confidence intervals. Unweighted denominators: 16-19 year-olds, n=284; 20-24 year-olds, n=425.



Figure 8-7 shows Pgp3 seroprevalence by year of age and birth cohort defined by exposure to widespread chlamydia screening. Although data were only available on those aged up to 19 years in the group who had high exposure to screening (due to the number of HSE survey years available since 2008), there was no indication of a difference in the age-specific seroprevalence by birth cohort, with similar age curves seen in each group. Figure 8-7: Pgp3 seroprevalence by birth cohort and year of age (16 to 24 yearold women, HSE 1994 to 2012)\*



\*Unweighted denominators: High (born 1992-1996), n=185; Partial (born 1984-1991), n=853; Limited (born 1976-1983), n=1349; Limited (born 1966-1975), n=929. See Table 8-3 for details of denominators by year of age.

## 8.5 Discussion

#### 8.5.1 Key findings

In 2010/12, one quarter of 16 to 44 year-old and one in three 30 to 34 year-old women had evidence of a previous chlamydia infection as indicated by the presence of detectable Pgp3 antibodies in serum. Being Pgp3 seropositive was strongly associated with increasing age, years since first sex and numbers of lifetime sexual partners, as would be expected from a marker of previous infection with a sexually-transmitted infection. Three quarters of under-25 year-olds with evidence of previous infection did not report a previous chlamydia diagnosis, suggesting a high level of undiagnosed infections. There was no

significant trend in age-specific Pgp3 seroprevalence over time or between birth cohorts exposed to different levels of opportunistic chlamydia screening.

The proportion of the population who acquire chlamydia infection by their mid-40s is much greater than suggested by prevalence studies of *current* infection as measured by NAATs. For example in Natsal-3, which was conducted at a similar time, the prevalence of current chlamydia infection was estimated to be 1.5% in women and 1.1% in men aged 16 to 44 years.<sup>23</sup> Studies that focus on current infection underestimate the lifetime risk of chlamydia infection.

### 8.5.2 Strengths and limitations

The major strengths of this analysis are that it used data and stored sera from a series of nationally-representative samples and applied novel serological tests to investigate age-specific cumulative incidence of chlamydia. For HSE2010/12 participants Pgp3 seroprevalence could be investigated by sociodemographic and behavioural factors at an individual level. The assays used in the testing strategy have demonstrated higher sensitivity than commercially-available assays.<sup>209,269</sup>

There were some limitations. The findings may be affected by who agreed to take part in the survey or provide a blood sample. Non-response weights were applied to account for non-participation in the overall survey and provision of a blood sample. HSE2010/12 participants who contributed to this study were comparable to the overall HSE population on a range of sociodemographic and variables, however unmeasured bias remains feasible. A further limitation is that all behavioural data were self-reported. Sensitive items were collected using the self-completion booklet to minimise social desirability bias. However, as

questions were completed using a booklet rather than the best-practice of computer-assisted self-interview and as household members could be present during booklet completion, underreporting remains possible.<sup>272</sup>

Estimated sensitivity of both the indirect and the double-antigen ELISAs to detect a previous chlamydia infection is less than 100% (e.g. 74% and 83%) respectively in women). The percent Pgp3 seropositive among individuals who had ever been diagnosed with chlamydia was comparable to the reported sensitivity of the indirect, but slightly lower than that of the double-antigen ELISA. Pgp3 seroprevalence will therefore be a biased estimator of age-specific cumulative incidence of *C. trachomatis* due to misclassification arising from test performance (i.e. misclassification of presence/absence of antibodies) or patterns of seroconversion and subsequent loss of antibodies (i.e. absence of antibodies does not always reflect an absence of previous infection). Evidence from a study by Horner et al suggests that anti-Pgp3 antibodies are more likely to be detectable in the first few weeks/months of an infection and are more likely to remain detectable in women with repeat versus those with first infections.<sup>100</sup> In that study, sera from women attending two GUM clinics with at least one known previous chlamydia diagnosis were tested using the indirect Pgp3 ELISA. The percentage Pgp3 seropositive declined with increasing time since last detection up to ~6 months and remained relatively stable thereafter. The percent seropositive in women with more than one known previous chlamydia diagnosis remained higher for a longer period than in women with only one previous diagnosis. Thus interpreting Pgp3 seroprevalence as a measure of age-specific cumulative incidence of C. trachomatis infection will be complicated by test performance, time since infection and patterns of reinfection. Nevertheless, Pgp3 seroprevalence represents a lower bound of age-

specific cumulative incidence; the proportion of the population ever infected with *C. trachomatis* would be even higher than estimated Pgp3 seroprevalence presented in this chapter. While this misclassification error is important to understand and ideally quantify, measurement of Pgp3 seroprevalence is a promising approach for surveillance and for impact evaluation of chlamydia control programmes given the ability to measure minimum age-specific cumulative incidence.

Pgp3 antibodies were detected among people who had never had sexual intercourse. This may be due to under-reporting of sexual experience or to false positives that would be expected from low prevalence populations. Interestingly, the percentage Pgp3 seropositive among those with no lifetime sexual partners when measured on the indirect ELISA only was higher than when estimated using the testing strategy of combined indirect and double-antigen Pgp3 ELISAs (5.7% versus 4.0%). This may suggest a higher specificity of the double-antigen ELISA and supports its use within the testing strategy.

Interpreting Pgp3 seroprevalence patterns among men is complicated by lower test sensitivity to detect a previous infection, which may reflect differences in immune response following infection by gender and site of infection.<sup>209</sup> The lower Pgp3 seroprevalence also limits the power to investigate risk factors for Pgp3 seropositivity given the smaller number of men with detectable antibodies. The more gradual increase in Pgp3 seroprevalence with age in men may partly reflect age-mixing patterns whereby women tend to have older male sexual partners. Thus as the prevalence of infection in men is higher in those aged 20 to 24 than in 16 to 19 year-olds,<sup>23</sup> women may be at higher risk of incident infection at sexual debut than men. However, the more gradual increase in men

might also reflect the importance of repeated exposure in developing a persistent, detectable Pgp3 antibody response.

#### 8.5.3 Comparison to other studies

STI incidence is thought to have been low in the early 1990s following changes in sexual risk-taking behaviour after sexual health campaigns aimed at HIV prevention in the early- to mid-1980s.<sup>273</sup> From the mid-1990s there were increases in reports of bacterial STI<sup>273</sup> and sexual risk behaviours increased between 1990 and 2000,<sup>235</sup> suggesting a likely rise in STIs. It might therefore be expected that Pgp3 seroprevalence would have increased between the mid-1990s and early 2000s, but there was no significant difference between these periods. One possible explanation for this is that STI incidence had already started to increase by the mid-1990s. Chlamydia incidence may also have been less affected by changes in sexual behaviour over the period as it is more widely distributed in the population than STI such as gonorrhoea or syphilis<sup>23</sup>. It is also feasible that participation bias not reflected in the survey weights changed over this period.

More interestingly for the evaluation of chlamydia screening is that no significant change was found in age-specific seroprevalence between 2008 and 2010 and there was no difference between birth cohorts exposed to different levels of opportunistic chlamydia screening. This is perhaps surprising given the increase in chlamydia control efforts over the last decade. Possible reasons for this and comparisons with findings from other studies are discussed in detail in the following chapter.

In this analysis, women with greater numbers of sexual partners and who reported younger age at first sex were more likely to have evidence of a previous infection. This is consistent with findings from a cohort study of women in New Zealand whose sera were tested using the double-antigen ELISA, among whom lifetime sexual partners and age at first sex were both associated with being seropositive.<sup>269</sup> The steep increase in seroprevalence from age 16 to age 25 also suggests a particularly high rate of infection among young women soon after commencing sexual debut. Based on the observed seroprevalence at 10-11 years after first heterosexual sex (31%), the average annual incidence of chlamydia infection in women can be estimated to be at least 3% per year in the ten years following sexual debut. This will be an underestimate of the overall rate of transmission, as repeat infections are common<sup>48-57</sup> but cannot be quantified from seroprevalence data. This emphasises the importance of adolescence and young adulthood as periods when chlamydia prevention and screening activities are needed to prevent development of adverse consequences.

Pgp3 seroprevalence estimated among HSE participants was higher than that reported from a recent population-based study in the Netherlands reported by Van Aar et al, which used the Medac CT IgG ELISA (9.8% in women and 5.7% in men aged 15 to  $39^{132}$ ). Van Aar *et al* also found a less marked relationship with age. This may be due to the difference in assay sensitivity (Medac: 46-2% versus double-antigen ELISA: 82-9%),<sup>269</sup> or to country differences in the epidemiology of chlamydia.

## 8.5.4 Implications for evaluation of chlamydia control

The clear relationship between Pgp3 seroprevalence and number of lifetime sexual partners, especially among women, further supports the use of age-specific seroprevalence as a marker of previous infection. The insight gained from these results demonstrates the strengths of this approach to outcome measurement.

The findings from this analysis do not support there having been a decrease in age-specific cumulative incidence among 16 to 24 year-old women following the implementation and national roll-out of the NCSP. As discussed in more detail in the next section, the evidence for high lifetime cumulative incidence and suggestion of high fraction of undiagnosed infections may offer a partial explanation for the absence of a decline in age-specific cumulative incidence.

## 8.5.5 Summary

A decrease in age-specific cumulative incidence of chlamydia following the national implementation of opportunistic chlamydia screening has not yet been demonstrated. The implications of these findings and possible reasons for an apparent absence of any decline are explored in detail in the next chapter.

## 9 Summary and discussion

In the previous five chapters I examined four different approaches to outcome measurement for the evaluation of chlamydia screening. In this chapter I discuss the meaning of my findings in the context of the original aims of the thesis, which were:

1. to identify and appraise outcome measures, and methods of their measurement, for the purpose of evaluating the impact of opportunistic chlamydia screening on the incidence and prevalence of infection (or related measures); and

2. using the outcome measures and methods identified in (1), to examine whether widespread opportunistic chlamydia screening, as it has been delivered in practice, has led to a reduction in the incidence or prevalence of chlamydia among young adults in England up to 2012 that would otherwise have been seen in the absence of opportunistic chlamydia screening.

I make recommendations about whether and how this range of methods should be incorporated into ongoing evaluation of chlamydia screening and discuss the implications of my findings for the future development of chlamydia control policies in England. Finally, I set out suggestions for future research given the questions raised by my research that, as yet, remain unanswered. 9.1 Critical appraisal of outcome measures for the evaluation of the impact of chlamydia screening on incidence and prevalence of infection

## 9.1.1 Trends in percentage testing positive for chlamydia using surveillance data

As discussed in Chapter 3, it is well established that, as a large proportion of chlamydia infections are asymptomatic<sup>10,11,20</sup> the number of chlamydia diagnoses reported and proportion of tests which are positive are highly dependent on the population tested. Thus reported diagnoses do not represent the true burden of incident infections and percentage testing positive does not equate to population prevalence.<sup>201</sup>

My analysis using surveillance data from populations accessing chlamydia testing showed that the percentage testing positive varies according to testing venue type, and that significant differences between venues in the odds of testing positive remained even after adjustment for sexual behaviour variables. This suggests that the sexual behaviour and demographic data items that were available in the NCSP dataset could not fully explain the difference in risk of having a current chlamydia infection between populations attending different services. Given that the available sexual and behavioural variables were not sufficient to capture the difference in risk of infection between people attending different venues, it is also possible that these variables would be insufficient to capture difference in risk arising from a different profile of people accessing testing over time, even in analyses stratified by testing venue type. This potential for unmeasured confounding would be especially true in the presence of changes in testing policy, for example if a venue were to move from universal to targeted testing. Thus trends in percentage testing positive from populations

accessing chlamydia testing do not provide definitive measures of change in the underlying population prevalence of infection.

As NAATs have been widely available since at least 2006 in the UK,<sup>128</sup> then there will have been minimal measurement error in the trends in percentage testing positive introduced by detection strategies used. However it is important to remain vigilant to changes in test technology that might affect surveillance data as changes in tests used have repeatedly been shown to affect observed trends in percentage testing positive for chlamydia.<sup>192,200,203,204,274</sup> Future developments in test technologies and testing pathways including point of care testing<sup>275</sup> and remote testing via mobile devices<sup>276</sup> are on the horizon and may in the future present a shift in test performance that would be reflected in cases reported and estimates of percentage testing positive. Maintaining consistency in test technologies is beneficial to permit comparisons over time, but the need to deliver the best diagnostic services means that changes will likely occur. Future analysis of longitudinal trends should therefore consider changes in test sensitivity and specificity. As with changes in testing strategies within the Natsal surveys (Chapter 6), head to head comparisons of different test technologies used in routine practice as developments occur will be useful to provide data for comparisons over time.

It was not possible to measure uptake of testing (i.e. the percentage of eligible individuals attending services who were tested) among those attending all venues. This would have been possible with data from GUM clinics as a clinicattending denominator is available. However as the offer and uptake of a chlamydia test is less likely to vary in GUM settings this was not investigated in the analysis presented in Chapter 4. Developments in the GUMCADv2

surveillance system, whereby comparable information to that collected in GUM clinics is now collected for all commissioned 'level 2' (i.e. non-GUM) sexual health services, mean that a measure of uptake of testing within sexual and reproductive health services and some GP clinics will be available in coming years. The extent to which percentage testing positive varies with test uptake within a setting would be useful to explore in future analyses.

Some of the issues of using surveillance data may be addressed if chlamydia screening were to be delivered as a register-based screening programme, where a defined population were invited to participate. By standardising the offer of screening, this might make the resulting data less susceptible to confounding arising from differences in who is offered a test. However, even register-based screening programmes face challenges in reliable outcome measurement. As described in Chapter 3, the CSI project in the Netherlands was a register-based programme and invited all 16 to 29 year-olds resident in three areas of the Netherlands to be tested for chlamydia once a year for three years. In their write-up of findings from the CSI project, van den Broek et al noted that the population tested each year appeared to be at higher risk, on average, in each subsequent year.<sup>88</sup> Thus even in the context of a RCT of register-based screening the study's findings remain open to question due to the potential for unmeasured confounding. In another (ongoing) RCT in Australia, the Australian Chlamydia Control Effectiveness Pilot 'ACCEPt' aims to assess the feasibility, acceptability and effectiveness of chlamydia testing in GP settings.<sup>277</sup> The primary outcome of the trial is chlamydia 'prevalence' measured among women and men attending GP services. However, as this is based on a clinic-attending population this would be more akin to percentage testing positive according to the definitions applied throughout this thesis.

Prevalence will be compared between clinics in the intervention arm where clinics have been asked to offer chlamydia testing to all sexually active under-30 year-olds to that measured in control clinics. While the baseline prevalence survey achieved high response rate (69.7%),<sup>278</sup> participants were provided with their results. It is therefore feasible that follow up collections might be affected by previous offers of testing meaning that participation bias may vary between the intervention and control arms. While the results of this study are not yet available, this further illustrates the potential limitations of using percentage testing positive as an outcome measure even when utilised as part of a RCT.

In summary, trends in percentage testing positive should not be interpreted as trend in underlying burden of disease without first ruling out or addressing sources of selection bias and confounders affecting trends over time.

## 9.1.2 Repeat cross-sectional anonymous postal surveys of chlamydia prevalence

As described in Chapter 4, the pilot of a postal survey of young women with anonymous testing for chlamydia resulted in low participation rates. Both provision of a home sampling kit and offering a small financial incentive increased participation, and reduced the cost per sample received. Although the sample size in the pilot was not sufficient to provide strong evidence for, or to rule out selection bias, given the low participation rates achieved, future surveys using this methodology would be open to substantial and potentially varying selection bias. There was some indication that demographic characteristics varied between the groups, suggesting the very real potential for selection bias using this methodology. The piloted approach is therefore not feasible or

suitable for population-based monitoring of chlamydia prevalence among young women in England.

The maximum achieved participation rate of 14% (in the group with the voucher) was comparable with other population based approaches using postal invitation for chlamydia testing. In a pilot of a postal offer of chlamydia screening, Bracebridge *et al* achieved a participation rate of 13% among women aged 18 to 24.<sup>158</sup> In the Netherlands, the CSI project achieved a 21% participation rate among 15 to 29 year-old women in year one, which fell to 13% in year 3.<sup>88</sup> Studies such as Natsal and NHANES, which invited participants to provide a sample during a face to face interview, have achieved response rates of upwards of 50%.<sup>141,172</sup> This suggests that high response rates can be achieved, but that a far more resource-intensive recruitment approach would be needed. Additionally, surveys with a broader focus than chlamydia prevalence may be of more interest to the eligible participants meaning a larger and potentially more representative population could be recruited.

## 9.1.3 Repeat cross-sectional surveys of chlamydia prevalence embedded within sexual behaviour or general health surveys

The Natsal surveys have achieved much higher response rates than were observed in my pilot of a chlamydia-specific postal survey of chlamydia prevalence.<sup>172</sup> As with other probability surveys among general population samples,<sup>7,135,141</sup> participation in Natsal is far from universal and there therefore remains some, theoretical, potential for non-participation bias. However these surveys are far more robust than postal surveys for chlamydia with regard to representativeness of the population.

My analyses presented in chapter 6 showed that, while changing approaches to prevalence measurement pose challenges for comparisons over time, adjustments can be made to allow for differences in diagnostic tests used in each survey where the relative sensitivity and specificity of tests is known. The adjustments for detection strategy were made using data from two separate studies where the tests and specimen collection devices used in Natsal-2 had been compared to those used in Natsal-3. As estimated sensitivity and specificity are dependent on the gold standard used, such head to head comparisons are invaluable. Given ongoing technological developments it is possible that tests for chlamydia (and other infections) used in any subsequent Natsal survey may differ from those in Natsal-3. In anticipation of this, maintaining a panel of residual biological samples for the purposes of validating using any future tests will be useful to strengthen the possibility for future comparisons.

While adjustments for different detection strategies could be made, the comparisons between age- and gender-specific measures of chlamydia prevalence were ultimately limited by the sample size available in each survey. The Natsal-2 and Natsal-3 surveys were not powered to detect changes in chlamydia prevalence, so it is not perhaps surprising that these comparisons were limited by the number of participants in specific groups. Other surveys have also been limited by the sample size available. In NHANES in the US, estimates by age, gender and race/ethnicity become somewhat unstable due to the small denominators in some cases, and small numerators in groups with very low measured prevalences.<sup>141</sup>

Given the limitations in sample size and survey methodologies, the real strength of the Natsal surveys for understanding effectiveness of chlamydia screening was in providing a source of data where chlamydia infection, testing and diagnoses within the same population could be investigated incorporating individual-level data on both outcomes and demographic and behavioural details. The implications of these findings for chlamydia control are discussed below (section 0). In summary, the combination of data available made it possible to investigate how chlamydia screening was being implemented in relation to risk, to identify potentially unmet need and to hypothesise why chlamydia screening up to 2012 may not have had a substantial impact on chlamydia incidence or prevalence. I therefore propose that future use of prevalence surveys should continue to assess prevalence but do so in the context of screening and sexual behaviours rather than focus on direct comparisons of prevalence alone.

Natsal has been conducted decennially since 1990. HSE is conducted every year, and therefore presents an opportunity for more regular monitoring of chlamydia-related outcomes. As described in Chapter 8 HSE successfully collected data on sexual behaviour in 2010 and 2012. To date, it has not collected data on prevalent chlamydia infections. However this may be feasible, as several biological measures are taken, including an array of tests using urine specimens.<sup>7</sup> Prah *et al* have shown that some sexual risk behaviours and STI-related factors were less-commonly reported in HSE2010 than in Natsal-3. This is likely due to differences in the administration of the surveys (household versus personal interviews) and to differences in participation in general health versus sexual attitude and lifestyle-specific surveys.<sup>272</sup> However survey-derived estimates are broadly consistent, which supports the use of both sources for

future monitoring of sexual health parameters including chlamydia prevalence, history of chlamydia testing and diagnosis and uptake of chlamydia screening. Given the major limitations due to available sample size in comparisons between Natsal-2 and Natsal-3, future surveys would also benefit from a larger sample size to allow sufficient power for comparison of outcomes across repeated surveys.

## 9.1.4 *C. trachomatis* antibody seroprevalence as a marker of agespecific cumulative incidence

The use of C. trachomatis antibody seroprevalence as an outcome measure for evaluating chlamydia screening faces some of the same challenges as surveys that measure prevalence of current infection and surveillance data, in that bias may arise from the population sampled and changes in the population who contribute serum samples for testing may lead to confounding of the relationship between testing antibody seropositive and time. The analysis presented in Chapter 8 used data from a nationally-representative probability survey with standardised data collection over time in order to minimise the impact of these potential problems. However separating the effects of sexual behaviour from those of increased access to chlamydia screening remains an issue when looking at antibody seroprevalence. Sexual behaviour data were not available in all years in the Health Survey for England thus it was not possible to adjust for sexual behaviour as a potential confounder of the relationship between seroprevalence and exposure to chlamydia screening, which remains a limitation of this analysis. Future serum collections should prioritise the collection of sexual behaviour data to allow more detailed analysis of trends over time in relation to factors that may influence cumulative incidence of infection.

Another problem in common with surveys of chlamydia prevalence is that of power. Although antibody seroprevalence is several-fold higher than that of prevalence of current chlamydia infection, the analysis presented in Chapter 8was restricted to the number of serum samples that were available from previous HSE surveys and may have limited the ability of the analysis to detect a significant decline in Pgp3 seroprevalence among 16 to 24 year-olds. This is explored further in Table 9-1 below, which shows the significance of the prevalence ratio that would have been observed had the same point estimates been observed in each year, but a larger sample size had been recruited. This suggests that a sample size between two to three times larger in each year may have resulted in a conclusion of there having been a statistically significant (at p<0.05 level) decline in Pqp3 seroprevalence among 16 to 24 year-old women between 2008 and 2012. However, although power may be an issue, it is important to note that a larger sample would not necessarily have achieved the same point estimates and that the downward (non-significant) trend in seroprevalence measured in HSE participants was especially driven by seroprevalence among 20 to 24 year-olds in 2012 (Figure 8-6). This may be an early indication of falling seroprevalence, but additional years of data would be needed to investigate this further.

# Table 9-1: Hypothetical scenario showing point estimates, 95%CI and p values for prevalence ratio of Pgp3 seroprevalence achieved with different sample sizes (16 to 24 year-old women, HSE2008 to HSE2012)

Prevalence ratios, p values and 95%Cl are based on the actual sample size available (as presented in section 8.4.2) and three hypothetical scenarios based on two to four times the number of participants recruited (maintaining the balance of positive and negative results).

Point estimate	Samp	le size	95%CI	P value		
(prevalence ratio	In relation to	Total N				
per year)	actual					
0.94	Actual size	707	0.85-1.04	0.250		
0.94	x 2	1414	0.87-1.01	0.103		
0.94	х З	2121	0.88-1.00	0.046		
0.94	x 4	2828	0.89-0.99	0.021		

Data from the analysis of HSE participants presented in this paper, and the two other analyses using the Pgp3 indirect ELISA among residual sera from women attending GUM clinics<sup>214</sup> and those submitting sera for routine investigations<sup>213</sup> now provide a wealth of data to indicate the seroprevalence that would be expected in future studies in different settings, which can be used to power future surveys or collections of residual sera.

Measurement error remains a major limitation for using *C. trachomatis* antibody seroprevalence as an outcome measure for evaluating chlamydia control programmes. While the double-antigen Pgp3 ELISA demonstrates higher sensitivity in women than the indirect Pgp3 ELISA and other commercial assays, estimated sensitivity to detect a previous known infection remains well under 100% at 82.9%.<sup>269</sup> Thus Pgp3 seroprevalence remains an imperfect measure of cumulative incidence of infection by any given age. It is unclear whether this imperfect sensitivity is due in part to some women not developing serum antibodies following genital infection with C. trachomatis, or only to waning of antibodies over time and/or assay performance, although a combination of these factors is likely. Waning of antibodies has been demonstrated using the indirect Pgp3 ELISA<sup>100</sup> and other commercial assays.<sup>210</sup> The ability of the double-antigen Pgp3 ELISA to detect antibodies appears to be less affected by time since infection.<sup>269</sup> However, further work is needed to better quantify the relationship between having detectable antibodies and time since infection, to allow future estimates using *C. trachomatis* antibody assays to adjust for this measurement error.

A further limitation with *C. trachomatis* antibody seroprevalence assays is that repeat infections cannot be distinguished from first infections. As is discussed in

more detail below, re-infection may play an important part in preventing chlamydia screening from having a substantial impact on chlamydia prevalence. Being able to measure rates of re-infection among the general population would therefore be informative. However it is not yet possible to use serological studies to measure rates of re-infection.

Among men, my analysis of HSE2010/12 participants demonstrated that the percentage Pgp3 seropositive correlated with numbers of sexual partners, as would be expected as a marker of previous infection with a STI. However, the lower sensitivity of the assay to detect a previous infection among men limits the use of Pgp3 seroprevalence as an outcome measure for the evaluation of chlamydia control. The extent to which this lower sensitivity reflects a difference in the immune response to *C. trachomatis* infection between women and men or by anatomical site of infection or differences in antibody waning by sex is unclear. However, as discussed in section 9.3 below, as men have an important part to play in the transmission of infection and in understanding the challenge of controlling the incidence and prevalence of chlamydia, a more detailed understanding of the relationship between *C. trachomatis* infection, antibody response and protective immunity in men is warranted.

Despite these limitations, the application of the Pgp3 indirect and double antigen ELISAs present a powerful tool for the evaluation of chlamydia screening programmes in England and elsewhere. As with data from Natsal-3, the analysis of Pgp3 seroprevalence among HSE2010/12 participants provided important insights into the epidemiology of chlamydia in relation to testing patterns. Specifically, my analysis showed that chlamydia is widespread through the population among women and that a high percentage of infections

had, up to 2012, gone undiagnosed. The use of *C. trachomatis* antibody tests combined with information on history of diagnoses provides a valuable and novel outcome measure, as a high level of undiagnosed infections represents both opportunities for transmission and potential development of complications arising from untreated infections. Future serum collections should incorporate data on history of chlamydia diagnoses wherever feasible.

## 9.2 Summary of recommendations for future evaluation of the impact of chlamydia control programmes in England

In summary, I propose the following recommendations for the ongoing and future evaluation of the impact of chlamydia screening on chlamydia incidence, prevalence and related measures:

- Repeat cross-sectional postal surveys of chlamydia prevalence in the general population are open to bias and are unlikely to provide value for money. They should not be pursued as a means to monitor chlamydia prevalence in England.
- Surveys such as Natsal, which are nationally representative, have relatively
  high response rates and collect data on a range of topics, will present better
  value for money than cross-sectional, chlamydia prevalence-specific postal
  surveys. Analysis of such surveys should focus on assessing prevalence in
  the context of reported screening and diagnoses and sexual behaviours
  rather than on direct comparisons of prevalence alone.
- The potential to incorporate the following measures into the HSE on a routine basis should be considered: current chlamydia infections (as measured using NAAT tests of urogenital specimens); previous infection with *C. trachomatis* (as measured using antibody tests of serum specimens); sexual behaviour and reported chlamydia testing and diagnosis history.
- Boosting the available sample size among young adults in both future rounds of the HSE and Natsal surveys would increase the power available

for comparative analyses and should be considered, using available data to determine the sample size required.

- Trends over time in percentage testing positive for chlamydia among populations accessing testing should not be interpreted as trends in the underlying burden of disease without first ruling out or addressing sources of selection bias and confounders. Future analyses should also consider any changes in test technology that may affect the sensitivity and specificity of tests used.
- Estimates of Pgp3 seroprevalence by birth cohort should be extended. In addition to continued use of sera from HSE participants, routine collections of residual sera from sentinel groups should be pursued, with appropriate consideration of the potential bias in each (e.g. sera collected for routine microbiological investigations, GUM clinic attenders, antenatal populations, blood donors).
- Future serum collections should incorporate measures of previous chlamydia diagnoses and sexual behaviour where possible to allow the measurement of the undiagnosed fraction of *C. trachomatis* infections and to reduce reduce confounding of the relationship between chlamydia control interventions and antibody seroprevalence arising from changes in sexual behaviour over time.

## 9.3 Has chlamydia screening effected a reduction in the incidence or prevalence of chlamydia infection?

I turn now to the second aim of the PhD, which was to address whether chlamydia screening as it has been delivered in practice in England has led to a reduction in the incidence or prevalence of chlamydia among young adults.

There is some evidence from analyses presented in this thesis and from other sources to support there having been a decrease in the transmission of chlamydia infection since the national implementation of the NCSP in 2008. Specifically, my analysis of surveillance data (Chapter 7) showed a consistent pattern of a declining trend in percentage testing positive in most subgroups (with the exception of tests among MSM). There remained a declining trend over time even after adjusting known confounders. This is consistent with the hypothesis that chlamydia prevalence among heterosexual young adults has declined in recent years. However, as discussed above, there was the potential - and some evidence - for unmeasured confounding. It is not, therefore, possible to reach a definitive conclusion of declining prevalence using these data alone. The second piece of supporting evidence comes from the study by Horner *et al*, which estimated Pgp3 seroprevalence (using the indirect Pgp3 ELISA) among 17 to 24 year-old women in England using residual sera submitted for routine microbiological or biochemical investigations,<sup>213</sup> which found a significant decrease in Pgp3 seroprevalence from 20% in 2007 to 15% in 2010 (p<0.001).<sup>213</sup> This observed decline is consistent with there having been a decrease in age-specific cumulative incidence in the years following the national implementation of the NCSP.

However, other analyses from population-based probability samples presented in this thesis do not support there having been a decrease in incidence or prevalence of infection following the national implementation of the NCSP. Firstly, comparisons between population prevalence as measured in Natsal-2 and Natsal-3 (Chapter 6) did not provide evidence to support there having been a decrease in prevalence in the decade between the surveys. There was no significant difference between prevalence in Natsal-3 compared to Natsal-2, although the utility of direct comparisons of age-specific prevalence was ultimately limited by the sample size available. Adjustments to correct for differences between the surveys suggested a trend towards there having been an increase, rather than a decrease in prevalence over the last decade. My

analysis of Natsal-3 data (Chapter 7) also showed evidence of ongoing transmission in 2010-12 as incident infections were found among those recently tested.

Secondly, my analysis using stored sera from participants in the Health Survey for England (Chapter 8) showed that Pgp3 seroprevalence among 16 to 24 year-old women decreased between 2008 and 2012, but that this observed decrease was not statistically significant. Importantly, there was no notable difference in age-specific Pgp3 seroprevalence between birth cohorts exposed to high levels of opportunistic screening and those who became sexually active before widespread screening. This absence of a decline in seroprevalence is in contrast to Horner et als study mentioned above, where a significant decrease in Pgp3 seroprevalence between 2007 and 2010 was found.<sup>213</sup> The difference between my findings from HSE participants and those of Horner et al may be due to the slightly different time-periods investigated or the sources of sera. Horner et al used residual sera submitted to laboratories for routine investigations rather than a probability sample. While such sera are considered broadly representative of the general population with respect to relatively common infections, it is not known whether the data are representative with respect to STI.<sup>213</sup> It is feasible that some bias may have arisen due to possible oversampling of antenatal sera in women of childbearing age and changes in the way that samples from GUM clinics were recorded (and thus excluded) may have affected the trends that were observed, as it is possible that not all GUM samples were excluded from the earlier panels. Another possible reason for difference in results is that my analysis may have been limited by the number of residual sera available. Horner *et al* had a sample size of 2,519 between 2007 and 2011 versus the 707 available from 16 to 24 year-old female HSE

participants between 2008 and 2012. Thus a larger sample size may have resulted in the observed trend being statistically significant.

In summary, based on my analyses and the available literature there is no strong empirical evidence to support the hypothesis that chlamydia screening, as delivered in practice, has led to a reduction in either the incidence or prevalence of chlamydia infection among young adults up to 2012.

Given the increases in screening over the last decade (Chapter 2), this lack of empirical evidence is perhaps surprising. Screening has certainly led to a large number of diagnoses among young adults. Among Natsal-3 participants, around a half of 16 to 24 year-olds diagnosed in the last year were considered to be 'screen-detected' infections as defined by the given reason for test (section 7.4.1.1) Surveillance data show large increases in testing and diagnoses over the decade up to 2012 (Figure 2-5; Figure 2-6). These data also show that ~60% of diagnoses made in 2010 to 2012 among 15-24 year-olds were reported from non-specialist GUM settings,<sup>279</sup> showing the expansion of chlamydia screening in recent years. So long as the detected infections were adequately treated, this increase in detection of chlamydia infections should have reduced their duration (relative to the counterfactual of there having been no national screening programme). As set out in section 2.2, reducing the average duration of infection is expected to reduce the number of transmission events (given the relationship  $R_0=\beta cD$ ) and also decrease the prevalence of infection (given prevalence=incidence x duration). The absence of evidence for any such reductions may be due to the limitations of the specific outcome measures used, as described above (section 9.1). In particular, my analyses using population-based samples may not have had sufficient power to detect

changes over time in the outcome measures of interest. However, if the opportunistic chlamydia screening efforts among young adults had really not effected a reduction in the incidence or prevalence of infection by 2012, this begs the question 'Why not?'

### 9.3.1 Insufficient time since implementation of screening

One possible explanation is that screening might not have been in place for long enough to have had a meaningful effect on transmission dynamics of infection in the population. As described in Chapter 2, opportunistic screening efforts increased substantially between in the first few years after the NCSP had achieved national implementation. Between 2008 and 2010 the estimated percentage of 15 to 24 year-olds tested in each year (assuming one test is equivalent to one person) increased from 26% to 44% in women and 11% to 24% in men.<sup>189</sup> Data up to 2012 therefore represent only three years of opportunistic screening at levels of the current programme. The mathematical model reported by Turner et al that explored the potential impact of chlamydia screening in England on the prevalence of infection demonstrated a reduction could be achieved over ten years after introducing screening (Figure 2-1).<sup>81</sup> so it is perhaps unreasonable to expect there to have been a notable impact on prevalence by 2012. However it should be noted that Turner et als model predicted the greatest declines to be in the first two to three years following screening. The implications of my findings for mathematical modelling are discussed in more detail below.

My analysis of age-specific Pgp3 seroprevalence is perhaps more limited by the available time between national implementation of screening and measurement. The presence or absence of antibodies to infection will depend on the full

history of chlamydia infection up to the time of antibody measurement. Thus exposure to chlamydia screening should be considered from the time of sexual debut. Women who have turned 16 since the national implementation of the NCSP were a maximum of age 20 in the analysis using data up to HSE2012, which limited the comparison of age-specific Pgp3 seroprevalence by birth cohort. Additional years of data are needed to extend these analyses to maintain robust monitoring of chlamydia in coming years.

# 9.3.2 Opportunities for transmission arising from undiagnosed infections and untested groups

An alternative hypothesis is that chlamydia screening up to 2012 was not diagnosing a high enough proportion of infections to lead to substantial interruptions in transmission. My findings from the Pgp3 seroprevalence study showed that the proportion of the population ever infected with chlamydia was much greater than suggested by studies that have measured prevalence of *current* infection. Additionally, screening patterns up to 2012 had left a high proportion of infections undiagnosed, as shown by the high proportion of women and men with evidence of a previous infection who did not report a previous diagnosis (75.5% of 16 to 24 year-old, see section 8.4.1).

Comparative analysis of factors associated with testing and with prevalent infection presented in Chapter 6 also highlighted gaps in testing coverage that present opportunities for ongoing transmission among young adults. Although testing rates were generally higher in those reporting risk factors for chlamydia, in Natsal-3 at least one quarter of women and around half of men reporting a risk factor associated with prevalent infection had not been recently tested. Testing rates were relatively low in men, especially among 20-24 year-olds who contributed the majority of prevalent infections and diagnoses. Therefore the interpretation of existing testing strategies also leaves room for transmission, for example, from men to their (often younger)<sup>268,280</sup> partners and from people with multiple sexual partners who are not accessing testing.

## 9.3.3 Frequency and timing of screening

Testing frequency in relation to time since infection and rate of partner change may also have been insufficient to meaningfully interrupt transmission. The impact of screening on incidence will be diminished if onward transmission has already occurred before being tested, for example if someone has had one or more new sexual partners between the time of infection and testing. Given that in some cases untreated chlamydia infections persist for over a year,<sup>18</sup> if testing is not accessed soon in the course of infection then this presents a potentially long time for an infected individual to be at risk of passing the infection on. This would be especially relevant in those groups reporting high rates of partner change or frequency of non-condom use. This has been explored in the context of partner notification by Althaus et al, who used a mathematical model to explore the effect of notifying partners on reducing transmission, comparing current to previous sexual partners.<sup>117</sup> They found that the longer the look-back period, the more likely it was that infections would already have been passed on, thus models predicted testing of non-current partners to have negligible effect on transmission. This highlights that the timing of testing in relation to time since infection could substantially affect the expected benefits of screening in terms of interrupting transmission.

The NCSP recommends that young adults be screened on change of sexual partner, with the aim of reducing the period between time of infection and time

of treatment. The extent to which this recommendation is emphasised in health promotion activities is unclear, as is our understanding of whether clinicians and young adults interpret this recommendation as a need to test at the end of a relationship or to test soon after having a new sexual partner. However, data from Natsal-3 suggests there is substantial room for improvement. Among sexually-experienced 16 to 24 year-olds in Natsal-3 36% of women and 59% of men with at least one new sexual partner in the last year had not been tested for chlamydia in the last year. Again, this points towards gaps between screening activity and infection risk such that ongoing transmission would be expected.

#### 9.3.4 Re-infection, partner notification and treatment

The precise relationship between chlamydia infection and subsequent immunity is unclear, although studies to date suggest that a previous infection with chlamydia confers, at best, only partial immunity.<sup>47</sup> In a modelling study using data from case rates in British Columbia, Brunham *et al* hypothesised that high levels of testing would lead to higher rates of repeat infections, if treatment prevents development of protective immune response. This notion of 'abrogated immunity' is a contested hypothesis,<sup>47</sup> and immunological evidence for this is yet to be provided.

However, even if treatment does not affect the development of protective immunity, repeat infections may partially account for high diagnosis rates without a corresponding reduction in prevalence. Returning to the relationship between prevalence and incidence, if the average duration of infection has been reduced (by screening and treatment of infection) but the prevalence has remained the same, then mathematically the rate of new infections must have increased, given the relationship prevalence=incidence x duration. However, this does not necessarily mean new infections in previously uninfected individuals. Instead, re-infection of an individual may counteract any reduction in duration, which in turn would limit the impact of screening on the prevalence of infection. Given the acknowledged risk of re-infection, in 2013 the NCSP introduced a recommendation that young adults who test positive for chlamydia be re-tested around three months after treatment.<sup>119</sup>

My analyses and other previously published studies<sup>48-57</sup> demonstrate that reinfection is common. In my analysis of surveillance data presented in Chapter 4, the odds of testing positive were higher in people with a known chlamydia diagnosis in the last year compared to those without, even after adjusting for other available demographic and sexual behavioural variables (AOR 1.13/1.04; 1.27/1.15 for women and men respectively in the NCSP/GUMCAD datasets, section 4.4.2). In a complementary analysis of surveillance data from 2010, I have previously shown that 10% of chlamydia diagnoses made in GUM clinics in 15 to 24 year-olds were among individuals with at least one known previous diagnosis that calendar year.<sup>120</sup>

Re-infection may occur due to incomplete treatment of sexual partner(s) or reinfection due to continuing risk behaviour. Although my analysis did not incorporate trends over time in partner notification rates (Chapter 4), available data suggest that, on average, fewer than 0.5 partners per index case are tested,<sup>281,282</sup> leaving a high proportion of those diagnosed with chlamydia exposed to re-infection. Opportunities for re-infection from new partners are also indicated by the high rates of partner change in the age group targeted for screening, with one fifth of sexually-experienced 16 to 24 year-old women and one quarter of men reporting two or more new sexual partners in the last year.

While these data indicate that re-infection is common and demonstrate opportunities for re-infection, they do not provide definitive evidence of there having been a change in re-infection rates over time that might counteract any reduction in the duration of infection. Thus, while re-infection is likely to be important, the relative contribution of repeat infections to overall annual incidence rates is not known.

Incomplete treatment of the index case would also reduce the expected impact of screening on incidence and prevalence. Infections that are diagnosed but not treated would not contribute to a reduction in the average duration of infection. Data on treatment rates are also limited, such that trends in treatment rates could not be incorporated into my analyses. However, data from clinical audits suggest that around 5% to 10% of diagnoses go untreated.<sup>89,283</sup>

While data on re-infection rates, partner notification and treatment are not sufficient to allow a quantitative analyses of their impact in the context of my findings, I hypothesise that reductions in incidence and prevalence that could be expected by high diagnosis rates will have been attenuated by re-infections due to incomplete treatment of sexual partner(s) and new sexual partners and by incomplete treatment of a proportion of those diagnosed with chlamydia.

### 9.3.5 Underlying changes in sexual behaviour

Another possible explanation for an absence of change in chlamydia prevalence or age-specific cumulative incidence is that the underlying risk of transmission
due to sexual behaviour may have changed over the period of expansion of chlamydia screening. My analyses of data from the Natsal-2 and Natsal-3 surveys (Chapter 6) present a mixed picture. There was no notable increase in reported numbers of sexual partners among women or men between the surveys. However, the proportion reporting sexual debut before age 16 increased with subsequent birth cohorts which may suggest increased sexual activity or opportunities for transmission over the last decade. In their comparison of sexual behaviours between Natsal surveys, Mercer et al reported an increase in proportion of participants reporting heterosexual oral and particularly anal sex<sup>235</sup> between Natsal-2 and Natsal-3. Although not necessarily a risk factor for transmission, this may indicate some increase in higher risk sexual behaviours and practices. However, if the underlying risk of STI has changed, this was not reflected in differences in the prevalence of non-vaccine HPV types, which did not differ between surveys. In summary, there is no strong evidence to support or to rule out there having been an increase in underlying transmission risk in recent years. It remains possible that in the absence of widespread chlamydia screening, incidence and prevalence may have been even higher.

In summary, there are several features of the relationship between chlamydia infection and approaches to screening that provide possible explanations as to why high rates of testing might not have resulted in a fall in transmission. It remains to be seen whether additional years will have a more dramatic and lasting effect. I now turn to consideration of my findings in relation to how chlamydia control efforts might be improved and developed in future years.

#### 9.4 Implications for chlamydia control policies

#### 9.4.1 Targeted versus universal screening strategies

The extent to which public health interventions are targeted among specific groups is an important question for policymakers. Different approaches have implications for costs, effectiveness and equity of potential benefit.<sup>267,284,285</sup>

Current chlamydia screening recommendations in England strike a balance between universal and targeted approaches. Screening is recommended to all sexually-active under 25 year-olds. However, from a cost efficiency perspective, it might be tempting to pursue a more targeted approach. Since 2012 the Health Protection Agency, and then Public Health England, have prioritised the detection of infections over testing volumes by recommending that local authorities aim to achieve a diagnosis rate of at least 2,300 diagnoses per 100,000 population.<sup>121</sup> This placed an implicit emphasis on testing among populations likely to be infected. Shifting screening activity to higher risk populations where infections are more likely to be detected might therefore be appealing in terms of detecting as many infections as possible for the fewest number of tests. One way to achieve this would be through the use of selective screening criteria using a combination of behavioural and demographic characteristics to determine an individual's eligibility for screening. Such criteria have been suggested and used elsewhere<sup>88,286,287</sup> and data from Natsal-3 could theoretically be used to develop such a scoring system for use in England.

However there are some fundamental disadvantages to such an approach for chlamydia screening. Selective screening criteria would be unlikely to reduce the eligible population by a substantial amount, as infections – at least among

women - are relatively widespread. Data from Natsal-3 show that 52% of prevalent infections and 31% of reported diagnoses in the last year were among women who reported fewer than two sexual partners in the last year (Table 7-3a). Selective screening may be more appropriate in men where infections are more concentrated (e.g. 80% of prevalent infections and 77% of reported diagnoses in the last year were in men with ten or more sexual partners by the time of the interview, Table 7-3b). Possibilities for different screening approaches by gender are discussed in the next section. However, even if a scoring system could be developed, there would likely be substantial challenges in implementing individual risk-based criteria. Risk-scores based on sexual behaviour would require accurate sexual history taking that may go beyond that usually provided, or that would be easily obtained, in normal practice. Furthermore, targeted interventions in sexual health have been shown to have adverse consequences by increasing the stigmatisation of those targeted.<sup>91</sup> It is also important to note that restricting screening to high risk groups would not necessarily achieve full coverage in those groups, so would not in itself improve diagnosis rates or reduce the opportunities for transmission presented by undiagnosed infections in relatively under-tested groups.

A further argument against selective screening criteria is that they would limit the proportion of all infections that could be diagnosed as infections are not restricted to high risk populations. This would restrict the potential direct benefit of screening – i.e. the potential reduction in the risk of developing complications – to the target population. In the interests of achieving an equitable approach, an alternative would be to implement screening so that the chance of having an infection diagnosed is balanced across different groups. This would support the

distribution of screening according to the risk of infection without restricting screening to high risk groups.

My analysis of Natsal-3 data showed that in 2010-12, screening distributions were broadly aligned to risk, but that more could be done to align these distributions (Chapter 6). Specifically, under the screening distributions in 2010-12 women and men with a chlamydia infection living in deprived areas would have less of a chance of having their infection diagnosed than those with an infection living in less deprived areas. Strategies that achieve higher testing in women living in more deprived areas - but not to the exclusion of those living in low risk areas - or those at higher risk due to other behavioural factors would provide a more equitable approach to diagnosing chlamydial infections and should be considered.

#### 9.4.2 Screening in men

In contrast to many other countries with recommendations for asymptomatic chlamydia testing, the NCSP does not have separate screening recommendations for women and men. As set out in Chapter 2, in the late 1990s when screening policies were under development in England, men were not considered to be part of the target population. However when the NCSP was implemented in 2003 men were included, partly to prevent stigmatisation of women and in recognition of men's role in transmitting infection to their female sexual partners.<sup>79</sup> As the majority of chlamydia-related complications are thought to occur in women, the suitability of asymptomatic screening in men has been questioned.<sup>288</sup>

The research presented in this thesis did not aim to address the question of screening in men, but my findings raise some important issues that warrant consideration. Firstly, in Natsal-3, peak prevalence in men was among 20 to 24 year-olds, with no prevalent infections detected among <19 year-olds. However, testing rates in this age group were lower than among 16 to 19 year-old men. Low levels of testing in men may, therefore, be contributing to ongoing transmission to women. Secondly, my analysis showed that chlamydia infections in men appeared to be more focussed among those with higher numbers of sexual partners, whereas in women infections were more evenly distributed in the population. Non-condom use was an important risk factor for infection among men, but low testing rates were also seen in those who reported recent non-condom use. Therefore focussing screening and prevention efforts on 20 to 24 year-old men and those with higher numbers of sexual partners.

The optimum means of achieving this is not clear. Achieving high rates of screening in men is notoriously challenging<sup>79</sup> and increased dedication of resources would not necessarily achieve higher testing.<sup>289</sup> As chlamydia appears to be more focussed in those with more sexual partners, risk-based screening criteria may be a way to focus resources more appropriately in men. However, these would still face the difficulties of operationalising such scoring systems as discussed above. An alternative would be to place more emphasis on the control of chlamydia at a partnership level, rather than an individual level. Turner *et al* have proposed focussing on partner notification as a more cost-efficient alternative to increasing screening in men with regard to the cost per diagnosis. This is attractive in terms of preventing re-infection and in the potential to identify partners of the partner.<sup>290</sup> However, Turner *et al* did not take

into account the potential benefit of identifying infections in women who may not have accessed testing themselves so there remains a question about comparative cost effectiveness of different approaches. In times of limited resources it is also difficult to argue for increased screening in men when the majority of chlamydia-related complications are thought to be among women. Determining the optimum approach to screening in men is beyond the remit for this PhD, but it is perhaps time for some of these questions to be revisited to consider screening at the partnership level.

In the previous section I described how inadequate treatment and partner notification can attenuate the potential benefit of screening in terms of transmission and that re-infection is an important feature to consider in the transmission dynamics of chlamydia. Therefore, independent of the distribution or level of chlamydia screening implemented, robust case management of individuals diagnosed with infection is needed to maximise the potential benefit afforded by each diagnosis. Timely and effective treatment, partner notification and re-testing should remain priorities for service delivery.

#### 9.5 Future research

The research presented in this thesis raises a number of questions that warrant further consideration.

## 9.5.1 Validation of the relationship between chlamydia infection and the development and persistence of detectable antibodies

Given the challenges of chlamydia seroepidemiology set out in section 9.1.4, further work is needed to develop serological assays so that they can distinguish between first and repeat infections. This would in turn allow future studies to investigate the relative contributions of first infections, re-infections and treatment to *C. trachomatis* antibody seroprevalence over time. The relationship between antibody-inducing chlamydia infection and complications also requires further elucidation to understand the meaning of serological findings for understanding the impact of chlamydia screening on long-term morbidity. Future outcome monitoring could feasibly comprise monitoring of first, repeat and complication-causing infections. A study using clinical samples characterised by previous diagnosis history is now ongoing and aims to develop the use of the indirect and double-antigen Pgp3 ELISAs in this way.<sup>291</sup>

#### 9.5.2 Implications for mathematical modelling

The most recently published mathematical modelling study to have examined the potential impact of chlamydia in England is that by Turner *et al.*<sup>80,81</sup> The model results suggested that opportunistic chlamydia screening in GP practices of women or women and men aged <25 years could at least halve the prevalence of chlamydia within ten years (with substantial reductions seen in the first few years after screening implementation as shown in Figure 2-1), providing screening is offered to the entire eligible population when they attend a GP, that 50% of those offered screening accept the invitation and that 20% of partners are treated.<sup>81</sup> As Turner *et al* point out, the predictions derived from the model do not represent the truth, but instead present "the likely outcome, if our description of reality is accurate".<sup>81</sup> Findings from Natsal-3 showed that 62.5% of sexually-experienced women and 43.2% of sexually-experienced men reported having a chlamydia test in the last year and surveillance data suggest similar levels of testing were in place between 2009 and 2012 (Figure 2-5; Figure 2-6). However the dramatic reductions in prevalence predicted by the

model do not yet appear to have been borne out, thus raising some queries about the model predictions. There are a number of infection- and testingrelated parameters in the model that are challenged by my findings.

The model was fitted to chlamydia prevalence among women estimated from a meta-analysis of studies among GP attenders or those registered with a GP<sup>123</sup>. The resulting baseline prevalence is substantially higher than that estimated in Natsal-2 and suggests the baseline prevalence in the model may have been high. For example in 16 to 19 year-old women, chlamydia prevalence estimated in Natsal-2 was 3.9% (Table 6-4), but the model baseline prevalence was over 6%. My analysis of Natsal-3 data also shows that the age distribution of infection and diagnoses varies by gender, and that screening patterns vary by age, gender and, sexual behaviour (Chapter 6). However, the Turner model assumed the same distribution of infection and diagnosis in both women and men and assumed that acceptance of screening was not associated with age, gender or numbers of sexual partners. Given interactions between model parameters, the full impact of these departures between the model assumptions and empirical data are unclear without further runs of an amended model.

Findings from the serological study presented in Chapter 8 provide new parameters which may be useful either for fitting or validation of mathematical models. *C. trachomatis* antibody seroprevalence by age can be used as a marker of minimum age-specific cumulative incidence and the prevalence of reporting a chlamydia diagnosis among seropositive individuals represents a proportion of those ever infected who have ever been diagnosed.

While increasing the complexity of mathematical models may not always result in an improvement in their explanatory or predictive capability,<sup>292</sup> the impact of varying infection and testing by age, gender and sexual behaviour simultaneously and the use of newly available parameters should be considered in future mathematical models of chlamydia screening in England and elsewhere.

#### 9.5.3 Screening early in the course of infection.

I have hypothesised that timing of testing in relation to time since infection may have been insufficient to prevent transmission. As well as potentially reducing onward transmission, testing early in the course of infection has the potential to reduce an individual's risk of developing complications.<sup>39</sup> Increasing testing rates either at the end of a sexual partnership or following change of sexual partner therefore warrants attention. Specifically, future research to examine current adherence to recommendations of testing on change of sexual partner should be considered. This could include both quantitative analyses among those accessing testing and qualitative work in young adults to understand attitudes towards current recommendations. This work could contribute to developing and evaluating an intervention to increase testing soon after partner change with the aim of reducing both onward transmission and the risk of developing complications in those with chlamydia.

#### 9.6 Concluding statement

In summary, there is some evidence from my analysis of trends in the percentage testing positive for chlamydia using surveillance data and Horner *et al*'s analysis of *C. trachomatis* antibody seroprevalence to support there having

been a decrease in chlamydia incidence and prevalence in recent years. However, findings from population-based studies using probability sampling do not support there having been such a decline. Thus, taken together there is no strong empirical evidence that chlamydia screening, as it has been delivered in practice in England, has had an effect on the incidence or prevalence of chlamydia infection among under 25 year-olds.

This absence of any such evidence may be due to insufficient power to detect changes in the outcome measures investigated due to available sample size or to insufficient time since the national implementation of the NCSP in 2008. Alternatively, a combination of high lifetime risk of infection, high proportion of infections that go undiagnosed, insufficient testing levels or frequency among high risk groups and – potentially – re-infection due to incomplete treatment of sexual partners may have limited the impact of chlamydia control efforts.

Future evaluation of the impact of chlamydia control programmes should focus on applying *C. trachomatis* antibody assays to population-based serum samples and the use of existing population-based surveys to monitor chlamydia screening in relation to the risk of infection.

#### 10 References

- 1 Scholes D, Stergachis A, Heidrich FE, Andrilla H, Holmes KK, Stamm WE. Prevention of pelvic inflammatory disease by screening for cervical chlamydial infection. N Engl J Med 1996; 334(21):1362-1366.
- 2 European Centre for Disease Prevention and Control. Chlamydia control in Europe: literature review. 2014. <u>http://ecdc.europa.eu/en/publications/Publications/Chlamydia-control-europe.pdf</u>
- 3 Redmond SM, Alexander-Kisslig K, Woodhall SC, van den Broek IV, van BJ, Ward H et al. Genital Chlamydia prevalence in europe and non-European high income countries: systematic review and meta-analysis. *PLoS One* 2015; 10(1):e0115753.
- 4 Erens, B., McManus, S., Field, J., Korovessis, C., Johnson, A. M., Fenton, K., and Wellings, K. National Survey of Sexual Attitudes and Lifestyles II: Technical Report. 2001. <u>http://doc.ukdataservice.ac.uk/doc/5223/mrdoc/pdf/5223\_techreport.pdf</u>
- 5 Erens, B., Phelps, A., Clifton, S., Hussey, D., Mercer, C. H., Tanton, C., Sonnenberg, P., Macdowall, W., Copas, A. J., Field, N., Mitchell, K., Datta, J., Hawkins, V., Ison, C., Beddows, S., Soldan, K., Coelho da Silva, F., Alexander, S., Wellings, K., and Johnson, A. M. The Third National Survey of Sexual Attitudes and Lifestyles (Natsal-3): Technical Report. London: National Centre for Social Research; 2013. <u>www.natsal.ac.uk</u>
- 6 Savage EJ, Mohammed H, Leong G, Duffell S, Hughes G. Improving surveillance of sexually transmitted infections using mandatory electronic clinical reporting: the genitourinary medicine clinic activity dataset, England, 2009 to 2013. *Euro Surveill* 2014; 19(48):20981.
- 7 Mindell J, Biddulph JP, Hirani V, Stamatakis E, Craig R, Nunn S et al. Cohort profile: the health survey for England. *Int J Epidemiol* 2012; 41(6):1585-1593.
- 8 Low N, Stroud K, Lewis DA, Cassell JA. Mind your binomials: a guide to microbial nomenclature and spelling in Sexually Transmitted Infections. Sex Transm Infect 2015; 91(3):154-155.
- 9 Public Health England. Table 8: Number and rates of selected STI diagnoses in the UK, 2008-2012. Available from: <u>www.gov.uk/government/statistics/sexually-transmittedinfections-stis-annual-data-tables</u> Date accessed: 2014
- 10 Low N, McCarthy A, Macleod J, Salisbury C, Campbell R, Roberts TE et al. Epidemiological, social, diagnostic and economic evaluation of population screening for genital chlamydial infection. *Health Technol Assess* 2007; 11(8):iii-xii, 1.
- 11 Stamm WE. *Chlamydia trachomatis* infections of the adult. In: Holmes KK, Sparling PF, Mardh P-A, ., Lemon SM, Stamm WE et al., editors. Sexually Transmitted Diseases. New York: 1999.
- 12 Gottlieb SL, Martin DH, Xu F, Byrne GI, Brunham RC. Summary: The natural history and immunobiology of Chlamydia trachomatis genital infection and implications for Chlamydia control. *J Infect Dis* 2010; 201 Suppl 2:S190-S204.
- 13 Price MJ, Ades A, Welton NJ, Macleod J, Turner K, Simms I et al. How much tubal factor infertility is caused by Chlamydia? Estimates based on serological evidence corrected for sensitivity and specificity. *Sex Transm Dis* 2012; 39(8):608-613.
- 14 Price MJ, Ades AE, De AD, Welton NJ, Macleod J, Soldan K et al. Risk of pelvic inflammatory disease following Chlamydia trachomatis infection: analysis of prospective studies with a multistate model. *Am J Epidemiol* 2013; 178(3):484-492.

- 15 Westrom L, Joesoef R, Reynolds G, Hagdu A, Thompson SE. Pelvic inflammatory disease and fertility. A cohort study of 1,844 women with laparoscopically verified disease and 657 control women with normal laparoscopic results. *Sex Transm Dis* 1992; 19(4):185-192.
- 16 Schachter J. Biology of Chlamydia trachomatis. In: Holmes KK, editor. Sexually Transmitted Diseases. New York: McGraw-Hill; 1999.
- 17 European Centre for Disease Prevention and Control. Annual epidemiological report 2014 - Sexually transmitted infections, including HIV and blood-borne viruses. 2014. <u>http://ecdc.europa.eu/en/publications/\_layouts/forms/Publication\_DispForm.aspx?List=4</u> <u>f55ad51-4aed-4d32-b960-af70113dbb90&ID=1262</u>
- 18 Price MJ, Ades AE, Angelis DD, Welton NJ, Macleod J, Soldan K et al. Mixture-ofexponentials models to explain heterogeneity in studies of the duration of Chlamydia trachomatis infection. *Stat Med* 2012.
- 19 British Association of Sexual Health and HIV. 2006 UK National Guideline for the Management of Genital Tract Infection with *Chlamydia trachomatis*. 2006.
- 20 Simms I, Talebi A, Riha J, Horner P, French RS, Sarah R et al. The English National Chlamydia Screening Programme: variations in positivity in 2007/2008. *Sex Transm Dis* 2009; 36(8):522-527.
- 21 Persson K. The role of serology, antibiotic susceptibility testing and serovar determination in genital chlamydial infections. *Best Pract Res Clin Obstet Gynaecol* 2002; 16(6):801-814.
- 22 van den Broek IV, Land JA, van Bergen JE, Morre SA, van der Sande MA. Chlamydia trachomatis Antibody Testing in Vaginal Mucosal Material versus Blood Samples of Women Attending a Fertility Clinic and an STI Clinic. *Obstet Gynecol Int* 2014; 2014:601932.
- 23 Sonnenberg P, Clifton S, Beddows S, Field N, Soldan K, Tanton C et al. Prevalence, risk factors, and uptake of interventions for sexually transmitted infections in Britain: findings from the National Surveys of Sexual Attitudes and Lifestyles (Natsal). *Lancet* 2013; 382(9907):1795-1806.
- 24 Haggerty CL, Schulz R, Ness RB. Lower quality of life among women with chronic pelvic pain after pelvic inflammatory disease. *Obstet Gynecol* 2003; 102(5 Pt 1):934-939.
- 25 Smith KJ, Tsevat J, Ness RB, Wiesenfeld HC, Roberts MS. Quality of Life Utilities for Pelvic Inflammatory Disease Health States. *Sexually Transmitted Diseases* 2008; 35(3):307-311.
- 26 Stratton, K. R., Durch, J. S., and Lawrence, R. S. Vaccines for the 21st Century: A tool for decisionmaking. Washington D.C.: National Academy Press; 2000.
- 27 Aghaizu A, Adams EJ, Turner K, Kerry S, Hay P, Simms I et al. What is the cost of pelvic inflammatory disease and how much could be prevented by screening for chlamydia trachomatis? Cost analysis of the Prevention of Pelvic Infection (POPI) trial. *Sex Transm Infect* 2011; 87(4):312-317.
- 28 Adams EJ, Turner KM, Edmunds WJ. The cost effectiveness of opportunistic chlamydia screening in England. Sex Transm Infect 2007; 83(4):267-274.
- 29 Roberts TE, Robinson S, Barton PM, Bryan S, McCarthy A, Macleod J et al. Cost effectiveness of home based population screening for Chlamydia trachomatis in the UK: economic evaluation of chlamydia screening studies (ClaSS) project. *BMJ* 2007; 335(7614):291.

- 30 Westaby DT, Wu O, Duncan WC, Critchley HO, Tong S, Horne AW. Has increased clinical experience with methotrexate reduced the direct costs of medical management of ectopic pregnancy compared to surgery? *BMC Pregnancy Childbirth* 2012; 12:98.
- 31 Brunham RC, Binns B, Guijon F, Danforth D, Kosseim ML, Rand F et al. Etiology and outcome of acute pelvic inflammatory disease. *J Infect Dis* 1988; 158(3):510-517.
- 32 Chow JM, Yonekura ML, Richwald GA, Greenland S, Sweet RL, Schachter J. The association between Chlamydia trachomatis and ectopic pregnancy. A matched-pair, case-control study. *JAMA* 1990; 263(23):3164-3167.
- 33 Ness RB, Soper DE, Richter HE, Randall H, Peipert JF, Nelson DB et al. Chlamydia antibodies, chlamydia heat shock protein, and adverse sequelae after pelvic inflammatory disease: the PID Evaluation and Clinical Health (PEACH) Study. Sex Transm Dis 2008; 35(2):129-135.
- 34 Robertson JN, Ward ME, Conway D, Caul EO. Chlamydial and gonococcal antibodies in sera of infertile women with tubal obstruction. *J Clin Pathol* 1987; 40(4):377-383.
- 35 Darville T, Hiltke TJ. Pathogenesis of genital tract disease due to Chlamydia trachomatis. *J Infect Dis* 2010; 201 Suppl 2:S114-S125.
- 36 Rours IG, Hammerschlag MR, Ott A, De Faber TJ, Verbrugh HA, de Groot R et al. Chlamydia trachomatis as a cause of neonatal conjunctivitis in Dutch infants. *Pediatrics* 2008; 121(2):e321-e326.
- 37 Jain S. Perinatally acquired Chlamydia trachomatis associated morbidity in young infants. *J Matern Fetal Med* 1999; 8(3):130-133.
- 38 Haggerty CL, Gottlieb SL, Taylor BD, Low N, Xu F, Ness RB. Risk of sequelae after Chlamydia trachomatis genital infection in women. *J Infect Dis* 2010; 201 Suppl 2:S134-S155.
- 39 Oakeshott P, Kerry S, Aghaizu A, Atherton H, Hay S, Taylor-Robinson D et al. Randomised controlled trial of screening for Chlamydia trachomatis to prevent pelvic inflammatory disease: the POPI (prevention of pelvic infection) trial. *BMJ* 2010; 340:c1642.
- 40 Rours GI, Duijts L, Moll HA, Arends LR, de Groot R, Jaddoe VW et al. Chlamydia trachomatis infection during pregnancy associated with preterm delivery: a population-based prospective cohort study. *Eur J Epidemiol* 2011.
- 41 Haggerty CL, Klebanoff MA, Panum I, Uldum SA, Bass DC, Olsen J et al. Prenatal infection increases the risk of preeclampsia. *Pregnancy Hypertens* 2013; 3(3):151-154.
- 42 Liu B, Roberts CL, Clarke M, Jorm L, Hunt J, Ward J. Chlamydia and gonorrhoea infections and the risk of adverse obstetric outcomes: a retrospective cohort study. *Sex Transm Infect* 2013; 89(8):672-678.
- 43 Thorne, C. UK National Screening Committee Policy Review Chlamydia screening in pregnancy: an evidence review. 2011. <u>http://www.screening.nhs.uk/chlamydiapregnancy</u>
- 44 Johnson LF, Lewis DA. The effect of genital tract infections on HIV-1 shedding in the genital tract: a systematic review and meta-analysis. Sex Transm Dis 2008; 35(11):946-959.
- 45 Silva J, Cerqueira F, Medeiros R. Chlamydia trachomatis infection: implications for HPV status and cervical cancer. *Arch Gynecol Obstet* 2013.

- 46 Shew ML, Ermel AC, Weaver BA, Tong Y, Tu W, Kester LM et al. Association of Chlamydia trachomatis Infection With Redetection of Human Papillomavirus After Apparent Clearance. *J Infect Dis* 2013.
- 47 Batteiger BE, Xu F, Johnson RE, Rekart ML. Protective immunity to Chlamydia trachomatis genital infection: evidence from human studies. *J Infect Dis* 2010; 201 Suppl 2:S178-S189.
- 48 Batteiger BE, Tu W, Ofner S, Van Der PB, Stothard DR, Orr DP et al. Repeated Chlamydia trachomatis genital infections in adolescent women. *J Infect Dis* 2010; 201(1):42-51.
- 49 Gotz HM, van dB, IV, Hoebe CJ, Brouwers EE, Pars LL, Fennema JS et al. High yield of reinfections by home-based automatic rescreening of Chlamydia positives in a largescale register-based screening programme and determinants of repeat infections. *Sex Transm Infect* 2013; 89(1):63-69.
- 50 Gotz HM, Wolfers ME, Luijendijk A, van den Broek IV. Retesting for genital Chlamydia trachomatis among visitors of a sexually transmitted infections clinic: randomized intervention trial of home- versus clinic-based recall. *BMC Infect Dis* 2013; 13(1):239.
- 51 LaMontagne DS, Baster K, Emmett L, Nichols T, Randall S, McLean L et al. Incidence and reinfection rates of genital chlamydial infection among women aged 16-24 years attending general practice, family planning and genitourinary medicine clinics in England: a prospective cohort study by the Chlamydia Recall Study Advisory Group. *Sex Transm Infect* 2007; 83(4):292-303.
- 52 Rietmeijer CA, Van Bemmelen R, Judson FN, Douglas JM, Jr. Incidence and repeat infection rates of Chlamydia trachomatis among male and female patients in an STD clinic: implications for screening and rescreening. *Sex Transm Dis* 2002; 29(2):65-72.
- 53 Turner KM, Horner PJ, Trela-Larsen L, Sharp M, May M. Chlamydia screening, retesting and repeat diagnoses in Cornwall, UK 2003-2009. Sex Transm Infect 2013; 89(1):70-75.
- 54 Walker J, Tabrizi SN, Fairley CK, Chen MY, Bradshaw CS, Twin J et al. Chlamydia trachomatis incidence and re-infection among young women--behavioural and microbiological characteristics. *PLoS One* 2012; 7(5):e37778.
- 55 Woodhall SC, Atkins JL, Soldan K, Hughes G, Bone A, Gill ON. Repeat genital Chlamydia trachomatis testing rates in young adults in England, 2010. Sex Transm Infect 2013; 89(1):51-56.
- 56 Hosenfeld CB, Workowski KA, Berman S, Zaidi A, Dyson J, Mosure D et al. Repeat infection with Chlamydia and gonorrhea among females: a systematic review of the literature. *Sex Transm Dis* 2009; 36(8):478-489.
- 57 Fung M, Scott KC, Kent CK, Klausner JD. Chlamydial and gonococcal reinfection among men: a systematic review of data to evaluate the need for retesting. *Sex Transm Infect* 2007; 83(4):304-309.
- 58 Brunham RC, Pourbohloul B MS, White R, Rekart ML. The Unexpected Impact of a Chlamydia trachomatis Infection Control Program on Susceptibility to Reinfection. *The Journal of Infectious Diseases* 2005; 192:1836-1843.
- 59 Brunham RC, Rekart ML. The arrested immunity hypothesis and the epidemiology of chlamydia control. *Sex Transm Dis* 2008; 35(1):53-54.
- 60 Raffle AE, Gray MJA. Screening: Evidence and Practice. Oxford; 2007.

- 61 Wilson, J. M. G. and Jungner, G. Principles and practice of screening for disease. Geneva: World Health Organization; 1968. <u>http://whqlibdoc.who.int/php/WHO\_PHP\_34.pdf</u>
- 62 Andermann A, Blancquaert I, Beauchamp S, Dery V. Revisiting Wilson and Jungner in the genomic age: a review of screening criteria over the past 40 years. *Bull World Health Organ* 2008; 86(4):317-319.
- 63 National Screening Committee. Criteria for appraising the viability, effectiveness and appropriateness of a screening programme. Available from: <u>http://www.screening.nhs.uk/criteria</u> Date accessed: 7-5-2015
- 64 Holland WW, Stewart S. Screening in Health Care: Benefit or Bane? Abingdon: The Nuffield Provincial Hospitals Trust; 1990.
- 65 Pimenta JM, Catchpole M, Rogers PA, Perkins E, Jackson N, Carlisle C et al. Opportunistic screening for genital chlamydial infection. I: acceptability of urine testing in primary and secondary healthcare settings. *Sex Transm Infect* 2003; 79(1):16-21.
- 66 Herzog SA, Heijne JC, Scott P, Althaus CL, Low N. Direct and indirect effects of screening for Chlamydia trachomatis on the prevention of pelvic inflammatory disease: a mathematical modeling study. *Epidemiology* 2013; 24(6):854-862.
- 67 Ostergaard L, Anderson B, Moeller JK, Olesen F. Home sampling versus conventional swab samples for screening of *Chlamydia trachomatis* in women: a cluster randomised 1 year follow up study. *Clinical and Infectious Disease* 2000; 31(4):951-957.
- 68 Andersen B, van V, I, Sokolowski I, Moller JK, Ostergaard L, Olesen F. Impact of intensified testing for urogenital Chlamydia trachomatis infections: a randomised study with 9-year follow-up. Sex Transm Infect 2010.
- 69 Anderson RM, May RM. Epidemiological parameters of HIV transmission. *Nature* 1988; 333(6173):514-519.
- 70 Brunham RC. Parran Award Lecture: insights into the epidemiology of sexually transmitted diseases from Ro = betacD. Sex Transm Dis 2005; 32(12):722-724.
- 71 Price MJ, Ades AE, De AD, Welton NJ, Macleod J, Turner K et al. Incidence of Chlamydia trachomatis infection in women in England: two methods of estimation. *Epidemiol Infect* 2014; 142(3):562-576.
- 72 Pimenta J, Catchpole M, Gray M, Hopwood J, Randall S. Evidence based health policy report. Screening for genital chlamydial infection. *BMJ* 2000; 321(7261):629-631.
- 73 Herrmann B, Egger M. Genital Chlamydia trachomatis infections in Uppsala County, Sweden, 1985-1993: declining rates for how much longer? *Sex Transm Dis* 1995; 22(4):253-260.
- 74 Hillis SD, Nakashima A, Amsterdam L, Pfister J, Vaughn M, Addiss D et al. The impact of a comprehensive chlamydia prevention program in Wisconsin. *Fam Plann Perspect* 1995; 27(3):108-111.
- 75 Chief Medical Officer's Expert Advisory Group. Main report of the CMO's Expert Advisory Group on *Chlamydia trachomatis*. London: Department of Health. 1998.
- 76 Department of Health. The national strategy for sexual health and HIV. London: Department of Health. 2001.
- 77 Sheringham J, Baraitser P, Simms I, Hart G, Raine R. Chlamydia screening in England: a qualitative study of the narrative behind the policy. *BMC Public Health* 2012; 12:317.

- 78 National Chlamydia Screening Programme. The NCSP: An overview. 2015 Available from: <a href="http://www.chlamydiascreening.nhs.uk/ps/overview.asp">http://www.chlamydiascreening.nhs.uk/ps/overview.asp</a>
- 79 National Chlamydia Screening Programme. Men too... 2007. http://www.chlamydiascreening.nhs.uk/ps/resources/guidelines/NCSPMens\_strat.pdf
- 80 Turner KM, Adams EJ, Gay N, Ghani AC, Mercer C, Edmunds WJ. Developing a realistic sexual network model of chlamydia transmission in Britain. *Theor Biol Med Model* 2006; 3:3.
- 81 Turner KM, Adams EJ, LaMontagne DS, Emmett L, Baster K, Edmunds WJ. Modelling the effectiveness of chlamydia screening in England. *Sex Transm Infect* 2006; 82(6):496-502.
- 82 European Centre for Disease Prevention and Control. Chlamydia control in Europe a survey of Member States. 2014. <u>http://www.ecdc.europa.eu/en/publications/Publications/chlamydia-control-surveyeurope-2012.pdf</u>
- 83 Scottish Intercollegiate Guidelines Network. Management of genital Chlamydia trachomatis infection. A national clinical guideline (109). Edinburgh: Scottish Intercollegiate Guidelines Network; 2009. <u>http://www.sign.ac.uk/pdf/qrg109.pdf</u>
- 84 Meyers DS, Halvorson H, Luckhaupt S. Screening for chlamydial infection: an evidence update for the U.S. Preventive Services Task Force. *Ann Intern Med* 2007; 147(2):135-142.
- 85 Low N, Bender N, Nartey L, Shang A, Stephenson JM. Effectiveness of chlamydia screening: systematic review. *Int J Epidemiol* 2009; 38(2):435-448.
- 86 Low N. Screening programmes for chlamydial infection: when will we ever learn? *BMJ* 2007; 334(7596):725-728.
- 87 Nicholson A, Rait G, Murray-Thomas T, Hughes G, Mercer CH, Cassell J. Management of first-episode pelvic inflammatory disease in primary care: results from a large UK primary care database. *Br J Gen Pract* 2010; 60(579):e395-e406.
- 88 van den Broek IV, van Bergen JE, Brouwers EE, Fennema JS, Gotz HM, Hoebe CJ et al. Effectiveness of yearly, register based screening for chlamydia in the Netherlands: controlled trial with randomised stepped wedge implementation. *BMJ* 2012; 345:e4316.
- 89 National Audit Office. Young People's Sexual Health: The National Chlamydia Screening Programme. London: The Stationery Office; 2009. www.nao.org.uk/report/department-of-health-young-peoples-sexual-health-the-nationalchlamydia-screening-programme/
- 90 HM Treasury. The Magenta Book. Guidance for evaluation. 2011. www.gov.uk/government/publications/the-magenta-book
- 91 Bonell C, Jamal F, Melendez-Torres GJ, Cummins S. 'Dark logic': theorising the harmful consequences of public health interventions. *J Epidemiol Community Health* 2015; 69(1):95-98.
- 92 Aral SO, Blanchard JF. The Program Science initiative: improving the planning, implementation and evaluation of HIV/STI prevention programs. Sex Transm Infect 2012; 88(3):157-159.
- 93 Moore GF, Audrey S, Barker M, Bond L, Bonell C, Hardeman W et al. Process evaluation of complex interventions: Medical Research Council guidance. *BMJ* 2015; 350:h1258.

- 94 Craig P, Dieppe P, Macintyre S, Michie S, Nazareth I, Petticrew M. Developing and evaluating complex interventions: the new Medical Research Council guidance. *BMJ* 2008; 337:a1655.
- 95 Bonell C, Fletcher A, Morton M, Lorenc T, Moore L. Realist randomised controlled trials: a new approach to evaluating complex public health interventions. Soc Sci Med 2012; 75(12):2299-2306.
- 96 Pawson R, Tilley N. Realistic Evaluation. London: SAGE; 1997.
- 97 Centers for Disease Control and Prevention. Framework for Program Evaluation in Public Health. *MMWR* 1999; 48(RR-11):1-40.
- 98 Medical Research Council. Developing and evaluating complex interventions: new guidance. 2008. <u>http://www.mrc.ac.uk/documents/pdf/complex-interventions-guidance/</u>
- 99 Johnson AM, Horner P. A new role for Chlamydia trachomatis serology? Sex Transm Infect 2008; 84(2):79-80.
- 100 Horner PJ, Wills GS, Reynolds R, Johnson AM, Muir DA, Winston A et al. Effect of time since exposure to Chlamydia trachomatis on chlamydia antibody detection in women: a cross-sectional study. *Sex Transm Infect* 2013; 89(5):398-403.
- 101 Bender N, Herrmann B, Andersen B, Hocking JS, van Bergen J, Morgan J et al. Chlamydia infection, pelvic inflammatory disease, ectopic pregnancy and infertility: cross-national study. *Sex Transm Infect* 2011.
- 102 Morris GC, Stewart CM, Schoeman SA, Wilson JD. A cross-sectional study showing differences in the clinical diagnosis of pelvic inflammatory disease according to the experience of clinicians: implications for training and audit. *Sex Transm Infect* 2014; 90(6):445-451.
- 103 Bonell CP, Hargreaves J, Cousens S, Ross D, Hayes R, Petticrew M et al. Alternatives to randomisation in the evaluation of public health interventions: design challenges and solutions. *J Epidemiol Community Health* 2011; 65(7):582-587.
- 104 Moore L, Moore GF. Public health evaluation: which designs work, for whom and under what circumstances? *J Epidemiol Community Health* 2011; 65(7):596-597.
- 105 Craig P, Cooper C, Gunnell D, Haw S, Lawson K, Macintyre S et al. Using natural experiments to evaluate population health interventions: new Medical Research Council guidance. J Epidemiol Community Health 2012; 66(12):1182-1186.
- 106 Cousens S, Hargreaves J, Bonell C, Armstrong B, Thomas J, Kirkwood BR et al. Alternatives to randomisation in the evaluation of public-health interventions: statistical analysis and causal inference. *J Epidemiol Community Health* 2011; 65(7):576-581.
- 107 McNulty CA, Hogan AH, Ricketts EJ, Wallace L, Oliver I, Campbell R et al. Increasing chlamydia screening tests in general practice: a modified Zelen prospective Cluster Randomised Controlled Trial evaluating a complex intervention based on the Theory of Planned Behaviour. *Sex Transm Infect* 2013.
- 108 Kretzschmar M, Turner KM, Barton PM, Edmunds WJ, Low N. Predicting the population impact of chlamydia screening programmes: comparative mathematical modelling study. Sex Transm Infect 2009; 85(5):359-366.
- 109 Estimating chlamydia tests and diagnoses by age group and gender over a period of developing surveillance systems in England between 2000 and 2011, Poster presentation at the 2014 Public Health England conference, 2014. 2014.

- 110 Moore G, Audrey S, Barker M, Bond L, Bonell C, Cooper C et al. Process evaluation in complex public health intervention studies: the need for guidance. *J Epidemiol Community Health* 2014; 68(2):101-102.
- 111 Moore G, Audrey S, Barker M, Bond L, Bonell C, Hardeman W, Moore L, O'Cathain A, Tinati T, Wight D, and Baird J. Process evaluation of complex interventions: Medical Research Council guidance. London: MRC Population Health Sciences Research Network; 2014. <u>http://decipher.uk.net/process-evaluation-guidance/</u>
- 112 National Chlamydia Screening Programme. Integrating the National Chlamydia Screening Programme within local sexual health economies. 2012.
- 113 Woodhall SC, Sile B, Talebi A, Nardone A, Baraitser P. Internet testing for Chlamydia trachomatis in England, 2006 to 2010. *BMC Public Health* 2012; 12:1095.
- 114 Public Health England. National Chlamydia Screening Programme Standards (7th edition). 2014. <u>http://www.chlamydiascreening.nhs.uk/ps/standards.asp</u>
- 115 Health and Social Care Act 2012. 2012 Available from: http://www.legislation.gov.uk/ukpga/2012/7/contents/enacted
- 116 Cassell JA, Dodds J, Estcourt C, Llewellyn CD, Lanza S, Richens J et al. The relative clinical effectiveness and cost-effectiveness of three contrasting approaches to partner notification for curable sexually transmitted infections: a cluster randomised trial in primary care. *Health Technology Assessment* 2015; 19(5).
- 117 Althaus CL, Heijne JC, Herzog SA, Roellin A, Low N. Individual and population level effects of partner notification for Chlamydia trachomatis. *PLoS One* 2012; 7(12):e51438.
- 118 Althaus CL, Turner KM, Mercer CH, Auguste P, Roberts TE, Bell G et al. Effectiveness and cost-effectiveness of traditional and new partner notification technologies for curable sexually transmitted infections: observational study, systematic reviews and mathematical modelling. *Health Technology Assessment* 2014; 18(2).
- 119 Public Health England. Position Statement: Routine offer of re-test to young adults testing positive for chlamydia. 2013. Available from: <u>http://www.chlamydiascreening.nhs.uk/ps/resources.asp</u>
- 120 Woodhall SC, Atkins JL, Soldan K, Hughes G, Bone A, Gill ON. Repeat genital Chlamydia trachomatis testing rates in young adults in England, 2010. *Sex Transm Infect* 2012.
- 121 Department of Health. Improving outcomes and supporting transparency. Part 1: A public health outcomes framework for England, 2013-2016. 2012.
- 122 National Chlamydia Screening Programme. Towards best practice for chlamydia screening in Reproductive and Sexual Health Services (RSH). 2010. Available from: http://www.chlamydiascreening.nhs.uk/ps/resources/guidelines/SRH\_guidance.pdf
- 123 Adams EJ, Charlett A, Edmunds WJ, Hughes G. Chlamydia trachomatis in the United Kingdom: a systematic review and analysis of prevalence studies. *Sex Transm Infect* 2004; 80(5):354-362.
- 124 Public Health England. Chlamydia Screening: 3Cs & HIV programme. Available from: http://www.chlamydiascreening.nhs.uk/ps/3cs\_hiv.asp
- 125 Health Protection Agency. Testing Times HIV and other Sexually Transmitted Infections in the United Kingdom: 2007. 2007. Available from: <u>http://webarchive.nationalarchives.gov.uk/20140714084352/http://www.hpa.org.uk/web/ HPAweb&HPAwebStandard/HPAweb\_C/1203084355941</u>

- 126 Scottish Executive. Respect and Responsibility. Strategy and Action Plan for Improving Sexual Health. 2005. Available from: http://www.scotland.gov.uk/Resource/Doc/35596/0012575.pdf
- 127 The National Assembly for Wales. A strategic framework for promoting sexual health in Wales. 2001. Available from: http://www.wales.nhs.uk/sitesplus/documents/888/WAGStrategicFramework.pdf
- 128 Skidmore S, Horner P, Mallinson H. Testing specimens for Chlamydia trachomatis. Sex Transm Infect 2006; 82(4):272-275.
- 129 Department of Health. Choosing Health: making healthier choices easier. Department of Health: London. 2004.
- 130 Cassell JA, Mercer CH, Sutcliffe L, Petersen I, Islam A, Brook MG et al. Trends in sexually transmitted infections in general practice 1990-2000: population based study using data from the UK general practice research database. *BMJ* 2006; 332(7537):332-334.
- 131 Department of Health. Sexual Health: 48 Hour Genitourinary Medicine Access Monthly Monitoring. Available from: <u>http://webarchive.nationalarchives.gov.uk/20130107105354/http://www.dh.gov.uk/en/Publicationsandstatistics/Statistics/Performancedataandstatistics/SexualHealth/index.htm</u>
- 132 van AF, de MM, Morre SA, van Bergen JE, van der Klis FR, Land JA et al. Chlamydia trachomatis IgG seroprevalence in the general population of the Netherlands in 1996 and in 2007: differential changes by gender and age. Sex Transm Infect 2014; 90(5):434-440.
- 133 Fenton KA, Korovessis C, Johnson AM, McCadden A, McManus S, Wellings K et al. Sexual behaviour in Britain: reported sexually transmitted infections and prevalent genital Chlamydia trachomatis infection. *Lancet* 2001; 358(9296):1851-1854.
- 134 Bozicevic I, Grgic I, Zidovec-Lepej S, Cakalo JI, Belak-Kovacevic S, Stulhofer A et al. Urine-based testing for Chlamydia trachomatis among young adults in a populationbased survey in Croatia: feasibility and prevalence. *BMC Public Health* 2011; 11:230.
- 135 Goulet V, de Barbeyrac B, Raherison S, Prudhomme M, Semaille C, Warszawski J. Prevalence of Chlamydia trachomatis: results from the first national population-based survey in France. *Sex Transm Infect* 2010; 86(4):263-270.
- 136 Haar K. Prävalenz von urogenitalen Chlamydia trachomatis-Infektionen bei Teilnehmern des bundesweiten Kinder- und Jugendgesunheitssurveys (KiGGS) [Berlin School of Public Health; 2012.
- 137 van Bergen JE, Fennema JS, van dB, IV, Brouwers EE, de Feijter EM, Hoebe CJ et al. Rationale, design, and results of the first screening round of a comprehensive, registerbased, Chlamydia screening implementation programme in the Netherlands. *BMC Infect Dis* 2010; 10:293.
- 138 Klavs I, Rodrigues LC, Wellings K, Kese D, Hayes R. Prevalence of genital Chlamydia trachomatis infection in the general population of Slovenia: serious gaps in control. *Sexually Transmitted Infections* 2004; 80:121-123.
- 139 Ku L, St LM, Farshy C, Aral S, Turner CF, Lindberg LD et al. Risk behaviors, medical care, and chlamydial infection among young men in the United States. *Am J Public Health* 2002; 92(7):1140-1143.
- 140 Miller WC, Ford CA, Morris M, Handcock MS, Schmitz JL, Hobbs MM et al. Prevalence of chlamydial and gonococcal infections among young adults in the United States. *JAMA* 2004; 291(18):2229-2236.

- 141 Datta SD, Torrone E, Kruszon-Moran D, Berman S, Johnson R, Satterwhite CL et al. Chlamydia trachomatis trends in the United States among persons 14 to 39 years of age, 1999-2008. Sex Transm Dis 2012; 39(2):92-96.
- 142 Ostergaard L, Andersen B, Olesen F, Moller J. Efficacy of home sampling for screening of *Chlamydia trachomatis*: randomised study. *BMJ* 1998; 317:26-27.
- 143 Munk C, Morre SA, Kjaer SK, Poll PA, Bock JE, Meijer CJ et al. PCR-detected Chlamydia trachomatis infections from the uterine cervix of young women from the general population: prevalence and risk determinants. *Sex Transm Dis* 1999; 26(6):325-328.
- 144 Bennedsen M, Nygard B, Berthelsen L, Jensen JS, Lind I. [Prevalence of Chlamydia among young men. A screening among men liable for military service and coming before the military board]. *Ugeskr Laeger* 2001; 163(34):4583-4586.
- 145 Andersen B, Olesen F, Moller JK, Ostergaard L. Population-based strategies for outreach screening of urogenital Chlamydia trachomatis infections: a randomized, controlled trial. *J Infect Dis* 2002; 185(2):252-258.
- 146 Uuskula A, Kals M, Denks K, Nurm U, Kasesalu L, Dehovitz J et al. The prevalence of chlamydial infection in Estonia: a population-based survey. *Int J STD AIDS* 2008; 19(7):455-458.
- 147 van Valkengoed IG, Morre SA, van den Brule AJ, Meijer CJ, Deville W, Bouter LM et al. Low diagnostic accuracy of selective screening criteria for asymptomatic Chlamydia trachomatis infections in the general population. *Sex Transm Infect* 2000; 76(5):375-380.
- 148 Steen TW, Hjortdahl P, Storvold G, Vilimas K, Elstrom P, Esholdt I et al. [Prevalence of genital Chlamydia trachomatis infection in the age group 18-29 years in Oslo]. *Tidsskr Nor Laegeforen* 2005; 125(12):1637-1639.
- 149 Klovstad H, Grjibovski A, Aavitsland P. Population based study of genital Chlamydia trachomatis prevalence and associated factors in Norway: A cross sectional study. *BMC Infect Dis* 2012; 12:150.
- 150 Franceschi S, Smith JS, van den Brule A, Herrero R, Arslan A, Anh PT et al. Cervical infection with Chlamydia trachomatis and Neisseria gonorrhoeae in women from ten areas in four continents. A cross-sectional study. *Sex Transm Dis* 2007; 34(8):563-569.
- 151 Brannstrom M, Josefsson GB, Cederberg A, Liljestrand J. Prevalence of genital Chlamydia trachomatis infection among women in a Swedish primary health care area. *Scand J Infect Dis* 1992; 24(1):41-46.
- 152 Jonsson M, Karlsson R, Rylander E, Boden E, Edlund K, Evander M et al. The silent suffering women--a population based study on the association between reported symptoms and past and present infections of the lower genital tract. *Genitourin Med* 1995; 71(3):158-162.
- 153 Novak DP, Edman AC, Jonsson M, Karlsson RB. The internet, a simple and convenient tool in Chlamydia trachomatis screening of young people. *Euro Surveill* 2003; 8(9):171-176.
- 154 Novak DP, Lindholm L, Jonsson M, Karlsson RB. A Swedish cost-effectiveness analysis of community-based Chlamydia trachomatis PCR testing of postal urine specimens obtained at home. *Scand J Public Health* 2004; 32(5):324-332.
- 155 Domeika M, Oscarsson L, Hallen A, Hjelm E, Sylvan S. Mailed urine samples are not an effective screening approach for Chlamydia trachomatis case finding among young men. *J Eur Acad Dermatol Venereol* 2007; 21(6):789-794.

- 156 Pierpoint T, Thomas B, Judd A, Brugha R, Taylor-Robinson D, Renton A. Prevalence of Chlamydia trachomatis in young men in north west London. Sex Transm Infect 2000; 76(4):273-276.
- 157 Stephenson J, Carder C, Copas A, Robinson A, Ridgway G, Haines A. Home screening for chlamydial genital infection: is it acceptable to young men and women? *Sex Transm Infect* 2000; 76(1):25-27.
- 158 Bracebridge S, Bachmann MO, Ramkhelawon K, Woolnough A. Evaluation of a systematic postal screening and treatment service for genital Chlamydia trachomatis, with remote clinic access via the internet: a cross-sectional study, East of England. *Sex Transm Infect* 2012; 88(5):375-381.
- 159 Miller GC, McDermott R, McCulloch B, Fairley CK, Muller R. Predictors of the prevalence of bacterial STI among young disadvantaged Indigenous people in north Queensland, Australia. *Sex Transm Infect* 2003; 79(4):332-335.
- 160 Latif A, Smith K. STI screening conducted in NT Department of Health and Community Services and Community Controlled Health Services in Central Australia in 2004. *NT Dis Control Bull* 2004; 11(18):20.
- 161 Hocking JS, Willis J, Tabrizi S, Fairley CK, Garland SM, Hellard M. A chlamydia prevalence survey of young women living in Melbourne, Victoria. *Sex Health* 2006; 3(4):235-240.
- 162 Huang RL, Torzillo PJ, Hammond VA, Coulter ST, Kirby AC. Epidemiology of sexually transmitted infections on the Anangu Pitjantjatjara Yankunytjatjara Lands: results of a comprehensive control program. *Med J Aust* 2008; 189(8):442-445.
- 163 Hodgins S, Peeling RW, Dery S, Bernier F, LaBrecque A, Proulx JF et al. The value of mass screening for chlamydia control in high prevalence communities. *Sex Transm Infect* 2002; 78 Suppl 1:i64-i68.
- 164 Steenbeek A, Tyndall M, Sheps S, Rothenberg R. An epidemiological survey of chlamydial and gonococcal infections in a Canadian arctic community. *Sex Transm Dis* 2009; 36(2):79-83.
- 165 Corwin P, Abel G, Wells JE, Coughlan E, Bagshaw S, Sutherland M et al. Chlamydia trachomatis prevalence and sexual behaviour in Christchurch high school students. *NZ Med J* 2002; 115(1158):U107.
- 166 Baud D, Jaton K, Bertelli C, Kulling JP, Greub G. Low prevalence of Chlamydia trachomatis infection in asymptomatic young Swiss men. *BMC Infect Dis* 2008; 8:45.
- 167 Klausner JD, McFarland W, Bolan G, Hernandez MT, Molitor F, Lemp GF et al. Knockknock: a population-based survey of risk behavior, health care access, and Chlamydia trachomatis infection among low-income women in the San Francisco Bay area. *J Infect Dis* 2001; 183(7):1087-1092.
- 168 Turner CF, Rogers SM, Miller HG, Miller WC, Gribble JN, Chromy JR et al. Untreated gonococcal and chlamydial infection in a probability sample of adults. *JAMA* 2002; 287(6):726-733.
- 169 Eggleston E, Rogers SM, Turner CF, Miller WC, Roman AM, Hobbs MM et al. Chlamydia trachomatis infection among 15- to 35-year-olds in Baltimore, MD. Sex *Transm Dis* 2011; 38(8):743-749.
- 170 Wellings K, Field J, Johnson AM, Wadsworth J, Bradshaw S. Sexual Behaviour in Britain: The National Survey of Sexual Attitudes and Lifestyles. London: Penguin Books Ltd; 1994.

- 171 Johnson AM, Mercer CH, Erens B, Copas AJ, McManus S, Wellings K et al. Sexual behaviour in Britain: partnerships, practices, and HIV risk behaviours. *Lancet* 2001; 358(9296):1835-1842.
- 172 Erens B, Phelps A, Clifton S, Mercer CH, Tanton C, Hussey D et al. Methodology of the third British National Survey of Sexual Attitudes and Lifestyles (Natsal-3). *Sex Transm Infect* 2014; 90(2):84-89.
- 173 Rothman KJ, Greenland S, Lash TL. Validity in epidemiologic studies. In: Rothman KJ, Greenland S, Lash TL, editors. Modern Epidemiology. Philadelphia: Lippincott Williams & Wilkins; 2008.
- 174 Groves RM. Nonresponse Rates and Nonresponse Bias in Household Surveys. *Public Opinion Quarterly* 2006; 70(5):646-675.
- 175 Gill ON, Adler MW, Day NE. Monitoring the prevalence of HIV. *BMJ* 1989; 299(6711):1295-1298.
- 176 Groves RM, Presser S, Dipko S. The Role of Topic Interest in Survey Participation Decisions. *Public Opinion Quarterly* 2004; 68(1):2-31.
- 177 McCadden A, Fenton KA, McManus S, Mercer CH, Erens B, Carder C et al. Chlamydia trachomatis testing in the second British national survey of sexual attitudes and lifestyles: respondent uptake and treatment outcomes. *Sex Transm Dis* 2005; 32(6):387-394.
- 178 Fenton KA, Mercer CH, McManus S, Erens B, Wellings K, Macdowall W et al. Ethnic variations in sexual behaviour in Great Britain and risk of sexually transmitted infections: a probability survey. *Lancet* 2005; 365(9466):1246-1255.
- 179 Greenland KE, Op de Coul EL, van Bergen JE, Brouwers EE, Fennema HJ, Gotz HM et al. Acceptability of the Internet-Based Chlamydia Screening Implementation in the Netherlands and Insights Into Nonresponse. *Sex Transm Dis* 2011.
- 180 Op de Coul EL, Gotz HM, van Bergen JE, Fennema JS, Hoebe CJ, Koekenbier RH et al. Who participates in the Dutch Chlamydia screening? A study on demographic and behavioral correlates of participation and positivity. *Sex Transm Dis* 2012; 39(2):97-103.
- 181 Uuskula A, Kals M, McNutt LA. Assessing non-response to a mailed health survey including self-collection of biological material. *Eur J Public Health* 2011; 21(4):538-542.
- 182 Riha J, Mercer CH, Soldan K, French CE, Macintosh M. Who is being tested by the English National Chlamydia Screening Programme? A comparison with national probability survey data. *Sex Transm Infect* 2011; 87(4):306-311.
- 183 Greenland S, Lash TL. Bias analysis. In: Groves RM, Dillman DA, Eltinge JL, Little RJA, editors. Survey Nonresponse. New York: John Wiley & Sons; 2001.
- 184 Brick JM. The Future of Survey Sampling. *Public Opinion Quarterly* 75(5):872-888.
- 185 van Bergen J, Gotz HM, Richardus JH, Hoebe CJ, Broer J, Coenen AJ. Prevalence of urogenital Chlamydia trachomatis increases significantly with level of urbanisation and suggests targeted screening approaches: results from the first national population based study in the Netherlands. Sex Transm Infect 2005; 81(1):17-23.
- 186 Audit commission. The national duplicate registration initiative 2009/10. 2012.
- 187 Prevalence of *N. gonorrhoeae* in U.S., Ages 14-39, 1999-2006. Poster presentation at the 18th Conference of the International Society for STD Research. 2009.

- 188 Monteiro EF, Lacey CJ, Merrick D. The interrelation of demographic and geospatial risk factors between four common sexually transmitted diseases. *Sex Transm Infect* 2005; 81(1):41-46.
- 189 Health Protection Agency. Genital *Chlamydia trachomatis* diagnoses in young adults in England, 2011. 2012. Available from: <u>http://webarchive.nationalarchives.gov.uk/20140714084352/http://www.hpa.org.uk/hpr/archives/2012/hpr2212.pdf</u>
- 190 Centers for Disease Control and Prevention. Sexually Transmitted Disease Surveillance, 2010. 2011.
- 191 Satterwhite CL, Grier L, Patzer R, Weinstock H, Howards PP, Kleinbaum D. Chlamydia positivity trends among women attending family planning clinics: United States, 2004-2008. Sex Transm Dis 2011; 38(11):989-994.
- 192 Fine D, Dicker L, Mosure D, Berman S. Increasing chlamydia positivity in women screened in family planning clinics: do we know why? *Sex Transm Dis* 2008; 35(1):47-52.
- 193 O'Rourke KM, Fairley CK, Samaranayake A, Collignon P, Hocking JS. Trends in Chlamydia positivity over time among women in Melbourne Australia, 2003 to 2007. *Sex Transm Dis* 2009; 36(12):763-767.
- 194 Vodstrcil LA, Fairley CK, Fehler G, Leslie D, Walker J, Bradshaw CS et al. Trends in chlamydia and gonorrhea positivity among heterosexual men and men who have sex with men attending a large urban sexual health service in Australia, 2002-2009. *BMC Infect Dis* 2011; 11:158.
- 195 Ali H, Guy RJ, Fairley CK, Wand H, Chen MY, Dickson B et al. Understanding trends in genital Chlamydia trachomatis can benefit from enhanced surveillance: findings from Australia. *Sex Transm Infect* 2012.
- 196 Lyttle PH. Surveillance report: disease trends at New Zealand sexually transmitted disease clinics 1977-1993. *Genitourin Med* 1994; 70(5):329-335.
- 197 Hiltunen-Back E, Haikala O, Kautiainen H, Paavonen J, Reunala T. A nationwide sentinel clinic survey of chlamydia trachomatis infection in Finland. *Sex Transm Dis* 2001; 28(5):252-258.
- 198 National Institute for Public Health and the Environment. Sexually transmitted infections, including HIV, in the Netherlands in 2010. 2011.
- 199 Bakken IJ, Nordbo SA. Chlamydia trachomatis infection in central Norway--testing patterns and prevalence. *Tidsskr Nor Laegeforen* 2007; 127(24):3202-3205.
- 200 Satterwhite CL, Tian LH, Braxton J, Weinstock H. Chlamydia Prevalence Among Women and Men Entering the National Job Training Program: United States, 2003-2007. Sex Transm Dis 2009.
- 201 MIller WC. Epidemiology of chlamydial infection: are we losing ground? *Sexually Transmitted Infections* 2008; 84:82-86.
- 202 Satterwhite CL, Gray AM, Berman S, Weinstock H, Kleinbaum D, Howards PP. Chlamydia trachomatis infections among women attending prenatal clinics: United States, 2004-2009. *Sex Transm Dis* 2012; 39(6):416-420.
- 203 Burckhardt F, Warner P, Young H. What is the impact of change in diagnostic test method on surveillance data trends in Chlamydia trachomatis infection? *Sex Transm Infect* 2006; 82(1):24-30.

- 204 Dicker LW, Mosure DJ, Levine WC, Black CM, Berman SM. Impact of switching laboratory tests on reported trends in Chlamydia trachomatis infections. *Am J Epidemiol* 2000; 151(4):430-435.
- 205 Gotz H, Lindback J, Ripa T, Arneborn M, Ramsted K, Ekdahl K. Is the increase in notifications of Chlamydia trachomatis infections in Sweden the result of changes in prevalence, sampling frequency or diagnostic methods? *Scand J Infect Dis* 2002; 34(1):28-34.
- 206 Jit M, Vyse A, Borrow R, Pebody R, Soldan K, Miller E. Prevalence of human papillomavirus antibodies in young female subjects in England. *Br J Cancer* 2007; 97(7):989-991.
- 207 Nardone A, Anastassopoulou CG, Theeten H, Kriz B, Davidkin I, Thierfelder W et al. A comparison of hepatitis B seroepidemiology in ten European countries. *Epidemiol Infect* 2009; 137(7):961-969.
- 208 Land JA, van Bergen JE, Morre SA, Postma MJ. Epidemiology of Chlamydia trachomatis infection in women and the cost-effectiveness of screening. *Hum Reprod Update* 2010; 16(2):189-204.
- 209 Wills GS, Horner PJ, Reynolds R, Johnson AM, Muir DA, Brown DW et al. Pgp3 antibody enzyme-linked immunosorbent assay, a sensitive and specific assay for seroepidemiological analysis of Chlamydia trachomatis infection. *Clin Vaccine Immunol* 2009; 16(6):835-843.
- 210 Yamamoto T, Moji K, Kusano Y, Kurokawa K, Kawagoe K, Katamine S. Trend in Chlamydia trachomatis infection among pregnant women in the past ten years in Japan: significance of Chlamydia trachomatis seroprevalence. Sex Transm Dis 1998; 25(10):516-521.
- 211 Lyytikainen E, Kaasila M, Koskela P, Lehtinen M, Patama T, Pukkala E et al. Chlamydia trachomatis seroprevalence atlas of Finland 1983-2003. *Sex Transm Infect* 2008; 84(1):19-22.
- 212 Lyytikainen E, Kaasila M, Hiltunen-Back E, Lehtinen M, Tasanen K, Surcel HM et al. A discrepancy of Chlamydia trachomatis incidence and prevalence trends in Finland 1983-2003. *BMC Infect Dis* 2008; 8:169.
- 213 Horner P, Soldan K, Vieira SM, Wills GS, Woodhall SC, Pebody R et al. C. trachomatis pgp3 Antibody Prevalence in Young Women in England, 1993-2010. *PLoS One* 2013; 8(8):e72001.
- 214 Seroepidemiology to evaluate chlamydia screening programmes: Results from two surveys of pgp3 antibody in residual stored serum samples in England. Poster presentation at the Joint Meeting of the 20th ISSTDR and 14th IUSTI Meeting. 2013.
- 215 World Health Organization. Public Health Surveillance. Available from: http://www.who.int/topics/public\_health\_surveillance/en/
- 216 M'ikantha NM, Lynfield R, Julian KG, Van Beneden CA, de Valk H. Infectious disease surveillance: a cornerstone for prevention and control. In: M'ikantha NM, Lynfield R, Julian KG, Van Beneden CA, de Valk H, editors. Infectious Disease Surveillance. London: Blackwell Publishing Ltd; 2013.
- 217 Public Health England. NCSP Data. Available from: http://www.chlamydiascreening.nhs.uk/ps/info-management.asp
- 218 Health Protection Agency. NCSP: Five Years. The fifth Annual Report of the National Chlamydia Screening Programme 2007/08. 2008.

- 219 Health Protection Agency. Genitourinary Medicine Clinic Activity Dataset (GUMCAD): Access, specifications, guidance materials and policies. Available from: http://www.hpa.org.uk/gumcad#Guidelines\_and\_specifications
- 220 Noble M, McLennan D, Wilkinson K, et al. The English Indices of Deprivation 2007. www.communities.gov.uk/documents/communities/pdf/733520.pdf [ 2008
- 221 Office for National Statistics. Summary Quality Report for Super Output Area Population Estimates. Newport: Office for National Statistics; 2010.
- 222 McNutt LA, Wu C, Xue X, Hafner JP. Estimating the relative risk in cohort studies and clinical trials of common outcomes. *Am J Epidemiol* 2003; 157(10):940-943.
- 223 Satterwhite CL, Tian LH, Braxton J, Weinstock H. Chlamydia prevalence among women and men entering the National Job Training Program: United States, 2003-2007. *Sex Transm Dis* 2010; 37(2):63-67.
- 224 Public Health England. Table 13: Number of contacts & STI diagnoses made through partner notification at GUM clinics in England, 2012. Available from: <u>http://www.hpa.org.uk/stiannualdatatables</u>
- 225 Stone J, Berrington A, Falkingham J. The changing determinants of UK young adults' living arrangements. *Demographic Research* 2011; 25:629-666.
- 226 Health Protection Agency. Further increase in chlamydia screening coverage in 2010/11. *Health Protection Report* 2011; 5(24).
- 227 Robinson, C., Nardone, A., Mercer, C., and Johnson, A. M. Sexual Health (in: Health Survey for England, 2010). 2011.
- 228 Data from the National Chlamydia Screening Programme. 2012.
- 229 National Chlamydia Screening Programme. NCSP Core Requirements (5th Edition). http://www.chlamydiascreening.nhs.uk/ps/resources/corerequirements/1\_Core\_Reqs\_5th\_edition.pdf\_2010
- 230 Field N, Tanton C, Mercer CH, Nicholson S, Soldan K, Beddows S et al. Testing for sexually transmitted infections in a population-based sexual health survey: development of an acceptable ethical approach. J Med Ethics 2012.
- 231 British Associaton of Sexual Health and HIV. Chlamydia trachomatis UK Testing Guidelines. 2010.
- 232 Johnson TP, Cho YI, Campbell RT, Holbrook AL. Using Community-Level Correlates to Evaluate Nonresponse Effects in a Telephone Survey. *Public Opinion Quarterly* 2006; 70(5):704-719.
- 233 Wisniewski CA, White JA, Michel C-EC, Mahilum-Tapay L, Magbuana JPV, Nadala Jr ECB et al. Optimum method of collection of first-void urine for diagnisis of *Chlamydia trachomatis* infection in men. *Journal of Clinical Microbiology* 2008; 46(4):1466-1469.
- 234 Johnson AM, Mercer CH, De Silva N, Desai S, Howell-Jones R, Carder C et al. Epidemiology of, and behavioural risk factors for, sexually transmitted human papillomavirus infection in men and women in Britain. Sexually Transmitted Infections 2012; 88(3):212-217.
- 235 Mercer CH, Tanton C, Prah P, Erens B, Sonnenberg P, Clifton S et al. Changes in sexual attitudes and lifestyles in Britain through the life course and over time: findings from the National Surveys of Sexual Attitudes and Lifestyles (Natsal). *Lancet* 2013; 382(9907):1781-1794.

- 236 General Register Office for Scotland. 2011 Census. Available from: <u>http://www.gro-scotland.gov.uk/census/censushm2011</u>
- 237 Office for National Statistics. Census data. Available from: <u>http://www.ons.gov</u> uk/ons/guide-method/census/2011/census-data/index.html;
- 238 Hadgu A, Dendukuri N, Wang L. Evaluation of screening tests for detecting Chlamydia trachomatis: bias associated with the patient-infected-status algorithm. *Epidemiology* 2012; 23(1):72-82.
- 239 Schachter J, Chow JM, Howard H, Bolan G, Moncada J. Detection of Chlamydia trachomatis by nucleic acid amplification testing: our evaluation suggests that CDCrecommended approaches for confirmatory testing are ill-advised. *J Clin Microbiol* 2006; 44(7):2512-2517.
- 240 Hadgu A, Dendukuri N, Hilden J. Evaluation of nucleic acid amplification tests in the absence of a perfect gold-standard test: a review of the statistical and epidemiologic issues. *Epidemiology* 2005; 16(5):604-612.
- 241 Miller WC. Bias in Discrepant Analysis. When Two Wrongs Don't Make a Right. *Journal* of Clinical Epidemiology 1998; 513(3):219-231.
- 242 Schachter J, Chow JM. The fallibility of diagnostic tests for sexually transmitted diseases: the impact of behavioral and epidemiologic studies. *Sex Transm Dis* 1995; 22(3):191-196.
- 243 Rogan WJ, Gladen B. Estimating prevalence from the results of a screening test. *Am J Epidemiol* 1978; 107(1):71-76.
- 244 Martin DH, Nsuami M, Schachter J, Hook EW, III, Ferrero D, Quinn TC et al. Use of multiple nucleic acid amplification tests to define the infected-patient "gold standard" in clinical trials of new diagnostic tests for Chlamydia trachomatis infections. *J Clin Microbiol* 2004; 42(10):4749-4758.
- 245 Hadgu A. Discrepant analysis: a biased and an unscientific method for estimating test sensitivity and specificity. *J Clin Epidemiol* 1999; 52(12):1231-1237.
- 246 Food and Drug Administration. Guidance for Industry and FDA Staff. Statistical Guidance on Reporting Results from Studies Evaluating Diagnostic Tests. US Department of Health and Human Services; 2007. Available from: <u>http://www.fda.gov/downloads/MedicalDevices/DeviceRegulationandGuidance/Guidance eDocuments/ucm071287.pdf</u>
- 247 Grun L, Tassano-Smith J, Carder C, Johnson AM, Robinson A, Murray E et al. Comparison of two methods of screening for genital chlamydial infection in women attending in general practice: cross sectional survey. *BMJ* 1997; 315(7102):226-230.
- 248 Diggle PJ. Estimating prevalence using an imperfect test. *Epidemiology Research International* 2011; 2011(2011):Article ID 608719.
- 249 Gaydos CA, Theodore M, Dalesio N, Wood BJ, Quinn TC. Comparison of Three Nucleic Acid Amplification Tests for Detection of Chlamydia trachomatis in Urine Specimens. *J Clin Microbiol* 2004; 42(7):3041-3045.
- 250 Skinner CJ, Holt D, Smith TMF (eds). Analysis of Complex Surveys. Chichester: John Wiley; 1989.
- 251 Nelson HD, Helfand M. Screening for chlamydial infection. Am J Prev Med 2001; 20(3 Suppl):95-107.

- 252 Cook RL, Hutchison SL, Ostergaard L, Braithwaite RS, Ness RB. Systematic review: noninvasive testing for Chlamydia trachomatis and Neisseria gonorrhoeae. *Ann Intern Med* 2005; 142(11):914-925.
- 253 Boyadzhyan B, Yashina T, Yatabe JH, Patnaik M, Hill CS. Comparison of the APTIMA CT and GC Assays with the APTIMA Combo 2 Assay, the Abbott LCx Assay, and Direct Fluorescent-Antibody and Culture Assays for Detection of Chlamydia trachomatis and Neisseria gonorrhoeae. J Clin Microbiol 2004; 42(7):3089-3093.
- 254 Head-to-head study comparing performance of the Gen-probe Aptima Combo 2, Abbott LCx probe, Becton Dickinson BDProbeTec ET, and Roche Amplicor Nucleic Acid Amplification assays in detecting Chlamydia trachomatis from first catch urine. Poster presentation C-175. American Society for Microbiology. 2002.
- 255 Comparison of Gen-Probe APTIMA Combo 2 (AC2) assay for C. trachomatis (CT) and N. gonorrhoeae (GC) with ProbeTec and LCx on urine swab specimens using a rotating standard. Poster presentation C-174. American Society for Microbiology. 2002.
- 256 Simultaneous detection of Chlamydia trachomatis and Neisseria gonorrhoeae in the APTIMA Combo 2 assay by testing cervical swabs previously tested by LCx or ProbeTec assays. Poster presentation P1174. European Society for Clinical Microbiology. 2002.
- 257 Wheeler CM, Castellsague X, Garland SM, Szarewski A, Paavonen J, Naud P et al. Cross-protective efficacy of HPV-16/18 AS04-adjuvanted vaccine against cervical infection and precancer caused by non-vaccine oncogenic HPV types: 4-year end-ofstudy analysis of the randomised, double-blind PATRICIA trial. *Lancet Oncol* 2012; 13(1):100-110.
- 258 Mesher D, Soldan K, Howell-Jones R, Panwar K, Manyenga P, Jit M et al. Reduction in HPV 16/18 prevalence in sexually active young women following the introduction of HPV immunisation in England. *Vaccine* 2013; 32(1):26-32.
- 259 Dukers-Muijrers NH, Morre SA, Speksnijder A, van der Sande MA, Hoebe CJ. Chlamydia trachomatis test-of-cure cannot be based on a single highly sensitive laboratory test taken at least 3 weeks after treatment. *PLoS One* 2012; 7(3):e34108.
- 260 Vodstrcil LA, McIver R, Huston WM, Tabrizi SN, Timms P, Hocking JS. The Epidemiology of Chlamydia trachomatis Organism Load During Genital Infection: A Systematic Review. J Infect Dis 2015; 211(10):1628-1645.
- 261 Lanjouw E, Ossewaarde JM, Stary A, Boag F, van der Meijden WI. 2010 European guideline for the management of Chlamydia trachomatis infections. *Int J STD AIDS* 2010; 21(11):729-737.
- 262 Sheringham J, Mann S, Simms I, Stafford M, Hart GJ, Raine R. It matters what you measure: a systematic literature review examining whether young people in poorer socioeconomic circumstances are more at risk of chlamydia. *Sex Transm Infect* 2013; 89(2):175-180.
- 263 Cook RL, Clark DB. Is there an association between alcohol consumption and sexually transmitted diseases? A systematic review. *Sex Transm Dis* 2005; 32(3):156-164.
- 264 Payne R.A., Abel GA. UK indices of multiple deprivation a way to make comparisons across constiuent countries easier. *Health Statistics Quarterly* 2012; 53.
- 265 Johnson SA, Simms I, Sheringham J, Bickler G, Bennett CM, Hall R et al. The implementation of chlamydia screening: a cross-sectional study in South East of England. *Sexually Tranmitted Infections* 2010; 86(3):217-221.
- 266 Public Health England. Sexually transmitted infections and chlamydia screening in England, 2012. 2013. Available from:

http://webarchive.nationalarchives.gov.uk/20140714084352/http://www.hpa.org.uk/hpr/a rchives/2013/hpr2313.pdf

- 267 Marmot, M., Allen, J., Goldblatt, P., Boyce, T., McNeish, Di., Grady, M., and Geddes, I. Fair Society, Healthy Lives (The Marmot Review). The Marmot Review; 2010. Available from: <u>http://www.instituteofhealthequity.org/projects/fair-society-healthy-lives-themarmot-review</u>
- 268 Mercer CH, Copas AJ, Sonnenberg P, Johnson AM, McManus S, Erens B et al. Who has sex with whom? Characteristics of heterosexual partnerships reported in the second National Survey of Sexual Attitudes and Lifestyles. *International Journal of Epidemiology* 2009; 38:206-2014.
- 269 Horner PJ, Wills GS, Righarts A, Vieira S, Samuel D, Winston A et al. Sensitive detection of *Chlamydia trachomatis* Pgp3 antibody demonstrates antibody persistence and correlates with self-reported infection and behavioural risks in a blinded cohort study. *SUBMITTED: Lancet Infectious Diseases* 2015.
- 270 Health & Social Care Information Centre. Health Survey for England 2012. Volume 2 Methods and documentation. 2013. available from: <u>http://www.hscic.gov.uk/catalogue/PUB13218/HSE2012-Methods-and-docs.pdf</u>
- 271 Health & Social Care Information Centre. Health Survey for England 2008. Volume 2 Methods and documentation. 2009.Available from: <u>http://www.hscic.gov.uk/catalogue/PUB00430/heal-surv-phys-acti-fitn-eng-2008-rep-v3.pdf</u>
- 272 Prah P. Asking about Sex in General Health Surveys: Comparing Data Collected By the 2010 Health Survey for England with the Third Natsal Survey of Sexual Attitudes and Lifestyles. Oral presentation (3F4) at the 2014 STD Prevention Conference. Atlanta, GA, USA. *Sexually Transmitted Diseases* 2014; 41(June, Supplement 1).
- 273 Nicoll A, Hughes G, Donnelly M, Livingstone S, De Angelis D, Fenton K et al. Assessing the impact of national anti-HIV sexual health campaigns: trends in the transmission of HIV and other sexually transmitted infections in England. *Sex Transm Infect* 2001; 77(4):242-247.
- 274 Woodhall SC, Torrone E, Fine D, Salomon SG, Natsukasa-Ono W, Soldan K et al. How do changes in the population tested for chlamydia over time affect observed trends in chlamydia positivity? Analysis of routinely collected data from young women tested for chlamydia in family planning clinics in the Pacific Northwest (USA), 2003 to 2010. *Sexual Health* 2015; Under review (as at April 2015).
- 275 Brook G. The performance of non-NAAT point-of-care (POC) tests and rapid NAAT tests for chlamydia and gonorrhoea infections. An assessment of currently available assays. *Sex Transm Infect* 2015.
- 276 Aicken C, Estcourt CS, Gibbs J, Sonnenberg P, Mercer CH, Tickle L et al. 001 pp: online clinical management pathways for Chlamydia treatment: enriching formative evaluation of a complex e-health intervention. *BMJ Open* 2015; 5(4):UCLSymposiumAbstracts12.
- 277 Hocking JS, Spark S, Guy R, Temple-Smith M, Fairley CK, Kaldor J et al. The Australian chlamydia control effectiveness pilot (ACCEPt): first results from a randomised trial of annual chlamydia screening in general practice. Oral presentation (O8) at the 4th joint BASHH-ASTDA meeting in Brighton, UK, 27-29 June 2012. Sexually Tranmitted Infections 2012; 88(Supplement 1).
- 278 Yeung AH, Temple-Smith M, Fairley CK, Vaisey AM, Guy R, Law MG et al. Chlamydia prevalence in young attenders of rural and regional primary care services in Australia: a cross-sectional survey. *Med J Aust* 2014; 200(3):170-175.

- 279 Public Health England. Table 3: Selected STI diagnoses & rates by gender, sexual risk & age group, 2009-2013. Available from: <u>www.gov.uk/government/uploads/system/uploads/attachment\_data/file/340434/Table\_3</u> <u>Selected\_STI\_diagnoses\_and\_rates\_by\_gender\_sexual\_risk\_and\_age\_group.pdf</u>
- 280 Prah P, Copas AJ, Mercer CH, Nardone A, Johnson AM. Patterns of sexual mixing with respect to social, health and sexual characteristics among heterosexual couples in England: analyses of probability sample survey data. *Epidemiol Infect* 2014;1-11.
- 281 McClean H, Carne C, Bunting P, Bhaduri S, Fernandes A, Dhar J et al. UK National Audit of chlamydial infection management in sexual health clinics. case notes audit: information-giving, partner notification and follow-up. *Int J STD AIDS* 2008; 19(7):473-476.
- 282 National Chlamydia Screening Programme. National Chlamydia Screening Programme (NCSP) data presentation: England April 2011-March 2012. Available from: <u>http://www.chlamydiascreening.nhs.uk/ps/resources/data-</u> <u>tables/DP\_England\_2011\_12.pdf</u>
- 283 McClean H, Carne C, Bunting P, Bhaduri S, Fernandes A, Dhar J et al. UK National Audit of Chlamydial Infection Management in Sexual Health Clinics. Case notes audit: demography, diagnosis and treatment. *International Journal of STD & AIDS* 2008; 19:469-472.
- 284 Rose G. Rose's Strategy of Preventive Medicine. Oxford: Oxford University Press; 2008.
- 285 Kneale D, Fletcher A, Wiggins R, Bonell C. Distribution and determinants of risk of teenage motherhood in three British longitudinal studies: implications for targeted prevention interventions. *J Epidemiol Community Health* 2013; 67(1):48-55.
- 286 Verhoeven V, Avonts D, Van RP, Weyler J, Wang X, Stalpaert M. Performance of a screening algorithm for chlamydial infection in 2 samples of patients in general practice. *Scand J Infect Dis* 2004; 36(11-12):873-875.
- 287 Falasinnu T, Gilbert M, Gustafson P, Shoveller J. Deriving and validating a risk estimation tool for screening asymptomatic chlamydia and gonorrhea. *Sex Transm Dis* 2014; 41(12):706-712.
- 288 Gift TL, Blake DR, Gaydos CA, Marrazzo JM. The cost-effectiveness of screening men for Chlamydia trachomatis: a review of the literature. *Sex Transm Dis* 2008; 35(11 Suppl):S51-S60.
- 289 Fuller SS, Mercer CH, Copas AJ, Saunders J, Sutcliffe LJ, Cassell JA et al. The SPORTSMART study: a pilot randomised controlled trial of sexually transmitted infection screening interventions targeting men in football club settings. *Sex Transm Infect* 2015; 91(2):106-110.
- 290 Turner K, Adams E, Grant A, Macleod J, Bell G, Clarke J et al. Costs and cost effectiveness of different strategies for chlamydia screening and partner notification: an economic and mathematical modelling study. *BMJ* 2010; 342:c7250.
- 291 UK Clinical Research Network. UK Clinical Research Network: Portfolio Database. Measurement of chlamydia antibodies using residual GUM specimens (UKCRN 18521). Available from: <u>http://public.ukcrn.org.uk/search/StudyDetail.aspx?StudyID=18521</u>
- 292 Althaus CL, Turner KM, Schmid BV, Heijne JC, Kretzschmar M, Low N. Transmission of Chlamydia trachomatis through sexual partnerships: a comparison between three individual-based models and empirical data. *J R Soc Interface* 2012; 9(66):136-146.

# APPENDIX 1: Study paperwork used in pilot postal survey to measure the population prevalence of chlamydia among young women

### **Relevant to Chapter 5**

The following documents are available in the printed version of the thesis:

- Patient Information leaflet (Leaflets varied by group. The leaflet for group A is provided as an example)
- Study questionnaire
## APPENDIX 2: Comparison of maximal and parsimonious multivariable models of prevalent chlamydia infection, recent diagnosis and recent testing

#### Relevant to Chapter 7

As set out in Chapter 7, two approaches to multivariable modelling to investigate associations between sociodemographic and behavioural variables and prevalent chlamydia infection, recent diagnosis and recent testing were explored: maximal and parsimonious models were constructed for prevalent chlamydia infection, self-reported chlamydia diagnosis in the last year and selfreported chlamydia test in in the last year.

To fit parsimonious models, variables found significant at p<0.1 in univariable analyses were entered into a multivariable model selection process based on backwards selection whereby variables were removed from the model if the statistical significance of their association with the outcome had an associated p value of >0.05. Only those variables significantly associated with the outcome were retained in the final parsimonious models. With two exceptions, all variables included in univariable models were included in multivariable maximal models: number of sexual partners in the last year was not included due to collinearity with other sexual partnership variables; age left school was not included as data were unavailable for 16 year-olds.

Table 9-2below shows results from univariable models along with maximal (Multivariable model 1) and parsimonious (Multivariable model 2) multivariable models for women for the three outcomes of interest (prevalent infection, recent diagnosis, recent testing). Table 9-3 shows results for men.

Adjusted ORs in the maximal models tended to be closer to 1 than those in parsimonious models, reflecting some confounding of associations between predictor and outcome variables. Factors which remained statistically significant were largely comparable between models, with some exceptions. Firstly, in men the maximal model found deprivation of area of residence to be significantly associated with prevalent infection and non-condom use was not, whereas in the parsimonious model non-condom use remained a significant predictor in the model whereas deprivation did not. Age group and number of lifetime sexual partners were significant in both models. The adjusted odds ratio (OR) for prevalent infection among those reporting non-condom use was similar between the models (3.59 versus 4.43 in the maximal and parsimonious models respectively). The difference between the models likely reflects, in part, the sample size available meaning that maximal models would have reduced power to detect associations after adjusting for multiple confounders. In the maximal models, no factors were statistically significant predictors of recent diagnosis among those tested, reflecting the smaller sample size available when limiting to those recently tested and when entering several covariates into the model.

# Table 9-2: Percentage with a) prevalent chlamydia infection, b) self-reported diagnosis in the last year and c) self-reported testing and unadjusted and adjusted odds ratios for prevalent chlamydia infection by sociodemographic and behavioural variables (16-24 year-old sexually-experienced <u>women</u>)

-		1a) Pr	evale	nt infectior	n dete	ected i	n urine (n=	=992)				
	%	(95%CI)	I	Univariable	•	Multiv	ariable mo	del 1	l Mult	ivariable m	odel	Denominator
										2		(wt, unwt)
			OR	(95%CI)	р	$AOR^{\flat}$	(95%CI)	р	$AOR^{b}$	(95%CI)	р	
Age group												
16-19	3.8%	(2.2-6.3)	1.00	-	0.36	1.00	-	0.50				214, 395
20-24	2.7%	(1.7-4.3)	0.71	(0.35 - 1.46)		0.71	(0.27-1.87)					383, 597
Country <sup>c</sup>												
England	2.9%	(2.0-4.3)	1.00	-	0.53	1.00	-	0.48				504, 817
Scotland	3.1%	(1.1-8.6)	1.08	(0.35 - 3.33)		1.34	(0.43-4.14)					56, 103
Wales	5.3%	(1.9-13.8)	1.87	(0.63 - 5.60)		1.88	(0.63 - 5.54)					37, 72
IMD quintile of LSOA	of res	idence <sup>d</sup>										
2 least deprived	1.3%	(0.5-3.4)	1.00	-	0.01	1.00	-	0.01	1.00	-	<0.01	213, 355
Middle quintile	1.8%	(0.8-4.2)	1.37	(0.38 - 4.90)		1.40	(0.39-4.98)		1.45	(0.40-5.20)		111, 174
2 most deprived	4.9%	(3.3-7.3)	3.82	(1.35 - 10.8)		4.23	(1.53-11.6)		4.11	(1.45-11.7)		273, 463
Age left school <sup>e</sup>												
17+	3.2%	(2.1-4.8)	1.00	-	0.88							445, 700
16	3.4%	(1.9-6.0)	1.06	(0.50 - 2.22)								120, 229
Age at first heterosex	ual se	x										
17+	1.6%	(0.7-3.7)	1.00	-	0.15	1.00	-	0.42				188, 246
16	3.9%	(2.1-6.9)	2.52	(0.86 - 7.36)		2.20	(0.67-7.17)					178, 304
<16	4.0%	(2.4-6.6)	2.65	(0.95 - 7.36)		1.82	(0.60-5.42)					213, 415
Number of sexual par	tners	in the last	year									
0 or 1	2.5%	(1.5-4.0)	1.00	-	0.03							387, 600
2	3.9%	(1.8-8.5)	1.62	(0.63 - 4.15)								90, 161
3 to 4	1.9%	(0.7-5.1)	0.75	(0.24 - 2.33)								63, 127
5+	8.3%	(3.9-16.8)	3.57	(1.39 - 9.17)								49, 93
Number of new sexua	al part	ners in the	last y	/ear								
0	2.2%	(1.3-3.7)	1.00	-	0.03	1.00	-	0.70				313, 495
1	2.8%	(1.2-6.3)	1.26	(0.47-3.41)		1.17	(0.38-3.52)					160, 263
2+	5.9%	(3.5-9.8)	2.73	(1.26-5.93)		1.65	(0.50-5.39)					118, 225
Number of sexual par	tners	in the last	year	without a c	ondo	om						
0	2.9%	(1.2-7.1)	1.00	-	0.03	1.00		0.18				120, 173
1	2.2%	(1.4-3.6)	0.76	(0.26-2.17)		0.34	(0.10-1.10)					368, 606
2+	6.3%	(3.5-11.2)	2.25	(0.73-6.93)		0.49	(0.12-1.83)					108, 212
Number of lifetime se	xual p	bartners										
1 to 4	2.4%	(1.4-4.2)	1.00	-	0.14	1.00	-	0.67				321, 482
5 to 9	2.8%	(1.4-5.2)	1.15	(0.48-2.75)		0.87	(0.32-2.32)					150, 267
10+ Oser dama and familia	5.4%	(2.9-9.7)	2.29	(0.97-5.39)		1.39	(0.45-4.18)					121, 234
Condom used for mos	st rec			st recent p		r 4 00		0.00				000 044
res	2.8%	(1.5-5.3)	1.00	-	0.53	1.00	-	0.29				202, 314
NU Concurrent northereb	3.0% inc in	(2.4-5.4)	1.20	(0.59-2.60)		1.59	(0.67-3.74)					332, 013
Concurrent partnersn		last year	4 00		0.40	4 00		0.05				400 700
NO	2.1%	(1.8-4.2)	1.00	-	0.19	1.00	-	0.85				439,706
100	0.3%	(2.3-13.4)	2.40	(0.34-0.15)		1.34	(0.40-3.70) (0.20.2.24)					72 107
	J.1%	(1.3-1.2) a	1.14	(0.43-3.04)		1.13	(0.30-3.31)					13, 121
never / <monthly< td=""><td>2 50/</td><td>9 (15_/1)</td><td>1 00</td><td></td><td><b>~</b>0 01</td><td>1 00</td><td>_</td><td>0.01</td><td>1 00</td><td>_</td><td>&lt;0.01</td><td>373 500</td></monthly<>	2 50/	9 (15_/1)	1 00		<b>~</b> 0 01	1 00	_	0.01	1 00	_	<0.01	373 500
monthly	2.0%	$(1.0^{-4}.1)$	0.10	-	-0.01	0.46	- (0 13-1 51)	0.01	0.5/	- (0 17-1 69)	<b>\U.U</b>	130 226
Swookly	7 00/	(0.3-3.3)	2 25	(1 55-7 25)		2 51	(0.10-1.01) (1.08-5.76)		3 72	(0.17-1.00) (1.73_8.09)		05 169
	1.9%	eriencelco	0.00	(1.00-7.20)		2.01	(1.00-0.70)		5.15	(1.75-0.00)		55, 100
No	3 10/2	(2 1-4 4)	1 00	-	0.92	1 00	-	0 49				473 750
Yes	3 20/2	(-1, -1, -1, -1)	1.00	(0 42-2 63)	0.02	0.74	(0 30-1 76)	0.40				124 242
	0.270	A 117 1 1 1 1	1.00	10.72 2.00)		0.74	(0.00 1.70)					127, 272

### Table 9-2 continued.

		1b) [ (2m	year								
	%	(95%Cl)	ong t	Univariable	а III с Э	Multivariable mo	odel 1	l Multiv	ariable mo	del 2	Denominator
	,.	()	OR	(95%CI)	р	AOR <sup>b</sup> (95%CI)	р	AOR⁵	(95%CI)	р	(wt, unwt)
Age group				· · /	· ·	, ,			· /		
16-19	6.0%	6(3.8-9.2)	1.00	-	0.62	1.00-	0.58				193, 375
20-24	5.1%	5(3.4-7.6)	0.86	(0.46-1.60)		0.80(0.35-1.78)					329, 565
Country <sup>c</sup>											
England											469, 832
Scotland											30, 58
Wales		d d									24, 50
IMD quintile of LSOA	of res	sidence"									
2 least deprived	4.8%	6(2.8-8.1)	1.00	-	0.23	1.00-	0.36				183, 319
Middle quintile	3.5%	6(1.6-7.3)	0.71	(0.27-1.86)		1.06(0.37-3.04)					102, 176
2 most deprived	6.8%	5(4.6-10.0)	1.46	(0.73-2.93)		1.70(0.73-3.91)					236, 445
Age left school											
17+	5.2%	(3.6-7.5)	1.00	-	0.58						387, 658
16	6.2%	5(3.7-10.5)	1.21	(0.62-2.37)							109, 228
Age at first heterosex	ual s	ex									
1/+	4.1%	b(1.8-8.9)	1.00	-	0.37	1.00-	0.93				137, 215
16	5.0%	(2.8 - 8.9)	1.24	(0.44-3.55)		0.80(0.26-2.42)					154, 272
<16	6.9%	6(4.8-9.9)	1.76	(0.70-4.41)		0.89(0.35-2.25)					220, 429
Number of sexual par	tners	in the last	year								
0 or 1	3.0%	b(1.7-5.4)	1.00	-	0.01						291, 507
2	6.6%	6(3.7-11.6)	2.25	(0.98-5.18)							93, 178
3 to 4	8.8%	6(5.2-14.6)	3.08	(1.34-7.05)							76, 146
5+	11.6%	6.2-20.8)	4.19	(1.69-10.4)							58, 101
Number of new sexua	al part	iners in the	last y	/ear							
0	2.8%	b(1.5-5.0)	1.00		<0.01	1.00-	0.11	1.00 ·		0.02	226, 397
1	4.8%	(2.5-9.1)	1.76	(0.71-4.38)		1.89(0.69-5.16)		1.82	0.74-4.50)		156, 287
2+	10.7%	o(7.3-15.6)	4.23	(2.04-8.79)		3.09(1.07-8.86)		3.07	(1.41-6.67)		137, 249
Number of sexual par	tners	in the last	year	without a c	ondo	om L a a					70 400
0	4.2%	b(1.5-11.6)	1.00	-	<0.01	1.00-	0.84				76, 130
1	3.8%	(2.4-6.1)	0.90	(0.27-2.94)		0.72(0.15-3.21)					319, 567
2+	10.3%	b(7.0-15.0)	2.60	(0.80-8.46)		0.90(0.18-4.49)					126, 241
Number of lifetime se	xual	partners	4 00			4.00		4 00			040 004
1 to 4	2.1%	(1.1-4.0)	1.00	-	<0.01	1.00-	0.08	1.00		<0.01	218, 391
5 to 9	6.2%	(3.7-10.3)	3.07	(1.33-7.07)		2.40(0.90-6.37)		2.54	1.06-6.12)		161, 289
10+	9.9%	6.5-14.7	5.12	(2.31-11.3)		3.76(1.19-11.8)		3.81	(1.67-8.68)		141, 254
Condom used for mo	St rec	ent sex wit	n mo	st recent p	arthe	r 1.00	0 1 4				470 000
Yes	3.4%	(1.6-7.0)	1.00	-	0.14	1.00-	0.14				170, 303
NO	6.3%	6(4.4-8.9)	1.90	(0.81-4.49)		1.88(0.81-4.32)					328, 592
Concurrent partners			4 00		0 4 2	1.00	0.46				204 020
NO	4.3%	(2.9-6.4)	1.00	-	0.13	1.00-	0.40				301, 039
Yes	8.1%	(4.1-15.3)	1.95	(0.84 - 4.54)		0.75(0.27-2.00)					78, 146
	7.8%	6(4.3-13.9)	1.88	(0.90-3.94)		1.46(0.62-3.38)					76, 144
Frequency of binge a			4 00		0 02	4.00	0.00				040 504
never / <monthly< td=""><td>3.1%</td><td>(2.4-5.7)</td><td>1.00</td><td></td><td>0.03</td><td>1.00-</td><td>0.28</td><td></td><td></td><td></td><td>312, 561</td></monthly<>	3.1%	(2.4-5.7)	1.00		0.03	1.00-	0.28				312, 561
montnly	6.6%	o(3.6-11.8)	1.85	(0.85-4.01)		1.91(0.74-4.89)					112, 200
	9.4%	o(5.2-16.3)	2.69	(1.25-5.80)		2.06(0.75-5.61)					97,177
Ever nad any same se	ex exp		ntact		0.70	1.00	0.44				000 704
NO Vac	5.3%	(3.7-7.5)	1.00	-	0.72	1.00-	0.44				388,704
res	5.9%	o(3.5-10.0)	1.13	(0.57-2.24)		0.71(0.30-1.68)					134, 236

#### Table 9-2 continued.

#### 1c) Tested for chlamydia in the last year (n=1,375)

	0/_	(05%CI)	I	Univariable	e	Multiv	ariable mo	odel 1	Multi	ivariable n 2	nodel	Denominator
	70	(35/801)	OR	(95%CI)	р	AOR⁵	(95%CI)	р	AOR⁵	_ (95%CI)	р	(wt, unwt)
Age group				, ,			, ,			, ,		
16-19	56.6%	(52.5-60.6)	1.00	-	0.16	1.00	-	0.15				343, 672
20-24	52.8%	(49.2-56.4)	0.86	(0.69-1.06)		0.82(	0.62-1.07)					623, 1064
Country <sup>c</sup>		````		、 ,			,					
England	57.1%	(54.1-60.1)	1.00	-	<0.01	1.00	-	<0.01	1.00-	-	<0.01	823, 1452
Scotland	32.4%	(24.4-41.5)	0.36	(0.24-0.54)		0.29(	0.18-0.45)		0.31(	(0.21 - 0.47)		91, 178
Wales	45.6%	(36.2-55.4)	0.63	(0.42-0.94)		0.53	0.32-0.85)		0.54	0.34-0.86)		52, 106
IMD quintile of LSOA	A of res	idence <sup>d</sup>		````			,			,		
2 least deprived	54.2%	(49.5 - 58.8)	1.00	-	0.99	1.00	-	0.94				338, 595
Middle quintile	54.4%	(48.0-60.7)	1.01	(0.74-1.38)		1.03(	0.71-1.48)					189, 324
2 most deprived	54.0%	(49.8-58.2)	0.99	(0.77-1.27)		0.97(	0.73-1.29)					439, 817
Age left school <sup>e</sup>												
17+	54.3%	(51.0-57.6)	1.00	-	0.70							715, 1217
16	55.5%	(50.1 - 60.8)	1.05	(0.82-1.35)								196, 405
Age at first heterose	xual se	x		. ,								
17+	43.9%	(38.7 - 49.1)	1.00	-	<0.01	1.00	-	0.05	1.00-	-	<0.01	313, 489
16	56.4%	(51.6 - 61.1)	1.66	(1.25-2.20)		1.39(	0.99-1.92)		1.48(	(1.09-2.01)		273, 503
<16	63.9%	(59.8 - 67.9)	2.27	(1.75 - 2.94)		1.44(	1.05-1.97)		1.66(	(1.24-2.21)		344, 680
Number of sexual pa	artners	in the last	year									
0 or 1	46.6%	(43.3-50.0)	1.00	-	<0.01							624, 1096
2	65.2%	(58.8-71.1)	2.15	(1.59-2.89)								143, 275
3 to 4	69.7%	(62.0-76.4)	2.63	(1.82-3.80)								111, 210
5+	74.8%	(64.4 - 83.0)	3.40	(2.05 - 5.65)								77, 135
Number of new sexu	al part	ners in the	last y	/ear								
0	45.6%	(41.9-49.3)	1.00	-	<0.01	1.00	-	<0.01	1.00-	-	<0.01	495, 873
1	59.2%	(54.0-64.2)	1.73	(1.34-2.23)		1.69(	(1.25-2.27)		1.94(	(1.48-2.54)		264, 485
2+	70.0%	(63.8-75.6)	2.79	(2.03-3.84)		1.46(	(0.95-2.21)		2.03(	(1.40-2.94)		197, 359
Number of sexual pa	artners	in the last	year \	without a c	ondo	m						
0	36.4%	(30.8-42.3)	1.00	-	<0.01			0.05	1.00-	-	<0.01	210, 361
1	54.7%	(51.4-58.0)	2.11	(1.61-2.78)		1.52(	(1.03-2.24)		1.74(	(1.29-2.35)		585, 1049
2+	74.2%	(68.3-79.4)	5.04	(3.47-7.32)		1.86(	(1.09-3.15)		2.19(	(1.43-3.36)		169, 322
Number of lifetime s	exual p	partners										
1 to 4	43.5%	(40.0-47.1)	1.00	-	<0.01	1.00	-	<0.01	1.00-	-	<0.01	503, 894
5 to 9	63.8%	(58.5-68.7)	2.29	(1.76-2.96)		1.96(	(1.43-2.69)		1.90(	(1.43-2.54)		252, 453
10+	69.6%	(63.9-74.7)	2.97	(2.22-3.98)		2.11(	(1.41-3.13)		2.00(	(1.42-2.82)		202, 373
Condom used for me	ost rec	ent sex wit	h mos	st recent p	artne	r						
Yes	51.6%	(46.9-56.3)	1.00	-	0.03	1.00	-	0.97				330, 586
No	58.0%	(54.6-61.3)	1.30	(1.03-1.63)		1.01(	(0.74-1.35)					567, 1027
Concurrent partners	hips in	last year										
No	51.0%	(47.8-54.2)	1.00		<0.01	1.00	-	0.05				710, 1256
Yes	74.6%	(67.1-80.8)	2.81	(1.93-4.09)		1.46(	0.93-2.28)					105, 196
Unknown	64.6%	(57.5-71.2)	1.76	(1.27-2.42)		1.53(	1.02-2.28)					117, 226
Frequency of binge	drinkin	g										
never / <monthly< td=""><td>52.1%</td><td>(48.7-55.5)</td><td>1.00</td><td>-</td><td>0.01</td><td>1.00</td><td>-</td><td>0.08</td><td></td><td></td><td></td><td>601, 1085</td></monthly<>	52.1%	(48.7-55.5)	1.00	-	0.01	1.00	-	0.08				601, 1085
monthly	52.4%	(46.6-58.2)	1.01	(0.78-1.32)		0.75(	0.54-1.02)					214, 375
<u>&gt;</u> weekly	64.4%	(57.6-70.7)	1.66	(1.22-2.27)		1.16(	0.81-1.64)					151, 274
Ever had any same s	sex exp	erience/co	ntact			4.00		·				740 40-0
NO	51.9%	(48.8-55.0)	1.00	-	<0.01	1.00	-	0.24				/49, 1352
Yes	61.9%	(56.0-67.5)	1.51	(1.15-1.98)		1.21(	<u>U.88-1.65)</u>					218, 384

<sup>a</sup>N in column headings shows unweighted denominators. Total denominators by characteristic and in multivariable models vary due to item-missingness. <sup>b</sup>AOR: Adjusted odds ratios, adjusted for all variables shown. <sup>c</sup>Results for recent diagnosis are not reported due to small sample size in Scotland and Wales when limited to those tested. <sup>d</sup>IMD: Index of multiple deprivation of LSOA (lower super output area) of residence. IMD scores for England, Scotland and Wales were adjusted before being combined and assigned to quintiles, using the method described by Payne and Abel<sup>264</sup>. <sup>e</sup>Excludes 16 year-olds. <sup>f</sup>Among those with at least 1 sexual partner in last year.

# Table 9-3: Percentage with a) prevalent chlamydia infection, b) self-reported diagnosis in the last year and c) self-reported testing and unadjusted and adjusted odds ratios for prevalent chlamydia infection by sociodemographic and behavioural variables (16-24 year-old sexually-experienced <u>men</u>)

		2a) Pr	evale	nt infection	n dete	ected in	n urine (n	=840)				
	%	(95%CI)	l	Univariable	9	Multiv	ariable m	odel 1	Multi	ivariable m 2	odel	Denominator
	70	(00/001)	OR	(95%CI)	р	AOR⁵	(95%CI)	р	$AOR^{b}$	(95%CI)	р	(wt, unwt)
Age group												
16-19	0.3%	(0.1-1.4)	1.00	-	<0.01	1.00	-	0.02	1.00-	-	0.03	234, 343
20-24	3.4%	(2.2-5.2)	10.6	(2.40-46.3)		7.54(	(1.37-41.3	5)	5.71(	(1.23-26.6)		391, 497
Country												
England	1.9%	(1.2-3.0)	1.00	-	0.12	1.00	-	0.27				532, 719
Scotland	5.7%	(2.1-14.3)	3.13	(1.04-9.45)		3.16(	(0.78-12.8	5)				60, 72
Wales	1.7%	(0.2-12.1)	0.88	(0.11-7.02)		1.20(	(0.18-7.63	5)				33, 49
IMD quintile of LSOA	of res	idenceª										
2 least deprived	1.3%	(0.4-3.6)	1.00	-	0.14	1.00	-	0.04				241, 315
Middle quintile	1.6%	(0.6-4.4)	1.24	(0.26-5.88)		1.01(	(0.15-6.68	5)				114, 164
2 most deprived	3.4%	(2.1-5.6)	2.71	(0.83-8.82)		3.75 (	(1.11-12.5	5)				269, 361
Age left school <sup>e</sup>												
17+	1.6%	(0.9-2.7)	1.00	-	0.01							439, 568
16	5.0%	(2.7-9.2)	3.28	(1.38-7.82)								143, 206
Age at first heterosex	ual se	x										
17+	1.0%	(0.3-2.8)	1.00	-	0.03	1.00	-	0.67				210, 245
16	1.5%	(0.5-4.6)	1.49	(0.31-7.15)		1.14(	(0.27-4.74	)				148, 205
<16	4.0%	(2.4-6.5)	4.18	(1.31-13.3)		1.65 (	(0.55-4.90	))				238, 352
Number of sexual par	tners	in the last	year									
0 or 1	1.5%	(0.7-3.0)	1.00	-	0.01							359, 466
2	1.3%	(0.4-4.2)	0.86	(0.21-3.56)								123, 159
3 to 4	3.1%	(1.1-8.5)	2.16	(0.60-7.78)								70, 110
5+	7.5%	(3.7-14.6)	5.47	(1.89-15.9)								67, 100
Number of new sexua	al part	ners in the	last y	vear								
0	1.8%	(0.9-3.8)	1.00	-	0.01	1.00	-	0.49				263, 335
1	0.8%	(0.2-2.5)	0.42	(0.10-1.72)		0.33 (	(0.05-2.06	5)				203, 270
2+	5.1%	(2.9-8.8)	2.87	(1.08-7.63)		0.47 (	(0.09-2.45	5)				152, 229
Number of sexual par	tners	in the last	year \	without a c	ondo	m						
0	0.3%	(0.1-1.3)	1.00		<0.01	1.00	-	0.09				205, 248
1	1.7%	(0.8-3.6)	5.26	(1.05-26.2)		1.23(	(0.09-15.2	2)				287, 396
2+	6.5%	(3.9-10.9)	21.3	(4.67-97.2)		4.95 (	(0.42-57.9	))				130, 194
Number of lifetime se	xual p	artners										
1 to 4	0.4%	(0.1-1.5)	1.00		<0.01	1.00	-	0.03	1.00-	•	<0.01	332, 412
5 to 9	1.2%	(0.3-4.0)	3.21	(0.47-21.9)		1.78(	0.20-15.5	)	2.21(	(0.31-15.9)		141, 200
10+	7.6%	(4.8-11.7)	22.6	(4.92-104)		8.69 (	(1.21-62.0	))	14.2(	(3.02-66.7)		148, 224
Condom used for mo	st rec	ent sex wit	h mos	st recent p	artne	r						
Yes	0.7%	(0.2-2.0)	1.00		<0.01	1.00		0.10	1.00-		<0.01	301, 391
No	4.1%	(2.6-6.4)	6.03	(1.87-19.4)		3.59 (	(0.77-16.6	5)	4.43(	(1.32-14.9)		283, 398
Concurrent partnersh	ips in	last year										
No	2.6%	(1.6-4.2)	1.00		0.74	1.00	-	0.05				418, 557
Yes	2.0%	(0.7-5.4)	0.75	(0.24-2.37)		0.18(	0.04-0.71	)				82, 121
Unknown	1.7%	(0.5-5.6)	0.64	(0.17-2.37)		0.60(	(0.11-3.00	))				92, 122
Frequency of binge d	rinkin	g										
never / <monthly< td=""><td>1.1%</td><td>(0.4-2.7)</td><td>1.00</td><td>-</td><td>0.08</td><td>1.00</td><td></td><td>0.62</td><td></td><td></td><td></td><td>322, 421</td></monthly<>	1.1%	(0.4-2.7)	1.00	-	0.08	1.00		0.62				322, 421
monthly	3.0%	(1.4-6.2)	2.79	(0.85-9.18)		1.47(	0.33-6.48	3)				125, 177
<u>&gt;</u> weekly	3.8%	(2.0-7.0)	3.58	(1.17-11.0)		2.03 (	(0.49-8.38	3)				177, 241
Ever had any same se	ex exp	erience/co	ntact									
No	2.3%	(1.5-3.5)	1.00	-	0.89	1.00	-	0.37				577, 762
Yes	2.0%	(0.4-9.0)	0.90	(0.18-4.48)		0.31 (	0.02-4.16	5)				47, 78

#### Table 9-3 continued

2b) Diagnosed with chlamydia in the last year (among those tested in the last year) (n=940) Univariable Multivariable model 1 <sup>Multivariable model</sup>													
			I	Univariable	)	Multiv	ariable m	odel 1	Multi	ivariable n	nodel	Denomina	
	% (	95%CI)	OR	(95%CI)	р	AOR⁵	(95%CI)	р	AOR	2 (95%CI)	р	tor (wt, unwt)	
Age group													
16-19	4.7%(2	.4-9.0)	1.00	-	0.41	1.00	-	0.61				151, 226	
20-24	6.7%(3	.9-11.1)	1.46	(0.59-3.58)		0.76(	0.26-2.15					192, 245	
Country												246 440	
England												316, 440	
Wales												20, 22 7 0	
IMD quintile of LSOA	of resid	ence <sup>d</sup>										1, 3	
2 least deprived	5 2%(2	5-10 5)	1 00	_	0.85	1 00	_	0.85				127 180	
Middle quintile	5.0%(1	.7-13.8)	0.96	(0.26-3.58)		0.68(	0.15-2.97					60, 86	
2 most deprived	6.5%(3	.6-11.4)	1.26	(0.48-3.33)		1.06(	0.42-2.64					155, 205	
Age left school <sup>e</sup>		- /		()		(	,						
17+	5.3%(3	.1-9.0)	1.00	-	0.49							233, 304	
16	7.2%(3	.6-13.8)	1.38	(0.55-3.45)								87, 134	
Age at first heterosex	kual sex												
17+	2.8%(0	.7-9.9)	1.00	-	0.26	1.00	-	0.74				87, 112	
16	4.7%(1	.8-11.8)	1.75	(0.33-9.27)		1.14(	0.24-5.30)	)				84, 108	
<16	7.8%(4	.7-12.6)	2.99	(0.70-12.7)		1.58(	0.37-6.62)					167, 243	
Number of sexual pa	rtners in	the last	year										
0 or 1	3.4%(1	.5-7.8)	1.00	-	<0.01				1.00-		<0.01	145, 196	
2	0.9%(0	.2-3.9)	0.27	(0.05 - 1.41)					0.18	(0.03 - 1.17)		74,99	
3 to 4	1.5%(0	.4-6. <i>2)</i>	0.44	(0.08 - 2.30)					0.23	(0.04 - 1.31)		57,83	
0+ Number of new covus	ZI.Z%(I al partno	2.9-32.1) rs in tho	last y	(2.00-21.3)					2.67(	(0.00-0.00)		63, 69	
	5 5%(2	3_12 6)	1 00		0 37	1 00	_	0 31				108 136	
1	4.0%(1	.3-12.0) 7-9.0)	0.71	- (0 21-2 47)	0.07	1.00	- 0 00-13 0	0.01				115 161	
2+	8.0%(4	5-13 7)	1 48	(0.212.47) (0.50-4.35)		2 87(	0.05 10.0					115, 170	
Number of sexual pa	rtners in	the last	vear	without a c	ondo	m	0.20 00.1					110, 170	
0	1.8%(0	.4-8.4)	1.00	-	0.04	1.00	-	0.45				88. 115	
1	4.9%(2	.6-8.9)	2.75	(0.50-15.0)		0.78(	0.19-3.15	)				160. 222	
2+	10.9%(6	.0-19.0)	6.51	(1.19-35.6)		0.46(	0.11-1.83					95, 134	
Number of lifetime se	exual par	tners		. ,			,						
1 to 4	1.0%(0	.3-3.1)	1.00	-	<0.01	1.00	-	<0.01	1.00-		0.01	133, 176	
5 to 9	3.8%(1	.3-10.8)	4.15	(0.82-21.0)		4.87(	0.58-40.2)		4.25(	(0.61-29.7)		84, 123	
10+	12.3%(7	.6-19.2)	14.6	(3.89-54.5)		19.80(	3.03-129.)	)	8.30(	(1.77-38.8)		121, 167	
Condom used for mo	st recen	t sex wit	h mo	st recent pa	artnei	r							
Yes	4.3%(2	.0-8.9)	1.00		0.22	1.00	-	0.91				163, 221	
No	7.4%(4	.5-12.0)	1.79	(0.70-4.59)		1.06(	0.35-3.21)					172, 237	
Concurrent partners	nips in la	st year	4 00		0.40	4 00		0.07				040 000	
NO Vac	6.7%(4	.1-10.8)	1.00	-	0.10	1.00	- 0 4 0 4 70	0.07				219, 298	
res Linknown	0.0%(2	.7-14.0) 2 5 5)	0.90	(0.34 - 2.72)		0.60(	0.19-1.79					67,92 52,75	
Erequency of binge d	T.3%(U	.3-5.5)	0.19	(0.04-0.66)		0.00(	0.00-0.71					52,75	
never / <monthly< td=""><td>3 5%/1</td><td>6-7 7)</td><td>1 00</td><td>_</td><td>0 07</td><td>1 00</td><td>-</td><td>0.46</td><td></td><td></td><td></td><td>146 204</td></monthly<>	3 5%/1	6-7 7)	1 00	_	0 07	1 00	-	0.46				146 204	
monthly	4 0%(1	5-10 3	1 15	(0.31-4.18)	0.07	0.60/	0 13-2 67	0.10				80 113	
>weekly	9.8%(5	.6-16.6)	2.98	(1.07-8.34)		1.23(	0.48-3.14					117, 154	
Ever had any same s	ex exper	ience/co	ntact	(		0(	0.10 0.14					,	
No	5.8%(3	.8-8.9)	1.00	-	0.87	1.00	-	0.82				311, 427	
Yes	5.1%(1	.0-22.0)	0.87	(0.17-4.54)		0.79(	0.10-6.09)	)				31, 44	

#### Table 9-3 continued

2C) Tested for Chiamydia in the last year (n=1,736)													
	%	(95%CI)		Univariabl	е	Multiv	ariable mo	odel 1	Multi	ivariable m 2	odel	Denominator	
	70	(00700)	OR	(95%CI)	р	$AOR^{b}$	(95%CI)	р	$AOR^{b}$	(95%CI)	р	(wt, unwt)	
Age group 16-19 20-24 Country <sup>c</sup>	40.4% 31.1%	(35.9-45. (27.8-34.)	l) 1.00 7) 0.67	) - 7 (0.52-0.86)	<0.01	1.00 0.53(	- 0.37-0.73)	<0.01	1.00- 0.55	(0.40-0.77)	<0.01	374, 582 629, 793	
England Scotland Wales	37.3% 22.2% 12 8%	(34.3-40.3 (14.0-33.3	3) 1.00 5) 0.48 0.25	) - 8 (0.27-0.85) 5 (0 13-0 48)	<0.01	1.00 0.33( 0.19(	- 0.16-0.64) 0.08-0.40)	<0.01	1.00- 0.37	(0.19-0.73) (0.09-	<0.01	859, 1181 89, 111 55, 83	
	12.070	d	0.20	,(0.10 0.10)					0.220	0.50)		00, 00	
2 least deprived Middle quintile 2 most deprived Age left school <sup>e</sup>	34.5% 33.3% 35.2%	(30.0-39.2 (27.4-39.9 (31.1-39.9	2) 1.00 9) 0.95 5) 1.03	) - 5 (0.67-1.35) 8 (0.78-1.36)	0.89	1.00 1.04( 1.13(	- 0.70-1.52) 0.82-1.53)	0.75				369, 509 183, 263 450, 603	
17+ 16	33.6% 37.8%	(30.4-37.)	l) 1.00 5) 1.20	) - )(0.91-1.58)	0.21							703, 927 230, 334	
Age at first heterose	xual se	ex.	,	,									
17+ 16 <16 Number of sexual pa	25.6% 33.4% 45.3%	(21.4-30.3 (27.9-39.4 (40.7-49.9	3) 1.00 4) 1.46 9) 2.40	) - 5(1.03-2.06) )(1.78-3.23)	<0.01	1.00 1.13( 1.53(	- 0.75-1.67) 1.07-2.19)	0.05	1.00- 1.10( 1.52(	(0.75-1.62) (1.06-2.16)	0.045	340, 431 253, 351 376, 539	
0 or 1 2 3 to 4 5+	26.0% 40.3% 43.0% 60.9%	(22.6-29. (33.2-47. (35.5-50.9 (51.8-69.3	7) 1.00 7) 1.92 9) 2.15 3) 4.42	) 2 (1.34-2.75) 5 (1.49-3.10) 2 (2.93-6.67)	<0.01				1.00- 1.63 1.44 2.43	(1.11-2.39) (0.93-2.22) (1.49-3.98)	<0.01	568, 768 185, 251 134, 194 103, 146	
Number of new sexu	ial part	ners in th	e last	year									
0 1 2+	26.0% 36.7% 46.3%	(22.0-30.5 (31.8-41.5 (40.7-52.0	5) 1.00 3) 1.64 0) 2.45	) 	<0.01	1.00 1.28( 1.06(	- 0.88-1.85) 0.67-1.68)	0.37				416, 540 323, 452 251, 366	
		(22 / 22 ·	tyear	without a c	20NaC	1 00		0 49				221 /50	
1 2+	34.3% 48.8%	(30.2-38.)	7) 1.41 4) 2.59	, (1.02-1.96) (1.81-3.70)		1.12( 1.37(	0.72-1.71) 0.80-2.34)	0110				475, 640 194, 281	
Number of lifetime s			) 1 OC	,	-0 01	1 00		~0.01	1 00		~0.01	E24 706	
5 to 9 10+	39.6% 49.2%	(33.6-45.9 (43.2-55.2	2) 1.93 2) 2.86	; 3(1.39-2.69) 5(2.09-3.92)	<b></b>	1.50( 2.23(	- 1.01-2.21) 1.45-3.42)	<0.01	1.49 2.03	(1.01-2.20) (1.31-3.13)	20.01	222, 314 247, 342	
Condom used for me	OSt rec	ent sex w		st recent p	artne	r 1 00		0 07				101 671	
res	33.5%	(23.0.42)	) 1.00 1) 1.20	, - 2 (0.05, 1.56)	0.13	1.00	- 0 70 1 24)	0.07				491,071	
Concurrent partners	shins in	last vear	f) 1.22	.(0.95-1.50)		0.97 (	0.70-1.34)					430, 021	
No	32.9%	(29.6-36.3	3) 1.00	) -	<0.01	1.00	-	0.24				676, 916	
Yes Unknown	49.7% 38.5%	(41.0-58.4 (30.9-46.	4) 2.02 7) 1.28	2(1.37-2.96) 8(0.88-1.86)		1.52( 1.18(	0.92-2.50) 0.77-1.80)					136, 188 137, 193	
Frequency of binge	drinkin	g			.0.04	4 00		0.04	4 00		0.04	507 715	
never / <u>&lt;</u> monthly monthly <u>≥</u> weekly	28.1% 39.9% 43.2%	(24.6-32.0 (33.4-46. (37.5-49.	0) 1.00 7) 1.69 1) 1.94	) (1.21-2.37) (1.44-2.62)	<0.01	1.00 1.48( 1.50(	- 1.01-2.16) 1.04-2.15)	0.04	1.00- 1.51( 1.42(	(1.05-2.19) (1.00-2.00)	0.04	527, 715 202, 289 273, 370	
Ever had any same s	sex exp	erience/c	ontac	t		4 00						000 1000	
No Yes	33.9% 42.4%	(31.1-36.8 (32.2-53.4	3) 1.00 1) 1.44	- - 	0.11	1.00 2.08(	- 1.12-3.84)	0.02				922, 1260 80, 115	

<sup>a</sup>N in column headings shows unweighted denominators. Total denominators by characteristic and in multivariable models vary due to item-missingness..<sup>b</sup>AOR: Adjusted odds ratios, adjusted for all variables shown. <sup>c</sup>Results for recent diagnosis are not reported due to small sample size in Scotland and Wales when limited to those tested. <sup>d</sup>IMD: Index of multiple deprivation of LSOA (lower super output area) of residence. IMD scores for England, Scotland and Wales were adjusted before being combined and assigned to quintiles, using the method described by Payne and Abel<sup>264</sup>. <sup>e</sup>Excludes 16 year-olds. <sup>f</sup>Among those with  $\geq$ 1 sexual partner in last year. Appendix 3: Factors associated with prevalent infection, recent diagnosis or recent testing among 16 to 24 year-olds resident in England (Natsal-3)

### Relevant to Chapter 7

Table 9-4a and Table 9-4b overleaf present equivalent tables to those presented in Table 7-2a and Table 7-2b with the population restricted to participants resident in England as opposed to the population resident in Britain.

Table 9-4a: Percentage, unadjusted and adjusted odds (OR/AOR) ratios for prevalent chlamydia infection, self-reported diagnosis in the last year and self-reported testing by sociodemographic and behavioural factors (16-24 year-old sexually-experienced <u>women</u> resident in <u>England</u> Natsal-3)

	Prevalent chlamydia infection detected in urine (n=992)								Diagnose	d with	chlamydi	a in the	last year			Tested	for chla	amydia in	the last	t year		D (weigh	enominate	or abted) <sup>a</sup>
	0/ (050/ 0		- -	(II=992)				0/					(050(OI)	_	0/		( )	(050( QI)			_	(weight	Diamaria	Tracks d
-	% (95%C	I) OF	≺ (!	95%CI)	AUR	(95%CI)	р	%	(95%CI)	UR	(95%CI)	AUR	(95%CI)	р	%	(95%CI)	UR	(95%CI)	AUR	(95%CI)	р	Intection	Diagnosis	lested
Age group																						1		
16-19	3.6% (2.0-6.5	) 1.(	00	-			0.35	5.89	%(3.6-9.2)	1.00	-	1.00	-	0.62	61.2%	6(56.7-65.5)	1.00	-	1.00	-	0.07	181, 332	177, 341	291, 566
20-24	2.5% (1.5-4.2)	) 0.0	68 (0	.30-1.53)	0.61	(0.21-1.74	4)	5.5%	%(3.6-8.2)	0.94	(0.49-1.81	) 0.80	0.34-1.89)		54.9%	(50.9-58.8)	0.77	(0.61-0.98	) 0.76(	(0.56-1.02	)	323, 485	292, 491	532, 886
IMD quintile of LSOA	of residence																					1		
2 least deprived	1.2% (0.4-3.6	) 1.(	00	-	1.00	-	0.01	4.7	%(2.6-8.2)	1.00	-	1.00	-	0.23	56.9%	6(51.9-61.8)	1.00	-	1.00	-	0.660	179, 296	166, 289	292, 510
Middle quintile	1.3% (0.5-3.4)	) 1.(	02 (0	.23-4.44)	0.95	(0.21-4.13	3)	3.29	%(1.3-7.5)	0.66	(0.22-1.96	) 0.92	0.28-3.00)		58.0%	(50.9-64.7)	1.04	(0.74-1.48	) 1.20(	(0.80-1.79	)	92, 140	92, 154	159, 267
2 most deprived	4.9% (3.1-7.5)	) 4.0	08 (1	.26-13.3)	4.40	(1.32-14.6	5)	7.49	%(4.9-10.9)	1.61	(0.77-3.36	) 1.83	0.77-4.31)		56.9%	(52.2-61.5)	1.00	(0.76-1.32	) 1.03 (	(0.75-1.39	)	233, 381	210, 389	371, 675
Age at first heterosex	ual sex																					1		
17+	1.3% (0.4-3.8)	) 1.(	00	-	1.00	-	0.51	4.49	%(1.9-9.6)	1.00	-	1.00	-	0.94	47.8%	(42.1-53.6)	1.00	-	1.00	-	0.13	156, 198	127, 195	267, 406
16	3.7% (1.9-7.1)	) 3.0	00 (0	.82-11.0)	2.38	(0.54-10.3	3)	5.89	%(3.2-10.2)	1.34	(0.47-3.83	) 0.93	0.30-2.79)		57.7%	(52.4-62.8)	1.49	(1.09-2.04	) 1.21 (	(0.85-1.71	)	152, 245	133, 230	231, 413
<16	3.9% (2.2-6.8)	) 3.1	11 (0	.89-10.9)	1.79	(0.48-6.66	5)	6.5%	%(4.4-9.7)	1.53	(0.60-3.89	) 0.85	0.32-2.24)		67.4%	(63.0-71.5)	2.26	(1.70-3.01	) 1.42	(1.01-2.00	)	181, 351	198, 385	295, 579
Number of sexual part	tners in the last	year	witho	out a con	dom																	1		
0	2.3% (0.7-7.7)	· 1.(	00	-	1.00	-	0.26	4.5%	%(1.6-12.1)	1.00	-	1.00	-	0.79	40.3%	(34.1-46.8)	1.00	-	1.00	-	0.160	101, 143	72, 121	181, 306
1	2.0% (1.1-3.5	0.8	85 (0	.22-3.33)	0.41	(0.09-1.78	3)	3.89	%(2.2-6.3)	0.83	(0.25-2.79	) 0.79	0.17-3.60)		57.2%	(53.5-60.7)	1.98	(1.48-2.65	) 1.41 (	(0.92-2.14	)	309, 492	282, 494	495, 869
2+	6.6% (3.5-12.	0) 2.9	93 (0.	.72-11.9	0.81	(0.18-3.63	3)	10.99	%(7.3-16.1)	2.62	(0.80-8.65	) 1.12	0.22-5.63)		77.7%	(71.5-83.0)	5.18	(3.42-7.85	) 1.73	(0.96-3.10	)	94, 181	113, 215	146, 274
Number of new sexua	I partners in the	alast	vear	,		·	,		. ,							. ,		•				1		
0	2.2% (1.2-3.9	) 1.0	00	-	1.00	-	0.99	2.99	%(1.6-5.4)	1.00	-	1.00	-	0.22	47.8%	(43.8-51.9)	1.00	-	1.00	-	0.003	263, 400	201, 348	421, 727
1	2.6% (1.0-6.7	) 1.2	22 (0	.39-3.81)	1.06	(0.28-3.89	9)	5.29	%(2.7-9.9)	1.83	(0.72-4.62	) 2.04	0.71-5.80)		62.5%	(56.9-67.8)	1.82	(1.38-2.41	) 1.71 (	(1.23-2.36	)	140, 225	143, 257	229, 411
2+	5.4% (3.0-9.7	2.6	60 (1.	.08-6.27)	1.06	(0.28-3.98	ý 3)	10.59	%(6.9-15.7)	3.91	(1.82-8.42	) 2.76	0.86-8.82)		74.1%	(67.2-80.0)	3.12	(2.16-4.51	1.74	(1.09-2.79	)	97, 186	122, 222	166, 301
Number of lifetime sex	xual partners		(	,		<b>(</b>	,		,			,	,			(,					,	1		
1 to 4	2 1% (1 1-4 1	) 1(	00	-	1 00	-	0 48	2.39	%(1 2-4 3)	1 00	-	1 00	-	0.08	47.3%	(43 4-51 3)	1 00	-	1 00	-	0 000	274, 401	203, 357	430, 751
	2	,					00	,	, ( , , _ , , , , , , , , , , , , , , ,					0.00		(1011 0110)					9			
5 to 9	2 2% (1 0-4 7	) 1(	03 (0	36-2 92)	0.87	(0 28-2 72	2)	6.89	%(4 0-11 4)	3 16	(1 36-7 34	) 2 60	0 96-6 98)		66.3%	(60 6-71 7)	2 1 9	(1 64-2 93	181	(1 28-2 55	)	126, 217	142, 251	214, 376
10+	6.1% (3.2-11)	3) 3(	02 (1	17-7.81)	1 71	(0 54-5 30	-/ 2)	9.80	%(6 2-15 1)	4 70	(2 07-10 7	) 3.71	1 12-12 2)		71 5%	(65 2-77 1)	2.80	(2 01-3 89	1 92	(1 24-2 95	)	100, 191	122, 219	170.310
Condom used for mos	st recent sex wi	th mo	st rec	cent nart	ner	(0.0 + 0.0)	')	0.07	/0(0.2 10.1)		(2.07 10.7	, 5.71			1.070	(00.2 77.1)	2.00	(2.01 0.00	, 1.52	(1.212.00	/		, = · •	.,
Yes	2 3% (1 0-5 1	1 (	00	-	1 00	_	0.36	3 79	%(1 8-7 6)	1 00	_	1 00	-	0 24	55 1%	(49 9-60 2)	1 00	_	1 00	-	0 920	173, 262	156, 273	284, 494
No	3.6% (2.3-5.5)	, 1.0 ) 1 <i>!</i>	54 (N	60-3 93)	1 49	(0.63-3.5(	))	6.5	%(4 4-9 3)	1.80	(0 76-4 31	) 1.67	0 70-3 94)	0.24	60.9%	(57 2-64 5)	1 27	(0 99-1 63	1.00	(0 73-1 40	)	298, 505	293, 521	482, 858

#### Table 9-4a continued.

Concurrent partnership	s in last year <sup>t</sup>												
No	2.6% (1.6-4.2)	1.00 -	1.00 - 0.46	4.5%(2.9-6.8)	1.00 -	1.00 - 0.56	54.0%(50.5-57.5)	1.00 -	1.00 -	0.12	369, 574	325, 564 (	603, 1045
Yes	7.2% (3.3-15.1)	2.92 (1.10-7.75)	1.60 (0.55-4.64)	8.6%(4.2-16.6)	2.00 (0.83-4.86)	0.72(0.25-2.02)	77.5%(69.6-83.7)	2.92 (1.92-4.44)	1.47 (0.90-2.39)		58, 117	70, 131	90, 169
Unknown	1.6% (0.5-5.2)	0.63 (0.18-2.24)	0.66 (0.15-2.75)	7.4%(3.8-14.0)	1.72 (0.78-3.81)	1.34(0.55-3.25)	67.3%(59.5-74.3)	1.75 (1.23-2.51)	1.46 (0.93-2.29)		60, 105	67, 126	99, 190
Frequency of binge													
drinking													
Never / <monthly< td=""><td>2.3% (1.3-4.1)</td><td>1.00 -</td><td>1.00 - 0.02</td><td>3.9%(2.5-6.0)</td><td>1.00 -</td><td>1.00 - 0.27</td><td>55.1%(51.4-58.7)</td><td>1.00 -</td><td>1.00 -</td><td>0.01</td><td>321, 502</td><td>289, 512</td><td>527, 934</td></monthly<>	2.3% (1.3-4.1)	1.00 -	1.00 - 0.02	3.9%(2.5-6.0)	1.00 -	1.00 - 0.27	55.1%(51.4-58.7)	1.00 -	1.00 -	0.01	321, 502	289, 512	527, 934
Monthly	1.1% (0.3-3.3)	0.46 (0.12-1.70)	0.43 (0.11-1.67)	6.8%(3.5-12.8)	1.80 (0.78-4.12)	2.08(0.75-5.72)	53.4%(46.8-59.8)	0.93 (0.70-1.25)	0.66 (0.47-0.93)		108, 181	95, 165	178, 302
>weekly	8.2% (4.6-14.1)	3.80 (1.60-9.04)	2.39 (0.97-5.89)	9.7%(5.2-17.6)	2.66 (1.18-5.99)	2.15(0.74-6.22)	71.7%(64.0-78.3)	2.06 (1.42-2.99)	1.35 (0.89-2.04)		76, 134	84, 153	117, 214
Ever had any same													
sex experience/contact													
No	2.7% (1.7-4.2)	1.00 -	1.00 - 0.64	5.5%(3.7-7.9)	1.00 -	1.00 - 0.36	54.9%(51.4-58.3)	1.00 -	1.00 -	0.33	397, 609	346, 618 (	633, 1120
Yes	3.7% (1.6-8.2)	1.39 (0.53-3.60)	0.80 (0.31-2.04)	6.0%(3.4-10.4)	1.10 (0.54-2.27)	0.65	64.6%(58.2-70.6)	1.50 (1.11-2.03)	1.19 (0.83-1.68)		107, 208	122, 214	190, 332

Table 9-4b: Proportion, unadjusted and adjusted odds ratios for prevalent chlamydia infection, self-reported diagnosis in the last year andself-reported testing by sociodemographic and behavioural factors (16-24 year-old sexually-experienced men resident in England, Natsal-3)

Prevalent chlamydia infection detected in urine (n=992)							e	(;	Diagnose among the	d wit ose te	h chlamydia ested in the	a in the last yea	last year r) (n=940)			Tested	for ch	lamydia ir (n=1,736)	the las	st year		D (weight	enominato ed, unweig	or ghted) <sup>a</sup>
	%	(95%CI)	OR	(95%CI)	AOR <sup>b</sup>	(95%CI)	р	%	(95%CI)	OR	(95%CI)	$AOR^{b}$	(95%CI)	р	%	(95%CI)	OR	(95%CI)	$AOR^{b}$	(95%CI)	р	Infection I	Jiagnosis	Tested
Age group																								
16-19	0.2%	(0.0-1.5	) 1.00	-	1.00	-	0.06	3.7%	(1.8-7.8)	1.0	0 -	1.00		- 0.91	44.3%	(39.6-49.2	2) 1.00	-	1.00	-	0.0001	202, 303	141, 218	319, 511
20-24	2.9%	(1.8-4.8	)13.85	(1.80-106)	9.72	(0.92-102)		5.9%	(3.3-10.6)	1.6	3(0.60-4.44)	0.93	(0.27-3.12	)	33.1%	(29.4-37.2	1) 0.62	(0.48-0.81)	0.50(	0.35-0.70)		330, 416	175, 222	540, 670
IMD quintile of L	SOA of	residend	e																					
2 least deprived	0.9% (	0.3-3.1)	1.00	-	1.00	-	0.04	4.7%	2.1-10.3)	1.0	0 -	1.00		0.54	37.1%(3	32.3-42.1)	1.00	-	1.00	-	0.64	214, 284	121, 171	327, 457
Middle quintile	0.8% (	0.2-3.5)	0.92	(0.14-6.04)	0.78	(0.06-9.66	)	3.4%	1.0-10.4)	0.7	0(0.17-2.96)	0.47(	0.11-1.94)		35.3%(2	28.6-42.6)	0.92	(0.63-1.34)	0.97(	0.64-1.45)		95, 137	53, 77	151, 217
2 most deprived	3.3% (	1.9-5.7)	3.69	(0.96-14.2)	4.13	(0.91-18.5	)	5.7%	3.0-10.7)	1.2	2(0.41-3.58)	0.97(	0.35-2.67)		38.2%(3	33.8-42.8)	1.05	(0.79-1.39)	1.13(	0.81-1.56)		222, 298	142, 192	381, 507
Age at first heter	osexua	al sex																						
17+	0.9% (	0.3-2.9)	1.00	-	1.00	-	0.64	3.1%	0.8-11.0)	1.0	0 -	1.00 ·		0.86	26.9%(2	22.5-31.9)	1.00	-	1.00	-	0.02	173, 207	78, 105	291, 371
16	1.7% (	0.5-5.3)	1.93	(0.35-10.5)	2.13	(0.43-10.5	)	4.5%	1.6-12.2)	1.4	8(0.27-8.27)	1.01(	0.18-5.60)		36.2%(3	30.2-42.6)	1.54	(1.07-2.20)	1.17(	0.78-1.74)		129, 179	80, 103	221, 308
<16	3.0% (	1.7-5.3)	3.47	(0.89-13.5)	1.21	(0.32-4.50	)	6.3%	3.6-10.8)	2.1	1(0.48-9.22)	1.38(	0.28-6.66)		49.4%(4	14.3-54.4)	2.64	(1.94-3.60)	1.64(	1.14-2.35)		205, 301	154, 225	318, 458
Number of sexua	ıl partn	ers in the	e last y	year withou	it a cond	lom																		
0	0.4% (	0.1-1.6)	1.00	-	1.00	-	0.12	0.0%-							28.3%(2	23.4-33.7)	1.00	-	1.00	-	0.08	167, 207	77, 107	276, 382
1	1.3% (	0.5-3.1)	3.20	(0.59-17.2)	0.41	(0.03-4.64	)	3.8%	1.9-7.4)	0.3	1(0.11-0.83)				36.4%(3	31.9-41.1)	1.45	(1.03-2.04)	1.25(	0.79-1.97)		259, 356	152, 209	424, 568
2+	5.9% (	3.3-10.5)	15.56	(3.32-73.0)	1.82	(0.12-26.4	)	11.3%	6.1-20.0)	1.0	0-				56.2%(4	18.8-63.3)	3.26	(2.20-4.81)	1.86(	1.06-3.25)		104, 154	88, 124	156, 227
Number of new s	exual p	partners i	in the	last year																				
0	1.2% (	0.5-3.1)	1.00	-	1.00	-	0.99	4.5%(	1.7-11.3)	1.0	0 -	1.00 ·		0.99	28.3%(2	24.0-33.1)	1.00	-	1.00	-	0.77	234, 298	105, 132	371, 477
1	0.9% (	0.3-3.0)	0.75	(0.16-3.49)	0.94	(0.15-5.80	)	3.0%	1.2-7.8)	0.6	7(0.16-2.71)	1.12(	0.24-5.21)		38.2%(3	32.8-43.8)	1.56	(1.13-2.16)	1.14(	0.77-1.66)		167, 226	99, 145	269, 381
2+	4.5% (	2.4-8.2)	3.74	(1.18-11.9)	0.92	(0.16-5.18	)	7.3%(	3.9-13.4)	1.6	7(0.50-5.51)	1.11(	0.29-4.12)		52.1%(4	46.0-58.1)	2.75	(1.97-3.84)	1.17(	0.72-1.87)		127, 190	108, 159	208, 309
Number of lifetin	ne sexu	al partne	ers																					
1 to 4	0.4%(	0.1-1.8)	1.00	-	1.00	-	0.03	0.7%	0.2-3.1)	1.0	0-	1.00		0.00	27.1%(2	23.4-31.2)	1.00	-	1.00	-	0.004	281, 357	122, 167	451, 616
5 to 9	0.3% (	0.0-2.2)	0.72	(0.06-8.26)	0.30	(0.02-3.45	)	2.0%	0.5-8.3)	2.8	6(0.35-23.2)	2.31(	).22-23.3)		43.2%(3	36.6-50.1)	2.04	(1.44-2.89)	1.52(	1.01-2.26)		125, 173	80, 118	193, 270
10+	6.9% (	4.1-11.3)	17.25	(3.68-80.8)	6.27	(0.88-44.2	)	12.0%	7.2-19.3)	19.2	3(3.87-95.5)	14.44(	2.19-94.9)		53.4%(4	46.8-59.9)	3.08	(2.21-4.31)	2.15(	1.36-3.38)		124, 186	110, 151	206, 284
Condom used fo	r most	recent se	ex with	h most rece	ent partn	er																		
Yes	0.3% (	0.1-1.4)	1.00	-			0.04	2.8%	1.1-7.0)	1.0	0 -	1.00 ·		0.14	36.1%(3	32.1-40.4)	1.00	-	1.00	-	0.59	252, 330	149, 209	417, 576
No	3.6% (	2.2-6.0)	10.99	(2.45-49.3)	9.41	(1.06-82.9	)	7.2%	4.2-12.1)	2.7	3(0.88-8.50)	2.47(	0.75-8.12)		40.8%(3	36.3-45.6)	1.22	(0.94-1.59)	0.91(	0.64-1.28)		245, 343	160, 218	395, 532

#### Table 9-4b continued.

Concurrent partn	erships in last y	'ear'													
No	2.0% (1.1-3.6)	1.00 -	1.00 -	0.09	6.1%(3.6-10.2)	1.00 -		0.05	36.0%(32.4-39.8)	1.00 -	1.00 -	0.72	357, 475	207, 281	582, 789
Yes	2.6% (0.9-7.0)	1.27 (0.39-4.21)	0.18 (0.03-0.84)		5.0%(1.8-13.5)	0.81(0.24-2.71)	1.00 -		51.5%(42.4-60.5)	1.89(1.26-2.82)	1.11(0.69-1.78)		62, 97	55, 80	106, 152
Unknown	1.4% (0.3-6.0)	0.70 (0.15-3.37)	0.62 (0.09-4.05)		0.6%(0.1-4.2)	0.09(0.01-0.73)	0.07(0.00-0.70)		41.6%(33.4-50.3)	1.26(0.86-1.87)	1.20(0.76-1.86)		83, 111	50, 73	123, 173
Frequency of bin	ge drinking														
Never / <monthly< th=""><th>0.8% (0.3-2.5)</th><th>1.00 -</th><th>1.00 -</th><th>0.49</th><th>2.8%(1.2-6.5)</th><th>1.00 -</th><th>1.00 -</th><th>0.18</th><th>30.3%(26.4-34.5)</th><th>1.00 -</th><th>1.00 -</th><th>0.06</th><th>279, 363</th><th>138, 194</th><th>463, 624</th></monthly<>	0.8% (0.3-2.5)	1.00 -	1.00 -	0.49	2.8%(1.2-6.5)	1.00 -	1.00 -	0.18	30.3%(26.4-34.5)	1.00 -	1.00 -	0.06	279, 363	138, 194	463, 624
Monthly	2.9% (1.3-6.2)	3.56 (0.88-14.5)	1.89 (0.31-11.4)		2.1%(0.7-6.7)	0.76(0.17-3.27)	0.48(0.09-2.36)		44.0%(36.8-51.5)	1.81(1.27-2.59)	1.52(1.01-2.28)		108, 154	75, 106	171, 246
>weekly	3.3% (1.6-6.6)	4.13 (1.07-16.1)	2.73 (0.51-14.3)		9.8%(5.3-17.5)	3.78(1.25-11.4)	1.78(0.61-5.16)		46.6%(40.5-52.8)	2.01(1.46-2.75)	1.41(0.97-2.04)		143, 201	103, 140	224, 310
Ever had any san	ne sex experiend	ce/contact													
No	2.0% (1.2-3.2)	1.00 -			5.4%(3.4-8.5)				36.4%(33.4-39.6)	1.00 -	1.00 -	0.02	499, 661	289, 401	799, 1093
Yes	0.9% (0.1-6.5)	0.44 (0.06-3.44)			0.0%-				48.4%(36.7-60.3)	1.64(1.01-2.68)	2.22(1.11-4.42)		33, 58	27, 39	60, 88

<sup>a</sup>N in column headings shows unweighted denominators. Total denominators by characteristic and in multivariable models vary due to item-missingness. <sup>b</sup>AOR: Adjusted odds ratios, adjusted for all variables shown. <sup>c</sup>Results for recent diagnosis are not reported due to small sample size in Scotland and Wales when limited to those tested. <sup>d</sup>IMD: Index of multiple deprivation of LSOA (lower super output area) of residence. IMD scores for England, Scotland and Wales were adjusted before being combined and assigned to quintiles, using the method described by Payne and Abel<sup>264</sup>. <sup>e</sup>Excludes 16 year-olds. <sup>f</sup>Among those with  $\geq$ 1 sexual partner in last year. Shaded areas indicate variables that were not entered into univariable or multivariable logistic regression due to small sample sizes.

## **Appendix 4: Thesis outputs**

#### Peer-reviewed publications

<u>Woodhall SC</u>, Soldan K, Sonnenberg P, Mercer CH, Clifton S, Saunders P, da Silva F, Alexander S, Wellings K, Tanton C, Field N, Copas AJ, Ison CA, Johnson AM. Is chlamydia screening and testing in Britain reaching young adults at risk of infection? Findings from the third National Survey of Sexual Attitudes and Lifestyles (Natsal-3). Sex Transm Infect. 2015 [Epub ahead of print]

<u>Woodhall SC</u>, Nichols T, Alexander S, da Silva FC, Mercer CH, Ison C, Gill ON, Soldan K. Can we use postal surveys with anonymous testing tomonitor chlamydia prevalence in young women in England? Pilot study incorporating randomised controlled trial of recruitment methods. Sex Transm Infect. 2015;91(6):412-4. doi: 10.1136/sextrans-2015-052067

Woodhall SC, Wills G, Horner P, Craig R, Mindell JS, Murphy G, McClure M, Soldan K, Johnson AM\*, Nardone A1\* [\*joint senior authors] Chlamydia trachomatis Pgp3 antibody population seroprevalence before and during an era of widespread opportunistic chlamydia screening in England (1994-2012). PLoS One (under review)

Redmond SM, Alexander-Kisslig K, <u>Woodhall SC</u>, van den Broek IV, van Bergen J, Ward H, Uusküla A, Herrmann B, Andersen B, Götz HM, Sfetcu O, Low N. Genital chlamydia prevalence in Europe and non-European high income countries: systematic review and meta-analysis. PLoS One. 2015 Jan 23;10(1):e0115753.

#### **Conference presentations**

<u>Woodhall SC</u>, Alexander S, Nichols T, Mercer CH, Ison CA, Gill N, Soldan K. Monitoring the prevalence of chlamydia in England. Oral Presentation. Health Protection Conference 2011. Warwick, UK.

<u>Woodhall SC</u>, Soldan K, Sonnenberg P, Mercer CH, Clifton SC, Saunders P, Coelho da Silva F, Alexander S, Wellings K, Tanton C, Field N, Copas AJ, Ison CA, Johnson AM. Chlamydia infection and control among young adults in Britain in 2010-2012: Findings from the 3<sup>rd</sup> British National Survey of Sexual Attitudes and Lifestyles (Natsal-3). Poster presentation: Third Joint Conference of the British HIV Association (BHIVA) with the British Association for Sexual Health and HIV (BASHH). 2014. Liverpool, UK.

<u>Woodhall SC</u>, Mercer CH, Gill ON, Macintosh M, Soldan K. How much can available data explain the decline in the proportion positive in the National Chlamydia Screening Programme in England? Poster presentation: 19<sup>th</sup> ISSTDR Conference 2011, Quebec City, Canada.

<u>Woodhall SC</u>, Mercer CH, Hughes G, Bone A, Gill ON, Johnson AM, Soldan K. Making sense of chlamydia surveillance data: understanding trends in routine data about testing and diagnoses of *Chlamydia trachomatis*. Poster presentation: MRC Conference, Population Health Methods and Challenges. 2012. Birmingham, UK. <u>Woodhall SC</u>. What can chlamydia serology tell us about chlamydia and control efforts in England? Oral presentation: British Association for Sexual Health and HIV (BASHH) Evening Scientific Meeting. 2015. London, UK.

<u>Woodhall SC</u>, Wills G, Horner P, Craig R, Mindell J, Murphy G, McClure M, Soldan K, Johnson AM, Nardone A. Seroprevalence of *Chlamydia trachomatis* among women and men in England, 1994 to 2012: repeated cross-sectional study using sera from a nationally representative household survey. Oral presentation: Public Health England Applied Epidemiology Scientific Meeting. 2015. Warwick, UK.

<u>Woodhall SC</u>, Soldan K, Sonnenberg P, Mercer CH, Clifton S, Saunders P, Coelho da Silva F, Alexander S, Tanton C, Field N, Copas A, Ison CA, Johnson AM. What can probability surveys tell us about changes in chlamydia prevalence in Britain? Evidence from the National Surveys of Sexual Attitudes and Lifestyles (Natsal). Abstract accepted for poster presentation at the 2015 World STI & HIV Congress, Brisbane, Australia (September 2015).

<u>Woodhall SC</u>, Gillian Wills, Horner P, Craig R, Mindell JS, Murphy G, McClure M, Soldan K, Nardone A,<sup>\*</sup> Johnson AM.<sup>\*</sup> [\*joint senior authors] Insights into *Chlamydia trachomatis* cumulative incidence in the context of widespread opportunistic chlamydia screening in England: Seroprevalence study using sera from a nationally-representative household survey. Abstract accepted for poster presentation at the 2015 World STI & HIV Congress, Brisbane, Australia (September 2015).

#### Letters

<u>Woodhall SC</u>, Ong KJ, Saunders J, Dunbar K. Mathematical modelling of costeffectiveness of chlamydia testing highlights complexities and uncertainties that should not be overlooked. Response to: Looker et al, Impact and costeffectiveness of chlamydia testing in Scotland: a mathematical modelling study. Theoretical Biology and Medical Modelling 2015,12:2